

Leah Leipold Senior Counsel Direct: 470.481.4218 leah@caiolarose.com

May 7, 2024

<u>VIA ELECTRONIC FILING</u> Wisconsin Department of Financial Institutions https://www.wdfi.org/apps/franchiseefiling/

RE: Enviro-Master International Franchise, LLC ("Enviro-Master Services") Registration: #633406 Application for Initial Franchise Registration

Dear Sir or Madam:

Uploaded with this letter please find one clean copy of Enviro-Master Services' Franchise Disclosure Document submitted on behalf of Enviro-Master Services to offer and sell its franchises in the State of Wisconsin.

A renewal fee of \$400.00 has been submitted on behalf of Enviro-Master Services through the online securities portal with this renewal application.

Please contact me at 470-481-4218 or franchise@caiolarose.com if you have any questions or comments on the enclosed application.

Sincerelv Senior Counsel

Enclosures

FRANCHISE DISCLOSURE DOCUMENT



Enviro-Master International Franchise, LLC 5200 77 Center Drive, Suite 500 Charlotte, NC 28217 www.enviro-master.com

An Enviro-Master International Franchise, LLC franchise (a "Franchised Business") provides restroom hygiene, drain line management, window cleaning, power washing, paper, and chemical products and services to customers that include restaurants, including quick service and traditional, hotels, schools, and other types of commercial establishments.

The total investment necessary to begin operation of a single-unit Franchised Business is estimated to be between \$96,260 to \$220,260. This includes \$69,960 to \$70,960, which must be paid to Enviro-Master International Franchise, LLC.

If a franchisee chooses to sign a Multi-Unit Development Addendum to Franchise Agreement, the total initial investment necessary to begin operation of two to five Franchised Businesses is \$265,060 to \$529,060. This includes \$125,920 to \$264,800, which must be paid to Enviro-Master International Franchise, LLC.

This disclosure document summarizes certain provisions of your franchise agreement and other information in plain English. Read this disclosure document and all agreements carefully. You must receive this disclosure document at least 14 calendar days before you sign a binding agreement with, or make any payment to, the Enviro-Master or an affiliate in connection with the proposed franchise sale. Note, however, that no government agency has verified the information contained in this document.

You may wish to receive your disclosure documents in another format that is more convenient for you. To discuss the availability of disclosure in different formats, contact Enviro-Master at 5200 77 Center Drive, Suite 500 Charlotte, NC 28217, (704) 302-1016.

The terms of your contract will govern your franchise relationship. Do not rely on the disclosure document alone to understand your contract. Read all of your contract carefully. Show your contract and this disclosure document to an advisor, like a lawyer or an accountant. Buying a franchise is a complex investment. The information in this disclosure document can help you make up your mind. More information on franchising, such as "*A Consumer's Guide to Buying a Franchise*," which can help you understand how to use this disclosure document, is available from the Federal Trade Commission. You can contact the FTC at 1-877-FTC-HELP or by writing to the FTC at 600 Pennsylvania Avenue, NW, Washington, DC 20580. You can also visit the FTC's home page at www.ftc.gov for additional information. Call your state agency or visit your public library for other sources of information on franchising.

There may also be laws on franchising in your state. Ask your state agencies about them.

Issued: May 6, 2024

#198798v1 Enviro-Master International Franchise, LLC FDD May 6, 2024

How to Use This Franchise Disclosure Document

Here are some questions you may be asking about buying a franchise and tips on how to find more information:

QUESTION	WHERE TO FIND INFORMATION
How much can I earn?	Item 19 may give you information about outlet sales, costs, profits or losses. You should also try to obtain this information from others, like current and former franchisees. You can find their names and contact information in Item 20 or Exhibits C and D.
How much will I need to invest?	Items 5 and 6 list fees you will be paying to the franchisor or at the franchisor's direction. Item 7 lists the initial investment to open. Item 8 describes the suppliers you must use.
Does the franchisor have the financial ability to provide support to my business?	Item 21 or Exhibit E includes financial statements. Review these statements carefully.
Is the franchise system stable, growing, or shrinking?	Item 20 summarizes the recent history of the number of company-owned and franchised outlets.
Will my business be the only Enviro-Master business in my area?	Item 12 and the "territory" provisions in the franchise agreement describe whether the franchisor and other franchisees can compete with you.
Does the franchisor have a troubled legal history?	Items 3 and 4 tell you whether the franchisor or its management have been involved in material litigation or bankruptcy proceedings.
What's it like to be an Enviro- Master franchisee?	Item 20 or Exhibits C and D lists current and former franchisees. You can contact them to ask about their experiences.
What else should I know?	These questions are only a few things you should look for. Review all 23 Items and all Exhibits in this disclosure document to better understand this franchise opportunity. See the table of contents.

What You Need To Know About Franchising Generally

Continuing responsibility to pay fees. You may have to pay royalties and other fees even if you are losing money.

Business model can change. The franchise agreement may allow the franchisor to change its manuals and business model without your consent. These changes may require you to make additional investments in your franchise business or may harm your franchise business.

<u>Supplier restrictions.</u> You may have to buy or lease items from the franchisor or a limited group of suppliers the franchisor designates. These items may be more expensive than similar items you could buy on your own.

Operating restrictions. The franchise agreement may prohibit you from operating a similar business during the term of the franchise. There are usually other restrictions. Some examples may include controlling your location, your access to customers, what you sell, how you market, and your hours of operation.

<u>Competition from franchisor.</u> Even if the franchise agreement grants you a territory, the franchisor may have the right to compete with you in your territory.

<u>Renewal.</u> Your franchise agreement may not permit you to renew. Even if it does, you may have to sign a new agreement with different terms and conditions in order to continue to operate your franchise business.

<u>When your franchise ends.</u> The franchise agreement may prohibit you from operating a similar business after your franchise ends even if you still have obligations to your landlord or other creditors.

Some States Require Registration

Your state may have a franchise law, or other law, that requires franchisors to register before offering or selling franchises in the state. Registration does not mean that the state recommends the franchise or has verified the information in this document. To find out if your state has a registration requirement, or to contact your state, use the agency information in Exhibit A.

Your state also may have laws that require special disclosures or amendments be made to your franchise agreement. If so, you should check the State Specific Addenda. See the Table of Contents for the location of the State Specific Addenda.

Special Risks to Consider About *This* Franchise

Certain states require that the following risk(s) be highlighted:

- 1. <u>Out-of-State Dispute Resolution.</u> The franchise agreement requires you to resolve disputes with the franchisor by mediation, arbitration and/or litigation only in North Carolina. Out-of-state mediation, arbitration, or litigation may force you to accept a less favorable settlement for disputes. It may also cost more to mediate, arbitrate, or litigate with the franchisor in North Carolina than in your own state.
- 2. <u>Supplier Control.</u> You must purchase all or nearly all of your inventory or supplies that are necessary to operate your business from the franchisor, its affiliates, or suppliers that franchisor designates, at prices the franchisor or they set. These prices may be higher than prices you could obtain elsewhere for the same or similar goods. This may reduce the anticipated profit of your franchise business.
- 3. <u>Mandatory Minimum Payments</u>. You must make minimum royalty or advertising fund payments, regardless of your sales levels. Your inability to make the payments may result in termination of your franchise and loss of your investment.
- 4. <u>Sales Performance Required.</u> You must maintain the minimum sales performance levels. Your inability to maintain these levels may result in loss of any territorial rights you are granted, termination of your franchise, and loss of your investment.

Certain states may require other risks to be highlighted. Check the "State Specific Addenda" (if any) to see whether your state requires other risks to be highlighted.

THE STATE OF MICHIGAN PROHIBITS CERTAIN UNFAIR PROVISIONS THAT ARE SOMETIMES IN FRANCHISE DOCUMENTS. IF ANY OF THE FOLLOWING PROVISIONS ARE IN THESE FRANCHISE DOCUMENTS, THE PROVISIONS ARE VOID AND CANNOT BE ENFORCED AGAINST YOU.

Each of the following provisions is void and unenforceable if contained in any documents relating to a franchise:

- 1. A prohibition on the right of a franchisee to join an association of franchisees.
- 2. A requirement that a franchisee assent to a release, assignment, novation, waiver, or estoppel which deprives a franchisee of rights and protections provided in this act. This shall not preclude a franchisee, after entering into a franchise agreement, from settling any and all claims.
- 3. A provision that permits a franchisor to terminate a franchise before the expiration of its term except for good cause. Good cause shall include the failure of the franchisee to comply with any lawful provision of the franchise agreement and to cure such failure after being given written notice thereof and a reasonable opportunity, which in no event need be more than 30 days, to cure such failure.
- 4. A provision that permits a franchisor to refuse to renew a franchise without fairly compensating the franchisee by repurchase or other means for the fair market value at the time of expiration of the franchisee's inventory, supplies, equipment, fixtures, and furnishings. Personalized materials which have no value to the franchisor and inventory, supplies, equipment, fixtures, and furnishings not reasonably required in the conduct of the franchise business are not subject to compensation. This subsection applies only if: (i) the term of the franchise is less than 5 years and (ii) the franchisee is prohibited by the franchise or other agreement from continuing to conduct substantially the same business under another trademark, service mark, trade name, logotype, advertising, or other commercial symbol in the same area subsequent to the expiration of the franchise or the franchise does not receive at least 6 months advance notice of franchisor's intent not to renew the franchise.
- 5. A provision that permits the franchisor to refuse to renew a franchise on terms generally available to other franchisees of the same class or type under similar circumstances. This section does not require a renewal provision.
- 6. A provision requiring that arbitration or litigation be conducted outside this state. This shall not preclude the franchisee from entering into an agreement, at the time of arbitration, to conduct arbitration at a location outside this state.
- 7. A provision which permits a franchisor to refuse to permit a transfer of ownership of a franchise, except for good cause. This subdivision does not prevent a franchisor from exercising a right of first refusal to purchase the franchise. Good cause shall include, but is not limited to:

(a) The failure of the proposed transferee to meet the franchisor's then-current reasonable qualifications or standards.

(b) The fact that the proposed transferee is a competitor of the franchisor or subfranchisor.

(c) The unwillingness of the proposed transferee to agree in writing to comply with all lawful obligations.

(d) The failure of the franchisee or proposed transferee to pay any sums owing to the franchisor or to cure any default in the franchise agreement existing at the time of the proposed transfer.

- 8. A provision that requires the franchisee to resell to the franchisor items that are not uniquely identified with the franchisor. This subdivision does not prohibit a provision that grants to a franchisor a right of first refusal to purchase the assets of a franchise on the same terms and conditions as a bona fide third party willing and able to purchase those assets, nor does this subdivision prohibit a provision that grants the franchisor the right to acquire the assets of a franchise for the market or appraised value of such assets if the franchisee has breached the lawful provisions of the franchise agreement and has failed to cure the breach in the manner provided in subdivision (c).
- 9. A provision which permits the franchisor to directly or indirectly convey, assign, or otherwise transfer its obligations to fulfill contractual obligations to the franchisee unless provision has been made for providing the required contractual services.

If the franchisor's most recent financial statements are unaudited and show a net worth of less than \$100,000.00, the franchisee may request the franchisor to arrange for the escrow of initial investment and other funds paid by the franchisee until the obligations, if any, of the franchisor to provide real estate, improvements, equipment, inventory, training or other items included in the franchise offering are fulfilled. At the option of the franchisor, a surety bond may be provided in place of escrow.

THE FACT THAT THERE IS A NOTICE OF THIS OFFERING ON FILE WITH THE ATTORNEY GENERAL DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION, OR ENDORSEMENT BY THE ATTORNEY GENERAL.

Any questions regarding this notice should be directed to:

Michigan Department of Attorney General G. Mennen Williams Building, 7th Floor 525 W. Ottawa St. P.O. Box 30212 Lansing, MI 48909 (517) 373-7117

TABLE OF CONTENTS

ITEM 1	ENVIRO-MASTER AND ANY PARENTS, PREDECESSORS AND AFFILIAT	ES 1
ITEM 2	BUSINESS EXPERIENCE	4
ITEM 3	LITIGATION	5
ITEM 4	BANKRUPTCY	6
ITEM 5	INITIAL FEES	6
ITEM 6	OTHER FEES	7
ITEM 7	ESTIMATED INITIAL INVESTMENT	14
ITEM 8	RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES	18
ITEM 9	FRANCHISEE'S OBLIGATIONS	22
ITEM 10	FINANCING	24
ITEM 11	FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS AND TRAINING	25
ITEM 12	TERRITORY	34
ITEM 13	TRADEMARKS	
ITEM 14	PATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION	
ITEM 15	OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISE BUSINESS	40
ITEM 16	RESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL	
ITEM 17	RENEWAL, TERMINATION, TRANSFER AND DISPUTE RESOLUTION	41
ITEM 18	PUBLIC FIGURES	46
ITEM 19	FINANCIAL PERFORMANCE REPRESENTATIONS	46
ITEM 20	OUTLETS AND FRANCHISEE INFORMATION	53
ITEM 21	FINANCIAL STATEMENTS	63
ITEM 22	CONTRACTS	63
ITEM 23	RECEIPTS	63

EXHIBITS

- A. Agent for Service of Process
- B. Franchise Agreement
- C. Form of Multi-Unit Development Addendum to Franchise Agreement
- D. Market Reservation and Deposit Agreement
- E. Promissory Note
- F. Operations Manual Table of Contents
- G. Confidentiality Agreement
- H. Current Franchisees
- I. Former Franchisees
- J. Financial Statements
- K. State Specific Addenda
- L. Franchisee Disclosure Questionnaire

ITEM 1

ENVIRO-MASTER AND ANY PARENTS, PREDECESSORS AND AFFILIATES

To simplify the language in this franchise disclosure document, the terms "we," "us," "our" and "Enviro-Master" means Enviro-Master International Franchise, LLC, the franchisor. "You," "your" or "Franchisee" means the person, corporation, partnership, limited liability company or other legal entity ("Legal Entity") who is considering buying the Franchised Business. If you are a Legal Entity, "you, "your" or "Franchisee" may include your shareholders, members, partners, officers and directors.

Predecessor, Parent and Affiliate.

Enviro-Master International Franchise, LLC is a North Carolina limited liability company, formed on December 30, 2013. Our predecessor is Enviro-Master Franchise, LLC, a North Carolina limited liability company formed on November 12, 2010 ("EMF"). EMF sold franchises from its inception in January 2011 until December 31, 2013, at which time its franchise business operations were transferred to us. EMF's principal business address is the same as ours. We conduct business as "Enviro-Master" and not under any other name.

We are an indirect subsidiary of EMP Enviro-Master HoldCo, LLC ("HoldCo"), a Delaware limited lability company. On March 29, 2022, HoldCo, an affiliate of Eagle Merchant Partners ("EMP"), an Atlanta-based private equity investment firm with its principal place of business at 3060 Peachtree Road, NW, Suite 360, Atlanta, GA 30305, acquired control of us through intermediate holding companies, including Enviro-Master Holdings, LLC. The creation of these holding companies did not result in a transfer of any Enviro-Master franchise agreements, trademarks or other intellectual property related to the operation of the System (defined below). We remain the franchisor of the System and provide all required support and services to franchisees under their franchise agreements.

Address and Agent for Service of Process.

Our official address and principal business address is 5200 77 Center Drive Suite 500, Charlotte, NC 28217. Our agents for service of process are shown on **Exhibit A** to this document.

Other Affiliates with Franchise Programs.

We currently have two affiliates that own and operate Franchised Businesses. First, our affiliate Enviro-Master Charlotte, LLC, with a principal business address of 5200 77 Center Drive Suite 500, Charlotte, NC 28217, currently owns and operates a Franchised Business in Charlotte, North Carolina ("EM Charlotte"). Additionally, our affiliate Enviro-Master Myrtle Beach, LLC, with a principal business address of 5200 77 Center Drive Suite 500, Charlotte, NC 28217, currently owns and operates a Franchised Business address of 5200 77 Center Drive Suite 500, Charlotte, NC 28217, currently owns and operates a Franchised Business in Myrtle Beach, South Carolina ("EM Myrtle Beach"). We consider these locations "Company" locations.

Through control with private equity funds managed by EMP we are also affiliated with the following franchise programs:

Code Ninjas, LLC ("Code Ninjas") is a franchisor of learning centers operating under the Code Ninjas trade name and business system providing child-focused educational programs

focused on computer programming skills. Code Ninjas has been franchising since November 2016 and, as of December 31, 2023, there were approximately 389 Code Ninjas franchised outlets operating in the United States. Code Ninjas' principal place of business is 2880 Broadway Bend Drive, Building #2, Pearland, TX 77584. Other than as described above, Code Ninjas has not offered franchises in any other line of business. Code Ninjas does not operate a Franchised Business.

TCB AmeriSpec, LLC ("AmeriSpec"), a Delaware limited liability company, franchises home and commercial inspection businesses under the AmeriSpec® mark. The principal address for AmeriSpec is 1650 Shelby Oaks Drive North, Suite 18, Memphis, Tennessee 38134. AmeriSpec and its predecessors began offering franchises in 1988. As of December 31, 2023, AmeriSpec had 157 franchises in the United States.

TCB Furniture Medic, LLC ("Furniture Medic"), a Delaware limited liability company, franchises furniture, wood, and cabinet restoration, repair, fabrication, and refinishing businesses under the Furniture Medic[®] mark. The principal address for Furniture Medic is 1650 Shelby Oaks Drive North, Suite 18, Memphis, Tennessee 38134. Furniture Medic and its predecessors began offering franchises in 1992. As of December 31, 2023, Furniture Medic had 124 franchises in the United States.

Renew Medic Franchising, LLC ("Renew Medic"), a Delaware limited liability company, franchises specialty mitigation and restoration businesses that perform residential and commercial cabinet repair, restoration and renewal services (primarily associated with the disaster restoration industry) under the Renew Medic[™] trademarks. The principal address for Renew Medic is 3060 Peachtree Road, Suite 970, Atlanta, Georgia 30305. As of December 31, 2023, Renew Medic did not yet have any franchises in the United States.

TCB Services Ltd. ("TCB Canada") offers franchises in Canada. The principal address for TCB Canada is 7181 Woodbine Avenue 238, Markham, ON, Canada. As of December 31, 2023, there were approximately 88 franchises in Canada under the AmeriSpec® and Furniture Medic® trade names and trademarks serving residential and commercial customers throughout Canada.

TCB Group Holdings Limited ("TCB UK") offers franchises in Great Britain. The principal address for TCB UK is 10 Temple Back, Redcliffe, Bristol BS1 6FL, United Kingdom. As of December 31, 2023, there were approximately 11 franchises in Great Britain that operate using the Furniture Medic® trade names and trademarks.

AmeriSpec, TCB Canada, and TCB UK have never offered franchises in any line of business (except as provided above), have never conducted a business of the type you will operate, and do not provide products or services to our franchisees.

The Business We Offer.

As a Franchisee, you will operate a business that specializes in the provision of restroom hygiene, drain line management, window cleaning, power washing, paper, and chemical products and other products and services approved by the franchisor or added to the Operations Manual (the "Manual"), to customers located in the territory described in Attachment A to your Franchise Agreement (the "Territory"), which include restaurants, including quick service and traditional, hotels, schools, and other commercial establishments. We also may authorize you to offer and sell additional products and services, which may involve your payment of additional fees.

As our franchisee, you will conduct business under the service mark "Enviro-Master" and any other identifying marks, trade names, logos and symbols that we use now, or that we later develop (the "Marks"), and use our unique system for the establishment, development and operation of a Franchised Business (the "System"). The System is characterized by a number of features including uniform systems and techniques for operation; software and computer programs; our selection of approved products that you may utilize, and certain products that we may require you to buy from us ("Approved Products"); and record keeping and reporting, personnel management, purchasing, sales promotion and advertising; customer service standards; and any guidelines, standards, specifications, rules, procedures, policies, methods, requirements, and directives we establish, including without limitation, our standards and specifications as to operations and equipment (the "Standards") set forth in the Manual and otherwise in writing. The Manual includes the document called "Operations Manual" and all other written, electronic, video, and audio recorded policies, procedures, techniques, memos, bulletins, newsletter, forms, guidelines, and other materials prepared by us in connection with the System or to assist you in the operation of your Franchised Business. We may change, improve, add to, and further develop the elements of the System from time to time.

The form Franchise Agreement we currently offer (**Exhibit B** to this disclosure document) may have terms different from the various forms of agreement we or our predecessor has used in the past. We reserve the right to change the form and terms of the Franchise Agreement in the future.

We may, in our sole discretion, offer you the opportunity to purchase multiple Franchised Businesses pursuant to a Multi-Unit Development Addendum ("MUDA") (the current form of which is attached as **Exhibit C** to this disclosure document). Currently, there is no maximum number of Franchised Businesses that may be signed in conjunction with a MUDA. If you sign a MUDA, you will simultaneously sign all of the Franchise Agreements granted with the MUDA. If you do not sign a MUDA, you will have no right to develop or operate a Franchised Business in more than one Territory unless you sign additional Franchise Agreements. If you purchase a single Franchised Business, you may operate out of a service vehicle, so long as you maintain a warehouse. If you purchase more than one Franchised Business pursuant to a MUDA, you must have a service vehicle, as well as a brick and mortar location that has both warehouse and office space. If you sign a MUDA, there may be efficiencies of cost, since some fees are only charged once. There may be other differences between purchasing a single Franchised Business or multiple Franchised Businesses pursuant to a MUDA, which are described within this disclosure document.

The Market.

We believe that the market for our services is well developed, broad, and expanding. Potential customers for our services include restaurants, hotels, schools, malls, convenience stores, drug stores, and other public buildings.

The Competition.

If you open a Franchised Business, your competition will include businesses offering restroom hygiene, window cleaning, power washing, paper, and chemicals, and companies that provide products used by commercial customers that do their own cleaning. Our biggest competitors are Cintas and Airmaster.

Prior Business Experience; Other Businesses.

Enviro-Master and our predecessor, EMF, has been offering franchises since January of 2011. EMF operated an Enviro-Master business in Charlotte, North Carolina from 2009 until August 2013, at which time this unit was transferred to a franchisee. In 2023, the Franchised Business in Charlotte, North Carolina became the EM Charlotte franchise. Neither we nor any affiliate has offered franchises in, or engaged in, other lines of business except for the operation of Code Ninjas, EM Charlotte and EM Myrtle Beach, and the sale to our franchisees of the products and services described in this disclosure document.

Applicable Regulations.

You must comply with federal, state, and local health and environmental safety regulations concerning the proper handling of cleaning fluids, and other products used in the Franchised Business. In addition, you will have to comply with federal, state, and local laws and regulations applicable to most businesses generally like workers' compensation. You will also have to comply with federal, state and local laws and regulations that are specific to this Franchised Business. For example, there may be local laws that require you to secure a license or permit in order to advertise or provide the products and services offered by your Franchised Business. Some states, like California, may require you to secure a pest control license in order to advertise certain products and services. You should investigate the application of these laws further.

Compliance with these laws and regulations, as they may be amended from time to time, can increase your operational costs and affect your bottom line.

ITEM 2

BUSINESS EXPERIENCE

Tod Bierling: President and Chief Executive Officer

Mr. Bierling has served as President and Chief Executive Officer of Enviro-Master International Franchise, LLC in Charlotte, North Carolina since December 2022. From January 2021 until December 2022, Mr. Bierling served as President and Chief Operating Officer of Enviro-Master International Franchise, LLC in Charlotte, North Carolina. From December 2016 to December 2020, Mr. Bierling served as Chief Operating Officer of Enviro-Master International Franchise, LLC in Charlotte, North Carolina. From December 2016 to December 2020, Mr. Bierling served as Chief Operating Officer of Enviro-Master International Franchise, LLC in Charlotte, North Carolina.

Stephen Krol: Chief Financial Officer

Mr. Krol has been Chief Financial Officer of Enviro-Master International Franchise, LLC in Charlotte, North Carolina since April 2023. From September 2020 to January 2023, Mr. Krol was Chief Financial Officer for New Market Waste Solutions, LLC in Charlotte, North Carolina. From July 2017 to September 2020, Mr. Krol was Vice President of Finance for SentryOne, LLC in Charlotte, North Carolina.

Steven Richey: Chief Revenue Officer

Mr. Richey has been Chief Revenue Officer of Enviro-Master International Franchise, LLC in Charlotte, North Carolina since June 2022. From November 2020 to May 2022, Mr. Richey was Vice President of Sales of Enviro-Master International Franchise, LLC in Charlotte, North Carolina. From October 2013 to October 2020, Mr. Richey was Regional Sales Manager for RSM US LLP in Atlanta, Georgia.

Larry Rushing III: Executive Vice President of Development

Mr. Rushing has been Executive Vice President of Development of Enviro-Master International Franchise, LLC in Charlotte, North Carolina since January 2022. From October 2018 to December 2021, Mr. Rushing served as Director of Franchise Development of Enviro-Master International Franchise, LLC in Charlotte, North Carolina.

Marty Hulse: Chief Operating Officer

Mr. Hulse has been Chief Operating Officer of Enviro-Master International Franchisee, LLC in Charlotte, North Carolina since February 2024. From October 2023 to February 2024, Mr. Hulse was Vice President of Operations for Vestis Uniform and Workplace Supplies (formally Aramark Uniform Services) in Charlotte, North Carolina. From April 2017 to October 2023, Mr. Hulse was Group Vice President of Operations for Aramark Uniform Services in Charlotte, North Carolina.

Carson Suppe: Director of Franchise Development

Mr. Suppe has been Director of Franchise Development of Enviro-Master International Franchise, LLC in Charlotte, North Carolina since September 2022. From January 2017 to September 2022, Mr. Suppe served as Senior Franchise Development Representative for Acti-Kare, Inc. in Tampa, Florida.

ITEM 3 LITIGATION

Seattle Sanitation Services, Inc., et al. vs. Enviro-Master International Franchise, LLC, et al, Superior Court of the State of Washington for the County of King, Case Number: 18-2-54869-1 SEA (King County, Washington, filed November 16, 2018). On November 17, 2018, we were notified that a lawsuit was filed against us, but had not been formally served on us, seeking a rescission of the Franchise Agreement and alleging violations of Washington State's Franchise Investment Protection Act. While we believed the case was entirely without merit, we agreed to settle this dispute by repurchasing the franchisee's territory for approximately one half of the Initial Franchise fee rather than incur litigation costs which would have been significantly greater. The action was dismissed on January 30, 2019 without ever having been served on us by the Franchisee.

Enviro-Master International Franchise, LLC v. Pro-Hygiene, LLC, et al., United States District Court for the Western District of North Carolina, Civil Action File No. 3:23-cv-00444. We filed this lawsuit on July 21, 2023 against a former franchisee and its guarantors seeking injunctive relief and damages for the former franchisee's failure to remove our trademarks and equipment following termination of the franchise agreement. Our claims included breach of contract,

trademark infringement, unfair competition and false designation under federal law, and unfair and deceptive trade practices under North Carolina law. Shortly after being served and before filing an answer, the defendants agreed to settle the case. On September 13, 2023, the parties entered into a settlement agreement whereby the defendants agreed to permanently remove our trademarks from their equipment, replace various equipment parts, pay us \$48,700 and assign us certain accounts receivables. In exchange for these terms and mutual releases, the parties dismissed the case with prejudice on September 19, 2023.

There is no other litigation that is required to be disclosed in this Item.

ITEM 4

BANKRUPTCY

No bankruptcy information is required to be disclosed in this Item.

ITEM 5

INITIAL FEES

Initial Franchise Fee

The initial franchise fee for a single Franchised Business is \$50,000, due upon execution of the Franchise Agreement. In addition, before you open your Franchised Business, you will need to purchase your initial inventory of supplies and products, which we estimate to be approximately \$5,500 to \$7,500. You must also purchase equipment (deep cleaning machine, mixing equipment, and power washer) at an estimated cost of \$7,900 to \$8,900. The inventory and equipment purchases are not refundable. Please see Item 10 for financing options for certain large equipment purchases.

We may, in our sole discretion, offer a discounted initial franchise fee for those franchisees who complete their due diligence (i.e., they have completed the franchisee application process and are approved to become a franchisee, and they have received the disclosure document and all waiting periods for execution have expired) prior to attending a Discovery Day and are approved to sign their Franchise Agreement and/or MUDA and pay their initial franchise fee at their Discovery Day. Discovery Day is an event held periodically that allows prospective franchisees to meet members of our team and to learn more about the Enviro-Master brand.

If you choose to sign a MUDA, then the initial franchise fee will be as follows: \$40,000 for the second Franchised Business, \$30,000 for the third Franchised Business, and \$20,000 for the fourth and any subsequent Franchised Businesses, payable in full upon the execution of the Franchise Agreement(s) and MUDA. The initial fees are deemed fully earned by us once paid and we will not refund you any part of the initial fees, including if you fail to develop any of the Franchised Businesses by the deadlines set forth in the Franchise Agreement(s) or if any Franchise Agreement(s) subject to the MUDA are terminated.

Payment and Other Fees

We will not accept any initial franchise fee payment from you unless and until we have approved you and you have signed the Franchise Agreement. Upon the signing of the Franchise Agreement, the initial franchise fee is fully earned and non-refundable. We will provide our initial training program ("Initial Training Program") to up to two individuals. If you would like other individuals to attend the Initial Training Program, you must pay us \$2,500 per additional attendee. Per Section 12.7 of the Franchise Agreement, you must maintain at least one sales associate for each Franchise Agreement signed. If we approve, you may be the sales associate for the first Franchised Business that you purchase. If you sign additional franchise agreements, we may require you to complete our Initial Training Program again for those franchise agreements.

You will be required to utilize our internal employees who assist franchisees in identifying potential new clients and setting sales calls (the "Inside Sales Department") for you and/or your sales associates for the first 90 days that you are in business. The cost of the inside sales activity is \$380 per week or \$4,560 for your first 3 months of operation.

We charge a \$5,000 refundable market deposit if you enter into a Market Reservation Agreement before signing a Franchise Agreement. A form of the Market Reservation Agreement is attached as **Exhibit D**. If you enter into a Franchise Agreement, the \$5,000 deposit will be credited toward your initial franchise fee. If you do not enter into a Franchise Agreement, the \$5,000 deposit will be refunded to you within thirty (30) days of your request for a refund.

Incentive Referral Program

From time to time, we may utilize incentive referral programs for our existing franchisees to refer franchise prospects to us. Under these programs, we may pay cash compensation or provide other benefits and inducements for the referral of qualified franchise prospects. The compensation or benefit will only be paid if the franchise prospect enters into a franchise agreement with us and the franchisee prospect is not represented by a broker. These referral programs may be altered, modified, suspended, or terminated at any time. This referral program is available in all states except the State of Washington.

ITEM 6

Name of Fee ¹	Amount	Due Date	Remarks
Royalty Fees	6% of Gross Revenues ²	Weekly	See Note 2 for definition of Gross Revenues. We may change the payment
	or the minimum royalty fee defined in Attachment B of the Franchise Agreement ³ , whichever is greater.		due date in the Manual or in a written notice to you. See Note 4 for payment procedure.
Admin/Service Fees	5% of Gross Revenues or \$125, whichever is greater.	Weekly	We provide you with most of the administrative services necessary to operate the Franchised Business (Item 11; see Section 4.3 of the Franchise Agreement). For these business services,

OTHER FEES

Name of Fee ¹	Amount	Due Date	Remarks
			you must pay us Admin/Service Fees. Your Admin/Service Fees equal 5% of Gross Revenues (see Note 2), or \$125, whichever is greater. We may make changes in the business services provided to you, and may reasonably reduce or increase the Admin/Service Fees based upon costs saved or additional costs incurred by us.
National Advertising Fee	2% of Gross Revenues	Weekly	You must contribute the National Advertising Fee to the National Advertising Fund. See Item 11.
Local Marketing Expenditure	2% of Gross Revenues	Weekly	You must contribute the Local Marketing Expenditure. See Item 11.
National/Region al Accounts Fee ⁵	Varies	When billed by us	We charge you a commission of 4% of gross amount of revenue generated from National/Regional Accounts during the first 12 months of each account's service agreement, and after the first 12 months, a commission or ongoing management fee of 3% of Gross Revenue generated by each account. These are the current commissions we charge, but they are subject to change.
Technology Fee	We currently charge \$250 per month	Monthly	The Technology Fee, billed monthly, covers, in part, our costs for procuring and implementing the Approved Software and any on-going updates to it (Item 11). We reserve the right to increase this fee over time. If you sign a MUDA for Franchise Agreements that cover adjacent Territories, then we will only charge you the Technology Fee for your first Franchised Business.
Consulting Fee	We currently charge all travel expenses plus \$150 per day for onsite consulting fees. We reserve the right to update or change this amount.	As incurred	We provide individual assistance at your request, subject to personnel availability.

Name of Fee ¹	Amount	Due Date	Remarks
Additional Training Fee	We currently charge \$325 per day, per person	As incurred	We provide supplemental and refresher training, as we determine necessary at times and locations designated by us.
Related Party Transfer Fee	\$2,500	At the time the transfer is approved	If you transfer the entire interest in the Franchised Business (i) to a corporation or limited liability company, owned wholly by you and formed for convenience of ownership, (ii) to an immediate family member (if you are an individual), or (iii) if you are a corporate entity, to your existing principals in the same ownership percentage as the principals owned the corporate entity, we will charge a Related Party Transfer Fee of \$2,500.
Franchise Agreement Transfer Fee	\$10,000	Before transfer is effective	If you transfer a minority interest in your Franchised Business, we will charge a Transfer Fee of \$10,000.
Franchise Resale Fee	\$50,000	Upon signing Franchise Agreement	When a territory is sold or franchisee has a change in control, the buyer will sign a new 5-year initial term Franchise Agreement and pay the Resale Fee.
Supplier Review Fee	Up to \$1,000	30 days after billing	Fee not to exceed our reasonable costs of inspecting and testing. We may require you to pay a fee if we determine that it is necessary to inspect the vendor's facilities or conduct tests to approve their product(s).
Products You Buy From Us	Varies	Upon Ordering	You must purchase certain products from us or our designated suppliers. We and our affiliates may earn a profit on the sale of products to you. See Item 8.
Renewal Fee	10% of the then- current initial franchise fee	Before renewal is effective	
Audits	Actual cost of audit, plus overdue amounts and interest	30 days after billing	Cost currently estimated at \$1,000 per day. If audit reveals that you have under- reported Gross Revenues ² by 2% or more, you must pay for the cost of the audit plus travel costs. Interest accrues from the due date of late payment.

Name of Fee ¹	Amount	Due Date	Remarks
National or Regional Conference Fee	Reasonable fees for attendance, plus costs of expenses such as food, travel and lodging	As incurred	Attendance at the National and Regional Conferences is mandatory for you (including your General Manager, if applicable), your operations manager and all sales persons.
Missed Conference Charge and Resulting Additional Training	\$3,000 per person failing to attend, plus actual cost to send trainers to train your staff	As Incurred	If any of your employees fails to attend a National or Regional Conference, we will charge you \$3,000 per each person not attending. We also may send our trainers to train you and your staff on training items covered at that meeting and require you to reimburse us for the trainers' wages and all travel expenses of the trainers.
Late Payment Interest	1.5% per month or the highest rate allowable by law, whichever is lower, on amounts past due	When payment is past due	Interest accrues on amounts from the date due until paid.
Insufficient Funds Charge	\$100	At time funds are rejected	
Purchase of products from alternate suppliers	12% of the dollar amount purchased from an approved supplier that is not us	Upon purchase	Franchisee purchases products from alternative suppliers, once approved.
Taxes and Other Payments	Varies	30 days after billing	See Note 7.
Insurance	Varies	On demand	If you fail to purchase required insurance, you must reimburse our cost to us if we purchase insurance on your behalf.
Indemnification	Varies	On demand	You must indemnify us and our affiliates and personnel from all losses and expenses (as defined in Section 9.4 of the Franchise Agreement) incurred in any claim involving acts, errors or omissions by you or your employees, agents or contractors, unless the claim results directly from your

Name of Fee ¹	Amount	Due Date	Remarks
			having operated in compliance with our written instructions.
Deficiencies	Varies	On demand	If you fail to correct any deficiencies discovered in an inspection, we may make the corrections at your expense and charge you a Repeated Inspection Fee (as described below).
Repeated Inspection Fee	\$500/per person/per day, plus all travel costs and expenses of inspection	On demand	If you fail an inspection and we or our representatives return to inspect you to confirm you have remedied the deficiency, you must pay us the cost of the inspection, including the travel and living expenses of our representatives.
Non- Compliance Fee	Up to \$500 each time you default and for each week (or portion thereof) that the default continues	As stated in our notice	Only required, at our option, if you have committed an event of default under the Franchise Agreement.
Temporary Management Fee	30% of Gross Revenues ¹ , plus actual fees and expenses	As incurred	We may, but are not required to, appoint a temporary manager of the Franchised Business if we reasonably determine that a material default of the franchise agreement exists or is imminent. See Franchise Agreement, Section 13.13.
Inside Sales Activity	\$380 per week for the first 90 days after you are open	Weekly	You will be required to utilize our Inside Sales department to set sales appointments for you and/or your sales associates for the first 90 days that you are in business. If you sign a MUDA for Franchise Agreements that cover adjacent Territories, then we will only charge you the Inside Sales activity fee for your first Franchised Business.
Local Digital Marketing Campaign	\$350/month	Monthly	You will be required to utilize our approved vendor and the current cost is \$350 per month payable to us for pay-per-click advertising and search engine optimization. At your election, you may spend more than the monthly required amount. We collect the \$350 monthly fee from all franchisees and remit it directly to the vendor. If you

Name of Fee ¹	Amount	Due Date	Remarks
			sign a MUDA for Franchise Agreements that cover adjacent Territories, then we will only charge you the Local Digital Marketing fee for your first Franchised Business.
Sales Training Program – Recruiting	\$3,000 for assistance in recruiting one sales associate role. This fee is subject to change.	As incurred	If you elect to participate in a sales training program, we will provide you with assistance in recruiting sales associates. If you participate in the Sales Training Program, we will withdraw fees from your account in the same manner as royalties.
Sales Training Program – Mentoring	\$750 per four-week period, per Sales Associate. This fee is subject to change.	As incurred	If you elect to participate in a sales training program for your Sales Associates(s), we will provide you with mentoring for your Sales Associates. If you participate in the Sales Training Program, we will withdraw fees from your account in the same manner as royalties.

Note 1. Unless otherwise expressly indicated, all of the fees listed in Item 6 are: nonrefundable; imposed by us; payable to us; and uniformly imposed. If you purchase more than one Franchised Business pursuant to a MUDA, the fees indicated in the chart above are the fees charged and/or incurred for each Franchised Business, unless otherwise indicated in the chart. We reserve the right to vary fees in our discretion. Currently, there are no cooperatives, and therefore, no voting by members of cooperatives.

Note 2. Gross Revenues. "Gross Revenues" is defined as all revenues and income earned, received or accrued by, as a result of or on behalf of the Franchised Business from any source, whether for cash, credit or barter, including without limitation, fees charged to customers such as "service fees," "cleanout fees," "delivery fees," "installation fees," and "installation deposits." "Gross Revenues" exclude sales taxes you pay to a governmental agency, and revenues not generated by the operations of the Franchised Business (e.g. investment income). You may deduct refunds, discounts or allowances from Gross Revenues only if you included such amounts in Gross Revenues, and only if you comply with our policies for reporting and taking the deductions.

Note 3. Minimum Royalties. You will pay us the greater of (i) the Royalty Fee equal to 6% of Gross Revenue, or (ii) the "Minimum Royalty" due under the Franchise Agreement. At the end of each calendar year of the term of the Franchise Agreement, we will invoice you for the difference between the annualized Minimum Royalty and the Royalty Fees that you actually paid that calendar year. The current annualized Minimum Royalty for the initial 5-year term of the Franchise Agreement for a single Franchised Business is as follows:

Year of Operation	Minimum Annual Royalties
1	None
2	\$6,000
3	\$12,000
4	\$18,000
5	\$24,000

These fees may be subject to change.

Note 4. Payment Procedures.

When Due. Each week, we will determine your franchisee fees due (royalties, admin/service and advertising) based upon your reports or on our own information documenting your Gross Revenues. We will provide an invoice to you showing our calculations of the amount due and the amount to be deducted from revenues we collect on your behalf. If we have not received sufficient customer collections on your behalf to offset our weekly franchise fee, we will then deduct the amount shown due on your weekly fee statement via an electronic funds transfer debit to your bank account (as described in (b) below) to pay all monies due.

Method of Payment. We will have the right to receive payment of your fees by electronic funds transfers from your bank account to ours. You must provide us with your bank's name, address and account number, along with a voided check from the account. You must sign and give to us and your bank all documents necessary for us to commence making electronic funds transfers, and for us to continue to do so throughout the term of your franchise. You must make deposits to your account as needed to make funds available for transfer to us, and may not close that account without our prior written consent.

Late Payments. If any payments are received by us after the date due (i) you must pay to us, monthly and in addition to the amount due, interest in the sum of 1.5% per month of the amount due or the highest rate allowable by law, whichever is lower, from the date due and until paid in full; and (ii) we can terminate the Franchised Business as provided under Section 17.1 of the Franchise Agreement.

Note 5. National/Regional Accounts Fee. You will pay a monthly commission for all accounts sold and managed by our accounts team. During the first 12 months, you will pay a 4% commission based on Gross Revenue generated by our accounts team for you. After the first 12 months, you will pay a 3% commission on Gross Revenue generated by our accounts team for you. During the term of the service agreement, if our accounts team upsells an additional product or service, then you will pay a 4% commission on the Gross Revenue generated from the additional product or service for each month for the following 12 months of the service agreement. These commissions are subject to change during the term of your service agreement.

Note 6. Refurbishment. Refurbishments may include structural changes, installation of new equipment and signs, remodeling, redecoration and modifications to existing improvements. We are unable to estimate your costs for future Refurbishments which will vary from Franchised Business to Franchised Business based on a number of factors like: (i) the market where your

Franchised Business is located; (ii) the size of your Franchised Business; (iii) when your Franchised Business was last refurbished, if applicable; (iv) the amount of ENVIRO-MASTER[®] franchise system changes since the last refurbishment; and (v) the overall condition of your Franchised Business site and equipment.

Note 7. Taxes. You must pay us or our affiliates (i) all sales taxes, corporate taxes, and any similar taxes paid by us on your behalf, imposed on us, or required to be collected by us on account of products or services we furnish to you (through sale, lease, or otherwise) or on account of our collection of any fee related to the Franchise Agreement; (ii) all franchise or similar taxes, whether based on gross receipts, gross revenues, Royalty Fees, advertising contributions, or otherwise, imposed on, required to be collected by, or paid by us; (iii) all marketplace facilitator or similar taxes imposed on, required to be collected by, or paid by us in connection with your use of our website, internet sites, applications, or online ordering platforms; (iv) all other amounts we pay or must pay for you for any reason; and (v) any other fees or expenses that we are entitled to collect from you.

ITEM 7

ESTIMATED INITIAL INVESTMENT

YOUR ESTIMATED INITIAL INVESTMENT: A SINGLE FRANCHISED BUSINESS

Type of Expenditure	Amount	Method of Payment	When Due	To Whom Payment is to be Made
Initial Franchise Fee ¹	\$50,000	Lump Sum	At Signing	Us
Equipment ²	\$7,900 to \$8,900	Lump Sum	Before Opening	Us
Equipment & Vehicle Lease ²	\$1,250	Monthly	Before Opening	Suppliers
Vehicle Down Payment ²	\$5,000	Lum Sum	Before Opening	Suppliers
Inside Sales Activity ³	\$4,560 (\$380/week)	Monthly as incurred	During Training	Us
Local Digital Marketing ⁴	\$1,050(\$350/month)	Monthly	During Training	Us
Opening Inventory ⁵	\$5,500 to \$7,500	Lump Sum	Before Opening	Us
Deposits and Prepaid Expenses ⁶	\$500 to \$1,000	Lump Sum	Before Opening	Suppliers
Permits and Licenses ⁷	\$200 to \$500	Monthly	Before Opening	Governmental Agencies
Insurance ⁸	\$2,300 to \$3,800	As Incurred	Before Opening	Insurance Carriers
Travel and Living expenses while training ⁹	\$4,000 to \$5,200	As Incurred	Before Opening	Airline, Hotel and Others

Type of Expenditure	Amount	Method of Payment	When Due	To Whom Payment is to be Made
Rent ¹⁰	\$500 to \$1,000	As Incurred	Upon Signing Lease + Monthly	Landlord
Professional fees	\$500 to \$2,500	As Incurred	As Required	Your Accountant, Attorney or other Professionals
Additional Funds ¹¹	\$13,000 to \$128,000	As Incurred	As Required	Employees, Vendors, etc.
Total	\$96,260 to \$220,260			

YOUR ESTIMATED INITIAL INVESTMENT: MULTIPLE FRANCHISED BUSINESSES UNDER A MUDA

Type of Expenditure	Amount	Method of Payment	When Due	To Whom Payment is to be Made
Initial Franchise Fee ¹	\$90,000 – \$160,000	Lump Sum	At signing of Franchise Agreement	Us
Initial Equipment, Signage and Furniture Package ²	\$15,800 to \$16,800	As incurred	Before Opening	Us
Equipment & Vehicle Lease ²	\$1,250	Monthly	Before Opening	Suppliers
Vehicle Down Payment ²	\$5,000	Lum Sum	Before Opening	Suppliers
Inside Sales Activity ³	\$4,560 (\$380/week)	Weekly	During Training	Us
Local Digital Marketing Campaign ⁴	\$1,050(\$350/month)	Monthly	During Training	Us
Opening Inventory ⁵	\$10,500 to \$16,000	Lump Sum	Before Opening	Us
Deposits and Prepaid Expenses ⁶	\$500 to \$1,000	Lump Sum	Before Opening	Suppliers
Permits and Licenses ⁷	\$200 to \$500	Lump Sum	Before Opening	Government Agencies
Insurance ⁸	\$2,300 to \$3,800	As Incurred	Before Opening	Insurance Carriers
Travel and Living Expenses while Training ⁹	\$4,000 to \$7,600	As Incurred	Before Opening	Airline, Hotel and Others

Rent ¹⁰	\$2,400 to \$4,000	As Incurred	Upon signing lease + Monthly	Landlord
Professional Fees	\$500 to \$2,500	As Incurred	As required	Your Accountant, Attorney or other Professionals
Additional Funds (12 Months) ¹¹	\$127,000 to \$305,000	As incurred	As required	Employees, Vendors, etc.
TOTAL ¹²	\$265,060 to \$529,060			

NOTES TO TABLES

1. See Item 5.

2. For a single Franchised Business, you will need one or more personal computers, and a handheld computer device (e.g. cellphone). For two or more Franchised Businesses, you will need office furniture, one or more personal computers, handheld computer device (e.g. cellphone), office telephone(s) with voicemail, shelves and files (see Item 11). For one or more Franchised Businesses, you will need to buy or lease at least one truck or van that meets our Standards, which may require a down payment as well as monthly payments. You must outfit the required vehicle and install our logo package, all as specified in our Manual or other written materials. For one or more Franchised Businesses, you will need a deep cleaning machine for Sani Scrub service, a mixing machine for hand soap, a power washer, a minimum of 2 electrostatic sprayers and any other items necessary to provide proscribed services to clients. All of these items are included in this estimate.

3. See Items 5 and 6. The cost of this is \$38 per appointment, or \$380 per week, which we estimate (but do not guarantee) will yield 10 appointments per week. If you sign a MUDA for Franchise Agreements that cover adjacent Territories, then we will only charge you the Inside Sales activity fee for your first Franchised Business.

4. See Item 6. If you sign a MUDA for Franchise Agreements that cover adjacent Territories, then we will only charge you the Local Digital Marketing fee for your first Franchised Business.

5. If you are a new franchisee, you will need to purchase initial inventory from us. All products used in the Franchised Business must meet our standards and specifications.

6. The estimate includes deposits that may be refundable to you at a later time.

7. You are responsible for obtaining all permits and license necessary for you to operate your business. You can obtain information from your local municipality, county and state authorities regarding the specific legal requirements for business licenses and related types of expenses in your local area.

8. You must obtain and maintain insurance coverage as we direct. Insurance costs may vary in different localities. The estimate above is for: (a) one year of an all risks policy, providing liability insurance coverage of \$1,000,000 per occurrence and \$2,000,000 in the aggregate; (b) workers' compensation insurance; (c) automobile liability insurance coverage of \$2,000,000, naming us as Additional Insured; (d) umbrella liability policy in the amount ranging from of \$1,000,000 to \$4,000,000 in the aggregate and each occurrence, depending on the revenue of the Franchised Business; (e) property insurance with a coverage amount of at least \$25,000; and (f) pollution insurance policy in the amount of \$500,000 per occurrence. We may require you to get and keep additional types of insurance and coverage, as provided under the Franchise Agreement.

9. This estimate does not include training fees which would apply if you decide to send more than two people to initial training. The estimate does not include costs of transportation, which would vary according to distance and mode of transportation. You should contact airlines or other carriers for information about fares from your area to Charlotte, North Carolina where the 10 days of initial training generally occurs, and where the field sales training portion of initial training generally occurs.

10. If you purchase a single Franchised Business, you must rent or purchase storage space that has adequate space to store a power washing trailer, with access to power and water. Your storage space should be approximately 400 square feet. The estimate for a single Franchised Business assumes monthly rent plus one month's rent as a deposit to your landlord. If you purchase two or more Franchised Businesses, you must rent or purchase office space that has adequate access for large trucks to make deliveries and is connected to sufficient warehouse space for the products and supplies your Franchised Businesses will use. The office likely will be located in a commercial or industrial office center. You may not operate a Franchised Business out of your home. If you purchase two or more Franchised Businesses, you will incur costs to set up an office, including rent and the cost to furnish the office in accordance with the Standards. Your office should also contain approximately 2,500 combined square feet of office and warehouse space, with sturdy shelves for storing product inventory. The estimate for two or more Franchised Businesses assumes monthly rent plus one month's rent as a deposit to your landlord. You may be able to negotiate multiple months of free rent depending on your market and the leasing environment for warehouse and mixed-use space.

11. These are estimates only of the ranges of additional funds you may need during the initial phase of your Franchised Business. We estimate that, in general, you may require operational funds for approximately twelve months after starting the Franchised Business. This estimate does not include the owner's salary, but does include an estimate of \$100,000 per year as a salary for a General Manager, if applicable.

12. All figures in the above table are estimates based on our experience franchising Franchised Businesses. Your actual investment, expenditures and initial cash outlay will vary depending on many factors, including the warehouse size and location, whether you choose to use a van for your Franchised Business and whether you follow the Standards. The fees you pay to us are not refundable. The fees you pay to a third-party vendor may be refundable, in the sole discretion of the vendor. We do not offer direct or indirect financing for these items.

ITEM 8

RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES

Required Purchases.

You are required to obtain certain products and services from Enviro-Master or suppliers that we designate (an "Approved Supplier"), including restroom hygiene chemicals, drain line management products, paper products, uniforms, marketing materials, and any other items which are added to the Enviro-Master unit including initial opening inventory. We will periodically designate, through the Manual or other written notice, the products which must be purchased from Approved Suppliers, and will notify you of the Approved Supplier for each product. We may keep lists of Approved Suppliers for certain products or services, and we may update the lists and issue them to you. You may not directly or indirectly purchase products from unapproved suppliers, except in emergency circumstances and with our prior written consent.

You must purchase or lease at your expense the following: supplies, products, inventories, fixtures, and/or, equipment relating to the establishment and operation of your Franchised Business. You must also purchase the following on-going business services: (1) provision of Approved Software (see Item 11); (2) invoicing of all credit/charge accounts; (3) preparation of monthly statements of account; and (4) centralized ordering of supplies (collectively, the "Business Services").

Other than office supplies, and the items listed in the table below, all of these products and services must be purchased from us, or Approved Suppliers, including the Initial Equipment Package (described below), the Business Services, restroom hygiene chemicals, drain line management products, paper products and product inventory. The Initial Equipment Package consists of: (1) Mixing Tank, (2) Mixer, (3) Kaivak floor Scrubbing machine (4) 3 Electrostatic spray guns, (5) 1 backpack Sprayer and (6) power washer.

We estimate that the proportion of required purchases and leases to the total purchases and leases necessary to establish and open the Franchised Business is approximately 80%. Required purchases in the ongoing operation of the Franchised Business will depend upon many factors, including general economic conditions and the growth of your customer base. Your required purchases will escalate as your product sales increase with the growth of your Franchised Business. Except as specified for inventory, administrative services, sales training program and ongoing product purchases, as well as certain accounting, bookkeeping and financial services including payroll, you can purchase the below items from any supplier you choose:

Computer Hardware	You must purchase (if you do not already own) a personal computer. You must also obtain and use the Hand-Held Device described in Item 11. We will provide you with specifications for the Hand-Held Device in the Manual and/or materials.
Computer Software	You need to purchase computer operating software for your personal computer, which will be at your expense.

	We will provide you with the Approved Software (described in Item 11), either directly or through a third-party provider, and well as any updates. The cost for the Approved Software and updates is covered through your monthly Admin/Service Fee described in Item 6.
Internet Access	You must obtain Internet access for business communications.
Vehicle	You are required to buy or lease a new truck or van that complies with our specifications. You must outfit it and install our logo package, all as specified in our Manual or other written materials.
Equipment	For a single Franchised Business, equipment includes a computer, deep cleaning machine, mixing equipment, printer/scanner/fax machine, and electrostatic sprayers. Conference room furniture and other office furniture as required. You will need to buy a power washer from our designated supplier. For two or more Franchised Business, equipment also includes a desk, telephone, voice mail, shelving and files, conference room furniture and other office furniture as required.
Opening Inventory	You must purchase your opening inventory from us. See Item 7.
Ongoing product and service purchases	You will buy all or nearly all of your products from us, as well as administrative services.
Insurance	You will need to purchase insurance from a carrier we approve.
Accounting Services	You will need to purchase accounting services from our designated vendor, currently Top 2 Bottom Business Solutions.

Required and Approved Suppliers.

As of the date of this franchise disclosure document, we are the only Approved Supplier for all products and supplies used in your business (other than office supplies and as otherwise described in the table above) such as restroom hygiene chemicals, drain line management products, and paper products and for the initial opening inventories. You must also purchase your Initial Equipment Package from us. As of the date of this franchise disclosure document, we are the only Approved Supplier of the Business Services.

Approval of Alternative Suppliers.

You may propose a new supplier for our consideration. We may keep a list of criteria for reviewing proposed new suppliers, and we may require inspections and performance reviews of proposed suppliers. We will approve or disapprove suppliers based on our criteria. We may approve new suppliers or revoke past approvals of suppliers upon reasonable written notice to you.

You may request in writing our review of a proposed supplier. If we determine that it is necessary to inspect the supplier's facilities or conduct tests, we may require you to pay a Supplier

Review Fee not to exceed its reasonable costs (see Item 6). We currently charge a fee of up to \$2,500 to you to cover our costs in reviewing and approving any alternative suppliers. We will attempt to provide you with written notification of a decision regarding the designation of a supplier which you have proposed within 60 days after receipt of your request, or within 30 days after completing any inspections or testing, whichever is later. We will not unreasonably withhold approval of a proposed supplier, but we may impose reasonable limits on the number of Approved Suppliers in our System at any time, and may designate ourselves and our affiliates as the sole suppliers of particular products. The sale of products is a significant source of income to Franchisor. Thus, if we approve a new supplier that you have proposed and you purchase products from that supplier, then you will pay to us a twelve percent (12%) mark-up fee or the mark up equal to that charged by us for like product, whichever is greater, on all such items purchased.

We may require that you use our Approved Supplier for the Local Digital Marketing Campaign that is mandatory for all franchisees, and we reserve the right to not approve any alternative supplier for this service because the campaigns are most effective if coordinated nationally, and because it will be unduly burdensome for any then-currently Approved Supplier to be required to coordinate with an alternative vendor that you suggest we approve.

Interest in Approved Suppliers.

Currently, neither Enviro-Master nor any officer has any interest in any company, nor do we have any affiliate that provides products or services to our franchisees. Our affiliate EM Capital Services, LLC has provided financial services and sold acquired customer lists to franchisees in the past, but it is not currently offering any of these services to new franchisees.

Issuance of Standards and Specifications.

We maintain written lists of specifications that we have formulated for products. We update our lists and issue the updated lists to you. We furnish our specifications to proposed designated suppliers on request, but only on a confidential basis. We may modify specifications upon reasonable written notice to you. We will consider your written request for a modification of a specification, if you explain the reason for the requested modification (or for the approval of any product we have not previously approved) and provide us with sufficient technical data to enable us to evaluate your request. We may perform inspections or tests to determine if any product meets our specifications. We will provide you with notification of approval or disapproval within 60 days after receipt of your request, or 30 days after completing any inspections or testing, whichever is later. We will approve a request if it is determined that a modified specification is appropriate or that any product meets its specifications.

Insurance.

You must purchase all insurance from a company specified by us, which must have an A.M Best rating of A-VII or better. All insurance will provide coverage as specified in the Manual and the Certificate of Insurance and policy will name us as an additional named insured including (1) Workers Compensation & Employers Liability; (2) Automobile Liability insurance (including Hired & Non-Owned Auto Liability); (3) Comprehensive General Liability Insurance, including products and completed operations and personal injury protection; (4) Umbrella Liability; (5) Property Insurance; (6) Employment Practices liability policy; (7) Pollution Insurance Policy; and (8) other insurance coverage as required by applicable law in the greater of the amount set forth

below or as required by the state/province in which the Franchised Business operates, all in forms satisfactory to us and in at least the following minimum amounts:

Commercial General Liability (Occurrence Form)

	—
General Aggregate (other than Prod/Comp	Ops Liability) \$2,000,000
Products/Completed Operations Aggregate	\$2,000,000
Personal & Advertising Injury Liability	\$1,000,000
Each Occurrence	\$1,000,000
Commercial Automobile Liability	
Combined Single Limit	\$1,000,000 each accident
Hired and Non-Owned Auto	\$1,000,000 each accident
Workers Compensation and Employer's Liabilit	<u>х</u>
Workers' Compensation	State Statutory Limits
Employer's Liability	
Bodily Injury by Accident	\$1,000,000 each accident
Bodily Injury by Disease	\$1,000,000 policy limit
Bodily Injury by Disease	\$1,000,000 each employee
<u>Umbrella Liability (to overlay General Liability, A</u>	Auto Liability and Employers Liability
<u>coverages)</u>	
Each Occurrence and Aggregate	\$1,000,000 (less than \$2mm revenue)
Each Occurrence and Aggregate	\$2,000,000 (\$2mm - \$3mm revenue)
Each Occurrence and Aggregate	\$3,000,000 (\$3mm - \$4mm revenue)
Each Occurrence and Aggregate	\$4,000,000 (greater than \$4mm revenue)

Employment Practices Liability

In the minimum amount of \$1,000,000, or such other minimum amount we may require from time to time, and with the maximum required retainage we designate from time to time.

Insurance must, to the extent available at reasonable and standard rates, as determined by us, in our sole discretion, cover your indemnification obligations to us under the Franchise Agreement.

Pollution Liability

Each Occurrence

\$500,000

Property Insurance

The minimum coverage amount of \$25,000, or such other amount that we designate from time to time.

Revenue from Franchisee Purchases.

We receive revenue from the sale of supplies, inventory and equipment to franchisees. During the year ending December 31, 2023, we received approximately \$10,675,000.00 from franchisee purchases from us, representing 46% of our total revenues of approximately \$23,166,000.00. Required purchases or leases are estimated to make up 54% of the franchisee's total initial investment and 15% of a franchisee's annual operating expenses. We may earn a profit on our sale of such products to you, provided the product is competitively priced.

We currently receive a rebate of \$900 per power washer from our third party Approved Supplier, which increases to \$1,300 per power washer after 40 units are sold within a single year. We reserve the right to contract with other third parties for supplies, inventory and equipment, and in such case, may agree that the third-party supplier will provide us with a rebate, provided the product is competitively priced. As of the date of this disclosure document, none of our affiliates derive any revenue, rebates or other material consideration based on the required purchases or leases from franchisees.

Cooperatives; Negotiated Prices; Material Benefits.

We do not have any purchasing or distribution cooperatives. We negotiate purchase arrangements with suppliers for the benefit of franchisees, including volume purchase pricing. If deemed in the best interest of the franchise system, we may discontinue providing any product or service to you by providing you with at least 30 days' written notice. If deemed in the best interest of the franchise system, we may periodically change the Business Services that we provide and reasonably adjust the service fees that you must pay for the Business Services. If we change the Business Services offered to you, or incur reduced or increased costs in providing Business Services, we may reduce or increase the service fees based on the reasonable costs saved, or additional costs incurred. If we discontinue any Business Services involving the reporting of Gross Revenues or the collection of royalty and other fees, you must comply with our reasonable reporting and collection procedures as provided in the Manual. We will strive to offer products and services which are competitively priced. We currently require and will continue to require that you purchase accounting and bookkeeping services from our designated vendor.

We do not provide any material benefit to you (such as renewal of a franchise) if you purchase particular products or services, or use particular suppliers.

ITEM 9

FRANCHISEE'S OBLIGATIONS

This table lists your principal obligations under the franchise and other agreements. It will help you find more detailed information about your obligations in these agreements and other items in this disclosure document.

	Obligation	Section in Franchise Agreement; MUDA	Item in Franchise Disclosure Document
(a)	Site selection and acquisition/lease	1.3 – 1.5 of Franchise Agreement; Section 1 of MUDA	7
(b)	Pre-opening purchases/ leases	3.1, 9.1, 12.5,12.7, 13.3, 13.11 & 14.1	5, 6 and 7
(c)	Site development and other pre-opening requirements	1.3 & 13.3 of Franchise Agreement; Section 2 of MUDA	7
(d)	Initial & ongoing training	12	6, 7 and 11
(e)	Opening	1.3 & 13.3 of Franchise Agreement; Section 2 of MUDA	7
(f)	Fees	2.3(f) and (g), 3, 4.4, 11.4(b), 11.5, 12.5, 12.8, 13.11, 13.12, 13.13(b), 14.3, 14.3(b), 17.1(m), 17.1(o), & 22.2 of Franchise Agreement; Section 1 of MUDA	5, 6, 7 and 11
(g)	Compliance with standards & policies/ operations manual	1.2, 4.1(b)-(c), 7, 13.1-13.12, 14.1 & 14.2 of Franchise Agreement; Section 6 of MUDA	1, 8 and 11
(h)	Trademarks & proprietary information	5, 6, 7, & 15.1	13 and 14
(i)	Restrictions on products/services offered	1.2, 1.5, 1.6,1.8, 13.7, 13.11 & 14	16
(j)	Warranty & customer service requirements	13.9	Not applicable
(k)	Territorial development & sales quotas	1.4 - 1.7 & 12.7	12
(I)	Ongoing product/service purchases	4.3, 4.6, 9.1 & 14	6, 7 and 8
(m)	Maintenance, appearance and remodeling requirements	13 of Franchise Agreement;	6
(n)	Insurance	9.1, 9.2 & 9.5	6 and 7
(0)	Advertising	15	6, 7 and 11

Obligation	Section in Franchise Agreement; MUDA	Item in Franchise Disclosure Document
(p) Indemnification	9.3 - 9.7	6
(q) Owner's participation/ management/staffing	10, 12.1, 12.6, & 12.7 of Franchise Agreement;	15
(r) Records & reports	8	Not applicable
(s) Inspections & audits	8.7, 12.7 & 13.14	6
(t) Transfer	11 of Franchise Agreement; Section 5 of MUDA	17
(u) Renewal	2.2 & 2.3	17
(v) Post-termination obligations	18	15 and 25
(w) Non-competition covenants	20	14 and 15
(x) Dispute resolution	22	20
(y) Other: Temporary Management by Enviro-Master	13.13	8

ITEM 10

FINANCING

Except as disclosed below, we do not offer direct or indirect financing. We do not guarantee your note, lease or obligation.

We are not required to offer you any financing options. At our choice, in our sole discretion, we may offer you the opportunity to finance up to \$10,000 of your costs for certain large equipment purchases that are not part of the initial equipment package described in Item 5 and Item 7 of this franchise disclosure document, at an annual interest rate of 0%. The terms of any such financing will be governed by a Promissory Note between us and you, a copy of the form of Promissory Note is attached to this disclosure document as **Exhibit E** (the "Promissory Note").

The Promissory Note must be paid back in weekly installments over a 26-week period. If the Promissory Note is signed by a business entity, we will require a personal guarantee signed by the owners of the entity. We take a security interest in the equipment that you are financing through the Promissory Note, as well as your accounts receivable, until the balance of the Promissory Note has been paid off. You can pre-pay the Promissory Note without penalty. If you default on your obligations to pay the Promissory Note, if you default under the Franchise Agreement, if you file bankruptcy, are insolvent, or are dissolved, or if you terminate, transfer or do not renew the Franchise Agreement prior to the full payment of the Promissory Note, we may accelerate all principal and accrued interest due under the Promissory Note. You must pay all collection costs, including reasonable attorneys' fees and expenses that we incur.

The Promissory Note includes a general release from you in our favor. Other than the general release, the Promissory Note does not require you to waive defenses or other legal rights or bar you from asserting a defense against us. We do not intend to sell, assign or discount to a third party all or any part of the financing arrangement.

ITEM 11

FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS AND TRAINING

Except as listed below, Enviro-Master is not required to provide you with any assistance.

Pre-Opening Assistance. Before you open your Franchised Business, we will:

- a. Designate your Territory (Franchise Agreement, Section 1.1);
- b. Provide you and your managers with initial training (including if you acquire an existing Franchised Business), at times and locations we designate (Franchise Agreement, Section 4.1(a));
- c. Loan you a copy of our Manual and other manuals and training aids we designate for use in the System by all franchisees, as we may revise from time to time (Franchise Agreement, Section 7);
- d. Provide you with Approved Software (Franchise Agreement, Section 4.3);
- Provide you with our logo, make available for purchase on our storefront an initial supply of forms and business materials, and provide specifications for stationery and other printed materials you must use in the Franchised Business (Franchise Agreement, Section 4.1(c)); and
- f. Advise you in preparing to open the Franchised Business, including specifications and procedures for obtaining products and equipment from Approved Suppliers, providing Approved Products and services, advertising and promoting the business, and operating the business (Franchise Agreement, Sections 4.1(b) and 4.1(e)). We do not install any of the products or equipment for you.

Site Selection.

We do not approve or assist you in locating a site for the Franchised Business. We or our affiliates typically do not lease or sublease locations for a Franchised Business. You must conduct an independent evaluation of the site. Your site must be within your Territory and must be subject to a lease which requires our prior acceptance. You will be solely responsible for negotiation of the terms of your lease and performance under the lease. We disclaim any responsibility for the suitability of your lease. Our acceptance of the lease is solely based on the site and lease satisfying our minimum site selection criteria.

For a single Franchised Business, your warehouse location must meet our criteria as set forth in the Manual, including without limitation, criteria for location and appearance. If you purchase two or more Franchised Businesses, your leased location must meet the following criteria: (i) your warehouse location must be approved by our approved redistribution vendor to receive deliveries; (ii) your warehouse location must also meet our criteria as set forth in the Manual, including without limitation, criteria for location and appearance; (iii) the inside of your warehouse location must have both designated warehouse space for storage of equipment and products, and designated office space for meetings and performance of other office functions; (iv) you must build out and furnish your warehouse location in accordance with our specifications and trade dress.

Except as described in this Item 11, we do not provide assistance with (i) conforming your warehouse location to local ordinances and building codes, (ii) obtaining required permits, and/or constructing, remodeling or decorating your Franchised Business, and/or (iii) hiring and training employees.

Typical Length of Time to Open.

It typically takes between one to three months between the earlier of the date you sign the Franchise Agreement and pay the initial franchise fee and the date you open for business. The time it takes to open may be affected by many different factors, including the time it takes you to obtain financing, hire key personnel, complete training, your ability to obtain a business license, obtain a lease and move into your space, install equipment, and to purchase and fit a van, etc.

You must open your Franchised Business at the accepted location for your site within 90 days after the execution of your Franchise Agreement (the "Opening Deadline"). We may terminate your Franchise Agreement if you fail to meet the Opening Deadline.

If you sign a MUDA, unless otherwise approved by us in writing, your Opening Deadline will be the same for each Franchise Agreement you sign with the MUDA.

Continuing Assistance. During the operation of your Franchised Business, we will:

- Continue training, primarily your managers, at times, at locations and for fees we designate, currently about \$325 per day, per person. (Franchise Agreement, Section 4.2(b));
- b. Provide you with updated lists of Approved Products (by brand name and/or by specifications) and services, and updated lists of Approved Suppliers for such products and services (Franchise Agreement, Section 4.2(d)). We may, but are not obligated to, provide assistance with setting price for Products and Services that you sell to customers. If we require that certain Products or Services be sold to customers at a certain price (or minimum or maximum price), you will comply such price requirements to the extent permitted by law (Franchise Agreement, Section 4.6);
- c. Review new suppliers you propose, subject to your possible payment of Supplier Review Fees (see Items 8 and 9) (Franchise Agreement, Section 14.3);
- d. Provide other communications, resources and assistance that we develop and offer to all franchisees (Franchise Agreement, Section 4.2);
- e. Provide information regarding services and products (Franchise Agreement, Section 4.6);
- f. Facilitate credit card payments by customers (Franchise Agreement, Section 4.3);
- g. Manage and collect your accounts receivable (Franchise Agreement, Section 4.3); Invoice all credit/charge accounts (Franchise Agreement, Section 4.3);

- h. Prepare your monthly statements of account to customers (Franchise Agreement, Section 4.3);
- i. Provide centralized ordering of supplies always, subject to your local unit decisions on ordering (Franchise Agreement, Section 4.3);
- j. Provide support services we consider advisable. We may provide these services on-site, by telephone, or through other means. Timing will depend on the availability of our personnel. Support services may include procedures for administration, bookkeeping and inventory control, as well helping you resolve operational problems you may encounter (Franchise Agreement, Sections 4.3 and 4.4);
- k. Review and approve advertising, marketing and promotional materials (Franchise Agreement, Sections 4.1(c) & 4.1(e)); and
- I. Coordinate the local digital marketing campaign (Franchise Agreement, Section 15.2(d)).

National Advertising Fee.

We are not obligated to conduct any advertising or spend any amount on advertising in your Territory. We may periodically formulate, develop, produce, and conduct, at our sole discretion, advertising or promotional programs in such form and media as we determine to be most effective. We may make available to you for you to purchase approved advertising and promotional materials that we prepare.

We will allocate your National Advertising Fee to a fund for promotion of the Franchised Business, the Marks and the System (the "National Advertising Fund"). You must pay a National Advertising Fee to the National Advertising Fund in the amount of 2% of the weekly Gross Revenue (as defined in Section 3.5 of the Franchise Agreement) of the Franchised Business. Your National Advertising Fee is in addition to your National/Regional Accounts Fee and Digital Advertising Campaign contribution.

We currently do not, and are not required to, maintain the National Advertising Fund, your National Advertising Fees, or income earned from contributions to the National Advertising Fund, in a separate account from our other money. Our other franchisees may not be required to contribute to the National Advertising Fund, may be required to contribute to the National Advertising Fund, or may be required to contribute to a different advertising fund. Franchisor-owned outlets, if any, will not be required to contribute to the National Advertising Fund.

We or an affiliate will administer the National Advertising Fund. We have sole authority to direct all advertising programs and promotions and use of the National Advertising Fund, with sole control over the creative concepts, materials and media used in the programs, and the placement and allocation of advertising. We reserve the right to use any media, create any programs, and allocate advertising and promotional expenditures to any regions or locales we deem appropriate. The advertising may include promotions in areas where Enviro-Master businesses are not yet established. We may create the advertising in-house or use any outside agency, depending upon the type of media. We are not required to spend any amount from the National Advertising Fund on advertising or promotions in your Territory. We have no obligation to make expenditures from the National Advertising Fund that are equivalent or proportionate to your contributions, ensure that you benefit directly or proportionally or in any amount from the placement of advertising, or ensure that any advertising impacts or penetrates your Territory.

We may use the National Advertising Fund to meet the costs of administering, preparing and conducting national, regional or local advertising, promotional and brand building programs and initiatives or any kind (collectively, "Brand Development Activities"), which includes but is not limited to the cost of (i) preparing and conducting television, radio, magazine, podcasts, newspaper and digital advertising campaigns and other public relations activities (including, but not limited to, for purposes of brand reputation management), (ii) employing public relations firms and advertising agencies to assist in these activities, and (iii) conducting other activities that are directly or indirectly designed to attract national account customers, promote the System, its franchisees, and/or increase System sales, such as limited-time offerings, new product development, sales training programs, costs associated with the inside sales program, recruiting activities, annual and regional convention costs, franchisee incentives, customized materials, guest response programs, manager/employee recognition programs, quality assurance programs, mystery shopper programs, brand website and ordering platforms, brand applications, social media account administration and promotion, and equipment and technologies related to such marketing programs. We may use the National Advertising Fund to compensate us for our reasonable administrative costs and overhead we incur in connection with all Brand Development Activities, and any other activities related to advertising and promotional programs, including new product development; market research; preparing advertising and promotional materials; digital marketing; working with public relations firms, advertising agencies, advertising placement services and creative talent; reimbursing franchisees advisory council meeting expenses; developing and maintaining, and paying third parties for the development and maintenance of internet sites, applications, and other equipment and technology related to marketing programs. We may use the National Advertising Fund to pay for Brand Development Activities in certain limited territories only and not your Territory. We are not obligated to spend the National Advertising Fund on all territories equally, proportionately or at all. The National Advertising Fee is not utilized principally to solicit new franchise sales although we may state or reference in advertising that Franchises are available.

In 2023, we collected \$1,546,027.00 in the National Advertising Fund. All funds that were collected were expended; funds were expended in the following approximate proportions: National Account Staff and Direct Sales Costs (42%); Telemarketing (8%); Media Placement (25%) Marketing Personnel (20%); Dues, Subscriptions (3%); Trade Shows (0%); Marketing Software (2%); and Marketing Reserves (0%).

We are not required to have an independent audit of the National Advertising Fund completed. We will provide you with an annual summary of expenditures of the National Advertising Fund on your reasonable written request to us, but are not required to prepare financial statements for the National Advertising Fund. If any contributions to the National Advertising Fund, including any associated earnings, are not spent in the fiscal year in which they accrue, they remain set aside in the National Advertising Fund for use in later years. Any amounts we or our affiliate contribute to the National Advertising Fund will be considered an advance from us or our affiliates to the National Advertising Fund. We and/or our affiliates have the right to be reimbursed from the National Advertising Fund any amounts that we advanced to the National Advertising Fund. We may terminate the National Advertising Fund at any time, but will not do so until all monies collected in the National Advertising Fund have been spent or returned to contributors on a prorated basis.

Local Advertising.

You are required to participate in the Local Digital Marketing campaign disclosed in Item 6 and Item 7. This campaign contribution cannot be applied to your requirement to spend a minimum of 2% of your Gross Revenues on local marketing. If you purchase more than one Franchised Business, you will only pay the Local Digital Marketing campaign contribution for your first Franchised Business.

We may also require, as we currently do, that you list your business with local and other electronic media sites on the Internet.

Advertising Cooperatives.

Currently, there is no requirement that you participate in a local or regional advertising cooperative. In the future, we may form cooperatives or require cooperatives to be formed. We have no liability or obligation to you for maintaining any cooperative and each cooperative will be organized and governed in the form and manner that we determine in advance. We may change, dissolve, or merge any cooperative.

Franchise Marketing Committee.

The Franchise Marketing Committee was formed in 2018. The purpose of the committee is to assist the Franchisor in developing, implementing and communicating a comprehensive marketing strategy for the system. All franchisee members will be selected by the Franchisor. The committee currently consists of 2 franchisee members who are also members of the franchisee association.

The members of the committee serve until they resign or are removed.

The committee meets as needed. The committee serves in an advisory capacity only and has no operational or decision-making power, although its input into the strategy and planning process is of great importance to our business plans. We have the power to form, change, or dissolve the committee at any time in our sole discretion.

Internet and Electronic Media.

We currently operate a website related to the System at www.enviro-master.com (the "Website"). We have the right to designate a successor Website. Subject to the terms of the Franchise Agreement and Manual, we may make available to you a sub-page on the Website that will be located at a sub-domain of the Website to be specified by us (the "Subpage"). You will be permitted to upload content onto the Subpage solely to promote, and provide customers information related to, your Franchised Business. You may only upload content onto the Subpage in accordance with terms of the Franchise Agreement and any guidelines, directives or specifications (collectively, "Subpage Standards") issued by us. The Subpage may not contain content which references any other Franchised Business other than your Franchised Business. You may not upload, publish, display, or otherwise include or use any content on the Subpage without receiving our approval. Once we approve the initial content of the Subpage, you must submit any changes to us before you make any changes. We may, at any time, cease to make the Subpage available to you or the public. Upon the termination or expiration of the Franchise Agreement for any reason or a default under the Franchise Agreement for any reason, you may

not upload, content, onto, or otherwise use, the Subpage; the Subpage may immediately cease and we may cease to make the Subpage available to you.

No advertising or promotion may be conducted by you over the Internet or through other forms of electronic media, whether within or outside your Territory, without our express prior written consent, which we can withhold for any or no reason. (Franchise Agreement, Section 15.3.) This includes any use by you or your employees of any form of social media which references the Marks in any way; you are fully responsible for your employees with regard to social media and the Marks. You must comply with any social media policies in the Manual. You must submit to us for approval before use, all social media posts and replies including sites such as Facebook, Linked In, Instagram, Yelp! and other sites. Our right of approval for all such materials is necessitated by the fact that they will include and inextricably be linked with the Marks. You may only use material or postings which we have approved.

You will be required to run a Local Digital Marketing Campaign consisting of pay per click advertising on Google. This campaign is managed by our approved vendor, which is currently 20North Marketing. The cost of the management and ad buy is currently \$350/month. If you purchase more than one Franchised Business, you will only pay the Local Digital Marketing campaign contribution for your first Franchised Business. We collect this payment from each franchisee and remit one payment monthly to the vendor. This amount is subject to change.

Marketing Materials.

You may use and display promotional and advertising materials provided to you by us, in accordance with our standards and requirements specific in our Manual. You may not use or display any promotional or advertising materials without our prior written consent. All promotional and marketing materials that you propose to use must conform to our standards and requirements as specified in our Manual. You must provide us with all proofs of promotional and marketing materials for our approval before your proposed use of the materials. If you do not receive our approval within 30 business days, we will be deemed to have disapproved the materials. You cannot use any advertising or promotional plans or materials that we have not approved (Franchise Agreement, Section 15.2(a)-(c)).

Computer Systems.

<u>Hardware</u>. You must have or obtain a personal computer, which meets our standards and specifications, as outlined in the Manual. We estimate the current cost of the personal computer will be approximately \$1,000 to \$2,000, but this estimate may increase or decrease depending on market price.

In addition, you must acquire a hand-held computer device for each Service Hygiene Technician, for use on the job (the "Hand-Held Device"). We will provide you with specifications for the Hand-Held Device in the Manual and/or materials. You may obtain the hardware from any vendor you choose at your expense. We estimate the current cost of the handheld equipment to be approximately \$200 per technician, but this estimate may increase or decrease depending on market price. You may also incur a monthly charge from your local cellular carrier for each Hand-Held Device should you elect to provide cellular access to same. We will have independent access to all information stored on the hand-held device.

We estimate the total annual costs to franchisees for the maintaining and upgrading the required hardware will be \$0 to \$1,000. We may require you to update your hardware. There are no contractual limits on the cost or frequency of this obligation.

<u>Software</u>. We will provide you with specifications for required computer administrative software for your personal computer in the Manual and/or materials (e.g., Office 365, Quickbooks, etc.). You must update your operating software as we direct, and there are no contractual limitations on the frequency of this obligation. You must obtain Internet access at your expense for business communications.

In addition, you must use certain particular approved computer software particularly related to the Franchised Business (the "Approved Software"). The Approved Software will allow you to schedule service to customers, change route assignments for each employee, access accounts receivable and aging of accounts receivable, and to perform other functions. You may not substitute any other software for these functions. We purchase the Approved Software from a third party vendor and provide it to you. The cost of the Approved Software, including any updates or maintenance to the Approved Software, are covered by the monthly Technology Fee and weekly Admin/Service Fee that you pay us. We reserve the right to increase the Technology Fee and Admin/Service Fee to account for increased costs we incur from time to time for the implementation, procurement, and on-going updates/maintenance to the Approved Software. We may supplement, substitute, discontinue or modify the Approved Software and its conditions for use on reasonable notice to you. If you purchase more than one Franchised Business, we will only charge you a Technology Fee on your first Franchise Agreement.

Except as described in this Item, we will not guarantee, warranty, maintain or support any computer system. We may require you to agree to maintenance or support contracts.

We presently do not require you to obtain or use any electronic cash register. We reserve the right to require you to purchase new or different hardware or software during the term of the Franchise Agreement, which may result in costs to you that we currently are not able to estimate.

Conferences and Meetings.

We will host an annual conference and regional meetings for you (including your General Manager, if applicable), your operations manager and your sales personnel. Attendance at these conferences is mandatory for you (including your General Manager, if applicable), your operations manager and all sales persons. We currently charge a reasonable conference fee per attendee. In addition, we will charge you \$3,000 for each person who is required to attend a National or Regional Conference and fails to do so. You must pay travel and lodging expenses for you and your personnel.

Electronic Funds Transfer.

You must pay all fees or contributions due under the Franchise Agreement by automated bank draft or other reasonable means necessary to ensure we receive payment of all fees and contributions. You must comply with any of our payment instructions, including executing any forms which grant us the right to debit your account on a weekly basis for payment of royalty, management and advertising fees and contributions and other fees and contributions to be paid to us or required by us under the Franchise Agreement. (See Franchise Agreement, Section 3.3).

Operations Manual.

Exhibit F to this disclosure document is a table of contents to our Manual. Our Manual consists of 621 pages. The number of pages devoted to each subject are listed below:

•	Section I: Introduction, Understanding Proper Structure, Insurance:	19 pages
•	Section II: Sales: Gain	34 pages
•	Section III Products: Grow	58 pages
•	Section IV: Services	169 pages
•	Section V: Operations: Grow, Retain	157 pages
•	Appendix I: Sales Leadership and Customer Retention	37 pages
•	Appendix II: Office Management Guide	82 pages
•	Appendix III: ZOHO Guide (CRM)	39 pages
•	Appendix IV: Branding Guidelines	25 pages

The Manual contains mandatory and suggested specifications, standards and operating procedures. The Manual is confidential, remains our property, and must be kept secure. The Manual currently is provided electronically and hard copy at no cost to you.

Training Programs.

We offer an initial franchise training program ("Initial Training Program"). Before you open for business, you (or your General Manager, if applicable) and your operations manager must complete the Initial Training Program to our satisfaction. We may require any of your other principals or employees who are (or later become) actively involved in the operations and/or office management of the Franchised Business to satisfactorily complete Initial Training Program (Franchise Agreement, Section 12.2). If you or any of your employees do not satisfactorily complete any portion of the Initial Training Program, you may designate a substitute trainee acceptable to us. We have the right in our sole discretion to determine whether a trainee has successfully completed the Initial Training Program. The Initial Training Program is also available as an option to your employees who will perform our services (Franchise Agreement, Section 12.2). You must complete the Initial Training Program within 120 days after signing the Franchise Agreement.

The initial franchise fee includes the costs of the Initial Training Program for up to two individuals. You must pay a reasonable fee (currently \$2,500) if you elect to bring additional trainees. You or your employees must pay all personal expenses to attend the Initial Training Program, including expenses of transportation, lodging, meals, wages and employee benefits.

All training is scheduled after advance written notice to all franchisees. Scheduled training may be canceled by providing written notice fifteen (15) days prior to such scheduled training. All training is subject to space and time availability.

We generally conduct training in Charlotte, North Carolina about once a month. Training is eight days in Charlotte, North Carolina, and five days training in your market. We may adjust the training schedule based on demand. We will give you, upon request, a current training schedule. Currently, the Initial Training Program includes instruction in the following subjects:

	Appro	ximate		
Subject	Hours of I	nstruction	Location	
	Classroom	On-the-Job		
Field Service	1 hr.	8 days 2 days	Charlotte, NC/your Franchised Business	
General Management / computer operation	7-9 hrs.	0 days	Charlotte, NC/Online	
Financial Management	3 hrs.	N/A	Charlotte, NC/Online	
Business Planning	5-7 hrs.	N/A	Charlotte, NC/Online	
Field Sales	4 hrs.	11 days	Charlotte, NC/your Franchised Business	
Totals	20-24 hrs.	19-21 days	The days and hours may change depending on the number of trainees and their job responsibilities; on-the-job training subjects will substantially overlap each otherin other words, multiple subjects will be covered at the same time.	

TRAINING PROGRAM

Our trainers include, Scott George, Tracy Traficante, Mark Pinnell, Robert Skinner and Michael Lane. Mr. George has 13 years of experience in small business operations and 8 years of experience with us. Ms. Traficante has 16 years of experience in operating small businesses in the restroom hygiene industry and 7 years of experience with us. Mr. Pinnell has 11 years of experience in operating an Enviro-Master franchise and 11 years of experience with us. Mr. Skinner has 10 years of experience in the route services business and 2 year of experience with us. Mr. Lane has 11 years of experience in outside sales and 7 years of experience with us. Certain of these individuals provide training services as independent contractors to us, not as company employees.

We will provide you with the materials necessary to complete your training, which include the Manual and other business forms. While we train you, your managers and your first sales person, you are solely responsible for the training of the remainder of your staff.

Continuing Training.

We may offer continuing, supplemental, refresher or advanced training. Some of this training may be mandatory, and some may be optional. We may or may not charge you reasonable training fees and/or training material fees for this training (see Item 6). We will notify you of any fees before you or your employees enroll in any continuing training. Continuing training courses generally last from one to four working days each, depending on the subject, and are generally offered about once a year. Currently the fees for these courses are approximately \$325 per day, per person. These courses generally take place in Charlotte, North Carolina, and cover subjects such as product development, marketing techniques and similar business-related topics. You must attend (or cause your General Manager, if applicable, and your operations manager or other employees to attend) and satisfactorily complete all mandatory continuing training.

ITEM 12

TERRITORY

Territory Grant.

Attachment A to your Franchise Agreement describes your Territory. We will negotiate and agree upon your specific Territory with you. The Territory in which you are authorized to operate your Franchised Business will be based on a number of factors, including the number of Franchised Businesses you operate, geographic size and population of people and businesses. However, there is no minimum or maximum territory area. In certain densely populated metropolitan areas, a Territory may be considerably smaller, while franchisees operating in more rural areas may have a significantly larger Territory. We expect, but do not guarantee or represent, that each Territory will encompass approximately 10,000 to 25,000 businesses at the time you sign a Franchise Agreement for the Franchised Business. Once established, the boundaries of your Territory will not be adjusted regardless of whether the population of people and businesses in your Territory increases or decreases over time.

If you purchase a single Franchised Business, you may operate out of your designated vehicle and your warehouse within your Territory. If you purchase two or more Franchised Businesses, you must operate your Franchised Businesses from one brick and mortar location centrally located within your Territories.

Relocation.

We do not permit franchisees to relocate their Territory. For a single Franchised Business, you must propose a warehouse location to us and obtain our acceptance, which we may provide in our sole discretion. If you purchase two or more Franchised Businesses, you must propose a physical business location to us and obtain our acceptance, which we may provide in our sole discretion. You may not relocate your office or warehouse without our consent.

Territory Protection.

We will grant you a protected territory as described below. Since we do reserve some rights (see our reservation of rights below), you will not receive an exclusive territory. You may face competition from other franchisees, from outlets we own, or from other channels of distribution or competitive brands that we control. We will not operate or allow others to operate an Enviro-Master business within your territory.

You may not solicit or service accounts outside of your Territory without our prior written permission; under no circumstances may you solicit or service accounts in the territory of another franchisee. If you receive an unsolicited inquiry from a customer or prospective customer outside your Territory, you must comply with our instructions to refer the customer to us or to the franchisee responsible for servicing the customers in that area. We may grant you permission to service customers outside of your Territory; however, at all times we retain the right to sell franchises for areas other than your Territory, and if we sell a franchise for an area in which you are servicing any customer(s), you must immediately comply with our instructions for transferring the customer account(s) and all records of the account(s) to the new franchisee. In that case, you will have no right to receive compensation from any other franchisee or us for the transfer of any accounts for customers located outside of your territory. We resolve conflicts regarding territory and customers between our franchisees, its affiliates, and the affiliates' franchisees, either informally or by enforcement of the franchise agreements.

Under the Franchise Agreement, we and our affiliates retain all the rights that we do not specifically grant to you. For example, without limitation, we retain the following rights, without providing any rights or compensation to you:

- a. We and/or our affiliates may establish or license franchises and/or company or affiliateowned businesses offering products or services that are similar or identical to the Approved Products using the System or elements of the System under the Marks or any other marks anywhere outside of your Territory or in non-traditional locations inside or outside your Territory.
- b. We and/or our affiliates may, whether inside or outside of your Territory, produce and/or sell Approved Products or any other products or services, and authorize others to produce and/or sell Approved Products or any other products or services, using the Marks, the System, and any other marks and/or systems we desire through any alternative channel of distribution located anywhere, including to and through (a) supermarkets, convenience stores, club stores, and other retail facilities not dedicated to the sale of the Approved Products and (b) mail order and e-commerce channels.
- c. We and/or our affiliates may advertise, or authorize others to advertise, using the Marks anywhere, including inside and outside of your Territory.
- d. We may solicit, and you must service, National or Regional Accounts in your Territory as described below.
- e. We and/or our affiliates may acquire, be acquired by, or merge with another entity with existing businesses or franchises that are similar to or competitive with the Franchised Business anywhere (including inside and outside of your Territory) and (a) convert the other businesses to be Franchised Businesses operating under the Marks and the System (except inside your Territory), (b) permit the other Franchised Businesses to continue to operate under another name anywhere (including inside your Territory), and/or (c) permit the businesses to operate under another name another name and convert your Franchised Business and other existing Franchised Businesses to such other name. We may open or franchise new Franchised Businesses near your Franchised Business (but outside your Territory) without consulting you, giving you the first right to open them, or paying you any compensation. Other affiliate-owned or Franchised Businesses near your Franchised Businesses and services in close proximity to may (i) compete directly with you, (ii) provide products and services in close proximity to

your Franchised Business without compensating you, and (iii) possibly adversely affect the operation of your Franchised Business.

To benefit the System, if any of our affiliates, or their franchisees, offer products or services similar to those offered by us, then we and the affiliate, and the affiliate's franchisees, may share and use confidential information about the similar products and services.

National/Regional Accounts.

We may solicit National or Regional accounts in your Territory. To promote uniformity and consistency, we may set the prices to be charged for each National or Regional account. If we obtain a National or Regional account with a service location in your Territory, we will refer that business to you. If you fail to accept a National or Regional account in your Territory, and to service it at the prices we determine, we may designate another franchisee to service the account. You must comply with our conditions, standards, procedures, policies and pricing for servicing that account, as provided in the Manual or otherwise.

There are two types of National or Regional Accounts: (i) a "corporate account", which is a company which can make decisions on all locations, and which signs a national or regional agreement with Franchisor or its affiliate allowing Franchisor (or its affiliates and franchisees) to service the corporate account's various locations; and (ii) a "hunting license" which is a company where Enviro-Master has been designated as an approved vendor with the ability of Franchisor (or its affiliates or franchisees) to approach the specific locations of that company with the endorsement of that company's corporate headquarters.

We do not compensate you for soliciting or accepting orders from these National or Regional accounts in your Territory because you will be responsible for filling these orders. You will not expand the scope of work performed for National or Regional accounts beyond that allowed by us.

We further reserve the right to distribute products directly to accounts in your Territory. If we distribute products directly to accounts within your Territory, we may, but are not obligated to, split the profit that we earn on the difference between what you, as a franchisee, would pay for the products and what the account pays for the products. Currently, we split such profit with you on a 50/50 basis. For example, if you would be charged \$10 for a product by us, and we sell the same product to an account for \$12, then we would take \$1 of that profit and would distribute the other \$1 of that profit to you.

Failure to Exceed Minimum Royalties.

Beginning on the first anniversary of your Franchise Agreement, if at any time your Franchised Business becomes subject to minimum royalties, and remains subject to minimum royalties for a period of 12 months, we will have the right to reduce or remove the protection of your Territory described in this Item 12; in addition, this will be a default under the Franchise Agreement and we may terminate the Franchise Agreement.

Failure to Maintain Required Sales Staff/Sales Assistance Program.

Currently, franchisees are required to have at least one sales associate for each Franchised Business. For your first Franchised Business, if we approve, you may be the sales associate. If at any time during the term of your Franchise Agreement you fail to meet the

minimum threshold of sales associates for your number of Franchised Businesses, we will have the right to reduce or remove the protection of your Territory described in this Item 12; in addition, this will be a default under the Franchise Agreement, and we may terminate the Franchise Agreement.

Affiliated Programs.

Except as described in Item 1 and this Item 12, we do not operate or franchise, or currently plan to operate or franchise, any business under a different trademark that sells or will sell goods or services similar to those that our franchisees sell. However, our affiliates, including the affiliated programs described in Item 1 and other portfolio companies that currently are or in the future may be owned by private equity funds managed by Eagle Merchant Partners, may operate and/or franchise businesses that sell similar goods or services to those that our franchisees sell.

Item 1 describes our current affiliated programs that offer franchises, their principal business addresses, the goods and services they sell, whether their businesses are franchised and/or company-owned, and their trademarks. All of these other brands (with limited exceptions) maintain offices and training facilities that are physically separate from the offices and training facilities of our franchise network. Most of the affiliated programs are not direct competitors of our franchise network given the products or services they sell, although some are, as described in Item 1. All of the businesses that our affiliates and their franchisees operate may solicit and accept orders from customers near your business, including within your Territory. We do not expect any conflicts between our franchisees and our affiliates' franchisees regarding territory, customers and support, and we have no obligation to resolve any perceived conflicts that might arise.

ITEM 13

TRADEMARKS

Pursuant to the terms of the Franchise Agreement, we grant you the right and license to operate a Franchised Business pursuant to the System and using the Marks and related names and marks that may be developed in the future and used as part of the System. The following is a description of the principal trademarks and service marks that we will license to you.

We have applied to register the following Mark on the Principal Register of the United States Patent and Trademark Office (the "USPTO"). Until the USPTO grants a federal registration for this Mark, this Mark will not have the legal benefits and rights of a federally registered trademark. If our right to use this new Mark is challenged, you may have to change to an alternative trademark, which may increase your expenses.

TRADEMARK	REGISTRATION NUMBER	REGISTRATION DATE
	97,802,266 (serial number)	February 20, 2023 (filing date)

All of the Marks listed below are owned by us, have been registered on the Principal Register of the USPTO, and have been renewed at the proper time:

TRADEMARK	REGISTRATION NUMBER	REGISTRATION DATE
Enviro-Master	7,063,530	May 23, 2023
Enviro-Master	3,826,396	July 27, 2010 (renewed September 26, 2019)
WE KILL THE GERMS THAT KILL YOUR BUSINESS!	5,631,347	December 18, 2018
VIRUS VAPORIZER	6,179,516	October 20, 2020

The franchisor has filed all required affidavits. We intend to renew the registrations and file all appropriate affidavits for the Marks at the times required by law.

Other Marks have been, or may be, applied for or registered with the USPTO. The provisions of the Franchise Agreement apply to any and all other trademarks, service marks, and trade dress authorized and licensed for use by us to you during the term of your Franchise Agreement. We may specify the other Marks that you may use, if any, in writing from time to time.

You must comply with the proper use and marking of the Marks as we indicate in the Manual or otherwise. We update the Manual periodically, and add or delete Marks on a continuing basis.

We are the owner of all right, title and interest in and to the Marks and the goodwill associated with them. All goodwill associated with the Marks remains our exclusive property. All usage of the trademarks by you and any goodwill established will inure to our exclusive benefit.

There are no currently effective material determinations of the USPTO, the Trademark Trial and Appeal Board, or the trademark administrator of any state or any court, nor is there any pending infringement, opposition or cancelation proceeding. There is no pending material federal or state court litigation regarding our use or ownership rights in the Marks. There are no agreements currently in effect that significantly limit our rights to use or license the use of the Marks in any manner material to a franchisee. We have no actual knowledge of either superior prior rights or infringing uses that could materially affect your use of the Marks in any state.

Your rights to the Marks are derived solely from your Franchise Agreement. You will only use the Marks as we authorize. You have no right to apply for registration of any Mark. In using the Marks, you must strictly follow our standards, specifications, requirements, and instructions. You may not use any trademark or any words or designations similar to the Marks (i) as part of any corporate or legal business name, (ii) with any prefix, suffix or other modifying words, terms, designs or symbols (other than logos we have licensed to you), (iii) in selling any unauthorized services or products, (iv) as part of any domain name, electronic address, metatag, search engine keyword, social media account, or otherwise in connection with any website or other electronic medium without our consent, or (v) in any other manner we have not expressly authorized in writing. When your Franchise Agreement expires or terminates, all rights to use the Marks will revert to us automatically without payment to you and you will keep no rights in the Marks. You may not take any action to question or contests our rights or interest in the Marks and the goodwill in the Marks.

If litigation involving the Marks is initiated or threatened against you, you must notify us promptly and cooperate fully with us in defending or settling the litigation. We will decide whether to defend or settle any trademark litigation brought against you based upon our good faith assessment, exercised in our discretion, of your compliance with the Franchise Agreement and the Manual. If you are in compliance, we will indemnify you and hold you harmless from all claims, liabilities, costs, damages and reasonable expenses for which you are held liable arising out of your use of the Marks (including settlement amounts negotiated by us or with our expressed consent). If we decide to take either action, we will do so at our expense. We will have no obligation to defend or indemnify you if the claim against you relates to your use of the Marks in violation of the Franchise Agreement.

We may require you to modify or discontinue the use of any of the Marks, and to substitute different marks. If we exercise this right, you must make such changes at your sole cost and expenses with 180 days of our notice of the change.

ITEM 14

PATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION

We own no rights in or licenses to, any patents or copyrights that are material to the franchise, nor do we have any pending patent applications that are material to the franchise.

Enviro-Master International Franchise, LLC FDD May 6, 2024 #198798v1

Proprietary Information.

We will loan you a copy of our Manual after you purchase a franchise and before you begin your Initial Training Program. You must operate the Franchised Business strictly according to the current Manual, which we may amend periodically. You must at all times treat the Manual and the information in it as confidential. To protect the confidentiality of the Manual, you must sign a confidentiality agreement (**Exhibit G**) before you view the Manual. If you do not sign a confidentiality agreement, you may not view the Manual and you may not purchase a franchise.

ITEM 15

OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISE BUSINESS

We strongly believe that the success of the Franchised Business will depend to a large extent on your personal and continued efforts, supervision and attention. We require that either you, or a general manager that you designate (if we accept such person) (the "General Manager"), personally supervise the Franchised Business, and be involved in the day-to-day operation. Except as we expressly permit in writing, you (or your managing shareholder, partner, or General Manager) must devote full time, energy, and best efforts to the management of the Franchised Business, and must complete our Initial Training Program. At our option, before the General Manager is engaged, you must submit to us the proposed candidate's identity and qualifications, and we may accept or reject such candidate based on our commercially reasonable assessment of his/her management experience, qualifications and ability to maintain the Standards. We will not unreasonably withhold our acceptance if the General Manager meets our minimum qualifications and completes the Initial Training Program. If you purchase two or more Franchised Businesses that share adjacent Territories, then either you or the General Manager you designated (if we accepted such person) may personally supervise all of the Franchised Businesses.

If the General Manager fails to ensure that the Franchised Business satisfies the terms of the Franchise Agreement and complies with our Standards, then we may require you to hire a new General Manager. You, or the General Manager (as applicable) are solely responsible for hiring any personnel of the Franchised Business and determining the terms and conditions of their employment. You (or your General Manager) must hire and properly train all personnel. The General Manager must have at least a 10% equity interest in the business entity that is the franchisee of your Franchised Business.

Prior to allowing you to view our Manual, receive training or purchase the franchise, we will require that you, any General Manager, and each person who has ownership share in you or in the Franchised Business or receives training, deliver a confidentiality agreement. The agreement must be in a form satisfactory to us, and must provide that we are a third party beneficiary of, and have the independent right to enforce, the covenants (see **Exhibit G** for a sample covenant that is acceptable to us).

There is a covenant of non-competition in the Franchise Agreement. In addition, we may require that any person who has an ownership share in the Franchised Business, is a General Manager, or who receives training under the System deliver a confidentiality and system protection agreement in a form acceptable to us. If you are a Legal Entity, we may require each person who owns greater than a 10% ownership share in you or in the Franchised Business to sign a personal guaranty.

ITEM 16

RESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL

Limitations on Goods or Services.

You must offer and sell only those products and services that we authorize. You must offer and sell all products and services that we designate as required for franchisees. We maintain written lists of the products and services that are authorized in the Manual. You may not offer or sell any unauthorized product or service without our express prior written consent. You must discontinue offering or selling any product or service (whether or not we previously authorized it) promptly on notice from us. We may add to or modify authorized products or services as long as we provide you with prior notice. There are no other limits on our right to change the types of products and services you offer and sell.

Limitations on Customers.

There are no limitations on the customers to whom you may offer services, other than territorial (see Item 12).

ITEM 17

RENEWAL, TERMINATION, TRANSFER AND DISPUTE RESOLUTION

THE FRANCHISE RELATIONSHIP

This table lists important provisions of the franchise and related agreements. You should read these provisions in the agreements attached to this disclosure document.

	Provision	Section in Franchise Agreement	Summary
a.	Length of the franchise term	2.1	5 years starting on the date of signing of franchise agreement.
b.	Renewal or extension of the term	2.2	10 years.
C.	Requirements for you to renew or extend	2.3	180 to 240 days' notice; past and current compliance; updated location(s) and training; payments timely; general release (release may not apply under certain state laws); renewal fee; sign then-current franchise agreement which may contain materially different terms and conditions from the original contract; terms that will remain substantially the same are listed in section 2.3(g).

	Provision	Section in Franchise Agreement	Summary
d.	Termination by you	N/A	No provision in franchise agreement.
e.	Termination by us without cause	N/A	Enviro-Master cannot terminate your franchise without cause.
f.	Termination by us with cause	17.1	Enviro-Master can terminate your franchise only if you default.
g.	"Cause" defined – curable defaults	16.3	<u>30 days to cure</u> : you fail to perform or observe any covenant, duty or obligation in the Franchise Agreement; you fail to obtain Enviro-Master's approval when required; you fail to offer all services and products in accordance with our Standards; you interfere with our contractual or business relationships; you fail to operate the Franchised Business in accordance with our Standards; you fail to ensure all requires persons sign the Confidentiality Agreement or provide us a copy of the same; you fail to timely submit reports or other requested documents; you fail to pay, when due, any fee or other amount owed to Enviro-Master or its affiliates; you engage in any activity exclusively reserved to us; you fail to maintain required insurance or provide certificates of insurance to us; you conduct a business or engage in marketing products or services similar to those offered by your Franchised Business; you make unauthorized alterations to any invoice produced by our systems; we receive a written complaint from any Regional or National Account customer that intends to terminate service because of your poor service; you fail to service any required Regional or National Account; you fail to pay the fee for remedial training after you; you fail to pay the fee for remedial training after you; you fail to establish or maintain the Minimum Sales Associate Requirement (as defined in the Franchise Agreement); or your employees miss any regional or national training.
h.	"Cause" defined – non- curable defaults	16.2	Without notice: you make a material misrepresentation or omission in your application for the Franchised Business or otherwise to us; you violated a non-compete agreement by entering into

Provision	Section in Franchise Agreement	Summary
		the Franchise Agreement; you are convicted of or plead no contest to a felony or other crime; you engage in any action, behavior or conduct that is likely to affect our reputation; you misuse the Marks; you disclose our Confidential Information; you fail to open for business within the required timeline; you abandon or fail to operate the Franchised Business for 5 or more consecutive days; you relocate the Franchised Business without our consent; your landlord evicts you or notifies us that it is re-taking possession of the business location; you transfer the Franchised Business or your ownership interests without our consent; the Franchised Business is not maintained under the primary supervision of the approved General Manager during the 180 day period following death or incapacity; you impair or revoke our right to effect electronic funds transfer; you maintain false books or records or submit materially false statement or reports to us; you become insolvent, make an assignment for the benefit of creditors, make an admission of inability to pay obligations when due, file a bankruptcy petition, file a pleading seeking reorganization or dissolution, fail to contest any bankruptcy petition (or similar action) filed against you, or are adjudicated bankrupt or insolvent; receiver is appointed for your assets, or similar action is taken; you commit an act or omission that results in an immediate threat or danger to the health or safety of any person; you fail to comply with laws within 24 hours after being given notice of non- compliance; you fail to satisfactorily complete training; you are subject to minimum annual royalties for 12 or more consecutive months; you breach the Franchise Agreement 3 or more times in a 12 month period or the same obligation 2 or more times in any 6 month period; you default beyond applicable cure under any agreement with us or our affiliates; or you default beyond cure under any agreement related to the Franchised Business.

	Provision	Section in Franchise Agreement	Summary
İ.	Your obligations on termination/non- renewal	18	Cease operating; cease use of trademarks; cancel registrations and licenses; return to us the Manual, proprietary business materials, products purchased from us or our designated suppliers, and products bearing our trademarks, including but not limited to such products that are in the facilities of your former customers such as dispensers, urinal screens, and cleaning fluids; in lieu of returning products purchased from us you may provide us with documentation evidencing that you sold such products to another Enviro-Master franchisee; resolve customer account list issues; pay sums owed; de-identify vans and location(s) and, at Enviro-Master's option; assign telephone number; cease use of related social media and online business directories, cooperate with Enviro-Master in the return of all customers to Enviro-Master.
j.	Assignment of contract by us	11.9	Enviro-Master may assign the franchise agreement.
k.	"Transfer" by you - definition	11.1	Any transfer, sale, assignment, gift or encumbrance of any direct or indirect interest in the Franchise Agreement, the franchise or you, or of all or substantially all assets of the Franchised Business.
I.	Our approval of transfer by you	11.1 - 11.4 & 11.8	Your transfer without Enviro-Master's consent is void and an incurable default; Enviro-Master will not unreasonably withhold its consent to a transfer.
m.	Conditions for our approval of transfer	11.3 & 11.4	Qualified purchaser; you are in compliance; your payments are current; you and buyer agree to certain continuing obligations; you sign a general release; terms of sale are acceptable; update your van(s); allow on-site inspection at Enviro-Master's option; buyer satisfactorily completes training; Franchise Agreement Transfer Fee or Resale Fee paid, as applicable; buyer assumes franchise agreement.

	Provision	Section in Franchise Agreement	Summary
n.	Our right of first refusal to acquire your business	11.7	You must notify Enviro-Master when you have an offer, and Enviro-Master has 30 days to exercise its option to purchase the Franchised Business on the same terms.
0.	Our option to purchase your business	19	Enviro-Master may purchase any or all of the assets of the Franchised Business on termination, expiration or non-renewal of the Franchised Business.
p.	Your death or disability	11.6	Twelve months to transfer; Franchise Agreement Transfer Fee waived for qualified heirs.
q.	Non-competition covenants during the term of the franchise	20.2	You may not divert business or customers, or solicit employees of Enviro-Master or other franchisees; or engage in or have an interest in a similar business.
r.	Non-competition covenants after the franchise is terminated or expires	20.3	You may not engage in or have an interest in a similar business in your territory or in the territory of other Enviro-Master businesses, or within 25 miles of the outer boundary of your territory or of the territory of other Enviro-Master businesses, for three years after termination or expiration of the franchise.
s.	Modification of the agreement	21.4(d)	All changes must be in writing, signed by both parties.
t.	Integration/merger clause	21.4	Any representations or promises outside of the Franchise Disclosure Document and other agreements may not be enforceable.
u.	Dispute resolution	22.7	First mediation; then arbitration or litigation.
v.	Choice of forum	22.7	Charlotte, North Carolina (subject to applicable state law).
w.	Choice of law	22.1	North Carolina law (subject to applicable state law).
х.	Waiver of jury trial	22.8	You waive jury trial.
у.	Limitation on remedies	22.9	Right to settle by repurchase; waiver of punitive and tort damages.

<u>Note</u>: Certain states have laws that may supersede the Franchise Agreement and related agreements in your relationship with us, including the areas of termination and renewal of your franchise. Some states may have court decisions that may override the Franchise Agreement and related agreements in your relationship with us, including the areas of termination and renewal of your franchise.

ITEM 18

PUBLIC FIGURES

We do not use any public figure to promote our franchise.

ITEM 19

FINANCIAL PERFORMANCE REPRESENTATIONS

The FTC's Franchise Rule permits a franchisor to provide information about the actual or potential financial performance of its franchised and/or franchisor-owned outlets, if there is a reasonable basis for the information and if the information is included in the disclosure documents. Financial performance information that differs from that included in Item 19 may be given only if (1) a franchisor provides the actual records of an existing outlet you are considering buying; or (2) a franchisor supplements the information provided in this Item 19, for example, by providing information about possible performance at a particular location or under particular circumstances.

Historically, we granted our franchisees a single large territory without a MUDA ("Existing Franchises"). Existing Franchises only offered restroom hygiene services. As our franchise system has matured and our offerings have expanded to include power washing, window washing, and new products, our targeted territory per franchisee and the resulting market saturation has evolved. Currently, we offer territories that are smaller than what our Existing Franchisees have, but with a lower initial franchise fee ("New Franchises") and an option to develop multiple territories by signing our MUDA. New Franchises are single Franchised Businesses with territories encompassing approximately 10,000 to 25,000 businesses. For each table below, the right most column shows the number of businesses per territory for our Existing Franchises, at the time of the issuance of this Disclosure Document.

Background for Table 19-1.

The following table discloses the Gross Revenue for the calendar year 2023 of Existing Franchises who had been open and reporting revenue for the entire calendar years of 2022 and 2023. In the calendar year 2023, there were 96 Existing Franchises in the Enviro-Master network. Of those, 68 (or 71%) had been open and reporting revenue for the entire calendar years of 2022 and 2023. During 2022 and 2023, 2 Existing Franchises closed. All Existing Franchises in the following table 19-1 operate in one Territory. The table excludes New Franchises because there were no New Franchises operating during the period presented.

The Gross Revenue numbers are taken from the QuickBooks profit and loss statements for each Franchised Business and have not been audited. The business count is taken from real-time market analytics software programs.

Table 19-1

Gross Revenue (2023)

	Territory	Gross Revenue	Business Count in Territory
1	Charlotte	\$3,451,572.02	92,000
2	Saint Louis	\$2,786,602.20	92,000
3	Salt Lake City	\$2,641,918.89	65,000
4	Chicago	\$2,563,647.48	313,000
5	San Antonio	\$2,453,140.71	73,000
6	Raleigh	\$2,320,367.70	85,000
7	Orlando	\$2,048,656.71	113,000
8	Redondo Beach	\$1,841,360.20	169,000
9	Sacramento	\$1,728,756.40	111,000
10	Austin	\$1,700,825.23	104,000
11	Denver North	\$1,602,923.02	84,000
12	Denver South	\$1,599,444.43	91,000
13	Atlanta West	\$1,591,993.78	103,000
14	Ventura	\$1,562,481.90	91,000
15	Savannah	\$1,542,433.13	50,000
16	Columbia	\$1,456,317.99	63,000
17	District of Columbia	\$1,428,103.74	149,000
	Top Quartile Average	\$2,018,855.62	
	Top Quartile Median		
4.6		\$1,728,756.40	00.000
18	Baltimore	\$1,418,569.88	68,000
19	Portland	\$1,348,582.61	89,000
20	Greenville	\$1,280,238.68	76,000

21	Grand Rapids	\$1,260,808.75	52,000
22	San Bernardino	\$1,224,355.90	139,000
23	Miami	\$1,183,040.52	109,000
24	Indianapolis	\$1,163,768.08	77,000
25	Tampa	\$1,162,381.51	86,000
26	Phoenix East	\$1,131,491.84	75,000
27	Greensboro	\$1,123,293.16	60,000
28	Memphis	\$1,106,859.11	46,000
29	Van Nuys	\$1,099,715.26	180,000
30	Metro Detroit South	\$1,097,781.92	67,000
31	Cincinnati	\$1,034,356.65	73,000
32	Omaha	\$1,032,005.83	51,000
33	Chattanooga	\$1,002,572.66	74,000
34	Fort Lauderdale	\$970,009.34	39,000
	2nd Quartile Average	\$1,155,284.22	
	2nd Quartile Median	\$1,131,491.84	
35	Reno Lake Tahoe	\$959,869.14	29,000
36	Seattle	\$938,325.56	89,000
37	Columbus	\$915,955.36	62,000
38	Atlanta East	\$907,199.34	68,000
39	Kansas City	\$906,086.05	96,000
40	Nashville	\$864,308.18	65,000
41	Atlanta SE	\$844,767.15	69,000
42	Phoenix West	\$812,933.43	56,000
43	Northern Michigan	\$801,890.83 48	56,000

Enviro-Master International Franchise, LLC FDD May 6, 2024 #198798v1

<u> </u>			
44	Mobile	\$796,940.95	72,000
45	San Diego	\$771,285.47	143,000
46	Providence	\$739,523.51	95,000
47	Las Vegas	\$690,026.56	73,000
48	Milwaukee	\$627,175.94	80,000
49	Louisville	\$623,009.79	108,000
50	Albany	\$606,162.33	52,000
51	Westchester Rockland NY	\$601,263.83	53,000
	3rd Quartile Average	\$788,630.79	
	3rd Quartile Median	\$801,890.83	
52	Tulsa	\$572,795.00	62,000
53	Palm Beach	\$570,765.57	86,000
54	New Jersey North	\$542,473.35	145,000
55	New Jersey Central	\$513,622.22	86,000
56	Detroit	\$465,804.71	83,000
57	El Paso	\$454,477.23	28,000
58	Philadelphia	\$445,451.80	80,000
59	Fort Myers	\$438,180.85	54,000
60	Birmingham	\$421,343.69	93,000
61	Delaware	\$414,935.76	92,000
62	Sarasota	\$392,810.99	54,000
63	Central Pennsylvania	\$383,238.37	81,000
64	Albuquerque	\$352,357.78	64,000
65	Myrtle Beach	\$312,947.44	31,000

66	Madison	\$283,779.84	48,000
67	Space Coast Florida	\$258,412.09	24,000
68	Central Valley	\$208,184.87	81,000
	Last Quartile Average	\$413,622.44	
	Last Quartile Median	\$421,343.69	
	Total Average	\$1,094,098.27	
	Total Median	\$964,939.24	

The average Gross Revenue per Existing Franchise for 2023 was \$1,094,098.27. Of the 68 Franchised Businesses in the table above, 30 (or 44%) met or exceeded the average Gross Revenue for 2023 and 38 were below the average Gross Revenue for 2023. The highest number in this range was \$3,451,572.02. The lowest number in this range was \$208,184.87.

The median Gross Revenue per Existing Franchise for 2023 was \$964,939.24. Of the 68 Franchised Businesses in the table above, 34 (or 50%) met or exceeded the median Gross Revenue for 2023 and 34 were below the median Gross Revenue for 2023.

Background for Table 19-2.

The following table discloses the Gross Revenue for the calendar year 2023 of Multiple Territory Franchised Businesses who were open and reporting revenue for the entire calendar years of 2022 and 2023. Each of the Franchised Businesses in the table below are Existing Franchises. There are currently 6 Multiple Territory Franchised Businesses, only 5 of which were open and reporting revenue for the entire calendar years of 2022 and 2023. The table excludes New Franchises because there were no New Franchises operating during the period presented.

Table 19-2

Gross Revenue for Multiple Territories (2023)

	Territories	Gross Revenue	Business Count in Territory
1	San Jose Oakland / San Francisco	\$1,929,005.89	197,000
2	Dallas West / Denton / North Texas	\$1,885,167.20	221,000
3	Minneapolis / St. Paul	\$1,417,781.72	131,000
4	Pittsburgh / Cleveland	\$1,326,696.65	294,000
5	Boston North / South	\$1,223,311.33	234,000
	Average	<u>\$1,556,392.56</u>	

<u>Median</u>	<u>\$1,417,781.72</u>	

The average Gross Revenue per Multiple Territory Existing Franchise for 2023 was \$1,556,392.56. Of the 5 Multiple Territory Franchised Businesses in the table above, 2 (or 40%) met or exceeded the average Gross Revenue for 2023 and 3 were below the average Gross Revenue for 2023. The highest number in this range is \$1,929,005.89. The lowest number in this range is \$1,223,311.33.

The median Gross Revenue for per Multiple Territory Existing Franchise for 2023 was \$1,417,781.72. Of the 5 Multiple Territory Franchised Businesses in the table above, 3 (or 60%) met or exceeded the median Gross Revenue for 2023 and 2 were below the median Gross Revenue for 2023.

Background for Table 19-3.

The following table discloses the Gross Revenue for the 2023 calendar year of Existing Franchises who opened in 2022 and have been reporting revenue for the entire calendar year of 2023. The table excludes New Franchises because there were no New Franchises operating during the period presented. In the calendar year 2023, there were 96 Existing Franchises in the Enviro-Master network. Of those, 7 (or 7%) had opened in 2022 and have been reporting revenue for the entire 2023 calendar year. All Franchised Businesses in the following table operate in one Territory.

The Gross Revenue numbers are taken from the QuickBooks profit and loss statements for each Franchised Business and have not been audited.

	Territory	Gross Revenue	Business Count in Territory		
1	Fort Worth	\$689,247.04	69,000		
2	South Broward	\$680,256.01	79,000		
3	Pinellas	\$493,751.76	46,000		
4	Treasure Coast	\$321,496.62	57,000		
5	Wilmington	\$254,802.05	37,000		
6	Oklahoma City	\$185,360.32	69,000		
7	Richmond Virginia	\$89,713.72	57,000		
	Average	\$387,803.93			
	Median	\$321,496.62			

<u>Table 19-3</u>

Gross Revenue (2023)

The 2023 average Gross Revenue per Existing Franchise that opened in 2022 was \$387,803.93. Of the 7 Franchised Businesses in the table above, 3 (or 43%) Franchised Businesses met or exceeded the average Gross Revenue for 2023 and 4 Franchised Businesses were below the average Gross Revenue for 2023. The highest number in this range is \$689,247.04. The lowest number in this range is \$89,713.72.

The 2023 median Gross Revenue per Existing Franchise that opened in 2022 was \$321,496.62. Of the 7 Franchised Businesses in the table above, 4 (or 57%) Franchised Businesses met or exceeded the median Gross Revenue for 2023 and 3 Franchised Businesses were below the median Gross Revenue for 2023.

Background for Table 19-4.

The following table discloses the Gross Revenue for the 2023 calendar year of Multiple Territory Existing Franchised Businesses who opened in 2022 and were reporting revenue for the entire calendar year of 2023. The table excludes New Franchises because there were no New Franchises operating during the period presented.

The Gross Revenue numbers are taken from the QuickBooks profit and loss statements for each Franchised Business and have not been audited.

Table 19-4

Gross Revenue for Multiple Territories (2023)

	Territory	Gross Revenue	Business Count in Territory		
1	Northern Virginia	\$588,184.62	56,000		

The 2023 average Gross Revenue per Multiple Territory Existing Franchise that opened in 2022 was \$588,184.62.

The 2023 median Gross Revenue per Multiple Territory Existing Franchise that opened in 2022 was \$588,184.62.

Some outlets have earned these amounts. Your individual results may differ. There is no assurance that you'll earn as much.

Written substantiation for the financial performance representation will be made available to the prospective franchisee upon reasonable request.

Other than the preceding financial performance representation, we do not make any representations about a franchisee's future financial performance or the past financial performance of company-owned or franchised outlets. We also do not authorize our employees or representatives to make any such representations either orally or in writing. If you are purchasing an existing outlet, however, we may provide you with the actual records of that outlet. If you receive any other financial performance information or projections of your future income, you should report it to the franchisor's management by contacting Tod Bierling, 5200 77 Center

Drive, Suite 500, Charlotte, NC 28217, (704) 302-1016, the Federal Trade Commission and the appropriate state regulatory agencies.

ITEM 20

OUTLETS AND FRANCHISEE INFORMATION

Table No. 1

System wide Outlet Summary For years 2021-2023

Outlet Type	Year	Outlets at the Start of the Year	Outlets at the End of the Year	Net Change
Franchised	2021	80	85	+5
	2022	85	91	+6
	2023	91	94	+3
Company-Owned	2021	0	0	0
	2022	0	0	0
	2023	0	2	2
Total Outlets	2021	80	85	+5
	2022	85	90	+5
	2023	91	96	+5

Table No. 2

Transfers of Outlets from Franchisees to New Owners (other than Franchisor) For Years 2021-2023

State	Year	Number of Transfers
СА	2021	3
	2022	0
	2023	1
СТ	2021	0
	2022	0
	2023	0
FL	2021	2

State	Year	Number of Transfers
	2022	1
	2023	1
GA	2021	0
	2022	1
	2023	0
IL	2021	0
	2022	0
	2023	0
IN	2021	0
	2022	0
	2023	0
KS	2021	0
	2022	0
	2023	0
MA	2021	0
	2022	0
	2023	1
NJ	2021	0
	2022	0
	2023	1
NV	2021	0
	2022	0
	2023	1
ОН	2021	0
	2022	0
	2023	1
PA	2021	0
	2022	1
	2023	0

State	Year	Number of Transfers
RI	2021	0
	2022	0
	2023	0
ТХ	2021	0
	2022	2
	2023	0
WI	2021	0
	2022	0
	2023	0
Total	2021	4
	2022	5
	2023	6

Table No. 3

Status of Franchised Outlets For years 2021-2023

State	Year	Outlets at Start of Year	Outlets Opened	Termi- nations	Non- Renewals	Reacquired by Franchisor	Ceased Operations -Other Reasons	Outlets at End of the Year
AL	2021	2	0	0	0	0	0	2
	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
AZ	2021	2	0	0	0	0	0	2
	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
СА	2021	9	0	0	0	0	0	9
	2022	9	0	0	0	0	0	9
	2023	9	1	0	0	0	1	9
СО	2021	2	0	0	0	0	0	2
	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
СТ	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	1	0
	2023	0	0	0	0	0	0	0
DC	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1

State	Year	Outlets at Start of Year	Outlets Opened	Termi- nations	Non- Renewals	Reacquired by Franchisor	Ceased Operations -Other Reasons	Outlets at End of the Year
DE	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
FL	2021	9	0	0	0	0	0	9
	2022	9	3	0	0	0	0	12
	2023	12	1	0	0	0	0	13
GA	2021	4	0	0	0	0	0	4
	2022	4	0	0	0	0	0	4
	2023	4	0	0	0	0	0	4
IL	2021	2	0	0	0	0	0	2
	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
IN	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
KS	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
	2023	1	1	0	0	0	0	2
КҮ	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1

State	Year	Outlets at Start of Year	Outlets Opened	Termi- nations	Non- Renewals	Reacquired by Franchisor	Ceased Operations -Other Reasons	Outlets at End of the Year
	2023	1	0	0	0	0	0	1
LA	2021	0	0	0	0	0	0	0
	2022	0	0	0	0	0	0	0
	2023	0	0	0	0	0	0	0
MA	2021	2	0	0	0	0	0	2
	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
MD	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
МІ	2021	4	0	0	0	0	0	4
	2022	4	0	0	0	0	0	4
	2023	4	0	0	0	0	0	4
MN	2021	2	0	0	0	0	0	2
	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
MO	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
NE	2021	1	0	0	0	0	0	1

State	Year	Outlets at Start of Year	Outlets Opened	Termi- nations	Non- Renewals	Reacquired by Franchisor	Ceased Operations -Other Reasons	Outlets at End of the Year
	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
NJ	2021	2	0	0	0	0	0	2
	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
NM	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
NY	2021	1	1	0	0	0	0	2
	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
NV	2021	2	0	0	0	0	0	2
	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
NC	2021	4	0	0	0	0	0	4
	2022	4	1	0	0	0	1	4
	2023	4	1	0	0	1	0	4
ND	2021	0	1	0	0	0	0	1
	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1

State	Year	Outlets at Start of Year	Outlets Opened	Termi- nations	Non- Renewals	Reacquired by Franchisor	Ceased Operations -Other Reasons	Outlets at End of the Year
OH	2021	3	0	0	0	0	0	3
	2022	3	0	0	0	0	0	3
	2023	3	0	0	0	0	0	3
ОК	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
	2023	1	1	0	0	0	0	2
OR	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
PA	2021	3	0	0	0	0	0	3
	2022	3	0	0	0	0	0	3
	2023	3	0	0	0	0	0	3
RI	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
SC	2021	2	1	0	0	0	0	3
	2022	3	0	0	0	0	0	3
	2023	3	0	0	0	1	0	2
TN	2021	3	0	0	0	0	0	3
	2022	3	0	0	0	0	0	3

State	Year	Outlets at Start of Year	Outlets Opened	Termi- nations	Non- Renewals	Reacquired by Franchisor	Ceased Operations -Other Reasons	Outlets at End of the Year
	2023	3	0	0	0	0	0	3
ТХ	2021	7	0	0	0	0	0	7
	2022	7	1	0	0	0	0	8
	2023	8	2	0	0	0	1	9
UT	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
VA	2021	0	0	0	0		0	0
	2022	0	3	0	0	0	0	3
	2023	3	0	0	0	0	0	3
WA	2021	0	1	0	0	0	0	1
	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
WI	2021	1	1	0	0	0	0	2
	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
Totals	2021	80	5	0	0	0	0	
	2022	85	7	0	0	0	2	91
	2023	91	7	0	0	2	2	94

Note: These numbers do not include 8 franchised outlets operating in Canada as of December 31, 2023.

Table No. 4

State	Year	Outlets at Start of the Year	Outlets Opened	Outlets Reacquired From Franchisee	Outlets Closed	Outlets Sold to Franchisee	Outlets at End of Year
All states	2021	0	0	0	0	0	0
	2022	0	0	0	0	0	0
-	2023	0	0	2	0	0	2
Totals	2021	0	0	0	0	0	0
-	2022	0	0	0	0	0	0
-	2023	0	0	2	0	0	2

Status of Company-Owned Outlets For years 2021-2023

Table No. 5

Projected Openings As Of December 31, 2023

(Please review this table in conjunction with the notes that follow.)

State	Franchise Agreements Signed But Outlet Not Opened	Projected New Franchised Outlets In The Next Fiscal Year	Projected New Company-Owned Outlets In the Current Fiscal Year	
ТХ	2	2	0	
Total	2	2	0	

Exhibit H contains the names, addresses and telephone numbers of all Enviro-Master franchisees that were open as of December 31, 2023 and **Exhibit I** contains the same for the one franchisee who left the system during that time. If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system.

No other franchisees have ceased to do business under the franchise agreement, or had an outlet terminated, canceled or not renewed during the last fiscal year, or have not communicated with the franchisor within the last ten weeks.

One franchisee has signed a confidentiality agreement within the last three years, as a result of a settlement of a dispute. In some instances, current and former franchisees sign provisions restricting their ability to speak openly about their experience with Enviro-Master. You may wish to speak with current and former franchisees, but be aware that not all such franchisees will be able to communicate with you.

ITEM 21

FINANCIAL STATEMENTS

Exhibit J contains (i) the audited, consolidated fiscal year end financial statements of our indirect parent, Enviro-Master Holdings, LLC for the fiscal year ended December 31, 2023, (ii) our audited, consolidated financial statements for the successor period from March 29, 2022 through January 1, 2023 and for the predecessor period from January 3, 2022 through March 28, 2022, including for the predecessor period from January 3, 2022 through March 28, 2022, and (iii) our audited, consolidated fiscal year end financial statements for the periods ending December 31, 2021. Our indirect parent, Enviro-Master Holdings, LLC has guaranteed our performance with you. A copy of the Guaranty of Performance is included in **Exhibit J**.

Our fiscal year runs on a 364-day cycle, and is adjusted once every five years. For the year ended December 31, 2023, our fiscal year ran from January 2, 2023 to December 31, 2023.

ITEM 22

CONTRACTS

Attached are copies of the following agreements relating to the offer of the franchise:

Exhibit B	Franchise Agreement
Attachment A	Territory and Location(s)
Attachment C	Electronic Funds Withdrawal Authorization
Attachment D	Personal Guaranty
Attachment E	General Release
Exhibit C	Form of Multi-Unit Development Addendum to Franchise Agreement
Exhibit D	Market Reservation and Deposit Agreement
Exhibit E	Promissory Note
Exhibit G	Confidentiality Agreement
Exhibit L	Franchisee Disclosure Questionnaire

ITEM 23

RECEIPTS

The last four pages of this disclosure document contain two detachable documents, acknowledging receipt of this disclosure document by a prospective franchisee. You should sign

both copies of the Receipt. You should retain one signed copy for your records and return the other signed copy to:

Enviro-Master International Franchise, LLC 5200 77 Center Drive, Suite 500 Charlotte, North Carolina 28217 1-704-302-1016 EXHIBIT B

FRANCHISE AGREEMENT



Enviro-Master International Franchise, LLC FDD May 6, 2024 #198798v1

FRANCHISE AGREEMENT

TABLE OF CONTENTS

1.	GRANT OF FRANCHISE4
1.1	RIGHT AND LICENSE
1.2	SCOPE OF RIGHT AND LICENSE
1.3	LOCATION(S)
1.4	TERRITORIAL RIGHTS
1.5	TERRITORIAL RESTRICTIONS
1.6	NATIONAL AND REGIONAL ACCOUNTS
1.7	REFUSAL TO ACCEPT NATIONAL OR REGIONAL ACCOUNT.
1.8	NO RIGHT TO SUBLICENSE OR DELEGATE
2.	TERM AND RENEWAL
2.1	INITIAL TERM
2.2	Renewal Terms
2.3	RENEWAL REQUIREMENTS
3.	FEES
3.1	INITIAL FEES7
3.2	CONTINUING FEES AND COMMISSIONS
3.3	PAYMENTS
3.4	LATE PAYMENTS
3.5	GROSS REVENUES
4.	SERVICES BY FRANCHISOR
4.1	BEFORE OPENING10
4.2	After Opening
4.3	Administrative and Business Services
4.4	AVAILABILITY OF PERSONNEL

4.5	ADDITIONAL ASSISTANCE AND TRAINING.	13
4.6	SALE OF PRODUCTS.	13
4.7	NO WARRANTY.	13
5.	PROPRIETARY MARKS	13
5.1	WARRANTIES OF FRANCHISOR.	13
5.2	COVENANTS OF FRANCHISEE.	14
5.3	ACKNOWLEDGMENTS OF FRANCHISEE.	15
5.4	SUBSTITUTION OF MARKS.	16
5.5	Additional Intellectual Property.	16
6.	CONFIDENTIAL INFORMATION	16
6.1	DEFINITION OF "CONFIDENTIAL INFORMATION."	16
6.2	Exclusions.	16
6.3	CONFIDENTIAL TREATMENT.	16
6.4	Developments.	17
0		
7.	CONFIDENTIAL OPERATIONS MANUAL	17
	CONFIDENTIAL OPERATIONS MANUAL	
7.		17
7. 7.1	Manual.	17 17
7. 7.1 7.2	MANUAL	17 17 17
7. 7.1 7.2 7.3	Manual	17 17 17 18
7. 7.1 7.2 7.3 7.4	MANUAL. STANDARDS OF OPERATION. CONFIDENTIAL. PROTECTION OF THE MANUAL.	17 17 17 18 18
7. 7.1 7.2 7.3 7.4 7.5	MANUAL	17 17 17 18 18 18
7. 7.1 7.2 7.3 7.4 7.5 7.6	MANUAL. STANDARDS OF OPERATION. CONFIDENTIAL. PROTECTION OF THE MANUAL. REVISIONS. MASTER COPY.	17 17 18 18 18 18
7. 7.1 7.2 7.3 7.4 7.5 7.6 8.	MANUAL. STANDARDS OF OPERATION. CONFIDENTIAL. PROTECTION OF THE MANUAL. REVISIONS. MASTER COPY. ACCOUNTING AND RECORDKEEPING.	17 17 18 18 18 18 18 18
7. 7.1 7.2 7.3 7.4 7.5 7.6 8. 8.1	MANUAL. STANDARDS OF OPERATION. CONFIDENTIAL. PROTECTION OF THE MANUAL. REVISIONS. MASTER COPY. ACCOUNTING AND RECORDKEEPING. MAINTENANCE OF BOOKS AND RECORDS.	17 17 17 18 18 18 18 18 18
7. 7.1 7.2 7.3 7.4 7.5 7.6 8. 8.1 8.2	MANUAL. STANDARDS OF OPERATION. CONFIDENTIAL. PROTECTION OF THE MANUAL. REVISIONS. MASTER COPY. ACCOUNTING AND RECORDKEEPING. MAINTENANCE OF BOOKS AND RECORDS. SALES REPORTS.	17 17 17 18 18 18 18 18 18 18
7. 7.1 7.2 7.3 7.4 7.5 7.6 8. 8.1 8.2 8.3	MANUAL	17 17 17 18 18 18 18 18 18 18 18 18 18 18 18

8.7	EXAMINATIONS AND AUDITS	19
8.8	TAXES, LICENSES, TRADE ACCOUNTS	
9.	INSURANCE AND INDEMNIFICATION	21
9.1	INSURANCE REQUIREMENTS	21
9.2	PROOF OF INSURANCE.	
9.3	INDEMNIFICATION.	
9.4	"Losses and Expenses."	23
9.5	LACK OF INSURANCE COVERAGE	23
9.6	NOTICE OF CLAIMS.	24
9.7	DEFENSE OR SETTLEMENT OF CLAIMS.	24
10.	OWNERSHIP REQUIREMENTS	24
10.1	FRANCHISEE CORPORATION.	24
10.2	Ownership Information.	25
10.3	PERSONAL GUARANTIES.	25
10.4	COVENANTS OF FRANCHISEE'S OWNERS, OFFICERS AND EMPLOYEES	25
10.5	INCONSISTENT PROVISIONS	25
11.	TRANSFER OF INTEREST	25
11.1	PRIOR CONSENT REQUIRED.	25
11.2	Related Party Transfer	26
11.3	Non-Controlling Interest.	26
11.4	CONTROLLING INTEREST.	27
11.5	Remarketing by Franchisor.	27
11.6	DEATH OR INCAPACITY OF FRANCHISEE.	27
11.7	FRANCHISOR'S RIGHT OF FIRST REFUSAL.	
11.8	PARTITION OF TERRITORY.	
11.9	FRANCHISOR'S SUCCESSORS AND ASSIGNS.	
12.	TRAINING, MANAGEMENT, PERSONNEL AND MEETINGS	
12.1	Full Time and Best Efforts.	29
	D :::	

12.2	INITIAL TRAINING
12.3	SUPPLEMENTAL AND REFRESHER TRAINING
12.4	SUBSTITUTE TRAINEES
12.5	TRAINING COSTS AND EXPENSES
12.6	MANAGEMENT AND EMPLOYEES
12.7	MARKET DEVELOPMENT
12.8	REGIONAL AND NATIONAL MEETINGS
13.	OPERATIONS
13.1	COMPLIANCE WITH SYSTEM
13.2	CHANGES TO SYSTEM
13.3	OPENING FOR BUSINESS
13.4	APPEARANCE OF VAN(S)
13.5	MAINTENANCE
13.6	HOURS OF OPERATION
13.7	Authorized Products and Services
13.8	BUSINESS FORMS
13.9	RELATIONS WITH CUSTOMERS AND THE PUBLIC
13.10	INDIVIDUAL VARIANCES
13.11	COMPUTER REQUIREMENTS
13.12	BUSINESS PREMISES REFURBISHMENT
13.13	TEMPORARY MANAGEMENT
13.14	INSPECTIONS
14.	APPROVED PRODUCTS AND SERVICES, AND DESIGNATED VENDORS37
14.1	MINIMUM INVENTORY
14.2	DESIGNATED VENDORS
14.3	PROCEDURE FOR OBTAINING DESIGNATION
14.4	REVOCATION OF DESIGNATION

14.5	Service Providers	38
14.6	ACCESS TO CONFIDENTIAL INFORMATION.	38
14.7	UNAUTHORIZED PURCHASES.	39
14.8	APPLICABILITY TO BUSINESS SERVICES.	39
15.	ADVERTISING, MARKETING AND PROMOTION.	39
15.1	USE AND DISPLAY OF PROPRIETARY MARKS.	39
15.2	Advertising Products and Services.	39
15.3	WEBSITE AND INTERNET ADVERTISING.	42
16.	EVENTS OF DEFAULT	43
16.1	DEFINITION.	43
16.2	EVENTS OF DEFAULT WITH NO RIGHT TO CURE.	43
16.3	EVENTS OF DEFAULT WITH 30 DAY OPPORTUNITY TO CURE.	45
16.4	DEFAULT BY MANAGING OWNER.	47
17.	REMEDIES AND TERMINATION	47
17.1	REMEDIES UPON EVENT OF DEFAULT.	47
17.2	GENERAL PROVISIONS CONCERNING DEFAULT AND TERMINATION.	49
18.	OBLIGATIONS ON TERMINATION OR EXPIRATION	50
18.1	OBLIGATIONS OF FRANCHISEE.	50
18.2	Assignment.	51
18.3	DE-IDENTIFICATION	51
18.4	COOPERATION BY FRANCHISEE.	51
19.	OPTION TO PURCHASE THE ASSETS OF THE FRANCHISED BUSINESS.	52
20.	COVENANTS NOT TO COMPETE, SOLICIT AND HIRE	53
20.1	ACKNOWLEDGMENTS	53
20.2	IN-TERM COVENANTS	54
20.3	Post-Term Covenant.	54
20.4	Exception.	54
21.	INTERPRETATION OF AGREEMENT	56

1.1 INDEPENDENT CONTRACTOR	56
1.2 APPROVALS AND WAIVERS	56
1.3 Notices	56
1.4 Entire Agreement; Amendments5	57
1.5 REDUCTION IN SCOPE	57
1.6 INTERPRETATION	57
GOVERNING LAW AND ENFORCEMENT5	58
2.1 Applicable Law	58
2.2 Costs of Enforcement	58
2.3 Remedies	58
2.4 CLAIMS NO BAR TO ENFORCEMENT	59
2.5 LIMITATION ON ACTIONS	59
2.6 NEGOTIATION AND MEDIATION	59
2.7 ARBITRATION AND LITIGATION	61
2.8 ACKNOWLEDGMENT THAT ARBITRATION PREEMPTS JURY TRIAL RIGHTS	
2.9 WAIVER OF PUNITIVE AND TORT DAMAGES	63
ACKNOWLEDGMENTS AND REPRESENTATIONS OF FRANCHISEE6	63
3.1 ACKNOWLEDGMENTS. FRANCHISEE ACKNOWLEDGES THAT:	53
3.2 Representations	64

B-vi

ATTACHMENTS:

- A. Territory
- B. Minimum Annual Royalty Fees
- C. Electronic Funds Withdrawal Authorization
- D. Personal Guaranty
- E. General Release

ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC FRANCHISE AGREEMENT

This Franchise Agreement is entered into as of ______ by

Parties

ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC ("**Franchisor**") 5200 77 Center Drive Suite 500 Charlotte N.C. 28217 and ("**Franchisee**")

Background

A. As the result of expending time, skill, effort, and money, Franchisor has developed and owns a unique and distinctive system ("**System**") for establishing and operating businesses that offer and sell products and services;

B. The distinguishing features of the System include unique and specialized training, management, and marketing techniques and materials; procedures and methods of operation; uniform standards, specifications, and procedures for products and services; and advertising and promotional programs, all of which may be changed, improved, and further developed by Franchisor over time.

C. Franchisor identifies the System by means of certain trade names, service marks, trademarks, and other indicia of origin, including the name and mark "ENVIRO-MASTER" and such other names, marks and indicia as Franchisor may now or hereafter designate in writing for use in connection with the System ("**Proprietary Marks**").

D. Franchisor continues to develop, use, and control the use of the Proprietary Marks to identify for the public the source of products and services marketed under the Proprietary Marks and to represent the high standards of quality associated with them.

E. Franchisee desires to establish and operate a business in accordance with the System under the Proprietary Marks, and for that purpose wishes to obtain the right and license to do so from

Franchisor, and wishes to receive the training and other assistance provided in connection with the Franchise.

F. Franchisee understands and acknowledges the importance of Franchisor's uniform high standards of quality and service and the need for operating Franchisee's ENVIRO-MASTER FRANCHISE business strictly in accordance with Franchisor's standards and specifications.

NOW, THEREFORE, in consideration of the above premises and the parties' mutual commitments set forth in this Agreement, Franchisor and Franchisee hereby agree as follows:

Certain Defined Terms

- a. Amounts Due: See Section 3.3(a).
- b. Approved Software: See Section 13.11(b).
- c. Arbitrators: See Section 22.7..
- d. Business Location: See Section 1.3.
- e. Business Services: See Section 3.2(b).
- f. Confidential Information: See Section 6.1.
- g. Covenants Agreement: See Section 10.4.
- h. EFT: See Section 3.3(b).
- i. Event of Default: See Article 16.
- j. Franchise: See Section 1.1.
- k. Franchised Business: See Section 1.1.
- I. Franchisee: See Preamble.
- m. Franchisee's Debt: See Article 19.
- n. Franchisor: See Preamble.
- o. Gross Revenues: See Section 3.5.
- p. Hand-Held Device: See Section 13.11(a)
- q. Hazardous Substance: See Section 9.3(5).
- r. Indemnitees: See Section 9.3.
- s. Initial Term: See Section 2.1.
- t. Initiating Party: See Section 22.6(b).

- u. Losses and expenses: See Section 9.4.
- v. Management Fee: See Section 3.2(f).
- w. Manual (also called "Operations Manual"): See Section 7.1.
- x. Minimum Annual Royalties: See Attachment B.
- y. Local Advertising Fee : See Section 15.2(e).
- z. Minimum Sales Associate Requirement: See Section 12.7.
- aa. National Advertising Fee: See Section 3.2(c).
- bb. National or Regional Account: See Section 1.6.
- cc. Products: See Section 4.1(b)(1)
- dd. Proprietary Marks: See Background Paragraph C.
- ee. Purchased Assets: See Article 19.
- ff. Purchase Option: See Article 19.
- gg. Purchase Price: See Article 19.
- hh. Refurbishments: See Section 13.12.
- ii. Renewal Fee: See Section 2.3(f).
- jj. Renewal Term: See Section 2.2.
- kk. Repeated Inspection Fee: See Section 13.14.
- ll. Responding Party: See Section 22.6(b).
- mm. Royalty Fees: See Section 3.2(a)
- nn. Sales Commission: See Section 3.2(f).
- oo. Seller: See Section 11.3.
- pp. Service Fees: See Section 3.2(b).
- qq. Services: All services required by the Franchised Business or services to be performed by the Franchised Business, to the extent the context allows.
- rr. System: See Background Paragraphs A-B.
- ss. Termination Date: See Section 20.3.
- tt. Territory: See Section 1.1.

uu. Transfer: See Section 11.1.

vv. Accounting Period: One of thirteen 4-week periods per calendar year.

The above list is not all-inclusive. Other terms are defined in the text as well.

Basic Terms

1. GRANT OF FRANCHISE

1.1 Right and License. Franchisor grants to Franchisee the right and license ("**Franchise**"), and Franchisee undertakes the duty, to establish and operate a business providing restroom hygiene, drain line management, window cleaning, power washing, paper and chemical products and related services, to customers that include restaurants, including quick service and traditional establishments, hotels, schools, and other types of commercial establishments ("**Franchised Business**") under the System and the Proprietary Marks for customers located in the territory described in **Attachment A** to this Agreement ("**Territory**") for the term described in this Agreement, subject to Franchisee's compliance with the terms and conditions of this Agreement.

1.2 **Scope of Right and License.** Franchisee has the right and obligation to establish and operate the Franchised Business in accordance with the terms of this Agreement, and agrees to use the Proprietary Marks and the System solely in connection with that business. Franchisee will offer and sell restroom hygiene, drain line management, window cleaning, power washing, paper and chemical products, and related products and services, to customers in the Territory as part of the System. Franchisee has the right to use the complete System, as it exists or may be supplemented or modified by Franchisor during the term of the Franchise. Franchisee acknowledges that the System will need to evolve to meet changing market conditions, meet consumer demands and ensure the operational efficiency of the Enviro-Master franchise businesses generally. Accordingly, Franchisee agrees that Franchisor may, on notice, supplement or modify the System in any matter that Franchisor determines to be in the best interest of the System. Such changes may include the implementation or modification of minimum performance requirements and the introduction of new or different Products or Services that are enhancements or complements of the System as currently configured. Franchisee agrees to promptly implement such changes at its expense in connection with the operation of the Franchised Business.

1.3 Location(s). If Franchisee is purchasing a single Franchised Business to operate in one Territory, then Franchisee is only required to buy or lease warehouse space for storage of equipment and Products within Franchisee's Territory. If Franchisee is purchasing two or more Franchised Businesses, then the inside of the Business Location must be at least 2,500 square feet and have both designated warehouse space for storage of equipment and Products, and designated office space for meetings and performance of other office functions (in each case, the "Business Location"). The Business Location may not be moved outside the Territory except with Enviro-

Master's written consent. Franchisee will build out and furnish the Business Location in accordance with Franchisor's specifications and trade dress as set forth in the Manual, including paint, furniture, technician lockers and other required fixtures.

1.4 Territorial Rights. As long as Franchisee remains in compliance with Franchisee's obligations under this Agreement that relate to the Franchised Business, including the performance requirements set forth in Section 1.5, Franchisor will not operate or grant a franchise for, or allow an affiliate to operate or grant a franchise for, an Enviro-Master Franchise Business under any name within the Territory, except as otherwise described in this Agreement. Franchisor reserves the right at all times to operate, and to grant franchises to others to operate, Enviro-Master franchise and other businesses under any names outside the Territory.

1.5 Territorial Restrictions.

(a) Franchisee will not offer or sell any Products or Services to any customers who are located outside the Territory without the express prior written permission of Franchisor, which Franchisor may grant or withhold in its sole discretion. If Franchisee receives unsolicited inquiries from customers outside the Territory, Franchisee will notify Franchisor of the same and comply with Franchisor's instructions to refer them either to Franchisor or directly to the Enviro-Master franchisee who is responsible for servicing customers in that area.

(b) Franchisor will at all times have the right to sell franchises for areas other than the Territory. If Franchisor sells a franchise for an area outside of Franchisee's Territory in which the Franchised Business is servicing any customer(s) with Franchisors written consent, Franchisee will immediately comply with Franchisor's instructions for transferring the customer account(s) and all records of the account(s) to the new franchisee. Franchisee will have no right to receive compensation from Franchisor or any Enviro-Master franchisee for the transfer of any accounts for customers located outside of the Territory.

1.6 National and Regional Accounts. A customer that has multiple locations across the nation or a smaller geographic territory, and which has contracted with or otherwise authorized Franchisor or its franchisees to provide services across the customer's various locations is a "National Account" or "Regional Account" as applicable. There are two types of National or Regional Accounts: (i) a "corporate account", which is a company which can make decisions on all locations, and which signs a national or regional agreement with Franchisor or its affiliate allowing Franchisor (or its affiliates and franchisees) to service the corporate account's various locations; and (ii) a "hunting license" which is a company where Enviro-Master has been designated as an approved vendor with the ability of Franchisor (or its affiliates) to approach the specific locations of that company with the endorsement of that company's corporate headquarters.

Franchisor or its affiliates may solicit and provide services to National or Regional Accounts which have locations in Franchisee's Territory. If Franchisor obtains a National or

Regional account with a service location in the Territory, Franchisor will refer that service location's business to franchisee. Franchisee is obligated to accept any National or Regional Account's service location referred by Franchisor. Franchisee must service that location in accordance with Franchisor's conditions, standards, procedures, policies and pricing for servicing that account, as provided by Franchisor in the Manual or otherwise.

1.7 Refusal to Accept National or Regional Account. If Franchisee refuses to accept a National Account or a Regional Account, or service it according to the conditions established by Franchisor, to protect the system and preserve the National or Regional Account, Franchisor may in addition to all other remedies set forth herein as a result of Franchisee's default for failing to comply with Section 1.6, service that account in the Territory and/or may license or appoint another person, business or Franchisee to service that account in the Territory, without any compensation to Franchisee. Franchisor doing so shall not excuse Franchisee from the breach of Section 1.6 above.

1.8 Product Distribution to National or Regional Account. Franchisor reserves the right to distribute products directly to accounts in Franchisee's Territory. If Franchisor distributes products directly to accounts within Franchisee's Territory, Franchisor may, but is not obligated to, split the profit that it earns on the difference between what Franchisee would pay for the products from Franchisor and what the account pays for the products. Franchisor will split this profit with Franchisee on a 50/50 basis. For example, if Franchisee would be charged \$10 for a product by Franchisor, and Franchisor sells the same product to an account for \$12, then Franchisor would take \$1 of that profit and would distribute the other \$1 of that profit to Franchisee.

1.9 No Right to Sublicense or Delegate. Franchisee shall not grant sublicenses to other persons to use the Proprietary Marks or the System. Except as may be expressly approved in writing by Franchisor, Franchisee shall not delegate Franchisee's performance of any services comprising the Franchised Business except to employees of Franchisee acting under the direct supervision of Franchisee.

2. TERM AND RENEWAL

2.1 Initial Term. The initial term of the Franchise is 5 years ("**Initial Term**") starting on the date of execution of this Agreement by the last party hereto, unless the Franchise is terminated at an earlier date under Article 17.

2.2 Renewal Terms. Franchisee will have the right to renew the Franchise for an additional term of 10 years (the "**Renewal Term**") if Franchisee satisfies the renewal requirements in Sections 2.3.

2.3 Renewal Requirements. The following are conditions for renewal of the Franchise:

(a) **Notice.** Franchisee must notify Franchisor of Franchisee's intention to renew at least 180 days, but not more than 240 days, before the expiration of the Initial Term;

(b) **Compliance.** At all times during the expiring term and at the time of renewal, Franchisee must have complied in all material respects with the terms and conditions of this Agreement and all other agreements between Franchisee and Franchisor or any affiliates of Franchisor;

(c) **Training.** Franchisee and Franchisee's employees must successfully complete any training courses required by Franchisor to bring their training up to date;

(d) **Payments.** At all times during the Initial Term and at the time of renewal, Franchisee will have paid all amounts due to Franchisor and all affiliates of Franchisor on a timely basis;

(e) **General Release.** Franchisee must execute a General Release in the form attached hereto as **Attachment E**, releasing Franchisor and all affiliates of Franchisor from any claims Franchisee may have against them, whether known or unknown, by Franchisee regarding the Franchised Business or the franchise relationship;

(f) **Renewal Fee.** Franchisee must pay to Franchisor a renewal fee equal to 10% of the then-current Initial Franchise Fee (the "**Renewal Fee**"); and

(g) **Renewal Franchise Agreement.** For the Renewal Term, Franchisee must execute the then-current form of franchise agreement being offered by Franchisor at the time of renewal. Such then-current franchise agreement may contain terms and conditions that are substantially different from, or additional to, those in this Agreement, including without limitation, higher fees, including higher Minimum Annual Royalties, and different performance requirements; provided however that: (i) at the time of renewal, Franchisee will pay Franchisor a Renewal Fee in lieu of an additional Initial Franchise Fee; and (ii) the renewal franchise agreement will not increase the Royalty Fees, Service Fees or National Advertising Fee beyond what Franchisor is permitted to charge under this Franchise Agreement.

3. FEES

3.1 Initial Fees.

Initial Franchise Fee. On execution of this Agreement, Franchisee will pay to Franchisor, an Initial Franchise Fee of \$50,000.00. Franchisee acknowledges that, upon signing of this Agreement, Franchisor shall have fully earned the Initial Franchise Fee and it shall be non-refundable.

3.2 Continuing Fees and Commissions.

(a) **Royalty Fees**. During the term of the Franchise, Franchisee will pay weekly to Franchisor continuing royalty fees ("**Royalty Fees**") equal to 6% of the Gross Revenue (as defined in Section 3.5) of the Franchised Business. In the event that the annual Royalty Fees paid to Franchisor are less than the Minimum Annual Royalties set forth in Attachment B hereto, then at the end of each calendar year, Franchisor will invoice Franchisee for the difference between the Minimum Annual Royalties and the actual Royalty Fees paid that calendar year, and Franchisee shall pay the same within 10 days of the date of the invoice. Franchisee understands that failure to meet the Minimum Annual Royalties for twelve (12) consecutive months is a basis for default in accordance with Section 16.2(t) of this Agreement.

(b) Administrative and Business Service Fees. As an integral part of the System, Franchisor will provide certain business services to Franchisee as defined in Section 4.3 ("Business Services"). Franchisee will pay to Franchisor continuing service fees ("Service Fees") of 5% of Gross Revenue (as defined in Section 3.5) of the Franchised Business or \$125 per week, whichever is greater.

(c) **National Advertising Fee**. During the term of the Franchise, Franchisee will pay weekly to Franchisor a continuing national advertising and marketing fee ("**National Advertising Fee**") equal to 2% of weekly Gross Revenue (as defined in Section 3.5) of the Franchised Business.

Sales Commission and Management Fees. (d) Franchisee will pay Franchisor a sales commission for any service location of a National or Regional Account that Franchisor refers to Franchisee (the "Sales Commission"). If Franchisee was already providing Products and Services to that service location before Franchisor secured the National or Regional Account, Franchisee will only pay a Sales Commission on those Products and Services that are a result of Franchisor's referral and that are above and beyond those Products and Services that Franchisee was already providing. The Sales Commission shall be established in the Manual and may change from time to time. Currently the Sales Commission is 4%. The Sales Commission shall be paid for 12 months following the referral. After 12 months, the Sales Commission shall convert to a Management Fee to compensate Franchisor for its time managing the National and Regional Accounts (the "Management Fee"). The Management Fee shall be established in the Manual and may change from time to time. Currently the Management Fee is 3% of Gross Sales from National and Regional Accounts. The Management Fee will commence in the thirteenth month of the National or Regional account, and is based on all Gross Sales, regardless of whether Franchisee previously provided Products and Services to the location prior to Franchisor securing the National or Regional Account.

(e) **Technology Fee**. During the term of the Franchise, Franchisee will pay monthly to Franchisor a technology fee (the "**Technology Fee**"), that covers, in part, Franchisor's

costs for procuring and implementing the Approved Software and any on-going updates to it. Currently, the Technology Fee is \$250 a month, but Franchisor reserves the right to increase this fee over time. If Franchisee is simultaneously purchasing two or more Franchised Businesses under a Multi-Unit Development Agreement and such Franchised Businesses' Territories are adjacent, Franchisee must only pay the Technology Fee for the first Franchised Business.

(f) **Additional Fees.** Franchisee may be subject to additional fees as provided in this Agreement if Franchisee purchases Product or Services not approved by Franchisor or purchases from vendors not approved by Franchisor.

Franchisor will have the right to change, discontinue or add to any or all of the Business Services, and to adjust the Service Fees, as provided in Section 4.3(b). Franchisor and Franchisee agree that Service Fees may be increased over the life of the Agreement as follows: If Franchisor incurs additional costs in providing services to its franchisees, the net direct cost of the increase will be billed to Franchisee on a monthly basis.

3.3 Payments.

(a) When Due. As part of the services Franchisor provides, Franchisor manages and collects Franchisee's accounts receivable. Each week, Franchisor will determine the amount of Royalties, Service Fees, National Advertising Fee and other amounts due ("Amounts Due") based on Franchisee's reports and on Franchisor's own information about Franchisee's Gross Revenues. Franchisor will provide an invoice to Franchisee showing Franchisor's calculations of the Amounts Due and the amount Franchiser deducted from Franchisee's collections in satisfaction of the Amounts Due. If Franchisee does not have sufficient collections in order for Franchisor to deduct the Amounts Due as shown on the invoice, Franchisor will debit Franchisee's bank account (as described in (b) below) to pay all Amounts Due.

(b) **Method of Payment.** Franchisor will have the right to receive payment of Franchisee's Amounts Due by electronic funds transfers ("**EFTs**") from Franchisee's bank account to Franchisor's bank account. Franchisee must provide Franchisor with Franchisee's bank's name, address and account number, and a voided check from the account. Franchisee must sign and provide all documents necessary for Franchisor to commence making electronic funds transfers, and for Franchisor to continue to do so throughout the Franchise term. Franchisee must make deposits to Franchisee's account as needed to make funds available for transfer to Franchisor, and may not close such account without Franchisor's prior written consent. Franchisor will debit Franchisee's account in accordance with **Attachment C** hereto in the event there are not enough funds in Franchisee's account to pay all monies due.

(c) **Adjustments**. If it is later determined that the amount Franchisor has transferred from Franchisee's account is less than the Amounts Due by this Agreement, Franchisee will pay the difference within ten (10) days after Franchisor gives notice of the Amounts Due. If

it is determined that Franchisor has transferred more from Franchisee's account than the amount required by this Agreement, the difference shall be credited against Franchisee's future Royalties, Service Fees, National Advertising Fee and Minimum Local Marketing Expense contributions. In the event that an excess amount is paid to Franchisor, Franchisor shall have no other liability or obligation pertaining to the excess payment (other than to give such credit described above). Notwithstanding the above, any funds transferred by Franchisor which are more than the Amounts Due by Agreement for the month in question and the following month as reasonably determined by Franchisor shall be returned by Franchisor to Franchisee by wire or EFT to Franchisee's bank account.

3.4 Late Payments. If any payments are received by Franchisor after the date due (other than solely on account of Franchisor failing to timely initiate the EFT) then, (a) Franchisee will pay to Franchisor, monthly and in addition to the Amounts Due, interest in the sum of 1.5% per month on the amount due or the highest rate allowable by law, whichever is higher, from the date due until paid; and (b) Franchisee shall be in default of this Agreement and Franchisor may exercise any and all remedies in Article 17 of this Agreement. Franchisor also reserves the right to terminate the Franchise as provided under Article 17 if Franchisee fails to pay any amounts when due, whether or not interest is paid.

3.5 Gross Revenues. As used in this Agreement, the term "Gross Revenues" means all revenues and income earned, received or accrued by, through, as a result of, and/or on behalf of the Franchised Business from any source, whether for cash, credit or barter. "Gross Revenues" also specifically includes, without limitation, fees charged to customers and referred to as "service fees," "cleanout fees," "installation fees," and "installation deposits." "Gross Revenues" do not include revenues not generated by the operations of the Franchised Business, such as real estate rental and investment income.

(a) **Sales Taxes Not Included.** "Gross Revenues" does not include sales taxes that Franchisee is required to collect and pay, and does pay, to any governmental agency.

(b) **Approved Deductions.** Franchisee may deduct from Gross Revenues any refunds, discounts, or allowances only if the amounts deducted were included in Gross Revenues and if Franchisee has complied with Franchisor's policies and procedures for reporting and taking such deductions. Under no circumstances may any amount be deducted unless such deduction is expressly disclosed in the statement or report affected by the deduction.

4. SERVICES BY FRANCHISOR

4.1 Before Opening. Franchisor will provide the following services before opening the Franchised Business:

(a) **Training.** Before opening, Franchisor will provide initial training of Franchisee (and its General Manager, if applicable) and Franchisee's employees, at times and

B-10

locations designated by Franchisor, including other franchisee locations as specified in Section 12.2;

(b) **Designated Vendors for Products; Information on Approved Products** and Services.

(1) In order to maintain uniformity of concept and quality under the System, all equipment, inventory, supplies, tools, forms, promotional materials and other products and materials ("**Products**") required to be used or sold by Franchisee in providing restroom hygiene, drain line management, window cleaning, power washing, paper, chemical and other products and all Services, will be purchased from Franchisor (or its affiliates or designated vendors) in accordance with the terms and procedures set forth in the Manual. Franchisor will, to the extent feasible, provide information pertaining to sources from which proprietary and nonproprietary Products and Services which may be used under the System can be purchased. Franchisor agrees, however, that if Products are not or cannot be timely delivered and Franchisee has not been at fault or non-diligent in placing orders in a timely fashion, Franchisee can purchase the Products locally in such quantities necessary to fulfill immediate needs.

(2) Franchisor reserves the right to earn a profit on the sale of Products to Franchisee. The Franchisee agrees that it will not attempt to circumvent Franchisor by directly approaching or interfering with Franchisor's relationship with approved vendors.

(c) **Initial Supply of Printed Materials.** Before opening, (but not upon renewal), Franchisor will provide an initial supply of forms and marketing materials, and standards and specifications for stationery and other printed materials required to be used in the Franchised Business;

(d) **Confidential Operations Manual.** Before opening, Franchisor will loan Franchisee a copy of the Enviro-Master Operations Manual and other manuals and training aids designated by Franchisor for use in the System, as they may be periodically revised by Franchisor, and as are further described in the provisions of Article 7; and

(e) **Opening Assistance.** Before opening, Franchisor will provide advice and guidance to Franchisee in preparing to open the Franchised Business, including standards and procedures for obtaining Products, training employees, advertising, promoting and operating the Franchised Business.

4.2 After Opening. After Franchisee opens the Franchised Business, Franchisor will provide the following services:

(a) **On-Site Assistance.** On-site assistance on an as needed basis as determined by Franchisor after the execution of this Franchise Agreement;

(b) **Continuing Training.** Supplemental and refresher training, as determined necessary by Franchisor at times and locations designated by Franchisor, as specified in Section 12.3;

(c) **Updated Lists of Approved Products and Services, and of Designated Vendors.** Updated lists of Products (by brand name and/or by standards and specifications) Services, and designated vendors for approved Products; and

(d) Advice and Guidance. Periodic advice and guidance through meetings, printed materials and/or other media, as Franchisor makes available to all franchisees in the System.

4.3 Administrative and Business Services.

(a) In consideration of the payment of the Service Fee specified under Section 3.2(b), Franchisor will provide (either directly or through a third party vendor) to Franchisee, and Franchisee will use the Business Services which currently consist of the following:

- (1) collections service on delinquent accounts;
- (2) preparation of monthly statements of account for all charge accounts (i.e., net 30 days), including National and Regional Accounts;
- (3) centralized ordering of supplies;
- (4) management support and advice;
- (5) software and software support for the operation of the Franchised Business;
- (6) processing of customer credit card charges, and electronic payments; and
- (7) posting of charge customer payments.

(b) Franchisor does not guarantee that it will provide all of the above Business Services at all times during the Franchise term. Franchisor reserves the rights, in its sole judgment:

- (1) to discontinue any or all of the Business Services;
- (2) to substitute comparable service(s) for any of the Business Services;
- (3) to add other Business Services; and

(4) to make technological changes.

Without limiting its rights reserved under Section 3.2(b), if Franchisor makes changes to the Business Services offered to all franchisees, Franchisor may reduce or increase the Service Fees accordingly. Any increases will not be more than the additional net incremental direct cost to Franchisor for providing the additional services which benefit the franchisees.

4.4 Availability of Personnel. Franchisor's individualized assistance to Franchisee (whether in person or by telephone, fax or mail) is subject at all times to availability of Franchisor's personnel. Franchisor reserves the right to require Franchisee, as a condition of obtaining individualized, on-site assistance at any time after Franchisee opens for business, to pay the actual travel costs (transportation, lodging, and meals) incurred by Franchisor's employees or agents. Additionally, Franchisor reserves the right to charge a daily fee for on-site assistance, the amount of which is set forth in the Manual.

4.5 Additional Assistance and Training. Franchisor will not charge any additional fee for individualized assistance to Franchisee during Franchisee's first 30 days in operation; after that time, Franchisor reserves the right to charge a consulting fee for individualized assistance to Franchisee. Franchisor also reserves the right to charge reasonable fees for advanced training courses (not including the initial training course) and for training materials. Franchisor will notify Franchisee of any additional charges or fees before providing the services or materials to Franchisee. Franchisee is solely responsible for training Franchisee's personnel.

4.6 Sale of Products. Franchisor, its subsidiaries and affiliates, will provide Products that Franchisee will sell and/or use in the operation of the Franchised Business as provided in Section 4.1 (b) and (c) and in Article 14. Franchisor may, but are not obligated to, provide assistance with setting price for Products and Services that Franchisee sells to customers. If Franchisor requires that certain Products or Services be sold to customers at a certain price (or minimum or maximum price), Franchisee will comply such price requirements to the extent permitted by law.

4.7 No Warranty. Franchisor does not warrant or guarantee that the training, services, software and goods provided for in this Article 4 will be provided without flaw or defect. Franchisor shall not be in breach of this Article on account of any such flaw or defect or combination of flaws or defects.

5. **PROPRIETARY MARKS**

5.1 Warranties of Franchisor. Franchisor warrants with respect to the Proprietary Marks that:

(a) Franchisor is the owner of the Proprietary Marks, and owns the exclusive right to use and license others to use the Proprietary Marks in accordance with the terms and

conditions of this Agreement. Franchisor disclaims any warranty that other persons do not have or claim rights or interest in marks that may be similar or identical to one or more of Franchisor's Proprietary Marks;

(b) Franchisor will take actions reasonably necessary to preserve and protect the validity of the Proprietary Marks; and

(c) Franchisor will use and will permit Franchisee and other Enviro-Master franchisees to use the Proprietary Marks only in accordance with the System and the standards and specifications which underlie the goodwill associated with and symbolized by the Proprietary Marks.

5.2 Covenants of Franchisee. With respect to Franchisee's licensed use of the Proprietary Marks pursuant to this Agreement, Franchisee covenants that:

(a) Franchisee will use only the Proprietary Marks as are approved in writing by Franchisor for Franchisee's use, and will use them only in the manner authorized by Franchisor;

(b) Franchisee will use the Proprietary Marks only for the operation of the Franchised Business and only to sell goods and services expressly authorized in writing by Franchisor;

(c) Unless otherwise authorized or required by Franchisor, Franchisee will operate and advertise the Franchised Business under the name of "Enviro-Master Services" without prefix or suffix;

(d) Franchisee will not use the Proprietary Marks as part of Franchisee's corporate or other legal name. Franchisee will comply with Franchisor's instructions in filing and maintaining required trade name ("d/b/a") or fictitious name registrations. Franchisee agrees to execute, during or after the term of the Franchise, on Franchisor's request, any consents necessary for the registration of Franchisor's corporate name in the state where Franchisee conducts the Franchised Business;

(e) Franchisee will execute any documents deemed necessary by Franchisor or Franchisor's counsel to obtain protection for the Proprietary Marks or to maintain their continued validity and enforceability;

(f) Franchisee will not directly or indirectly contest the validity or ownership of the Proprietary Marks; and

(g) If litigation involving the Proprietary Marks is instituted or threatened against Franchisee, Franchisee will promptly notify Franchisor and will cooperate fully with Franchisor in defending or settling such litigation. Franchisor will decide whether to defend or

settle any trademark litigation. If Franchisor decides to take either action, it will do so at its expense, but Franchisee must cooperate with Franchisor. Franchisor will indemnify Franchisee and hold it harmless from and against any and all claims, liabilities, costs, damages, and reasonable expenses for which Franchisee is held liable in any proceeding arising of Franchisee's use of any of the Proprietary Marks (including settlement amounts negotiated by Franchisor or with its expressed consent), provided that the conduct of Franchisee and its principals with respect to such proceeding and use of the Proprietary Marks is in full compliance with the terms of this Agreement and any requirements imposed by the Operations Manual. Franchisor will decide the actions to be taken against the use of any of the Proprietary Marks by any persons or legal entities other than Franchisor. Any actions that Franchisor takes will be at Franchisor's expense. Franchisor may require Franchisee to modify or discontinue the use of any of the Proprietary Marks, and to substitute different marks. If Franchisor exercises this right, it will provide advance notice to Franchisee.

5.3 Acknowledgments of Franchisee. Franchisee expressly acknowledges that:

(a) The Proprietary Marks are valid and serve to identify the System and those who are licensed or franchised under the System;

(b) Franchisee's right to use the Proprietary Marks is limited to such uses as are authorized under this Agreement, and any unauthorized use will constitute an infringement of Franchisor's rights and a breach of this Agreement;

(c) Franchisor is the owner of all right, title, and interest in and to the Proprietary Marks and the goodwill associated with and symbolized by them;

(d) Franchisee's use of the Proprietary Marks pursuant to this Agreement does not give Franchisee any ownership interest or other interest in or to the Proprietary Marks or System, except the non-exclusive license granted herein;

(e) Franchisor retains all goodwill inherent in the Proprietary Marks or System and Franchisee shall make no claim thereto;

(f) The right and license of the Proprietary Marks granted to Franchisee under this Agreement is non-exclusive, and Franchisor, its subsidiaries and affiliates, subject to the terms of this Agreement (including Section 1.4) thus may:

(1) Grant other licenses and franchises for the Proprietary Marks, in addition to those licenses and franchises already granted;

(2) Use the Proprietary Marks in connection with marketing and selling any products and services as Franchisor sees fit; and

B-15

Enviro-Master International Franchise, LLC FDD May 6, 2024 #198798v1 (3) Develop and establish other systems for the same or similar Proprietary Marks, or any other proprietary marks, and grant licenses or franchises for the use of other systems or marks, without any obligation to offer, sell or grant Franchisee any rights in those systems or marks.

5.4 Substitution of Marks. Franchisor reserves the right to modify or discontinue the use of any of the Proprietary Marks and to substitute different names or marks for use in identifying the System and the businesses operating thereunder. If Franchisor requires Franchisee to change its primary trade name, Franchisee will make such changes at its sole cost and expenses within 180 days of Franchisor's notice of the change.

5.5 Additional Intellectual Property. As between the parties, Franchisor shall exclusively own all rights to all intellectual property used in the Franchised Business, including any inventions, proprietary processes, slogans, marketing materials, and other inventions, designs, or creative works, whether developed by Franchisor, Franchisee or otherwise. Franchisee agrees to execute from time to time such documents as Franchisor may reasonably request to assign rights in such intellectual property to Franchisor, and Franchisee agrees to cause its employees and agents to execute such assignments as well.

6. CONFIDENTIAL INFORMATION

6.1 Definition of "Confidential Information." Confidential Information includes all information, data and techniques designated or treated by Franchisor as confidential, and all information, data and techniques embodied in the System, and all records and copies of such information, data and techniques, in any form. Without limiting the foregoing, Confidential Information also includes, without limiting the foregoing, customer lists of Franchisee and the Franchised Business and customer lists developed by Franchisor or any Enviro-Master franchisee. Franchisee acknowledges that aspects of the System that are known to the public or that may be reverse-engineered are nevertheless Confidential as integrated parts of the System.

6.2 Exclusions. Confidential Information does not include information which Franchisee shows was already known generally to the public at the time when it was disclosed by Franchisor to Franchisee, or which, after the time it was disclosed by Franchisor to Franchisee, validly and legally has become generally known to the public.

6.3 Confidential Treatment. Franchisee hereby agrees that:

(a) Franchisee will not at any time disclose, copy or use any Confidential Information except as specifically authorized in writing by Franchisor;

(b) Franchisee's obligation to maintain all Confidential Information as confidential applies both during and after the term of the Franchise;

(c) Franchisee will use Confidential Information only for Franchisee's operations under this Agreement and for no other purpose;

(d) Franchisee will disclose Confidential Information only to Franchisee's employees or contractual agents as follows, and not to anyone else:

(1) Franchisee may disclose Confidential Information to its employees and contractual agents on a "need to know" basis only. Franchisee may disclose Confidential Information only to individuals who need to know it in order to carry out their jobs. Franchisee may disclose only so much of the Confidential Information as each individual needs to know in order to carry out that individual's job; and

(2) Franchisee will inform Franchisee's employees and agents that the Confidential Information is confidential and that they may not disclose, copy or use it in any unauthorized manner.

(e) Franchisee will obtain and deliver to Franchisor signed confidentiality and system protection agreements from all of Franchisee's employees, current and future, and any agents of Franchisee who have or may have access to Confidential Information. The confidentiality agreements must be in a form acceptable to Franchisor, and must provide that Franchisor is a third party beneficiary of, and has the independent right to enforce, the agreement.

6.4 Developments. Franchisee agrees that all information, data, techniques and knowhow developed or assembled by Franchisee or Franchisee's employees or agents during the term of the Franchise and relating to the System will be deemed a part of the Confidential Information protected under this Agreement.

7. CONFIDENTIAL OPERATIONS MANUAL

7.1 Manual. Franchisor will provide one copy of the Enviro-Master Manual to Franchisee, on loan, at or before the time when Franchisee begins initial training. As used in this Agreement, the term "Manual" also includes all other written, electronic, video, and audio recorded policies, procedures, techniques, memos, bulletins, newsletter, forms, guidelines, and other materials prepared by us in connection with the System or to assist you in the operation of your Franchised Business.

7.2 Standards of Operation. Franchisee will operate its Franchised Business strictly in accordance with the Manual as it may be revised by Franchisor.

7.3 Confidential. Franchisee will at all times treat the Manual and the information in it as Confidential Information, in accordance with the requirements of Article 6. Franchisee will not disclose, copy or use the Manual or the information in the Manual, except as specifically authorized by Franchisor.

7.4 Protection of the Manual. Franchisee will keep the Manual in a secure place at the Business Location at all times. The Manual remains the property of Franchisor and must be returned to Franchisor promptly on expiration or termination of the Franchise, or at any earlier time on Franchisor's request.

7.5 Revisions. Any revisions to the contents of the Manual will be deemed effective on issuance by Franchisor, unless otherwise specified by Franchisor. Revisions may be issued as inserts or replacements to the Manual, through physical or online updates, or may be in the form of informational communications from Franchisor's management. Revisions may be transmitted to Franchisee on paper by first class mail or courier, by facsimile or by electronic mail.

7.6 Master Copy. Franchisee will at all times ensure that Franchisee's copy of the Manual is kept current and up to date and readily available for reference by Franchisee. If there is any dispute about the contents of the Manual, the terms of the master copy of the Manual maintained by Franchisor at its headquarters or electronically will apply.

8. ACCOUNTING AND RECORDKEEPING

8.1 Maintenance of Books and Records.

(a) During the term of the Franchise and for a period of 3 years thereafter, Franchisee will maintain full and accurate books, records, reconciled bank statements and accounts. Franchisee will also timely enter financial data into QuickBooks or another bookkeeping software designated by Franchisor, maintained on Franchisor's server and to which Franchisor has remote access. Franchisee must produce accurate financial statements and records of its Franchised Business operation to Franchisor in a regular and timely fashion, and at any other times that Franchisor requests.

(b) Franchisee will keep all of its books, records and accounts for at least 3 years after the date when they were prepared either on-site or at a Business Location acceptable to Franchisor.

(c) Franchisor may require that Franchisee purchase accounting and bookkeeping services from Franchisor's designated vendor.

(d) Franchisee must use Franchisor's approved vendor for accounting services. At this time, the only approved accounting service provider is Top 2 Bottom Business Solutions.

8.2 Sales Reports. Franchisee will submit to Franchisor, promptly on request, sales reports and current information about prices charged by Franchisee to its customers.

8.3 Automated Reporting and Payments. Franchisee will install and use such computer programs and equipment as Franchisor may designate for the operation of the Franchised

Business. Franchisee will maintain and timely transmit data and reports from its computer system as required by Franchisor in the Manual or otherwise in writing.

8.4 Annual Financial Statements. Within 90 days after the end of each fiscal year of Franchisee, Franchisee, at its expense, will submit to Franchisor a financial statement consisting of a profit-and-loss statement showing the results of operations of the Franchised Business during the previous fiscal year and a balance sheet as of the end of the fiscal year. Each financial statement will be accompanied by a sworn statement signed by Franchisee or by Franchisee's treasurer or chief financial officer attesting that the items contained therein are true and accurate and prepared on a basis consistent with that of prior periods, that they completely and fully describe and disclose the information sought in such statements, and that the signer has made diligent and careful efforts to ascertain the truth, accuracy and completeness of such information. Franchisor reserves the right to provide the form of sworn statement described in this paragraph to be signed by Franchisee or its treasurer or chief financial officer.

8.5 Other Reports and Data. Franchisee will also submit to Franchisor all other forms, reports, records, statements, information, and data as Franchisor may reasonably require, as reasonably specified by Franchisor in the Manual or otherwise in writing.

8.6 Use of Data. Franchisee agrees that all financial and business data submitted by Franchisee to Franchisor may be used as Franchisor deems appropriate. Franchisor agrees that information designated by Franchisee as confidential will not be disclosed to third parties in a manner that identifies Franchisee individually as the subject or source of the information except (i) with Franchisee's permission, (ii) as may be required by law, (iii) in connection with audits or collections under this Agreement, and (iv) in the preparation of Franchisor's Franchise Disclosure Document.

8.7 Examinations and Audits. Franchisor or its designated agents will have the right at all reasonable times, with or without prior notice, to examine and copy the books, records, and tax returns of Franchisee and the Franchised Business, including but not limited to performing a field brand audit, which is designed to determine your compliance with our brand standards, and is broader than a financial audit. Franchisee agrees that Franchisee shall cause any affiliates of Franchisee (including spouse, family trust, minor children, employees who are relatives of Franchisee or closely-held businesses in which Franchisee or any such person has an interest) to produce tax returns and bank records for audit by Franchisor, if request is made for such items. Any records provided for such examination or audit shall be used solely for purposes of establishing amounts due under this Agreement and shall not be disclosed or used by Franchisor for any other purpose.

(a) If an examination or audit should reveal that any Gross Revenues of the Franchised Business have been understated in any report to Franchisor, then Franchisee will pay

Franchisor the amounts due on such understated Gross Revenues immediately on demand, together with interest as provided in Section 3.4

(b) In addition, if an examination or audit reveals that Gross Revenues of Franchisee were understated by 2% or more during the period audited, Franchisee will also pay Franchisor for all costs and expenses in connection with the audit.

(c) Franchisor also reserves the rights set forth under Article 17 if Franchisee understates any Gross Revenues of the Franchised Business, whether or not the applicable fees are later paid.

8.8 Taxes, Licenses, Trade Accounts.

(a) Franchisee will comply with all applicable laws, rules, and regulations, and will obtain all permits, certificates, or licenses necessary to conduct the Franchised Business.

(b) Franchisee will promptly pay when due all taxes applicable to Franchisee and the Franchised Business, including but not limited to withholding, unemployment and sales and property taxes.

(c) Franchisee will pay to Franchisor or its affiliates (i) all sales taxes, corporate taxes, and any similar taxes paid by Franchisor on Franchisee's behalf, imposed on Franchisor, or required to be collected by Franchisor on account of products or services Franchisor furnishes to Franchisee (through sale, lease, or otherwise) or on account of Franchisor's collection of any fee related to this Agreement; (ii) all franchise or similar taxes, whether based on gross receipts, gross revenues, Royalty Fees, advertising contributions, or otherwise, imposed on, required to be collected by, or paid by Franchisor; (iii) all marketplace facilitator or similar taxes imposed on, required to be collected by, or paid by Franchisor in connection with Franchisee's use of Franchisor's website, internet sites, applications, or online ordering platforms; (iv) all other amounts Franchisor pays or must pay for Franchisee for any reason; and (v) any other fees or expenses that Franchisor is entitled to collect from Franchisee.

(d) Franchisee will promptly pay all trade accounts and other indebtedness of Franchisee and the Franchised Business subject to the requirements of Section 3.4.

(e) If Franchisee is engaged in a bona fide dispute with any taxing authority or creditor as to Franchisee's liability for taxes or other indebtedness, Franchisee may defer payment of the amount in dispute while contesting the validity of the claim, but Franchisee will not permit, under any circumstances, any sale, foreclosure or seizure of any assets of the Franchised Business by any creditor. Franchisee must notify Franchisor in writing of the dispute to request the deferment, and must provide information adequate for Franchisor to either grant or deny the deferment.

9. INSURANCE AND INDEMNIFICATION

9.1 Insurance Requirements. Before beginning any activities or operations under this Agreement, Franchisee will obtain all required insurance coverage. Franchisee will thereafter maintain all required insurance coverage throughout the term of the Franchise or any extension thereof, at Franchisee's sole cost and expense. Franchisor may require franchisee to use an insurance broker designated by Franchisor.

(a) Franchisee's insurance must be issued by a company specified by Franchisor which has an A.M. Best rating of A-:VII or better and is licensed to do business in Franchisee's state.

(b) Franchisee's insurance will provide coverage as specified by Franchisor in the Manual or otherwise in writing, and the Certificate of Insurance and policy will name Franchisor as an Additional Insured including (1) Workers Compensation & Employers Liability; (2) Automobile Liability Insurance (including Hired & Non-Owned Auto Liability); (3) Comprehensive General Liability Insurance, including products and completed operations and personal injury protection; (4) Umbrella Liability; (5) Property Insurance; (6) Employment Practices Liability Insurance; (7) Pollution Insurance Policy; and (8) other insurance coverage as required by applicable law in the greater of the amount set forth below or as required by the state/province in which the Franchised Business operates, all in forms satisfactory to Franchisor and in at least the following minimum amounts:

<u>Commercial General Liability (Occurrence Form)</u>			
General Aggregate (other than Prod/Comp Ops Liability)	\$2,000,000		
Products/Completed Operations Aggregate	\$2,000,000		
Personal & Advertising Injury Liability	\$1,000,000		
Each Occurrence	\$1,000,000		
Commercial Automobile Liability			
Combined Single Limit	\$1,000,000 each accident		
Hired and Non-Owned Auto	\$1,000,000 each accident		
Workers Compensation and Employer's Liability State Statutory Limits			
Workers' Compensation			
Employer's Liability			
Bodily Injury by Accident	\$1,000,000 each accident		
Bodily Injury by Disease	\$1,000,000 policy limit		
Bodily Injury by Disease	\$1,000,000 each employee		

<u>Umbrella Liability (to overlay General Liability, Auto Liability and Employers Liability</u>

B-21

<u>coverages)</u>

Each Occurrence and Aggregate Each Occurrence and Aggregate

\$1,000,000 (less than \$2mm revenue) \$2,000,000 (\$2mm - \$3mm revenue)

Enviro-Master International Franchise, LLC FDD May 6, 2024 #198798v1 Each Occurrence and Aggregate Each Occurrence and Aggregate \$3,000,000 (\$3mm - \$4mm revenue) \$4,000,000 (greater than \$4mm revenue)

Employment Practices Coverage

The minimum coverage amount of \$1,000,000, or such other amount that Franchisor designates from time to time.

Pollution Liability

Each Occurrence \$500,000

Property Insurance

The minimum coverage amount of \$25,000, or such other amount that Franchisor designates from time to time.

(c) Franchisee's insurance will cover all indemnification obligations of Franchisee under Subsections 9.3(2), (4) and (5), to the extent that such insurance is readily available at reasonable and standard rates from a qualified insurer, as determined by Franchisor, in Franchisor's sole discretion;

(d) Franchisee's insurance will name all Indemnitees (defined in Section 9.3) as Additional Insured in accordance with Section 9.1;

(e) Franchisee's insurance will provide by endorsement that Franchisor will be entitled to written notice at least 30 days before any change or cancellation of the policy can take effect;

(f) Franchisee will use its best efforts to obtain "occurrence based" liability coverage, rather than "claims made" liability coverage. Franchisor requires that Franchisee obtain "occurrence based" liability coverage; and

(g) Franchisor may change or increase any required coverages upon sixty (60) days' notice to Franchisee.

9.2 Proof of Insurance. Franchisee will submit a certificate of insurance to Franchisor prior to attending initial training, at each annual renewal or change of Franchisee's insurance policy, and at any time upon written request of Franchisor. Franchisee will also provide a copy of the complete policy upon request by Franchisor at any time during or after the term of the Franchised Business.

9.3 Indemnification. Franchisee will, at all times, indemnify and hold harmless, to the fullest extent permitted by law, Franchisor and its affiliates, successors and assigns and the respective directors, officers, employees, agents and representatives of each (the "**Indemnitees**") from all losses and expenses (defined in Section 9.4) incurred in connection with any action, claim,

B-22

demand, or investigation (or any settlement thereof) which arises out of or is based on any of the following:

(1) Any actual or asserted violation by Franchisee of any law, regulation, order, or industry standard;

(2) Libel, slander or any other form of defamation by Franchisee;

(3) Any violation of any warranty, representation, covenant or obligation by Franchisee under this Agreement;

(4) Any act, error or omission of (a) Franchisee or (b) any of Franchisee's agents, servants, employees, partners, affiliates or representatives, or (c) any person with whom Franchisee or any of them may have a business relationship if the Loss or Expense to Franchisor is related to the business relationship with Franchisee; or

(5) Any actual or alleged violation of any law governing health, safety or the environment or the discharge, release, generation, transportation, storage, treatment or disposal of any hazardous substance. For purposes of this Agreement, the term "hazardous substance" means any matter giving rise to liability under the Resources Conservation Recovery Act, 42 U.S.C. § 6901 *et seq.*, the Comprehensive Environment Response Compensation and Liability Act, 42 U.S.C. § 9601 *et seq.*, the Clean Water Act, 33 U.S.C. § 1251 et seq., or generally any contaminant, asbestos, oil, radioactive or other material, the removal of which is required or the maintenance of which is regulated or prohibited or penalized by any local, state or federal agency, or other government authority. This indemnity includes, without limitation, all reasonable costs incurred to remove any and all hazardous substances, and to remedy any actual or alleged violation and bring the Franchised Business and/or the Business Location(s) into compliance with all applicable local, state and federal environmental laws and regulations.

This indemnification is not applicable where Franchisee is operating in compliance with Franchisor's written instructions.

9.4 "Losses and Expenses." As used in this Article 9, the phrase "losses and expenses" will include, without limitation, all losses, compensatory, exemplary or punitive damages, fines, charges, costs, lost profits, attorneys' fees, accountants' fees, expert witness fees, expenses, court costs, settlement amounts, judgments, compensation for damages to Franchisor's reputation and goodwill; costs of or resulting from delays, financing, costs of advertising material and media time/space, and costs of changing, substituting or replacing the same, and any and all expenses of recall, refunds, compensation, public notices and other such amounts incurred in connection with the matters described.

9.5 Lack of Insurance Coverage.

(a) Franchisee's obligation to indemnify the Indemnitees applies even if Franchisee's insurance does not cover the claims.

(b) If Franchisee fails to maintain adequate insurance coverage, Franchisor may purchase insurance at Franchisee's expense, but Franchisor has no obligation to do so.

(c) The Indemnitees have no obligation to maintain insurance coverage of their own. If the Indemnitees choose to rely on their own insurance coverage for claims subject to indemnification by Franchisee, Franchisee will remain liable for all losses and expenses incurred by the Indemnitees or their insurance carriers. The Indemnitees will have no obligation to defend or mitigate any claims subject to indemnification by Franchisee.

9.6 Notice of Claims. Franchisee will notify Franchisor in writing within 15 days after Franchisee learns of any claim covered by this Article or any other claim or action that may adversely affect the operation or financial condition of the Franchised Business.

9.7 Defense or Settlement of Claims. If Franchisor is or may be named as a party in any action against Franchisee, Franchisor will have the right (but no obligation) to undertake the defense and/or settlement of the claims. Franchisor's undertaking will not diminish in any way Franchisee's obligation to indemnify Franchisor and other Indemnitees for all losses and expenses incurred by them. Franchisor may, after consultation with Franchisee and/or Franchisee's insurer, take reasonable remedial or corrective action that Franchisor deems appropriate, if Franchisor reasonably determines that it must act expediently to protect persons or property or the reputation or goodwill of Franchisor or the system.

10. OWNERSHIP REQUIREMENTS

10.1 Franchisee Corporate Entity. If Franchisee is (or later becomes) a corporate entity (whether a corporation, partnership, limited liability company, or otherwise), the following provisions apply:

(a) The activities of the corporate entity must be limited exclusively to operation of the Franchised Business and to sale of goods and services authorized by Franchisor. On reasonable request by Franchisor, Franchisee will disclose all activities being conducted by Franchisee through its corporate entity; and

(b) Franchisee will maintain stop-transfer instructions against the transfer on its records of any equity interests except in accordance with the provisions of Article 11. If a corporation or other corporate entity that issues equity certificates, such certificates issued must be clearly marked with the following legend:

The transfer of these securities is restricted by the terms and conditions of an Enviro-Master Services Franchise Agreement dated [the date hereof].

10.2 Ownership Information. Franchisee will maintain a current list of all shareholders and all partners and all other persons who own any beneficial interest in Franchisee's corporate entity or in this Agreement. Franchisee will furnish the list of owners and copies of any governing documents of the corporate entity to Franchisor promptly on request.

10.3 Personal Guaranties. Each person who, directly or indirectly (a) controls Franchisee or the Franchised Business or (b) holds or controls any ownership interest in Franchisee or in the Franchised Business shall abide by the terms and conditions of this Agreement. Contemporaneously with this Agreement and concurrent with each change of ownership, Franchisee shall disclose the identities, addresses and contact information of all such persons to Franchisor and shall cause each person controlling and/or holding greater than a 10% ownership share in Franchisee or in the Franchised Business to execute a Personal Guaranty in the form attached hereto as **Attachment D**.

10.4 Covenants of Franchisee's Owners, Officers and Employees. Upon signing this Agreement and annually upon the request of Franchisor, Franchisee will obtain and deliver executed confidentiality and system protection agreements from each individual who has or may have an ownership share in Franchisee or in the Franchised Business (the "System Protection Agreement"). The System Protection Agreement must be in a form satisfactory to Franchisor, and must provide that Franchisor is a third party beneficiary of, and has the independent right to enforce, the covenants.

10.5 Inconsistent Provisions. Franchisee's Articles and Bylaws (or comparable governing documents) will expressly provide in the event of any conflict or inconsistency between their provisions and this Agreement, the terms and conditions of this Agreement will prevail.

11. TRANSFER OF INTEREST

11.1 Prior Consent Required. Franchisor has granted this franchise to Franchisee based on the personal financial capacity, moral character, and business background of Franchisee (including each individual partner or shareholder of Franchisee). In this Agreement, the term "transfer" includes any sale, assignment, transfer, gift, pledge, mortgage, encumbrance or hypothecation, by Franchisee or by any other person with an ownership interest in Franchisee or the Franchised Business, of (i) any direct or indirect interest in this Agreement, the Franchised Business or Franchisee, or (ii) all or substantially all of the assets of the Franchised Business. Any transfer of interest without the prior consent of Franchisor will be void and will constitute a default for which Franchisor may terminate the Franchised Business.

11.2 Related Party Transfer. If Franchisee is an individual or partnership, Franchisee will be entitled to transfer Franchisee's entire interest in this Agreement to a corporation or limited liability company formed for convenience of ownership; provided, that the stock in the resulting entity must be owned by the same individual(s) in the same proportionate ownership share(s) as existed before the transfer. Franchisee will also be entitled to transfer its interest (a) to an immediate family member of Franchisee (if Franchisee is an individual) or (b) if Franchisee is a corporate entity, to Franchisee's existing principals in the same ownership percentage as the principals owned the corporate entity. A transfer under this Section 11.2 will be subject to a \$2,500 related party transfer fee, provided that Franchisee complies with the applicable requirements of Article 10 and Section 11.3 (except for the Franchise Agreement Transfer Fee). Such transfer of ownership under this Section 11.2 will not be valid unless an amendment or modification to this Franchise Agreement is signed by Franchisor and Franchisee and appended to the Agreement.

11.3 Non-Controlling Interest. Franchisor will not unreasonably withhold its consent to any proposed transfer of less than the entire interest in this Agreement, the Franchised Business or Franchisee, which (alone or together with other past or current transfers) would not result in a change in control of the Franchised Business or Franchisee, but Franchisor will have the right to require any or all of the following conditions to be met by the individual or entity disposing of such interest ("Seller") before consenting to any transfer:

(a) **Personal Qualifications.** The purchaser must meet Franchisor's standard qualifications for new franchisees, including financial capacity, moral character, and business background;

(b) **Compliance.** Franchisee must be in substantial compliance with all material terms and conditions of this Agreement, and all other agreements between Franchisee and Franchisor or any affiliates of Franchisor;

(c) **Payments.** As of the time of transfer, Franchisee must pay all amounts due to Franchisor and all affiliates of Franchisor;

(d) **Seller's and Purchaser's Obligations.** Seller must remain liable for all obligations under this Agreement (and under any personal guaranty) up to the date of transfer, and the purchaser must assume all obligations of Seller after the date of transfer. If Seller will no longer own an interest in this Agreement, the Franchised Business or Franchisee, Seller must agree to be bound by all applicable post-termination obligations as provided in Article 18;

(e) **General Release.** Seller and Franchisee must release Franchisor and all affiliates of Franchisor from any claims that Seller and Franchisee may have against them, whether known or unknown by Seller and Franchisee regarding the Franchisor/Franchisee relationship which has not been asserted at the time of transfer (a claim for defense or indemnity as a result of

a claim by a 3rd party is not a claim regarding the Franchisor/Franchisee relationship and is not released under any circumstances); and

(f) **Franchise Agreement Transfer Fee.** The transferor or the transferee must pay to Franchisor a Transfer Fee in the amount of \$10,000 for the addition of a minority interest owner under this Section 11.3.

11.4 Controlling Interest. Franchisor will not unreasonably withhold its consent to any proposed transfer before consenting to any transfer which (alone or together with other past or current transfers) would result in a change in the control of Franchisee or the Franchised Business. However, Franchisor may require the following conditions to be met in addition to the conditions under Section 11.3:

(a) **Training.** The transferee and all employees must successfully complete any training courses required by Franchisor to bring their training up to date;

(b) **Franchise Agreement Resale Fee.** If a transfer is for the purpose of effecting the Franchised Business' resale or change in control thereof to a third party, the transferor or transferee shall execute a new Franchise Agreement for a five year initial term and pay a Resale Fee of \$50,000; and

(c) **Transferee's Franchise Agreement.** The transferee must execute Franchisor's then existing Franchise Agreement at the time of transfer.

11.5 Remarketing by Franchisor. Franchisor will have no obligation to assist Franchisee in identifying prospective buyers or negotiating the sale of Franchisee's business. If Franchisee requests and Franchisor is willing to do so, Franchisor may require Franchisee to pay a commission and/or agree to other terms and conditions under which Franchisor would agree to remarket the franchise.

11.6 Death or Incapacity of Franchisee. If Franchisee or any individual who is actively involved in the management of the Franchised Business dies or becomes incapacitated, the heir(s) or representative(s) of that person must notify Franchisor within 30 days.

(a) **Continuous Operation.** The Franchised Business must remain continuously in operation unless Franchisor grants written permission to close temporarily. The Franchised Business may continue to operate under management acceptable to Franchisor and in compliance with the terms and conditions of this Agreement.

(b) **Reasonable Time to Transfer.** The heirs or representative will have a reasonable time to transfer that person's interest to a qualified transferee.

(c) **No Transfer Fee to Qualified Heirs.** The transfer will be subject to the conditions in Section 11.3 (and Section 11.4 if applicable), except that Franchisor will not charge any transfer fee if the Franchise is transferred to the deceased person's qualified heirs or beneficiaries.

(d) **Franchisor's Right to Terminate.** If the interest is not disposed of by transfer to an acceptable transferee within 12 months, Franchisor may, at its option, terminate the Franchised Business upon 30 days' notice.

11.7 Franchisor's Right of First Refusal. If Franchisee intends to offer the Franchised Business for sale or transfer any of the Franchisee's ownership interests in or assets of the Franchised Business, Franchisee will notify Franchisor in writing of that intention and the price and terms that Franchisee will accept. Should Franchisee subsequently solicit offers for the Franchised Business, it shall provide any written bona-fide third party offer to Franchisor. This provision shall not apply to a sale or transfer to a husband, wife, mother, father, son, daughter, brother or sister of Franchisee who agrees to assume all obligations of the transferor under this Agreement.

(a) **Thirty (30) Days to Exercise Right of First Refusal.** Franchisor will have the right, exercisable within thirty (30) days after receipt of such written notice, to send written notice to the Franchisee that it intends to purchase the Franchised Business for the price and on the terms acceptable, or the price and terms offered in writing from a bona fide third party, whichever applies ("**Right of First Refusal**"). If Franchisor exercises its Right of First Refusal, Franchisor will be required to close the purchase within thirty (30) days after provision of written notice of intent to purchase, unless extended in writing and signed by both parties.

(b) **Financial and Business Information.** To enable Franchisor to determine whether it will exercise its Right of First Refusal, Franchisee and the seller will provide such information and documentation, including financial statements, as Franchisor may require, promptly on Franchisor's request.

(c) **Consent to Transfer Required.** If Franchisor does not exercise its Right of First Refusal as provided hereunder, Franchisee may for a 6-month period market and sell the interest, subject to Franchisor's consent as otherwise required under this Article 11, at any price and terms not more favorable to the buyer than those contained in the notice to Franchisor.

(d) **Changes in Offer.** If Franchisee will accept better price and terms offered by a 3^{rd} party than those in the notice to Franchisor, Franchisee will give a new notice to Franchisor, who will have 15 days to exercise its Right of First Refusal on the same terms and conditions as offered by the 3^{rd} party.

11.8 Partition of Territory. Franchisee will have no right to partition the Territory without the express prior written consent of Franchisor, which Franchisor may grant or withhold in its sole discretion.

11.9 Franchisor's Successors and Assigns. This Agreement will inure to the benefit of Franchisor, its successors and assigns. Franchisor will have the right to transfer and assign all or any part of its interest in this Agreement (including its Right of First Refusal under Section 11.7 and its right to substitute different Proprietary Marks under Section 5.4) to any person or legal entity.

12. TRAINING, MANAGEMENT, PERSONNEL AND MEETINGS

12.1 Full Time and Best Efforts. Except as Franchisor may otherwise expressly permit in writing, Franchisee (or a designated manager who has been trained by, and is acceptable to, Franchisor) will devote full time, energy, and best efforts to the management and operation of the Franchised Business. Either Franchisee or Franchisee's designated manager must live within 30 miles of the Territory at all times during the term of the franchise.

12.2 Initial Training. Before opening for business, Franchisee (Franchisee's General Manager, if applicable) and any operations manager, will attend and complete, to Franchisor's satisfaction, the initial training offered by Franchisor. Franchisor may require any other principal or employee of Franchisee who is (or later becomes) actively involved in the management of the Franchised Business to attend and satisfactorily complete all initial training as Franchisor may require. Franchisor may elect to make any portion of the initial training to be virtual by way of internet video or audio link.

12.3 Supplemental and Refresher Training. Franchisee (and its General Manager, if applicable) will, and will cause any operations manager and other employees to, attend and satisfactorily complete all mandatory supplemental and refresher training, as Franchisor may require. In its discretion, Franchisor may require Franchisee (and its General Manager, if applicable), its operations manager and/or other employees to complete such training to Franchisor's satisfaction, regarding Enviro-Master Products and Services, and any other Products or Services authorized or required by Franchisor under the System, before Franchisee may offer or sell those Products or Services.

12.4 Substitute Trainees. If Franchisee or any of its employees is unable to complete any required training to Franchisor's satisfaction, Franchisee may designate a substitute trainee acceptable to Franchisor. Franchisor will have no obligation to accept any person's prior experience or training received from other sources as a substitute for compliance with Franchisor's training requirements.

12.5 Training Costs and Expenses.

(a) Franchisee or its employees will be responsible for all personal expenses incurred by them in connection with training, including but not limited to transportation, lodging and meal expenses, and wages and employee benefits. Franchisee must pay a fee (currently \$2,500 per person) if Franchisee elects to bring more than two (2) trainees to training.

(b) Franchisor may require Franchisee to make reservations for Franchisee or its employees in advance of attending any training. Franchisor may charge a deposit in connection with such reservations (which may be refunded or applied toward any training fee on attendance), and may charge a cancellation fee if any reservation is cancelled.

(c) Franchisor reserves the right to charge reasonable fees for training materials in connection with training other than initial management training. Franchisor will notify Franchisee of any fees for materials before Franchisee enrolls in training.

12.6 Management and Employees.

(a) Franchisee (or if accepted by Franchisor, a general manager ("General Manager") appointed by Franchisee) will be personally involved in the Franchised Business. If Franchisee purchases two or more Franchised Businesses, then Franchisee (or the General Manager) will ensure that the Franchised Businesses are at all times under the management and supervision of a trained designated operations manager acceptable to Franchisor. Franchisee (or the General Manager) will also ensure it hires and maintains a competent, conscientious, and trained staff. Franchisee will ensure that its employees comply with all requirements under the System, including any standards for dress and appearance. Franchisee will ensure that its employees maintain good customer relations and follow the policies and procedures of the System. Franchisee is responsible for its own personnel policies, hiring and firing of its employees, compliance with labor and employment laws, and all other employee matters.

(b) At Franchisor's option, before a General Manager is engaged, Franchisee must submit to Franchisor the proposed candidate's identity and qualifications, and Franchisor may accept or reject such candidate based on its commercially reasonable assessment of his/her management experience, qualifications and ability to maintain the Standards. Franchisor will not unreasonably withhold acceptance if the General Manager meets the minimum qualifications and completes the initial training program. If the General Manager fails to ensure that the Franchised Business satisfies the terms of this Agreement and complies with Franchisor's Standards, then Franchisor may require Franchisee to hire a new General Manager. Franchisee, or the General Manager (as applicable) are solely responsible for hiring any personnel of the Franchised Business and determining the terms and conditions of their employment. The General Manager must have at least a 10% equity interest in the business entity that is the franchisee of your Franchised Business.

Market Development. Franchisee will develop its Territory by executing 12.7 Franchisor's growth strategies, including without limitation, the Retain-Grow-Gain strategy. Franchisee will employ and maintain the number of full-time sales associates as determined necessary by Franchisor in its sole discretion. Currently, Franchisor requires a minimum of one full-time sales associate for each Territory (the "Minimum Sales Associate Requirement"). Franchisee may also designate themselves as the full-time sales associated of one Territory. Franchisor may adjust and redefine the Minimum Sales Associate Requirement at any time and from time to time in its sole discretion. Franchisee will be required to hire its first full-time sales associate prior to opening for business. After receipt of written notice that Franchisor has increased its Minimum Sales Associate Requirement, Franchisee must then add an additional sales associate every 6 months until the Minimum Sales Associate Requirement is met. If a sales associate of Franchisee is fired, dies, voluntarily, resigns or otherwise terminates his or her employment or association with Franchisee for any reason, Franchisee will have sixty (60) days to meet the Minimum Sales Associate Requirement.

Franchisee (or its owner if Franchisee is an entity) may be included as one of the required sales associates provided Franchisee is selling a minimum of \$100 per week in new service revenue on average for any audited 8-week period or any other such minimum threshold that Franchisor sets forth in its Manual.

If this Agreement is signed due to the purchase of an existing Enviro-Master franchise, Franchisee will be required to hire their initial sales associate prior to taking over the business provided that the current franchisee does not have the required number of sales associates.

12.8 Regional and National Meetings. Franchisee (and Franchisee's General Manager, if applicable) and Franchisee's designated operations manager, and all sales personnel will attend all regional and national meetings scheduled by Franchisor for the benefit of the Enviro-Master Franchise System. Franchisor may require, in its sole discretion, that any or all of Franchisee's managers and employees attend Regional and National Meetings. Currently, Franchisor charges a reasonable attendance fee per attendee for the Regional and National Conferences to cover any and all costs associated with hosting such meetings. Franchisee and its employees will be responsible for all personal expenses incurred by them in connection with such meetings, including but not limited to transportation, lodging and meal expenses, and wages and employee benefits. Should Franchisee fail to have all required individuals attend any National or Regional Meeting, Franchisee must:

(a) Pay Franchisor \$3,000 per person who has failed to attend the meeting; and

(b) Reimburse Franchisor for the full cost of sending training personnel to train Franchisee's staff, including the cost of the trainer's time plus all travel, lodging and meal expenses incurred by the trainer(s).

This provision is in addition to Franchisor's rights and remedies under Article 17.

12.9 Inside Sales Department. Franchisee must utilize Franchisor's internal employees who assist franchisees in identifying potential new clients and setting sales calls (the "Inside Sales Department") for Franchisee and/or Franchisee's sales associates for the first 90 days that Franchisee is in business. Currently, the cost to utilize the Inside Sales Department is \$380 per week or \$4,560 for your first three months. If Franchisee is purchasing two or more Franchised Business simultaneously under a Multi-Unit Development Addendum, then Franchisee must only pay the Inside Sales Department fees for the first Franchised Business.

12.10 Sales Training Program. If Franchisee elects to participate in a sales training program for Franchisee's Sales Associates(s), Franchisor will provide Franchisee's Sales Associate(s) with either (A) recruiting assistance or (B) on-going sales mentoring. If Franchisee participates in the sales training program, Franchisor will withdraw fees from Franchisee's account in the same manner that royalties are remitted. Currently, we charge \$750 per four-week period, per Sales Associate for mentoring services and \$3,000 per role for sales associate recruiting services.

13. OPERATIONS

13.1 Compliance with System. Franchisee will operate the Franchised Business in strict conformity with all prescribed methods, procedures, policies, standards, and specifications of the System, and with required Products and Services, as set forth in this Agreement, in the Manual or in writing by Franchisor.

13.2 Changes to System. Franchisee understands and acknowledges that businesses change and evolve from time to time for a variety of reasons, including without limitation efforts to improve the business and in anticipation of and response to changes in market conditions. Franchisor may make additions to, deletions from, and modifications to the System and Manual at any time and from time to time in Franchisor's sole and absolute discretion, including without limitation, additions, deletions and modifications to the required Products and Services, the Marks, programs, policies, standards for operating the Franchised Business, the Manual, computer systems, equipment, and trade dress. Any or all of these changes may require Franchisee to incur additional expenses in their implementation. Franchisee agrees to immediately implement all such additions, deletions and modifications to the System and Manual as set forth by Franchisor and to pay all costs and expenses associated with such implementation.

13.3 Opening for Business.

(a) **When Due.** Within 90 days after execution of this Agreement, Franchisee must secure and open its Business Location at the location acceptable to Franchisor.

(b) **Lease.** Franchisor may require Franchisee to obtain a written agreement from the lessor of the Business Location including any or all of the following terms:

Franchisor must have the right (but not the obligation) to assume the lease if the lease or the Franchisee is terminated because of a default by Franchisee, and the right to assign the lease to any person who meets Franchisor's qualifications for a new franchisee. Regardless of any right of Franchisor to assume and/or assign the lease, Franchisee will remain responsible for the performance of all of its obligations under its lease.

13.4 Required Vehicle. Franchisee is required to buy or lease a new truck or van that meets Franchisor's standards for the operation of the Franchised Business. Franchisee must outfit the required vehicle with all equipment, logo packages and other items, all as specified in our Manual or other written materials. Franchisor reserves the right to modify the standards and specifications for the required vehicle from time to time at its discretion. If Franchisee is purchasing two or more Franchised Businesses simultaneously under a Multi-Unit Development Addendum and its territories are adjacent, then, if approved by Franchisor, Franchisee must only maintain one vehicle.

13.5 Maintenance; Refurbishment. At all times, Franchisee will maintain the vehicles used in connection with the Franchised Business in good, clean, attractive and safe condition. Franchisee will, at its expense, undertake all maintenance and make all repairs, replacements, alterations, and additions as may be required for that purpose. Franchisee must replace its required vehicle at least once every five years with a new vehicle.

13.6 Hours of Operation. Franchisee will maintain the minimum hours and days of operation as specified in the Manual, unless Franchisor expressly permits otherwise in writing.

13.7 Authorized Products and Services. Franchisee will offer and sell all Products and Services required by Franchisor to be sold under the System. Franchisee will not offer or sell any other Products or Services of any kind or character without the express prior written consent of Franchisor. Franchisee will discontinue offering any Products or Services (whether or not they were previously authorized by Franchisor) promptly upon notice from Franchisor.

13.8 Business Forms. In offering and selling Services and Products to customers, Franchisee will use only the standard forms approved by Franchisor and no other forms or documents except with the express prior written permission of Franchisor.

13.9 Relations with Customers and the Public. Franchisee will not engage or cooperate in any conduct that:

(a) reflects unfavorably on the reputation of Franchisee, Franchisor, the Franchised Business or the System;

(b) impairs the goodwill associated with the Proprietary Marks;

(c) might constitute a deceptive or unfair trade practice or otherwise violate applicable law or regulations; or

(d) interferes, or may interfere, in the contractual rights of other Enviro-Master franchisees or Franchisor.

13.10 Individual Variances. Franchisee hereby acknowledges that complete and detailed uniformity among Enviro-Master franchise businesses under varying conditions may be inadvisable, impractical or impossible. Therefore, Franchisee agrees that Franchisor, at its sole discretion, may modify or vary aspects of the System with respect to any franchisee or group of franchisees based on such factors as local conditions, sales potential, demographics, competition, or any other conditions or circumstances that Franchisor deems a reasonable basis for such variances. Franchisee further agrees that Franchisor will have no obligation to disclose or offer the same or similar variances to Franchisee.

13.11 Computer Requirements. Franchisee acknowledges that it may be necessary to use computer equipment for the efficient operation of the Franchised Business, and that such necessity is likely to change during the term of the Franchise according to the circumstances of the Franchised Business and the System. Franchisor and Franchisee therefore agree that:

(a) <u>Hardware</u>. Franchisee must have or obtain a personal computer. In addition, Franchisee must acquire a hand-held computer device for employees working outside the office (the "**Hand-Held Device**"). Franchisor will provide Franchisee with specifications for the Hand-Held Device in the Manual and/or materials. Franchisee may obtain the hardware from any vendor Franchisee chooses at Franchisee's expense. Franchisor may require Franchisee to update Franchisee's hardware. There are no limits on the cost or frequency of this obligation. Franchisor will have independent access to all information stored on the hand-held device.

(b) Franchisor will provide Franchisee with specifications for Software. required computer software for Franchisee's personal computer in the Manual and/or materials. Franchisee must update Franchisee's operating software as Franchisor directs, and there are no contractual limitations on the frequency of this obligation. Franchisee must also obtain Internet access at Franchisee's expense for business communications. In addition, Franchisee must utilize certain particular approved computer software particularly related to the Franchised Business (the "Approved Software"). The Approved Software will allow Franchisee to schedule service to customers, change route assignments for each employee, access accounts receivable and aging of accounts receivable, and to perform other functions. Franchisee may not substitute any other software for these functions. Franchisor will purchase the Approved Software from a third party vendor and provide it to Franchisee. The cost of the Approved Software and any updates is included in the Technology Fee and the Service Fees Franchisee pays Franchisor. Franchisor may supplement, substitute, discontinue or modify the Approved Software and its conditions for Franchisee on reasonable notice to Franchisee.

(c) Except as described above, Franchisor will not guarantee, warrant, maintain or support any computer system. Franchisor may require Franchisee to agree to maintenance or support contracts. Because Franchisor does not know when other computer requirements and updates may become necessary, Franchisor cannot estimate the cost at this time. Franchisee understands and acknowledges that: (a) it will have to maintain and update its computer hardware and software from time to time, at Franchisor's direction; (b) that the cost of doing so is Franchisee's responsibility; and (c) such costs are difficult to estimate.

(d) Franchisor may add to, remove, substitute, or modify computer requirements in its discretion, and Franchisee will comply with such changes at its own expense. Franchisor will provide Franchisee with standards and specifications for required computer products, instructions for use, and updates of the same in the Manual. Franchisor may require Franchisee to obtain Internet access for computer communications, and obtain maintenance and/or service contracts at Franchisee's expense. Franchisee will obtain only approved computer Products from designated vendors (as provided in Article 14), except that Franchisee may obtain computer hardware from any vendor it chooses. Franchisor may provide its own proprietary software to Franchisee.

13.12 Business Premises Refurbishment.

In addition to all your other obligations in the Franchise Agreement and Operations Manual related to repairing and maintaining the Business Location, at our request, but not more often than once every 5 years, unless sooner required by your lease, you must refurbish the Business Location at your expense, to conform to the business, trade dress, color schemes and presentation of the Marks in a manner consistent with the then-current image for new Enviro-Master businesses ("**Refurbishments**"). Refurbishments may include structural changes, installation of new equipment and signs, remodeling, redecoration and modifications to existing improvements. We are unable to estimate your costs for future Refurbishments which will vary from business to business based on a number of factors like: (i) the market where your Franchise Business is located; (ii) the size of your Franchised Business; (iii) when your Franchised Business was last refurbished, if applicable; (iv) the amount of System changes since the last Refurbishment; and (v) the overall condition of your Franchised Business site and equipment.

13.13 Temporary Management.

(a) **Right to Appoint Manager.** If at any time Franchisor reasonably determines that Franchisee is in default under this Agreement, or that default is imminent, Franchisor will have the right (but no obligation) to appoint a temporary manager for the Franchised Business. The temporary manager may be an employee of Franchisor or an independent contractor.

(b) **Income and Expenses.** All income received from the temporary operation of the Franchised Business while it is under temporary management will be deposited in a separate account, from which expenses will be drawn. While the Franchised Business is under temporary management, Franchisor may draw directly from such separate account a management fee up to 30% of the monthly Gross Revenues of the Franchised Business, in addition to Royalty Fees, Service Fees, National Advertising Fees and other debts and obligations of Franchisee to Franchisor and its affiliates.

(c) **No Fiduciary Obligation.** If Franchisor elects to appoint a temporary manager, Franchisor will have the right to discontinue the arrangement at any time, without prior notice. Franchisor will have no fiduciary obligation to Franchisee or the Franchised Business and, while the temporary management arrangement is in effect, will be obligated to exercise reasonable efforts only. Franchisor will at all times retain all of its rights under this Agreement, including the right to terminate the Franchise in accordance with Article 17.

(d) **Liability for Debts.** Franchisor will not be liable for any debts, losses or obligations of the Franchisee or the Franchised Business, whether incurred before, during or after the period of temporary management. If the income of the Franchised Business while under temporary management is insufficient to cover expenses, Franchisor will notify Franchisee (or Franchisee's executor, administrator or personal representative). Franchisee will be responsible for providing and maintaining a sufficient balance of funds for the operation of the Franchised Business.

13.14 Inspections. Franchisor or any of its authorized agents or representatives may, without notice, inspect the Franchised Business, during normal business hours to determine whether it is in compliance with this Agreement and with the System. Franchisor's right to inspect the Franchised Business includes, but is not limited to:

(a) the right to inspect interior and exterior of Franchisee's vans or other vehicles used in operating the Franchised Business; the right to inspect the interior and exterior of the Business Location;

(b) the right to observe and shadow Franchisee's meetings with current and prospective customers;

- (c) the right to observe and shadow Franchisee's service appointments; and
- (d) the right to inspect Franchisee's books and records as set forth in Section

8.7 hereof.

Franchisee agrees to cooperate with Franchisor in facilitating Franchisor's inspections. If Franchisee fails to correct any deficiencies discovered in an inspection, Franchisor may make the corrections at Franchisee's expense and charge Franchisee a Repeated Inspection Fee (as defined below). If Franchisee fails an inspection and Franchisor or its representatives return to inspect Franchisee to confirm Franchisee has remedied the deficiency, Franchisee must pay Franchisor the cost of the inspection, including the travel and living expenses of Franchisor's representatives (the "**Repeated Inspection Fee**"). Currently, Franchisor estimates the cost of inspection for its representatives to be \$500 per person, per day.

14. APPROVED PRODUCTS AND SERVICES, AND DESIGNATED VENDORS

14.1 Minimum Inventory. Franchisee will maintain at all times the minimum stock levels of inventory and supplies specified by Franchisor in writing in the Manual or otherwise.

14.2 Designated Vendors. Franchisor will periodically designate, through the Manual or other written notice, equipment, inventory, supplies, tools, forms, promotional materials, and other Products and Services that must be purchased from approved suppliers, which shall include Franchisor, and will notify its franchisees of the approved supplier for each product. Franchisee will purchase or lease all such required Products and Services only from approved suppliers. Franchisee acknowledges that from time to time it will be necessary to introduce new Products or Services. Franchisee acknowledges and agrees that Franchisee will begin offering the new Products and Services within a period of 90 days from their introduction by Franchisor.

As of the date of this Franchise Agreement, we are the only approved supplier for all products and supplies utilized in your Franchised Business (other than office supplies and the power washer), such as paper products, restroom hygiene chemicals and drain line management products. The power washer is supplied by our approved supplier. Franchisee acknowledges that there is a benefit to the System in maintaining standardization in both function and appearance. Accordingly, although alternate products or services may be functionally the same or similar as authorized Products or Services, there may be differences in reliability and differences in appearance. Differences in appearance may diminish the value of the System by creating the appearance of inconsistency and uncertain standards. Additionally, Franchisee acknowledges that Royalties paid by Franchisee are lower than they would otherwise be if Franchisor did not have revenue from Franchisee's purchase of Products. Franchisee agrees not to directly or indirectly purchase Products or Services from unapproved suppliers, except in emergency circumstances and with Franchisor's prior written consent.

Franchisor may provide the Products or Services, or it may enter into agreements with approved suppliers for such Products or Services. Franchisor may enter into agreements with approved suppliers, pursuant to which Franchisor is compensated or reimbursed in connection with such approval. Franchisor may discontinue providing any Product or Service to Franchisee by giving Franchisee 30 days' written notice. Franchisor may periodically change the Business Services that it provides and the fees that Franchisee must pay for the Business Services. If Franchisor changes the Business Services, or incurs reduced or increased costs in providing Business Services, Franchisor may reduce or increase the service fees based on costs saved, or additional costs incurred. If Franchisor discontinues any Business Services involving the reporting

of Gross Revenues or the collection of Royalty and other fees, Franchisee must comply with Franchisor's reasonable reporting and collection procedures as provided in the Manual.

14.3 Procedure for Obtaining Designation. If Franchisee desires to purchase any product from an undesignated vendor, Franchisee or the vendor will submit to Franchisor a written request for designation. Franchisee may approve or reject a vendor requested by Franchisee in its sole and absolute discretion. Franchisee acknowledges that the sale of Products is a significant source of income to Franchisor. Thus, if Franchisee purchases Products from alternate vendors, once designated as approved by Franchisor, Franchisee agrees to pay Franchisor a twelve percent (12%) mark-up fee or such other fees determined by Franchisor on all such items purchased.

(a) **Inspection and Testing.** Franchisor reserves the right to require that its representatives be permitted to inspect any proposed vendor's facilities and those samples of Products from the vendor delivered to Franchisor or its designee for testing.

(b) **Review Fee.** Franchisor may impose a charge for review not to exceed the reasonable costs of inspection and testing, which will be paid by Franchisee or the proposed vendor.

(c) **Notification.** Franchisor will attempt to provide Franchisee with written notification of its decision about the designation of a supplier that Franchisee has proposed within 60 days after receipt of Franchisee's request, or within 30 days after completing any inspections or testing, whichever is later. If no response is received from Franchisor within such time period, the request shall be deemed denied.

14.4 Revocation of Designation. Franchisor may approve new suppliers or revoke past approvals of suppliers upon written notice to Franchisee.

14.5 Service Providers. Franchisee may not use any Services from unapproved third party service providers. If Franchisee desires to obtain Services from any undesignated third party service provider, Franchisor will have the right to review the terms and conditions of any arrangement between Franchisee and the third party provider, and to require additional information from the third party provider about its business background and qualifications before Franchisee uses any Services. This may include personal interviews with individuals providing services. Franchisor may, in Franchisor's sole discretion, disapprove (at any time) any third party provider. This provision shall not apply to electrical, gas, fixed-site telephone, cable, waste removal, landscaping or security services relating to Franchisee's Business Location.

14.6 Access to Confidential Information. If, in supplying Products or providing Services to Franchisee, any third party may obtain access to Confidential Information, Franchisor may require, as a condition of its approval, that the vendor or provider and its personnel execute confidentiality agreements in forms satisfactory to Franchisor.

14.7 Unauthorized Purchases. Except to the limited extent expressly permitted in Section 4.1, if Franchisee purchases Products or Services that are not approved by Franchisor or purchases Products or Services from unapproved suppliers, Franchisor reserves the right to charge Franchisee an additional royalty equal to forty-five percent (45%) of the gross amount of such purchase. If charged, payment shall be made upon the earlier of (a) thirty (30) days after receipt of the product or service or (b) three (3) days after payment to the supplier of the product or service. It shall be the duty of Franchisee to accurately and timely report such purchases to Franchisor. If Franchisee substitutes Products or Services without prior authorization, Franchisee shall also be in default of this Agreement, and Franchisor may exercise its right and remedies under Article 17.

14.8 Applicability to Business Services. The Business Services described in Section 4.3 are not Products or Services under this Article 14.

15. ADVERTISING, MARKETING AND PROMOTION.

15.1 Use and Display of Proprietary Marks. Franchisee may use and display the Proprietary Marks only in connection with offering and selling authorized Products and Services under the System, and for no other purpose.

(a) In using or displaying any Proprietary Marks, Franchisee will use the appropriate marks or symbols (such as [®], TM or SM), as required by Franchisor.

(b) Franchisee will ensure that all stationery, signs, and other printed materials used or displayed in connection with the Franchised Business bear the Proprietary Marks in the form, colors, location and manner prescribed by Franchisor.

15.2 Advertising Products and Services. In connection with the operation of the Franchised Business, Franchisee will advertise, offer and sell only the Products and Services authorized by Franchisor under the System.

(a) **Use of Materials Provided by Franchisor.** Franchisee will use and display all advertising and promotional materials provided to Franchisee by Franchisor, in the manner and for the time periods designated by Franchisor.

(b) **Prior Approval for All Other Materials.** Franchisee will not use or display any other advertising, marketing or promotional materials, whether print, electronic, audio, visual or Internet, which includes setting up a website for the Franchised Business (including any materials prepared by Franchisee or its agent) in connection with the Franchised Business unless Franchisee has obtained Franchisor's prior written approval of those materials.

(1) If Franchisee desires to obtain Franchisor's approval of advertising materials, Franchisee must submit samples for review by Franchisor.

B-39

Enviro-Master International Franchise, LLC FDD May 6, 2024 #198798v1 (2) If Franchisor does not approve the samples within 30 business days after receipt, Franchisee may not use the materials.

(c) **Discontinue Non-Approved Advertising.** Franchisee will stop using or displaying any advertising materials immediately on notice from Franchisor. This applies even if the materials were previously approved.

(d) National Advertising Activities.

Franchisee will pay the National Advertising Fee to a fund for the (1)promotion of the Franchised Business, the Marks and the System (the "National Advertising Fund"). Franchisee's National Advertising Fee is in addition to Franchisee's Minimum Local Marketing Expense and the Local Digital Marketing Campaign. Franchisor or an affiliate will administer the National Advertising Fund. Franchisor reserves the right to use any media, create any programs, and allocate advertising and promotional expenditures to any regions or locales it deems appropriate. Franchisor may use the National Advertising Fund to meet the costs of administering, preparing and conducting national, regional or local advertising, promotional and brand building programs and initiatives or any kind (collectively, "Brand Development Activities"), which includes but is not limited to the cost of (i) preparing and conducting television, radio, magazine, newspaper and digital advertising campaigns and other public relations activities (including, but not limited to, for purposes of brand reputation management), (ii) employing public relations firms and advertising agencies to assist in these activities, and (iii) conducting other activities that are directly or indirectly designed to attract national account customers, promote the System, its franchisees, and/or increase System sales, such as limited-time offerings, new product development, sales training programs, costs associated with the inside sales program, franchisee incentives, customized materials, guest response programs, manager/employee recognition programs, quality assurance programs, mystery shopper programs, brand website and ordering platforms, brand applications, social media account administration and promotion, and equipment and technologies related to such marketing programs.

(2) Franchisee represents and acknowledges that, with respect to the National Advertising Fund:

(i) Franchisor is not obligated to conduct any advertising or spend any amount on advertising in Franchisee's Territory. Franchisor may periodically formulate, develop, produce, and conduct, at its sole discretion, advertising or promotional programs in such form and media as Franchisor determines to be most effective.

(ii) Franchisor has sole authority to direct all advertising programs and promotions and use of the National Advertising Fund, with sole control over the creative concepts, materials and media used in the programs, and the placement and

allocation of advertising. Franchisor reserves the right to use any media, create any programs, and allocate advertising and promotional expenditures to any regions or locales it deems appropriate.

(iii) Franchisor is not required to spend any amount from the National Advertising Fund on advertising or promotions in Franchisee's Territory. Franchisor has no obligation to make expenditures from the National Advertising Fund that are equivalent or proportionate to Franchisee's contributions, ensure that Franchisee benefits directly or proportionally or in any amount from the placement of advertising, or ensure that any advertising impacts or penetrates Franchisee's Territory.

(iv) Franchisor may use the National Advertising Fund to compensate Franchisor for its reasonable administrative costs and overhead it incurs in connection with all Brand Development Activities, and any other activities related to advertising and promotional programs, including new product development; market research; preparing advertising and promotional materials; digital marketing; working with public relations firms, advertising agencies, advertising placement services and creative talent; reimbursing franchisees advisory council meeting expenses; developing and maintaining, and paying third parties for the development and maintenance of internet sites, applications, and other equipment and technology related to marketing programs.

(v) Franchisor may use the National Advertising Fund to pay for Brand Development Activities in certain limited territories only and not Franchisee's Territory. Franchisor is not obligated to spend the National Advertising Fund on all territories equally, proportionately or at all.

(vi) If any contributions to the National Advertising Fund, including any associated earnings, are not spent in the fiscal year in which they accrue, they remain set aside in the National Advertising Fund for use in later years.

(vii) Any amounts Franchisor or its affiliates contribute to the National Advertising Fund will be considered an advance from Franchisor or its affiliates to the National Advertising Fund. Franchisor and/or its affiliates have the right to be reimbursed from the National Advertising Fund any amounts that Franchisor advanced to the National Advertising Fund.

(viii) Franchisor currently does not, and is not required to, maintain the National Advertising Fund, your National Advertising Fees, or income earned from contributions to the National Advertising Fund, in a separate account from Franchisor's other money. Franchisor's other franchisees may not be required to

contribute to the National Advertising Fund, may be required to contribute to the National Advertising Fund at a different rate than Franchisee, or may be required to contribute to a different advertising fund. Franchisor-owned outlets, if any, will not be required to contribute to the National Advertising Fund.

(ix) The advertising may include promotions in areas where Franchised Businesses are not yet established. Franchisor may create the advertising in-house or use any outside agency, depending upon the type of media.

(x) Franchisor is not required to have an independent audit of the National Advertising Fund completed.

Local Advertising Activities. Franchisee will pay for local advertising, (e) promotional and marketing activities in Franchisee's Territory in an amount determined by Franchisor, not to exceed 2% of Gross Sales (the "Minimum Local Marketing Expense"). All advertising, promotional, and marketing activities conducted by Franchisee in Franchisee's Territory will be subject to the prior approval of Franchisor. Franchisee will submit to Franchisor all local advertising, promotional and marketing plans as required by Franchisor. Franchisor will collect this payment from Franchisee and remit one payment monthly to the vendor. Franchisee will discontinue any local advertising plans or activities promptly on notice from Franchisor. All franchisees are required to participate in the local digital marketing campaign approved by the Franchisor (the "Local Digital Marketing Campaign"). Franchisee must utilize Franchisor's approved vendor and the current cost is \$350 per month payable to us for pay-per-click advertising and search engine optimization. At Franchisee's election, Franchisee may spend more than the monthly required amount. Franchisor collects the monthly fee from all franchisees and remits it directly to the vendor. The expense of such campaign does not apply to the Minimum Local Marketing Expense. We may require you to be listed locally or in other designated areas on electronic media, particularly certain websites on the Internet.

15.3 Website and Internet Advertising. No advertising or promotion by Franchisee shall be conducted on or through the Internet or other electronic transmission via computer without express prior written approval by Franchisor, including all social media sites. Franchisee shall ensure its employees do not make any website or social media post using the Marks without obtaining Franchisor's prior written approval. Without limiting the generality of the foregoing, Franchisee, without the express prior written approval of Franchisor, shall not operate, or permit to be operated on its behalf, any internet which incorporates any of the Marks or otherwise promotes the Franchised Business. All advertising and promotion by Franchisee must be factually accurate and shall not detrimentally affect the Marks or the Enviro-Master System, as determined in Franchisor's sole discretion. The approved Local Digital Marketing Campaign described in 15.2(e) has received the approval of the Franchisor. Additionally, there is an approved social media advertising provider approved by the Franchisor which can be hired by Franchisee if

Enviro-Master International Franchise, LLC FDD May 6, 2024 #198798v1 Franchisee chooses to do so to manage all of its social media accounts, subject to Franchisor's right of approval on any such social media advertising content and placement.

16. EVENTS OF DEFAULT.

16.1 Definition. The actions, occurrences, and breaches described in Sections 16.2 and 16.3 shall constitute an "Event of Default" under this Agreement.

16.2 Events of Default with No Right to Cure. Immediately upon the occurrence of any of the following events, all of which shall constitute an Event of Default, Franchisor may, but shall not be obligated to, exercise any of the remedies (or combination thereof) provided for in Article 17 including, but not limited to termination of this Agreement without providing Franchisee any notice or opportunity to cure:

(a) Franchisor discovers that Franchisee made any material misrepresentation or omission in its application for the Franchised Business opportunity or otherwise to Franchisor in the course of entering into this Agreement;

(b) Franchisor discovers that by entering this Agreement, Franchisee violated a non-competition agreement by which it is bound;

(c) Franchisee is convicted of or pleads no contest to a felony or other crime or offense that, in Franchisor's sole discretion and judgment, is likely to adversely affect the reputation of Franchisor, Franchisee, the System, the Franchised Business, or other Enviro-Master businesses;

(d) Franchisee engages in any action, behavior, or conduct that, in Franchisor's sole discretion and judgment, is likely to adversely affect the reputation of Franchisor, Franchisee, the System, the Marks, the Franchised Business, or other Enviro-Master businesses;

(e) Franchisee misuses or makes an unauthorized use of any of the Marks;

(f) Franchisee discloses, duplicates, or otherwise uses in an unauthorized manner any portion of the Manual or any other Confidential Information;

(g) Franchisee fails to open for business within the time required under this Agreement;

(h) Franchisee abandons, or fails or refuses to actively operate the Franchised Business for 5 or more consecutive days without Franchisor's express written permission;

(i) Franchisee relocates the Business Location without receiving Franchisor's express written permission;

(j) Franchisee's landlord notifies Franchisee that the landlord is re-taking possession of the Business Location, an eviction proceeding is filed against Franchisee, or Franchisee loses Franchisee's right to possession of the Business Location;

(k) Franchisee surrenders or transfers control of the operation of the Franchised Business without Franchisor's approval, makes or attempts to make an unauthorized direct or indirect assignment of the Franchised Business or an ownership interest in Franchisee, or fails or refuses to assign the Franchised Business or the interest in Franchisee of a deceased or incapacitated owner thereof, as herein required;

(1) Franchisee fails to maintain the Franchised Business under the primary supervision of an approved manager following the death or incapacity of Franchisee or any holder of a legal or beneficial interest in Franchisee or fails to transfer the Franchised Business to an approved transferee within 12 months following the death of incapacity of Franchisee, pursuant to Section 11.6;

(m) Franchisor's right to effect EFT pursuant to Section 3.3(b) is revoked or impaired as a result of any act or notice of Franchisee;

(n) Franchisee maintains false books or records or submits materially false statements or reports to Franchisor; without limiting this provision, any understatement of Gross Revenues by 2% or more during any period will automatically constitute a materially false statement by Franchisee;

(o) Franchisee:

(i) becomes insolvent by reason of Franchisee's inability to pay its debts as they become due;

(ii) makes an assignment for the benefit of creditors;

(iii) makes an admission of Franchisee's inability to pay its obligations as they become due;

(iv) files a voluntary bankruptcy petition;

(v) files any pleading seeking any reorganization, liquidation, dissolution, composition, or other settlement with creditors under any law;

(vi) admits or fails to contest the material allegations of any reorganization, liquidation, dissolution, composition pleading filed against Franchisee; or

(vii) is adjudicated a bankrupt or insolvent.

(p) A receiver or other custodian is appointed for the assets of Franchisee; or a final judgment remains unsatisfied or of record for ninety (90) days or longer (unless supersedeas bond is filed); or if execution is levied against any part of the assets of the Franchisee or suit to foreclose any lien or mortgage is instituted against the Franchisee and not dismissed within ninety (90) days; or if the real or personal property of the Franchisee is sold after levy of judgment thereupon by any sheriff, marshal or constable; or the claims of creditors of Franchisee are abated or subject to a moratorium under any law; or, in the event Franchisee is a single member limited liability company, and the sole member (individual or entity) files for bankruptcy or is adjudicated bankrupt;

(q) Franchisee commits any act or omission including, without limitation, violating any law or regulation or provision of this Agreement, that results in an immediate threat or danger to the health or safety of any person;

(r) Franchisee fails to comply with any law or regulation within twenty-four (24) hours after being given notice of non-compliance or within such other reasonable time applicable to such law or regulation;

(s) Franchisee fails to satisfactorily complete required training;

(t) Franchisee is subject to Minimum Annual Royalties as specified in Exhibit B for twelve months or more consecutive months.

(u) Franchisee breaches or fails to comply with: (i) any obligation under this Agreement, the Manual, or System three or more times in any 12 month period, or (ii) the same obligation under this Agreement, the Manual, or System two or more times in any six (6) month period, in either instance, (i) or (ii), regardless of whether Franchisor has provided notice of such previous breaches or failures and regardless of whether such previous breaches or failures were cured;

(v) Franchisee fails to establish or maintain the Minimum Sales Associate Requirement as set forth in Section 12.7;

(w) Franchisee, its affiliates or any owner of Franchisee defaults, beyond any applicable cure period, under any other agreement with Franchisor or its affiliates; or

(x) Franchisee, its affiliate, or any owner of Franchisee defaults, beyond any applicable cure period, under any agreement relating to or arising out of the operation of the Franchised Business, including without limitation, a lease, sublease, loan agreement, or security interest, with any third party.

16.3 Events of Default with 30 Day Opportunity to Cure. Immediately upon the occurrence of any of the following events (to the extent these events are not already set forth in

May 6, 2024 #198798v1 Section 16.2), all of which shall constitute an Event of Default, Franchisor may, but shall not be obligated to, exercise any of the remedies (or combination thereof) provided for in Section 17 including, but not limited to termination of this Agreement, provided that Franchisee fails to cure the Event of Default to Franchisor's satisfaction within thirty (30) days of written notice to Franchisee:

(a) Franchisee breaches or fails to perform or observe any covenant, duty or obligation contained in this Agreement; provided, however, that if Franchisee has commenced to diligently and expeditiously cure such default in the cure period, as determined in the sole discretion of the Franchisor, the cure period shall be extended for a period of time as is commercially reasonable under the circumstances (not to exceed ninety (90) days from the notice of breach) and for so long as Franchisee is diligently and expeditiously pursuing a cure;

(b) Franchisee fails to obtain Franchisor's prior approval or consent as required under this Agreement;

(c) Franchisee fails to offer all Services or Products as required by Franchisor, offers any unapproved services or products, or makes unauthorized changes to the Business Location;

(d) Franchisee tortiously interferes with Franchisor's contractual or advantageous business relationships;

(e) Franchisee fails to operate the Franchised Business in compliance with the System Standards;

(f) Franchisee fails to: (i) ensure that any person required by Section 6.3(e) executes a Confidentiality Agreement, or (ii) provide Franchisor with copies of such agreements required to be signed;

(g) Franchisee fails to submit reports or other information or supporting records when due;

(h) Franchisee fails to make any payment to Franchisor or its affiliates when

due;

(i) Franchisee engages in any activity exclusively reserved to Franchisor;

(j) Franchisee fails to maintain insurance as specified in Section 9.1 of this Agreement or provide certificates of insurance to Franchisor as required;

(k) Franchisee conducts any other business or engages in marketing any other products or services that Franchisor reasonably considers to be similar to the Products or Services

authorized under this Agreement, or uses any other name or marks that Franchisor reasonably considers to be similar to the Marks;

(1) Franchisee changes any invoice that is produced by Franchisor's proprietary software system and sent to Franchisee for delivery to any customer serviced by the Franchised Business unless Franchisee first makes a written request to Franchisor that is approved in writing by Franchisor. Unauthorized alterations include, but are not limited to: (i) changing the name or address of Franchisor, (ii) changing the return address to which the customer is to mail payment, (iii) changing the amount of the invoice in any manner not authorized by Franchisor, (iv) changing the manner, place or other terms of payment, (v) extending or otherwise changing the due date for payment, or (vi) directly or indirectly changing an invoice in any other manner (e.g., if a customer has a "cash on delivery/C.O.D." account as defined in the Manual, Franchisee has made an unauthorized alteration if Franchisee fails to collect cash from the customer on delivery, or if a customer has a charge account that is not properly reported to Franchisor as revenue, Franchisee has made an unauthorized alteration if Franchisee collects or attempts to collect cash or other payment from the customer). Any unauthorized alteration of an invoice is a default hereunder;

(m) Franchisor receives a written complaint from any Regional or National Account customer of Franchisor or Franchisee that it intends to terminate its contract in whole or in part on account of poor service by Franchisee;

(n) Franchisee fails to service any Regional or National Account customer in the manner required by this Agreement; or

(o) Franchisee fails to pay the \$3,000 per person charge and/or complete and pay for remedial training after Franchisee, or any of its managers, employees or sales people miss any regional or annual meeting or conference.

16.4 Default by Managing Owner. Any action or omission by any manager of Franchisee shall be deemed an action or omission by Franchisee for purposes of determining whether an Event of Default has occurred pursuant to this Section 16.

17. REMEDIES AND TERMINATION

17.1 Remedies Upon Event of Default. Upon the occurrence of any Event of Default, and subject to any applicable cure period, Franchisor may in its sole discretion, immediately exercise any or all of the following remedies, in addition to all other rights and remedies available to Franchisor under this Agreement or the law:

(a) Terminate this Agreement and any other agreement, including other franchise agreements, that Franchisor and Franchisee have executed, effectively immediately or effective upon a future date determined by Franchisor;

(b) Reduce the size of the Territory;

(c) Terminate Franchisee's protected rights in all or part of the Territory for the remainder of the term of the Franchise Agreement or such other time period Franchisor determines in its sole discretion, without refunding any fee paid for such rights;

(d) Suspend Franchisee's access to the intranet provided that Franchisee shall remain responsible for all costs of participation;

(e) Suspend Franchisee's access to any advertising or marketing materials or assistance provided for franchisees;

(f) Remove Franchisee from the Enviro-Master website;

(g) Remove Franchisee from any advertising materials;

(h) Suspend or terminate any fee reductions which Franchisor might have agreed to during the term of this Agreement;

(i) Require Franchisee to conduct an annual audit of its financials during the term of this Agreement at Franchisee's sole cost;

(j) Require Franchisee, its managers, or other employees of the Franchisee to participate in additional training;

(k) Suspend the provision of any operational support that this Agreement otherwise requires Franchisor to provide;

(1) Suspend the provision or supply of any Services or Products for which Franchisor is an approved supplier;

(m) Take any action to cure a breach or default on Franchisee's behalf and require Franchisee to reimburse Franchisor for all costs and expenses (including the allocation of any internal costs) for such action, plus a 10% administrative fee;

(n) Assume or appoint a third-party to assume the management of the Franchised Business as set forth in Section 13.13;

(o) Charge Franchisee a non-compliance fee up to \$500 for each time Franchisee defaults and \$500 per week for each week the Event of Default remains uncured;

(p) Eliminate any remaining renewal rights that Franchisee may have; or

(q) Reduce, modify, suspend, or otherwise terminate any other of Franchisee's rights under this Agreement while such Event of Default continues or for such other period of time that Franchisor, in its sole discretion deems appropriate, provided that Franchisee shall remain responsible for all fees and obligations under this Agreement.

17.2 General Provisions Concerning Default and Termination.

(a) **Reliance on Defaults.** In any arbitration or other proceeding in which the validity of any termination of this Agreement or Franchisor's refusal to enter into a renewal Franchise Agreement is contested, each party may cite to and rely upon all defaults or violations of this Agreement, not only the defaults or violations referenced in any written notice.

(b) **Notification to Third Parties**. Franchisee agrees that Franchisor has the right and authority (but not the obligation) to notify any or all of Franchisee's owners, lenders, landlords, creditors, vendors, or suppliers, if Franchisee commits an Event of Default or if Franchisor terminates this Agreement.

(c) **No Constructive Termination and Election of Remedies**. Unless Franchisor expressly terminates this Agreement, Franchisor's exercise of any of the foregoing remedies will not constitute an actual or constructive termination of this Agreement nor will it be Franchisor's sole and exclusive remedy for Franchisee's default or failure to comply with this Agreement.

(d) **Ongoing Payment of Fees.** During any suspension period, Franchisee must continue to pay all fees and other amounts due under, and otherwise comply with, this Agreement and all related agreements. If Franchisor restores any of Franchisee's rights, Franchisee will not be entitled to any compensation for any fees, expenses, or losses Franchisee might have incurred due to Franchisor's exercise of any of its rights provided in Section 17.1.

(e) **No Waiver.** If Franchisor exercises its right not to terminate this Agreement but to implement any remedies in Section 17.1, Franchisor may at any time after the appropriate cure period under the written notice has lapsed (if any) terminate this Agreement without giving Franchisee any additional corrective or cure period. Franchisor's exercise of its rights under Section 17.1 will not be a waiver by Franchisor of any breach of this Agreement.

(f) **Liquidated Damages.** If Franchisor terminates the Franchise Agreement prior to the expiration of the term, Franchisee shall pay Franchisor liquidated damages in an amount equal to the average Royalty Fees owed by you (even if not paid) per month over the 12month period preceding the date of termination (or, if the Franchised Business was not open throughout such 12-month period, then the average Royalty Fees earned per month for the period in which the Franchised Business was open), multiplied by the lesser of: (i) 24; or (ii) the number of months remaining in the term of the Franchise Agreement. Franchisee acknowledges and understand that the liquidated damages are not a penalty, but rather a reasonable pre-estimate of

B-49

the damages suffered by Franchisor due to Franchisee's failure to continue operating the Franchised Business for the remainder of the term of this Franchise Agreement. This liquidated damages provision will not limit Franchisor's rights to injunctive relief relating to any violations of this Franchise Agreement, nor limit any other damages available to us arising out of such violations, including without limitation brand damage.

18. OBLIGATIONS ON TERMINATION OR EXPIRATION.

18.1 Obligations of Franchisee. Upon the expiration or earlier termination of this Agreement, all of Franchisee's rights hereunder will terminate. Franchisee will comply with all requirements under this Agreement which expressly or by reasonable implication apply to Franchisee's conduct after termination or expiration, including but not limited to the following:

(a) Immediately cease to operate the Franchised Business and at no time thereafter represent itself to the public, directly or indirectly, as a current or former franchisee of Franchisor;

(b) Immediately and permanently cease to use, in any manner whatsoever, the System, and all Proprietary Marks, including all distinctive logos, symbols, slogans and graphics associated with the Proprietary Marks or System;

(c) Cancel any government registrations or licenses that Franchisee may hold in the name of "Enviro-Master Services" (or any other name designated by Franchisor for the System) and submit to Franchisor proof of compliance with this obligation within 30 days after termination or expiration of the Franchise;

(d) Immediately deliver to Franchisor or its designee the Manual and all other materials relating to operation of the Franchised Business (including but not limited to correspondence, drawings, files, manuals, handbooks, invoices, plans, programs, records, samples, standards and specifications, and all copies thereof, all of which are acknowledged to be Franchisor's property);

(e) Immediately deliver to Franchisor or its designee all products purchased from Franchisor or its designated suppliers and all products bearing the Proprietary Marks (including all distinctive logos, symbols, slogans and graphics associated with the Proprietary Marks or System), including but not limited to any such products that were installed in the facilities of Franchisee's customers such as dispensers, urinal shields, and cleaning fluid, or alternatively, provide to Franchisor sufficient documentation (as determined by Franchisor) evidencing that Franchisee sold such products to another Enviro-Master franchisee;

(f) Cease use of the customer lists, customer service contracts and related records and transfer possession of such items to Franchisor. Franchisee acknowledges that the customer base is a part of the overall system and is owned by the Franchisor;

B-50

(g) Immediately collect and deliver to Franchisor all advertising, brochures, literature, posters and any other items bearing the Proprietary Marks which have been provided, by or on behalf of Franchisee, to 3rd parties to distribute or display;

(h) Refrain from making any change, deletion or erasure to the computer system or systems (including all storage media or devices) used in the Franchised Business and immediately make available to Franchisor or its agents full and complete access to such systems (free of encryption or password protection) for the purpose of copying any information thereon; and

(i) Promptly pay all sums owing to Franchisor and its affiliates.

18.2 Assignment. At Franchisor's option, on termination or expiration of the Franchise, Franchisee will assign to Franchisor or its designee any interest which Franchisee has in any lease or sublease for the Franchised Business premises, if any, and any or all customer service agreements then in effect.

18.3 De-Identification.

(a) If Franchisor does not require Franchisee to assign the lease or sublease for the Business Location if any, Franchisee will ensure that all equipment, fixtures, other products and decor reflecting the Proprietary Marks and System (including any distinctive decorative features or color schemes) are removed from the Business Location, as Franchisor may consider necessary to prevent confusion, mistake, or deception.

(b) Franchisee will comply with Franchisor's instructions to cancel, forward or assign Franchisee's telephone number for the Franchised Business.

(c) Franchisee will remove any Enviro-Master trademark or logo or other information regarding the Franchised Business or the System from any vehicle which has been used for the Franchised Business.

18.4 Cooperation by Franchisee. Franchisee will assist and cooperate with Franchisor and its agents to accomplish the de-identification and/or the orderly transition of the formerly Franchised Business as required under this Article 18. Franchisee shall promptly provide Franchisor with access to all books and records of Franchisee and all information relating to employees, customers and prospects of the Franchised Business. If Franchisee fails or refuses to comply with the requirements of this Article 18, Franchisor and its agents will have the right to enter Franchisee's Business Location, if any, and make all required changes, at Franchisee's expense, which Franchisee agrees to pay on demand. Franchisee will indemnify Franchisor and its agents to de-identify the Business Location.

18.5 Stop Using Related Social Media and Online Business Directories.

(a) Franchisee must immediately cease operating all social media pages within Franchisee's control associated with, or previously associated at any time with, the Franchised Business and Enviro-Master®, including but not limited to Facebook, Instagram, YouTube, TikTok, and X (f/k/a Twitter). Franchisee must also immediately cease operating all online business directory listings within Franchisee's control associated with, or previously associated with, the Franchised Business and/or Enviro-Master®, including but not limited to Yelp, NextDoor, LinkedIn, Google, YP (Yellow Pages), or Angi. Franchisee must promptly provide Franchisor with all login credentials or other information necessary for Franchisor to assume exclusive control over each social media and business directory account, page or listing. To the extent that Franchisee is aware of or becomes aware of any social media or business directory account, page or listing associated with the Franchised Business that is not within Franchisee's control, Franchisee shall promptly notify Franchisor thereof in writing.

(b) Notwithstanding the foregoing, Franchisor may, in its exclusive discretion, demand that Franchisee delete, deactivate or otherwise modify such social media or business directory account or listing at any time. Franchisee must comply with any such demand immediately upon receipt.

(c) Franchisee further acknowledges and agrees that all consumer or other published reviews of the Franchised Business and/or any goods or services provided by the Franchised Business, are the exclusive property of Franchisor and/or its affiliates. Franchisee's right to use such reviews in any manner terminates concurrently with the expiration or termination of this Agreement. Franchisee is prohibited from advertising, promoting, quoting, or otherwise referring to such reviews in connection with any business or offer to conduct business upon expiration or termination of this Agreement.

(d) Franchisee further acknowledges and agrees that any violation of this Section 18.5 constitutes trademark infringement, service mark infringement, unfair competition, false advertising, and/or deceptive trade practices pursuant to federal, state, and common law, that such violation encroaches on the goodwill associated with Franchisor's brand, and that such violation is likely to cause confusion among reasonably prudent consumers.

19. OPTION TO PURCHASE THE ASSETS OF THE FRANCHISED BUSINESS.

(a) Upon expiration or earlier termination of the Franchise Agreement, Franchisor or Franchisor's designee (which may or may not be another affiliate or franchisee of Franchisor) shall have the option (but not the obligation) to purchase any or all of the assets of the Franchised Business (the "**Purchase Option**") using the following formulas to calculate the purchase price for the following assets: (i) the tangible assets of the Franchised Business other than inventory at a price for each item based on its purchase price, reduced one-fourth for depreciation for each twelve month period since its acquisition;

(ii) the inventory of supplies and Products of the Franchised Business at the price of ninety percent (90%) of Franchisee's cost (provided that obsolete or unsalable items shall be at twenty percent (20%) of cost);

(iii) all of the accounts receivable of the Franchised Business, at a price calculated at eighty-five percent (85%) of the face amount for all accounts receivable that have been outstanding for less than 90 days; and/or

(iv) and the other intangible assets of the Franchised Business for an aggregate amount of \$1.00

(Items (i)-(iv) above shall be collectively referred to as "**Purchased Assets**" or "**Purchase Price**" as applicable). Notwithstanding the foregoing: the Purchase Price shall be reduced by the amount of any indebtedness or liability (direct or indirect) the purchaser may assume as a result of purchasing the Purchased Assets; and if Franchisee owes an indebtedness to Franchisor at the time of purchase (the "**Franchisee's Debt**"), the purchaser shall reduce the Purchase Price by the Franchisee's Debt and pay the amount of Franchisee's Debt to Franchisor. It is agreed and acknowledged that no separate goodwill, customers or customer lists will be retained by Franchise, as such goodwill, customers or customer lists or goodwill of the Franchised Business, it being understood that Franchisor is the exclusive owner of all customers, customer lists and goodwill of the Franchised Business.

(b) **Manner of Exercising Option:** If Franchisor elects to exercise its Purchase Option, Franchisor will notify Franchisee in writing no later than 15 business days after the date of expiration or earlier termination of this Agreement. Closing will take place within 30 days after Franchisor has received all information, data and documentation from Franchisee necessary to calculate the Purchase Price. Franchisor or Franchisor's agent will have the right to operate the Franchised Business from the date of notice until closing.

20. COVENANTS NOT TO COMPETE, SOLICIT AND HIRE

20.1 Acknowledgments. Franchisee specifically acknowledges that Franchisee has no previous experience in the business embodied by the System and that, pursuant to this Agreement, Franchisee will receive valuable specialized training and information concerning the operational, sales, promotional, and marketing methods and techniques of Franchisor as embodied in the System. Accordingly, Franchisee agrees to the following covenants:

20.2 In-Term Covenants. Throughout the term of the Franchise, neither Franchisee nor any of its owners or affiliates will directly or indirectly:

(a) divert or attempt to divert any business or customer of the Franchised Business, the Franchisor, or of any franchisee to any competitor;

(b) solicit or hire any person who is at that time (or was within the previous 6 months) employed by Franchisor or by any other Enviro-Master franchisee without the prior express permission of the employer; or

(c) engage in, assist, have any interest in, or make loans or advances to, any business that is similar to the Franchised Business, or offers the same or similar Products or Service.

20.3 Post-Term Covenant. For a period of 2 years (the "**Restrictive Period**") after the transfer, assignment, termination or expiration of this Agreement (the "**Termination Date**") Franchisee will not, either directly or indirectly, individually or with another:

(a) engage in, assist, hold any ownership or financial interest in, or have any managerial or sales involvement in any business, which provides in whole or in part, products or services that are competitive with the Franchised Business or the System, within (i) the former Territory of Franchisee and within 25 miles of the outer boundary of the same or (ii) in the designated territory of any other Enviro-Master franchise or business in existence as of the Termination Date and within 10 miles of the outer boundary of such territories.

(b) solicit or divert, attempt to influence or take away any customers, Franchisees, vendors or patrons of Franchisor or any other franchisee of Franchisor which customers, franchisees, vendors or patrons were served by Franchisor or any franchisee of Franchisor at any time during the two (2) years before the Termination Date, to transfer or divert their business or patronage from Franchisor or any Franchisee to any other person engaged in any business the same or similar to the Franchised Business; or

(c) solicit or attempt to hire any person who was an employee of Franchisor or any Franchisee of Franchisor during the one year period ending on the Termination Date, or attempt to influence any such person to terminate his employment with Franchisor or any Franchisee of Franchisor.

20.4 Exception. Sections 20.2(c) and 20.3 do not apply to ownership of less than 3% of the shares issued and outstanding of a publicly traded corporation.

20.5 Franchisee agrees that the periods of time in these covenants and the geographical areas of restriction in this covenant are fair and reasonable and are reasonably required for the protection of Franchisor and its franchisees. Franchisee would desire at least this same protection

against competitive activities by another former franchisee. Franchisee agrees that, if a court or arbitrator should determine any part of this Section 20 to be overly broad, unenforceable, or invalid, the remaining parts hereof shall continue to be valid and enforceable as though the invalid portions were not a part hereof. If any of the provisions of this Agreement relating to the geographic area of restriction or the periods of time of the covenants shall be deemed to exceed the maximum area or periods of time which a court with jurisdiction would deem enforceable, the geographic area or periods of time shall, without further action on the part of any person, be deemed to be modified, amended and limited, to the maximum geographic area or time periods which a court of competent jurisdiction would deem valid and enforceable in any jurisdiction in which such court shall be convened or covenant applied. Any modification shall apply only in the jurisdiction of the deciding court or in the state where the arbitrator made the decision and that court or arbitrator is expressly empowered to make any modifications contemplated by this provision.

20.6 For purposes of this Agreement, all references to Franchisor shall be deemed to include: (a) any corporation or entity which acquires all, or substantially all, of the assets of Franchisor, whether by statutory merger or otherwise, (b) any corporation, partnership, or other entity directly or indirectly controlled by or under common control with Franchisor or its successor, and (c) any subfranchisor or other assignee of Franchisor.

20.7 Franchisee acknowledges that Franchisor shall have the right, in its sole discretion, to reduce the scope of any covenant in this Section 20 without Franchisee's consent, effective immediately upon receipt by Franchisee of written notice thereof. Franchisee agrees that it shall comply with any covenant as so modified, which shall be fully enforceable notwithstanding any other provision of this Agreement regarding the modification of the Agreement's terms.

20.8 Upon Franchisor's written request, Franchisee shall obtain signed covenants similar to those set forth in this Section 20 from any or all of the following persons: (1) all principles of Franchisee (if Franchisee is an entity), all managers of Franchisee, and any other personnel employed by Franchisee who have received or will receive access to confidential information including the Manual; (2) all officers, directors, and holders of a beneficial interest of 5% or more of the securities or ownership of Franchisee, and of any entity directly or indirectly controlling franchisee, if franchisee is an entity; (3) the general partners and any limited partners if franchisee is a partnership and (4) all sales personnel and technical personnel of Franchisee who have or will have material contact with Franchisee's customers, patrons, or vendors. Every covenant required by this subparagraph shall be in a form satisfactory to Franchisor and shall expressly name Franchisor as intended third party beneficiary.

20.9 The terms of this Section 20 shall survive the termination or expiration of this Agreement for any reason.

20.10 If, at any time during the Restrictive Period, Franchisee fails to comply with Franchisee's obligations under Section 20.3, then the Restrictive Period will be tolled during such period of non-compliance.

21. INTERPRETATION OF AGREEMENT

21.1 Independent Contractor. Nothing in this Agreement is intended to make either party an agent, legal representative, subsidiary, joint venturer, partner, employee, fiduciary or servant of the other party for any purpose whatsoever.

(a) During the term of the Franchise, Franchisee will hold itself out to the public as an independent contractor operating the Franchised Business under license. Franchisee will display in a conspicuous place in Franchisee's van (or business premises, if any) and on Franchisee's business stationery, forms and advertising, a sign or notice to the effect that:

This Business is Independently Owned and Operated by [Franchisee] under an Enviro-Master franchise.

(b) Nothing in this Agreement authorizes Franchisee to make any contract, agreement, warranty, or representation on behalf of Franchisor, or to incur any debt or obligation in the name of Franchisor. Franchisor will in no event be liable to any person as a result of any action by Franchisee.

21.2 Approvals and Waivers.

(a) Whenever this Agreement requires the prior approval, acceptance or consent of Franchisor, Franchisee will make its request in writing and submit it in time for Franchisor to review and respond effectively. All approvals, acceptances or consents of Franchisor must be in writing to be binding on Franchisor.

(b) Franchisor makes no warranties or guarantees to Franchisee. Franchisor will have no obligation or liability to Franchisee as a result of any approval, waiver, acceptance, consent or suggestion given to Franchisee, or any failure or delay in responding to any request of Franchisee, or the denial of any request by Franchisee.

(c) Franchisor will have the right to demand exact compliance by Franchisee with the terms and conditions of this Agreement at all times. No action or inaction by Franchisor constitutes a waiver of any of its rights under this Agreement. Any actual waiver by Franchisor is limited to the specific terms of the waiver and does not apply to any other past or future conduct by Franchisee.

21.3 Notices. All notices under this Agreement will be in writing and will be personally delivered or mailed by certified or registered mail, return receipt requested, to the respective parties

at the addresses first shown above, unless and until a different address has been designated by written notice to the other party.

21.4 Entire Agreement; Amendments.

(a) This Agreement (including all exhibits or attachments and any documents incorporated by reference) constitutes the complete, entire agreement between Franchisor and Franchisee concerning the subject matter covered in this Agreement.

(b) Franchisor and Franchisee are not bound by any previous discussions between them, whether on the subject matter of this Agreement or any other subject matter, unless the terms have been set down in writing signed by both parties.

(c) This Agreement supersedes all previous written or oral agreements between Franchisor and Franchisee on the subject matter of this Agreement.

(d) Except as specifically provided in this Agreement, no amendment or change to this Agreement will be binding on either party unless it is mutually agreed to by the parties and executed by their authorized officers or agents in writing.

(e) Notwithstanding the foregoing, nothing in this Agreement or any related agreement is intended to disclaim any representations made in the Franchise Disclosure Document delivered to Franchisee.

21.5 Reduction in Scope. Franchisee understands and acknowledges that Franchisor will have the right, in its sole discretion, to reduce the scope of any obligation or covenant of Franchisee under this Agreement, without Franchisee's consent, effective immediately on receipt by Franchisee of written notice. Thereafter, Franchisee agrees to comply with any such reduced covenant or obligation.

21.6 Interpretation.

(a) Each provision of this Agreement is separate and independent from every other provision. If any provision is found to be void, invalid or unenforceable, the remaining provisions of this Agreement will remain in full force and effect. If a provision is unenforceable as constructed but would be enforceable upon the striking of a phrase, clause or sentence, then such phrase, clause or sentence shall be stricken as may be necessary to render the remaining portion of the provision enforceable.

(b) If any provision imposing an obligation on Franchisee is found to be unreasonable or unenforceable because it imposes excessive requirements or restrictions, Franchisee agrees to comply with the basic obligation to the maximum extent permitted by law. (c) If any covenant in this Agreement is found to be void, invalid, unenforceable, or unreasonably broad, the court will be authorized to reduce or revise that covenant so that it will be effective to the maximum extent that would be permitted by law.

(d) Except as expressly provided in this Agreement, nothing in this Agreement is intended to give any person any rights or remedies under this Agreement except Franchisee, Franchisor and their respective successors and assigns (subject to Article 11).

(e) No right or remedy conferred on or reserved to Franchisor by this Agreement is intended to be, nor will be deemed, exclusive of any other right or remedy that is permitted or provided by law or equity, but each remedy will be cumulative of every other right or remedy.

(f) All headings in this Agreement are intended solely for convenience and will not affect the meaning of any provisions.

(g) All references involving gender and number will be construed to include the masculine, feminine, neuter, singular or plural, where the context so requires.

(h) This Agreement will be effective and binding on Franchisor only when executed on behalf of Franchisor by its President or other authorized senior officer.

(i) Each party has had the opportunity to consult with legal counsel before execution of this Agreement. In construing this Agreement there shall not apply any presumption that any provision of this Agreement will be construed against the party responsible for drafting the Agreement.

22. GOVERNING LAW AND ENFORCEMENT

22.1 Applicable Law. This Agreement will become effective when delivered in executed form to the Franchisor or acceptance at its principal place of business in Mecklenburg County, North Carolina. This Agreement will be governed by and interpreted under the laws of the State of North Carolina, without regard to North Carolina's choice of law rules.

22.2 Costs of Enforcement. Franchisee will pay all costs and expenses (including, without limitation, reasonable attorneys' fees, accounting fees and fees of other experts) incurred by Franchisor in connection with the enforcement of this Agreement against Franchisee.

22.3 Remedies. Franchisee acknowledges that Franchisee's violation of the terms of Article 6, Confidential Information, Article 18, Obligations on Termination or Expiration, or Article 20, Covenants not to Compete, would result in irreparable injury to Franchisor for which no adequate remedy at law may be available. Accordingly, Franchisee consents to the issuance of a temporary injunction and a preliminary injunction prohibiting any conduct by Franchisee in

violation of the terms of those Articles. Franchisee shall also be entitled to mandatory injunctive relief as may be necessary to continue the Franchised Business pursuant to Article 18 without detriment to the goodwill of the Franchise or detriment to customers. In either such event, injunctive relief may be obtained upon a showing of a prima facie case of violation, and a bond of \$1,000 is agreed to be sufficient to secure such injunction. Injunctive relief shall not be in lieu of damages but rather is acknowledged to be necessary to avoid irreparable harm that would occur in addition to damages that result from conduct prior to the injunctive relief taking effect.

22.4 Claims No Bar to Enforcement. Franchisee expressly agrees that the existence of any claims it may have against Franchisor, whether or not arising from this Agreement, will not constitute a defense to the enforcement by Franchisor of the covenants in Articles 6, 18 and 20.

22.5 Limitation on Actions. Notwithstanding any provision of law, Franchisee agrees that any demand or claim by Franchisee related to or arising under this Agreement will be brought within one year after the date when the facts giving rise to the demand or claim became known, or should have become known in the exercise of reasonable diligence, to Franchisee.

22.6 Negotiation and Mediation.

(a) **Agreement to Use Procedures**. Unless Franchisor is seeking immediate injunctive relief, neither Franchisor nor Franchisee shall file a demand for arbitration until they have attempted in good faith to resolve their dispute in accordance with the mediation process set forth in this Section 22.6.

(b) **Initiation of Procedures**. The party that initiates these procedures ("**Initiating Party**") must give written notice to the other party, describing in general terms the nature of the dispute, specifying the Initiating Party's claim for relief, and identifying one or more people with authority to settle the dispute for him, her, or it. The party receiving the notice ("**Responding Party**") has 10 business days within which to designate, by written notice to the Initiating Party, one or more people with authority to settle the dispute for him, her, by written notice to the Initiating Party, one or more people with authority to settle the dispute on the Responding Party's behalf. These people are called the "**Authorized People**."

(c) **Direct Negotiations**. The Authorized People may investigate the dispute as they consider appropriate, but shall meet within thirty (30) days from the date of the Initiating Parties' written notice to discuss resolving the dispute. The Authorized People may meet at any time any place and as often as they agree. If the Dispute has not been resolved within thirty (30) days after their initial meeting, either party may begin mediation procedures by giving written notice to the other party that it is doing so.

(d) **Selection of Mediator**. Within ten (10) business days from the date on which one party gives notice that he, she, or it, is beginning mediation, such party shall submit a list of ten certified mediators who is not associated with, or otherwise doing business with the party or such party's attorney (except as an adversary or as a previous mediator). Within ten (10)

business days from receiving such list, the Authorized Person for the other party shall select a name from the list. If the mediator is not available to serve, the Authorized Person who received the list shall select another mediator from the list. If three persons listed are unable or unwilling to serve after being selected, the Authorized Person who received the list shall provide a list of ten certified mediators, and the process shall be repeated with the Authorized Persons having exchanged roles.

(e) **Time and Place for Mediation**. In consultation with parties, the mediator shall promptly designate a mutually acceptable time and place for the mediation, which shall in any event be a place within 100 miles of Franchisor's principal place of business. Unless circumstances make it impossible, the time selected may not be later than thirty (30) days after selecting the mediator.

(f) **Exchange of Information**. If either party to this Agreement believes he, she, or it needs information in the possession of another party to this Agreement, to prepare for the mediation, all parties must attempt in good faith to agree on procedures for exchanging such information, with the help of the mediator if required.

(g) **Summary of Views**. At least seven (7) days before the scheduled mediation session, each party must deliver to the mediator and the other party a concise written summary of its views of the matter in dispute, and any other matters the mediator asks them to include. The mediator may also request that each party submit a confidential paper on relevant legal issues, which may be limited in length by the mediator.

(h) **Representatives**. In the mediation, each party must be represented by an Authorized Person and may be represented by counsel. In addition, each party may, with permission of the mediator, bring any additional people who are needed to respond to questions, contribute information, and participate in the negotiations.

(i) **Conduct of Mediation**. The mediator should advise the parties in writing of the format for the meeting or meetings. If the mediator believes it useful after reviewing the position papers, the mediator should give both the mediator and the Authorized People an opportunity for oral presentation of each party's views on the disputed matter. The mediator should assist Authorized People in negotiating resolution of the disputed matter, with or without the assistance of counsel or others. To this end, the mediators are authorized both to conduct the joint meetings, and to attend separate private caucuses with parties. All mediation sessions will be strictly private. The mediator must keep confidential all information learned in a private caucus with any party unless specifically authorized by that party to disclose information to the other party. The parties commit to participate in the proceedings in good faith with the intent of resolving the dispute if at all possible.

(j) **Termination of Procedure**. The parties agree to participate in the mediation procedure to its conclusion, as defined in this Section. The mediation may be concluded (1) by signing a settlement agreement by the parties; (2) by the mediator's declaration that the mediation is terminated; or (3) by a written declaration of either party, no earlier than at the conclusion of one full day's mediation, that the mediation is terminated. Even if the mediation is terminated without resolving the dispute, the parties agree not to terminate negotiations and not to begin any legal action proceedings or seek another remedy before the expiration of a five business-day waiting period following the mediation. A party may begin arbitration (or litigation if permitted under the terms hereof) within this period only if arbitration (or litigation if permitted under the terms hereof) might otherwise be barred by an applicable statute of limitations or to request an injunction to prevent irreparable harm.

(k) **Fees of Mediator; Disqualification**. The fees and expenses of the mediator must be shared equally by the parties. The mediator may not later serve as a witness, consultant, expert, counsel for any party concerning the dispute or any related or similar matter in which either of the parties is involved.

(1) **Confidentiality**. The mediation procedure is a compromise negotiation or settlement discussion for purposes of federal and state rules of evidence. The parties agree that no stenographic, visual, or audio record of the proceedings may be made. Any conduct, statement, promise, offer, view, or opinion, oral or written, made during mediation by the parties or their agents or employees, or the mediator, is confidential and shall be treated as privileged. No conduct, statement, promise, offer, views, or opinion made in the mediation procedure is discoverable or admissible in evidence for any purpose, not even impeachment, in any proceeding involving either of the parties. However, evidence that would otherwise be discoverable or admissible will not be excluded from discovery or made inadmissible simply because of its use in the mediation.

22.7 Arbitration and Litigation.

(a) **Arbitration.** All disputes and claims relating to or arising under this Agreement, any other agreement entered into between the parties, the rights and obligations of the parties, or the franchise relationship between the parties if not settled in mediation, shall be resolved by arbitration in Charlotte, North Carolina in accordance with the Commercial Arbitration Rules of American Arbitration Association ("**the Arbitrators**"), provided that there shall be no class action arbitration. The following shall supplement and, in the event of a conflict, shall govern any arbitration: If the claim is for less than \$30,000 then the matter shall be heard before a single arbitrator. If the claim, or a counterclaim, is for \$30,000 or more, the matter shall be heard before a panel of three (3) arbitrators (unless the parties agree on a single arbitrator). Each party must bear its own costs of arbitration; provided, however, that arbitrators' fee shall be shared equally by the parties. The arbitrators shall have no authority to determine class action claims and shall have no authority to amend or modify the terms of the Agreement. The arbitrators shall apply applicable law, subject to the terms hereof. The arbitrators shall deliver a written opinion with

Enviro-Master International Franchise, LLC FDD May 6, 2024 #198798v1 their award, setting forth the facts and the contractual authority for any award of money damages. Judgment upon the award of the arbitrator may be submitted for confirmation and entered in any court having competent jurisdiction. This agreement to arbitrate shall survive any termination or expiration of this Agreement.

(b) **Litigation.** Notwithstanding anything to the contrary contained in herein, in the event that Franchisor seeks injunctive or equitable relief, the Parties agree that Franchisor may bring such claims for injunctive or equitable relief, and all other claims, including claims for money damages in court, and not arbitration. In the event Franchisee has any counterclaims to the claims brought by Franchisor in court, Franchise shall file any such counterclaim in court. In the event of litigation in accordance with the terms herewith, Franchisee consents and agrees that:

(i) the following courts shall have personal jurisdiction over Franchisee in all lawsuits relating to or arising out of this Agreement and related agreements and hereby waives any defense Franchisee may have of lack of personal jurisdiction in any such lawsuit filed in these courts: (i) all courts included within the state court system of North Carolina; and (ii) all the United States District Courts sitting within North Carolina.

(ii) venue shall be proper in any of the following courts in all lawsuits relating to or arising out of this Agreement and related agreements and hereby waives any defense Franchisee may have of improper venue in any such lawsuits filed in these courts: (i) the magistrate, district or superior court of Mecklenburg County; and (ii) the United States District Court for the Western District of North Carolina (Charlotte Division). In the event any of these courts are abolished, Franchisee agrees that venue shall be proper in the state or federal court in North Carolina which most closely approximates the subject matter jurisdiction of the abolished court.

(iii) in all lawsuits relating to or arising out of the Agreement and related agreements, Franchisee consents and agrees to be served with process outside the State of North Carolina in the same manner as service may be made within the State of North Carolina by any person authorized to make service by the laws of the state, territory, possession, or country in which service is made or by any duly qualified attorney in such jurisdiction, and Franchisee hereby waives any defense Franchisee may have of insufficiency of service of process relating to such service. This method of service shall not be the exclusive method of service available in such lawsuits and shall be available in addition to any other method of service allowed by law.

(c) **Resolution of Dispute by Purchase**. Franchisee agrees that, at any time, Franchisor may cause a dispute regarding the Franchise or this Agreement (regardless whether the

dispute is the subject of mediation, arbitration, litigation, post-arbitration proceedings or otherwise) to be resolved fully, completely and summarily by tendering to Franchisee an amount which is two (2) times the Royalty Fees paid by Franchisee pursuant to Section 3.2(a) in the most recent twelve months in which such Royalty Fees were paid, but not less than twice the Franchise Fee paid pursuant to Section 3.1. Exercise of such right to terminate a dispute by tender shall be entirely at the option of Franchisor. Franchisee acknowledges that the amount described in this Section represents a generous estimate of the value of the Franchised Business, even in light of, an allowing substantial margin in favor of Franchisee for, claims that may arise in favor of either party. Upon exercise of the right, all then-pending disputes between Franchisor and Franchisee shall terminate, and the Franchise shall terminate. Upon a tender given pursuant to this Section, all pending actions and proceedings between Franchisor and Franchisee shall be dismissed, and the parties shall, to document the resolution for proof to third parties, execute a mutual release of all claims known and unknown between them and their affiliates; provided that (a) Franchisee shall nevertheless comply with all obligations under Sections 6, 18.1 - 18.4, and 20; and (b) the refusal of a party to execute a release shall not affect that the release is deemed given upon the tender hereunder being made.

22.8 ACKNOWLEDGMENT THAT ARBITRATION PREEMPTS JURY TRIAL RIGHTS. THE PARTIES HERETO HEREBY WAIVE ANY RIGHT TO JURY TRIAL OF DISPUTES TO WHICH FRANCHISOR AND FRANCHISEE ARE PARTIES. THIS PROVISION SHALL BE BINDING UPON ANY GUARANTOR OF THE OBLIGATIONS OF A PARTY HEREUNDER.

22.9 WAIVER OF PUNITIVE AND TORT DAMAGES. EXCEPT AS SPECIFIED IN SECTION 9.3, FRANCHISEE AND FRANCHISOR HEREBY WAIVE ANY RIGHT TO CLAIM OR RECEIVE PUNITIVE, EXEMPLARY, CONSEQUENTIAL OR ECONOMIC DAMAGES (E.G., LOST PROFITS, OPPORTUNITY COSTS OR DAMAGES FOR LOST OR DIMINISHED VALUE OF A BUSINESS) AS A RESULT OF ANY CLAIM BETWEEN FRANCHISOR AND FRANCHISEE, AND EACH PARTY WILL BE LIMITED TO THE RECOVERY OF ACTUAL DAMAGES SUSTAINED BY SUCH PARTY AND ANY COSTS AND FEES TO WHICH THE FRANCHISOR MAY BE ENTITLED PURSUANT TO SECTION 22.2.

23. ACKNOWLEDGMENTS AND REPRESENTATIONS OF FRANCHISEE

23.1 Acknowledgments. Franchisee acknowledges that:

(a) Franchisee received the Franchise Disclosure Document provided by

Franchisor;

(b) Franchisee has conducted an independent investigation of the Franchised

Business;

(c) Franchisor will not provide any financial assistance to Franchisee; and

(d) Franchisor has made no promise that Franchisor will buy back from Franchisee any equipment, fixtures or other products purchased by Franchisee in connection with the Franchised Business.

24. **REPRESENTATIONS.** To induce Franchisor to enter into this Agreement, Franchisee represents and warrants to Franchisor that all statements by Franchisee in connection with its application for a franchise were, and at the time of execution of this Agreement continue to be, true, complete and not misleading.

[signatures on following page]

IN WITNESS WHEREOF, the parties have duly executed, sealed and delivered this Agreement as of the date first written above.

Witness:	Franchisee:
	By: Title: Date:
Witness:	Franchisee:
	By: Title: Date:
Witness:	Franchisor: ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC
	By: Title: Date:

B-65

ATTACHMENT A

TERRITORY

The protected territory for the _____ Territory consists of the following area:

The Business Count for the territory is _____as of _____.

Date:

Franchisee Initial Here:

Franchisor Initial Here:

Date:

ATTACHMENT B

(New Franchisees)

MINIMUM ANNUAL ROYALTY FEES

Pursuant to Section 3.2 of this Franchise Agreement, during the Initial Term of the Franchise, Franchisee will pay Minimum Annual Royalties to Franchisor as follows for a single Franchised Business:

Year of Operation	Minimum Annual Royalties
1	None
2	\$6,000
3	\$12,000
4	\$18,000
5	\$24,000

ATTACHMENT B

(Renewing Franchisees)

MINIMUM ANNUAL ROYALTY FEES

During the term of this Franchise Agreement (which represents the renewal of Franchisee's first franchise agreement, and Years 6-15 of Franchisee's operations), Franchisee will pay to Franchisor the Minimum Annual Royalties calculated as set forth in this Attachment B. The amount of Franchisee's Minimum Annual Royalties shall be based upon (i) population level of businesses within the Territory, and (ii) Franchisee's year of operations.

Population Level:

Franchisee's Territory will be categorized according to the levels in the chart below, based on the most recent information from real-time market analytics software programs, on the population of businesses in the Territory:

Level	Business Population
1	0 to 25,000
2	25,001 - 50,000
3	50,001 - 75,000
4	75,001 - 100,000
5	100,001 - 125,000
6	125,001 - 150,000
7	150,001+

Year of Operations:

Franchisee will pay Franchisor the Minimum Annual Royalties that coincides with the Territory's Level and year of operations, as set forth in the charts below.

Year 6:

Level	Minimum Annual Royalties
1	\$30,000
2	\$60,000
3	\$90,000
4	\$120,000
5	\$150,000
6	\$180,000
7	\$210,000

Year 7:

Level	Minimum Annual Royalties
1	\$36,000
2	\$72,000
3	\$108,000
4	\$144,000
5	\$180,000
6	\$216,000
7	\$252,000

Level	Minimum Annual Royalties
1	\$42,000
2	\$84,000
3	\$126,000
4	\$166,000
5	\$208,000
6	\$250,000
7	\$292,000

Year 8:

Year 9:

Level	Minimum Annual Royalties
1	\$48,000
2	\$96,000
3	\$144,000
4	\$192,000
5	\$240,000
6	\$288,000
7	\$336,000

B-70

Year	10:
------	-----

Level	Minimum Annual Royalties
1	\$54,000
2	\$108,000
3	\$162,000
4	\$216,000
5	\$270,000
6	\$324,000
7	\$378,000

Year 11:

Level	Minimum Annual Royalties
1	\$60,000
2	\$120,000
3	\$180,000
4	\$240,000
5	\$300,000
6	\$360,000
7	\$420,000

2:

Level	Minimum Annual Royalties
1	\$66,000
2	\$132,000
3	\$198,000
4	\$264,000
5	\$330,000
6	\$396,000
7	\$462,000

Year 13:

Level	Minimum Annual Royalties
1	\$72,000
2	\$144,000
3	\$216,000
4	\$288,000
5	\$360,000
6	\$432,000
7	\$504,000

Year	14:
------	-----

Level	Minimum Annual Royalties
1	\$78,000
2	\$156,000
3	\$234,000
4	\$312,000
5	\$390,000
6	\$468,000
7	\$546,000

Year 15:

Level	Minimum Annual Royalties
1	\$84,000
2	\$168,000
3	\$252,000
4	\$336,000
5	\$420,000
6	\$504,000
7	\$588,000

B-73

ATTACHMENT C

ELECTRONIC FUNDS WITHDRAWAL AUTHORIZATION

Bank Name	:	
ABA#	:	
Acct. No.	:	
Acct. Name	:	

Effective as of the date of the signature below, [Franchisee Name] hereby authorizes ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC ("Enviro-Master") or its designee to withdraw funds from the above-referenced bank account, electronically or otherwise, to make the following payments to Company under the Franchise Agreement for the franchise located at _______: (1) all Royalty, National Advertising Fees Marketing and Admin/Business Service Fees. Such withdrawals shall occur on a weekly basis, or on such other schedule as Enviro-Master shall specify in writing. Enviro-Master is also authorized to deposit funds into the above-referenced account, electronically or otherwise. This authorization shall remain in full force and effect until terminated in writing by Enviro-Master. ______ [Franchisee Name] shall provide Enviro-Master, in conjunction with this authorization, a voided check from the above-referenced account.

AGREED:

ATTEST:

FRANCHISEE

By:_____

Print name: _____

Title:

ATTACHMENT D

PERSONAL GUARANTY

ENVIRO-MASTER :	Enviro-Master International Franchise, LLC
FRANCHISEE:	
DATE:	

NOTE: IF FRANCHISEE IS A CORPORATION, EACH OF ITS SHAREHOLDERS MUST EXECUTE THE FOLLOWING UNDERTAKING. IF FRANCHISEE IS A PARTNERSHIP, EACH OF ITS GENERAL PARTNERS MUST EXECUTE THE FOLLOWING UNDERTAKING. IF FRANCHISEE IS A LIMITED LIABILITY COMPANY, EACH OF ITS MEMBERS AND MANAGERS MUST EXECUTE THE FOLLOWING UNDERTAKING.

In order to induce Enviro-Master International Franchise, LLC ("Enviro-Master"), to enter into a franchise agreement ("Franchise Agreement") with the Franchisee named above ("Franchisee"), which each Guarantor acknowledges that Enviro-Master would not have executed in the absence of this Guaranty, and for and in consideration of the execution of the Franchise Agreement by Enviro-Master, each undersigned Guarantor, jointly and severally if two or more, hereby:

1) Represent to Enviro-Master that the undersigned are all of the shareholders of the Franchisee, or all of the general partners of the Franchisee, or all of the members and managers of the Franchisee, as the case may be.

2) Represents that he or she has carefully reviewed and considered the Franchise Agreement between Enviro-Master and Franchisee ("the Franchise Agreement");

3) Agrees, in consideration of benefits received and to be received by each of them, that they jointly and severally, and for themselves, their heirs, legal representatives and assigns: (a) are and shall be firmly bound by all of the terms, provisions and conditions of the Franchise Agreement, and any other agreement between the Franchisee and Enviro-Master and/or its affiliates, (b) hereby unconditionally guarantee the full and timely performance by the Franchisee of each and every financial and performance obligation of the Franchisee under the Franchise Agreement or other agreement, including, without limitation, any indebtedness of the Franchisee arising under or by virtue of the aforesaid Franchise Agreement and (c) will not permit or cause any change in the percentage of the Franchisee owned, directly or indirectly, by any person, without first obtaining the written consent of Enviro-Master prior to said proposed transfer, which consent must not be unreasonably withheld, and without first paying or causing to be paid to Enviro-Master the Franchise Agreement Transfer Fee provided for in said Franchise Agreement, if applicable, and without otherwise complying with the transfer provisions of the Franchise Agreement;

4) Agrees to be bound by the in-term and post-term covenants against competition of the Franchise Agreement and to be bound by the confidentiality mediation, arbitration and dispute resolution provisions of such Franchise Agreement.

The obligation of each Guarantor hereunder shall be binding upon such Guarantor without any further notice or acceptance thereof, and upon each and every default by Franchisee, without any notice to or demand upon such Guarantor, such Guarantor will pay to Enviro-Master the sum or sums in default and will comply with or perform all the terms, covenants and conditions of the Franchise Agreement imposed upon the Franchisee by the Franchise Agreement. In that regard, each Guarantor hereby waives the benefits of any provision of law requiring that Enviro-Master exhaust any right or remedy, or take any action, against the Franchisee, any other guarantor, any other person or property.

No extension, forbearance or leniency extended by Enviro-Master to the Franchisee shall discharge any Guarantor, or alter the obligation of any Guarantor hereunder, and each Guarantor agrees that at all times such Guarantor shall be liable under this Guaranty notwithstanding same and notwithstanding the fact that such Guarantor has had no notice of default or of any such extension, forbearance or leniency.

Enviro-Master and Franchisee, without notice to or consent by any Guarantor, may at any time or times, enter into any such modifications, extensions, amendments or other covenants respecting the Franchise Agreement, and Guarantor shall not be released from Guarantor's obligations under this Guaranty thereby. It is intended that any joinder, waiver, consent or agreement by Franchisee shall, by its own operation, be deemed to be a joinder, consent, waiver or agreement by Guarantor with respect thereto, and each Guarantor shall continue as a Guarantor with respect to the Franchise Agreement as so modified, extended, amended, or otherwise affected.

Neither the obligation of any Guarantor to make payment or render performance in accordance with the terms of this Guaranty nor any remedy for the enforcement thereof, shall be impaired, modified, or limited in any manner whatsoever by any impairment, modification, change or limitation of the liability of Franchisee or of Franchisee's estate in bankruptcy or of any remedy for the enforcement thereof resulting from the operation of the present or future provisions of the Bankruptcy Act of the United States or any other similar or succeeding statute or law or from the decision of any court.

EXECUTED UNDER SEAL, on the dates shown below.

DATE: SIGNATURES: (SEAL)

ATTACHMENT E

FORM OF GENERAL RELEASE

The following is the current general release language that we expect to include in any agreement that a franchisee may sign as a part of a renewal an approved transfer. We may in our sold discretion periodically modify the language.

RELEASE

THIS RELEASE ("**Release**") is executed on ______ day of ______, 202___ (the "**Effective Date**") by Enviro-Master International Franchise, LLC ("**Franchisor**") and _______ and _______ (collectively, "**Franchisee**"). Franchisor and Franchisee are parties to that certain Franchise Agreement, dated as of [insert date] (the "**Franchise Agreement**"). Under the Franchise Agreement, Franchisor has the right to condition the [sale/renewal] of Franchisee's Enviro-Master franchise located within the [insert location] territory (the "**Franchise**") on Franchisee's and Franchisee's owners' signing a release of claims. Franchisor is willing to approve the [sale/renewal] of Franchise if Franchise and its owners sign this Release.

1. Release by Franchisee. Franchisee, individually, collectively, and on behalf of their respective past and present parents, subsidiaries, successors, assigns, officers, directors, shareholders, managers, members, agents, employees, affiliates, and heirs, in each of their respective corporate and individual capacities (collectively, the "Releasors") hereby freely and without any influence forever release and discharge Franchisor and its past and present officers, directors, shareholders, managers, members, agents, and employees, in their corporate and individual capacities, and Franchisor's parents, subsidiaries, successors, assigns, and affiliates and each of their respective past and present officers, directors, shareholders, managers, members, agents, and employees, in their corporate and individual capacities, (collectively, the "Released Parties") from any and all claims, covenants, contracts, debts, demands, liabilities, suits, judgments, and causes of action of whatever kind or nature, whether known or unknown, vested or contingent, suspected or unsuspected (collectively, "Claims"), which any Releasors ever owned or held, now own or hold, or may in the future own or hold, including, without limitation, claims arising under federal, state, and local laws, rules, and ordinances and claims arising out of, or relating to, (i) the Franchise Agreement and all other agreements between any Releasors and Released Parties, (ii) the franchisor-franchisee relationship and the operation of the Franchise, and (iii) any act, omission or event occurring on or before the date of this Release, unless prohibited by applicable law.

2. **Risk of Changed or Additional Facts.** Releasors understand that the facts in respect of which the release in Section 1 is given may turn out to be different from or in addition to the facts now known or believed by them to be true. All parties hereby accept and assume the risk of the facts turning out to be different from or in addition to the facts now known or believed

to be true and agree that the release in Section 1 shall nevertheless be effective in all respects and not subject to termination or rescission by virtue of any such difference in facts or additional facts.

3. **Covenant Not to Sue.** Releasors covenant not to initiate, prosecute, encourage, assist, or (except as required by law) participate in any civil, criminal, or administrative proceeding or investigation in any court, agency, or other forum, either affirmatively or by way of cross-claim, defense, or counterclaim, against any of the Released Parties with respect to any Claim released under Section 1.

4. **Owners.** Under the Franchise Agreement, Franchisor is also entitled to a release and covenant not to sue from Franchisee's owners. By his, her, or their separate signatures below, Franchisee's owners likewise grant to Franchisor the release and covenant not to sue provided above.

5. **Representations.** Each Releasor or Franchisee, as applicable, represent and warrant that:

(i) it is the owner of all Claims and rights released in Section 1 and has not assigned or transferred, or purported to assign or transfer, to any person or entity, any Claim released under Section 1;

(ii) Franchisee has full and complete power and authority to execute this Release, and that the execution of this Release shall not violate the terms of any contract or agreement it is a party to or any court order it may be subject to;

(iii) it has read this Release in its entirety and understand all of its terms; and

(iv) Franchisee is signing this Release in exchange for good and valuable consideration and knowingly, freely and voluntarily assents to all of its terms and conditions.

6. **Complete Defense.** Each Releasor: (i) acknowledges that the release in Section 1 shall be a complete defense to any Claim released under Section 1; and (ii) consents to the entry of a temporary or permanent injunction to prevent or end the assertion of any such Claim.

7. **Successors and Assigns.** This Release will inure to the benefit of and bind the successors, assigns, heirs, and personal representatives of the Released Parties and each Releasor.

8. **Counterparts.** This Release may be executed in two or more counterparts (including by facsimile), each of which shall be deemed an original, and all of which shall constitute one and the same instrument.

9. **Capitalized Terms.** Any capitalized terms that are not defined in this Release shall have the meaning given them in the Franchise Agreement.

10. **Choice of Law.** This release shall be governed by the laws of the State of North Carolina without regard to its laws relating to conflicts of laws.

[SIGNATURES ON FOLLOWING PAGE]

B-80

IN WITNESS WHEREOF, the parties have executed this Release as of the date shown above.

WITNESS:	[Insert Franchisee Name]
By:	By:
Print Name:	Name:
	Title:
	Date:
WITNESS:	[Insert Franchisee's Owners Name]
By:	By:
Print Name:	Name:
	Date:
WITNESS:	[Insert Franchisee's Owners Name]
By:	By:
Print Name:	Name:
	Date:

B-81

EXHIBIT C

ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC MULTI-UNIT DEVELOPMENT ADDENDUM <u>TO</u> FRANCHISE AGREEMENTS

THIS **MULTI-UNIT** DEVELOPMENT ADDENDUM TO FRANCHISE **AGREEMENTS** (the "Addendum") is made and entered into as of , 20 (the "Effective Date") by and between Enviro-Master International Franchise, LLC, a North Carolina limited liability company with its principal office at 5200 77 Center Drive, Suite 500, Charlotte, North Carolina 28217 ("Franchisor", "we", "us" or "our") and ("Franchisee", you" or "your").

RECITALS:

WHEREAS, contemporaneously herewith, the parties have entered into franchise agreements listed in Section 1 of this Addendum (each, a "Franchise Agreement", or collectively, the "Franchise Agreements"); and

WHEREAS, the parties desire to amend the terms of the Franchise Agreements as set forth in this Addendum.

NOW THEREFORE, in consideration of the foregoing and the mutual promises set forth in the Franchise Agreements and this Addendum, the receipt and sufficiency of which the parties acknowledge, the parties hereby agree as follows:

TERMS AND CONDITIONS:

1. <u>Multi-Unit Development</u>. The parties agree that the chart below amends Section 3.1 of each the Franchise Agreements to provide that the initial franchise fee for the Franchised Business that is subject to the particular Franchise Agreement are as set forth below. The total of the initial franchise fees for all Franchised Businesses listed below is due and payable in full on the Effective Date, except as otherwise set forth herein. You expressly acknowledge that all initial franchise fees paid by you to us are non-refundable and fully earned upon receipt by us.

*Franchise Agreement No.	Franchisee Entity	Initial Franchise Fee
1		\$50,000
2		\$40,000
3		\$30,000
4		\$20,000
5		\$20,000

2. <u>Development Conditions</u>. In addition to the other terms and conditions set forth in the Franchise Agreements, your rights to develop and/or open Franchised Businesses are subject to your meeting the following conditions to our satisfaction:

- a you possess sufficient financial and organizational capacity to develop, open, operate, and manage the additional Franchised Businesses, as determined by us in our reasonable judgment; and
- b. you are in full compliance with all requirements at all of your Franchised Businesses, if any, and you are not in default under any of the Franchise Agreements or any other agreement between you (or your affiliate) and us (or our affiliate), if any.

If you do not meet the Opening Deadline for any Franchised Business(es) as set forth in Section 13.3 of the Franchise Agreement, then you will be in default of the Franchise Agreement(s) for any such Franchised Business(es), and, in addition to any and all other available remedies, we may, upon written notice to you, terminate the applicable Franchise Agreement(s) in accordance with Section 17 thereof. If we do so, you must comply with all post-termination obligations as set forth in Section 18 of the applicable Franchise Agreement(s).

3. <u>Transfers</u>. If we approve any Transfer by you or your owners (as described in Section 11.1 of the Franchise Agreements) related to a Franchised Business to which this Addendum applies, we may require, as a condition of our approval for such Transfer, that any rights granted to you under this Addendum be of no further force or effect, including any rights to develop, open and operate any other Franchised Businesses which have not yet opened for business. Any Franchised Business that has not yet opened for business will not be eligible to request our consent for a Transfer, unless the request relates to a Related Party Transfer pursuant to Section 11.2 of the Franchise Agreement.

4. <u>Confidentiality</u>. Franchisee agrees that the terms of this Addendum are Confidential Information under the Franchise Agreements.

5. <u>Effect of Addendum</u>. The parties acknowledge that this Addendum is an integral part of, and is incorporated into, each Franchise Agreement as the terms of this Addendum apply to the Franchised Business that is the subject of the particular Franchise Agreement. This Addendum will be attached to, incorporated in, and become a part of, the Franchise Agreements. The terms and conditions stated in this Addendum, to the extent they are inconsistent with the terms and conditions stated in the Franchise Agreements, will prevail over the terms and conditions of the Franchise Agreements. Capitalized terms used but not defined in this Addendum have the meanings set forth in the Franchise Agreements. Except as modified by this Addendum, the Franchise Agreements remain in full force and effect.

6. <u>Execution</u>. This Addendum may be executed electronically and in multiple counterparts, all of which together will constitute one and the same agreement. Notwithstanding the foregoing, this Addendum will only become effective upon execution by an authorized representative of Franchisor. Facsimile or electronic signatures will have the same legal effect as original signatures and may be used as evidence of execution.

[Signatures on following page.]

This Addendum is hereby executed by the parties as follows:

FRANCHISOR:

ENVIRO-MASTER INTERNATIONAL FRANCHISE, , LLC

By: _____ Title:

FRANCHISEE:

[INSERT NAME]

[INSERT ENTITY NAME, IF APPLICABLE]

OR

By:_____

[INSERT NAME, TITLE]

[SIGNATURE PAGE TO MULTI-UNIT DEVELOPMENT ADDENDUM TO FRANCHISE AGREEMENTS]

C-4

EXHIBIT D

MARKET RESERVATION AND DEPOSIT AGREEMENT

This Agreement is made this _____ of _____ by and between Enviro-Master International Franchise, LLC (the "Company") and ______ (Prospective Franchisees or "Prospect(s)").

Whereas the Prospect(s) desires that Company reserve a specific market area in which it may purchase a franchise from Company. Area is comprised of the _____ marketplace (the "Reserved Territory").

Whereas the Company is willing to reserve the Reserved Territory, under certain conditions and terms.

It is hereby agreed as follows: Prospect(s) will issue the Company a five thousand dollar (\$5,000.00) refundable deposit by ______, 20_____. In exchange for the above commitment the Company will immediately reserve the Reserved Territory and for an additional period of fourteen (14) days after its receipt of Prospect's deposit, such that no other prospective franchisee will be issued a franchise in the Reserved Territory during the above referenced period.

On or before the expiration of the fourteen (14) day reservation period after receipt of Prospect's deposit, Prospect(s) and Company will execute required closing documents in order to issue Prospect(s) an Enviro-Master franchise in said market (the "Franchised Business"). Should Prospect(s) elect not to execute closing documents and purchase a the Franchised Business, for any reason what so ever, Company shall do the following: (1) promptly refund the Prospect's five thousand dollar (\$5,000.00) deposit precisely fourteen (14) days from receipt of the refundable deposit; or (2) promptly refund the Prospect's five thousand dollar (\$5,000.00) deposit greet for refund at any time prior to expiration of the fourteen (14) day period. The fourteen (14) day reservation period may be extended by written mutual consent of Company and Prospect(s).

At any time prior to closing, should company elect not to execute closing documents and sell the Prospect(s) the Franchised Business, for any reason whatsoever, Company shall promptly refund the Prospect's five thousand-dollar (\$5,000.00) deposit.

Upon Company's return of Prospect's deposit, Company shall make the Reserved Territory available to other prospective franchisees and have no further obligations to Prospect(s) of any kind or nature.

Enviro-Master International Franchise, LLC	Prospective Franchisee:
Ву:	Ву:
Date:	Date:

D-1

EXHIBIT E

PROMISSORY NOTE AND SECURITY AGREEMENT FOR

EQUIPMENT PURCHASE

(Interest Free)

\$[X].00 Made In: Charlotte, NC

Date: [DATE]

DATE AND PARTIES. This Promissory Note and Security Agreement for Equipment Purchase ("Note") is made effective as of the <u>___day of ____</u>. This Note is given to evidence the obligations of Borrower to Lender/Franchisor for an equipment purchase advance from Lender and to secure those obligations with a lien on Borrower's Equipment Collateral as described below. Guarantor(s) provides a guaranty of payment as provided in the signature page. The parties and their addresses are:

BORROWER/FRANCHISEE: Name: DBA: Address:

Email: (hereinafter "Borrower") GUARANTOR(S): Name: <u>SAME AS BORROWER</u> Address: _____ City/ST/Zip: _____ Email: _____ (hereinafter individually and collectively "Guarantor")

LENDER/FRANCHISOR:

Enviro-Master International Franchise, LLC 5200 77 Center Drive, Suite 500 Charlotte, NC 28217

ATTN:Steve Krol skrol@enviro-master.com

Maturity Date will be on (26 weeks from the initial WFS date), payments to be by Weekly Fee Statement.

Equipment Collateral Description (include brand, model number, serial number, other identifying information:

In the event of default by Borrower, should the then value of the Equipment Collateral Not be sufficient to satisfy the remaining balance on this Note, the definition of Equipment Collateral shall additionally include the Borrower's Enviro-Master accounts receivable and hereby grants to

Lender a lien and security interest in and to the Borrower's right, title and interest in and to Enviro-Master accounts, until the remaining balance has been swept from Borrowers Weekly Fee Statement account. Lender, at is sole option, may file UCC-1 financing statements in the State of North Carolina, or in the state where the Equipment Collateral is located.

1. PROMISE TO PAY. FOR VALUE RECEIVED, Borrower promises to pay to Lender the principal sum of [LOAN AMOUNT] or such lesser amount as shall equal the unpaid principal amount of the loan made by Lender to Borrower pursuant to this Note, and to make payments of principal as noted on the weekly payment schedule set forth on Attachment A, beginning on the WFS date as shown on Attachment A and continuing weekly for a 26 week period (the "Term"). This Note is interest free so long as payments of principal are timely made. At the end of the Term, any remaining principal due under this Note will be fully due and payable plus any unpaid default interest or penalties (but in no event at a rate higher than that permitted by applicable law).

All payments shall be in lawful money of the United States of America, at Lender's address above or at such place as the legal holder hereof may notify Borrower in writing. Each installment shall be payable via the WFS. Enviro-Master International Franchise, LLC, as Franchisor under the Enviro-Master Franchise Agreement with Borrower (the "Franchise Agreement"), on behalf of Lender or the then note holder, will deduct any amount due and payable under the terms of this WC Note against payments due Borrower from the Borrower's WFS.

2. All parties to this Note, including maker and any sureties, endorsers, or guarantors hereby waive presentment, demand, protest, notice of protest, notice of dishonor, and notice of acceleration of maturity. They agree to continue to remain bound for the payment of principal and all other sums due under this Note notwithstanding any change or changes by way of release, surrender, exchange, modification, or substitution of any security for this Note, or by way of any extension or extensions of time for the payment of principal and interest. All such parties waive any notice of such change or changes and agree that they may be made without notice to or consent of any of them. Further, they agree that their liability under this Note shall be joint and several. If the Borrower is a corporation or limited liability entity, the individual executing this Note on its behalf by his or her execution hereof warrants individually that the Borrower is duly organized, validly existing, and in good standing under the laws of the jurisdiction of its incorporation or organization, and has taken all necessary actions and obtained all consents necessary to execute and deliver this Note and any related documents.

3. Default under this Note shall occur upon (i) the failure of the Borrower to pay principal or interest as either shall come due, unless such default is cured within 10 days of the due date, (ii) death, dissolution, or termination of the existence of Borrower, (iii) insolvency or bankruptcy of Borrower, (iv) default by Borrower of its franchise agreement with EMIF ("Franchise Agreement"), as such agreement may be from time to time amended, prior to payment in full of this Note; or (iv) termination, transfer, or nonrenewal of the Franchise Agreement prior to payment in full of this Note to be immediately due and payable, both as to principal and accrued interest. The holder may employ an attorney to enforce the holder's rights and remedies and the Borrower, principals, sureties, guarantors, and endorsers of this Note hereby agree to pay to the holder reasonable attorney's fees, plus all other reasonable expenses incurred by the holder in exercising any of the holder's rights and remedies upon default, whether or not suit is filed.

Borrower shall pay to the Note holder a late charge of Five (5%) Percent of any installment not received by the Lender within ten (10) days after any installment is due.

The rights and remedies of the holder as provided in this Note and in any other agreement or instrument securing or guaranteeing this Note shall be cumulative and may be pursued singly, successively, or simultaneously, and/or by setoff or sale of any funds or property of the Borrower held by or for the holder, whether held as security or otherwise, all in the sole discretion of the holder. The failure to exercise any right or remedy shall not be a waiver or release of such right or remedy or the right to exercise it at another time. Borrower agrees that upon and during the continuation of any default hereunder, the interest payable on the principal balance hereof shall be increased to a rate of 18% per annum (or the highest rate permitted by applicable law, if less) until such default is cured. Notwithstanding the foregoing, however, any lien or security interest created by this Note or agreements between Borrower and Lender or holder is secondary to any security lien or rights of EMIF.

4. This is a secured Note between the Borrower and the Lender conveying as security the Equipment Collateral, and the holder is entitled to said security. This Note represents indebtedness contracted for business purposes and is to be governed and construed in accordance with the laws of the State of North Carolina applicable to notes made and delivered in North Carolina. If legally permissible, the parties' consent to the jurisdiction of the state and federal courts of North Carolina with respect to any civil action arising from or related to any default hereunder. However, Lender may at its sole discretion apply the law, jurisdiction, and venue of the State where the Equipment Collateral is located in order to fully protect its security position. Notwithstanding anything contained in any provision in this Note, if the Franchise Agreement between Lender/Franchisor and Borrower/Franchisee is located in a state that pursuant to franchise disclosure law requires review by legal counsel of any jurisdiction, venue, waiver or release provisions in order to be valid, by signing below Borrower/Franchisee represents that it/he/she has had such provisions reviewed by legal counsel.

RELEASE. In consideration of the granting of this interest-free Note by Lender, and for 5. other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Borrower hereby irrevocably releases and forever discharges Lender and Lender's parent, subsidiaries, divisions, and affiliates and their respective successors, assigns, directors, officers, shareholders, and employees, or and from any claims, debts, liabilities, demands, obligations, costs, expenses, actions, and causes of action of every nature, character, or description, whether known or unknown, vested or contingent, that Borrower, Borrower's affiliates, or any of them, now has, or at any time heretofore had, or may at any time have, constituting, arising out of or related to any agreement, action, inaction, occurrence, product, thing, or state of facts of any kind or type whatsoever occurring or existing prior to or on the date of execution of this Note and Release. Borrower and each of them hereby warrants and represents that no third party has or claims an interest in any claim released by this Note and Release. Provided, however, nothing contained herein shall change or alter the terms, conditions, rights and obligations of the parties as contained in any modification thereof. All terms, conditions, rights and obligations of the parties as contained in the Enviro- Master franchise agreement between Lender and Borrower, are hereby ratified and reaffirmed. This Note and Release exclude unknown insurance claims.

IN TESTIMONY WHEREOF, the Borrower has executed and sealed this instrument, or caused this instrument to be executed in its name and its seal to be affixed.

BORROWER: _____

Ву:

Name:_____

Title:

GUARANTY

Guarantor guarantees the payment, when due, to any holder hereof, of all amounts from time to time owing under the Note. If suit is brought to enforce this Guaranty, the holder of the Note may employ an attorney to enforce its rights and remedies, and the Guarantor hereby agrees to pay the collection expenses and attorneys' fees as provided in the Note. Guarantor acknowledges that this is a guaranty of payment, not of collection, that the laws of the State of North Carolina apply to this Guaranty, without regard to conflicts of law and principles, and that the provisions of N.C.G.S. §26-7 are specifically waived, as well as similar laws existing in the state(s) where the collateral securing the Note is located, if legally permissible.

This _day of _, 20___.

GUARANTOR

Print Name:

EXHIBIT F

Operations Manual Index

[See attached]

#198798v1 Enviro-Master International Franchise, LLC FDD May 6, 2024



OPERATIONS MANUAL INDEX

Disclaimer. This Manual is not intended to constitute legal or other professional advice or professional opinion of any kind. Users of this Manual, in part or in whole, are advised to seek specific legal advice by contacting your attorney regarding any specific legal matters, including compliance with Federal, State, County, or Municipal laws and regulations, and your tax professional regarding tax matters. The information in this Manual is compiled from years of experience, as well as general practical application of procedures in multiple successful franchised locations throughout the United States and Canada. Laws may vary from state to state, region to region, county to county and city to city. Your franchised business is subject to your verification of certain Federal, State, County, Municipal and Employment laws and regulations through your legal, tax and employment law professionals.

SECTION I

INTRO	DUCTION, UN	DERSTANDING PROPER STRUCTURE, INSURANCE	
	CHAPTER 1	INTRODUCTION	1
	CHAPTER 2	UNDERSTANDING PROPER STRUCTURE	2
	CHAPTER 3	FRANCHISEE INSURANCE REQUIREMENTS	6
SECTION II			
SALES	S – GAIN		
	CHAPTER 4	SALES – GAIN	20
	CHAPTER 5	SALES TRAINING	21
	CHAPTER 6	THE SALES PRESENTATION	24
	CHAPTER 7	LEVERAGING INSIDE SALES	25
	CHAPTER 8	USE YOUR TECHNICIANS TO UPSELL	29
	CHAPTER 9	SERVICE QUALITY CONTROL	34
	CHAPTER 10	PROFILE OF A SALES REPRESENTATIVE	37
	CHAPTER 11	JOB DESCRIPTION FOR YOUR SALES REPRESENTATIVES	39
	CHAPTER 12	SALES COMPENSATION	42
	CHAPTER 13	MANAGING YOUR SALES STAFF	45
	CHAPTER 14	CANCELLATIONS	47
	CHAPTER 15	CORPORATE ACCOUNTS	50
	CHAPTER 16	RETAIN GROW GAIN LEADERSHIP PROGRAM	53
		APPENDIX I	
SECTION III			
PROD	UCTS – GROW		
	CHAPTER 17	UTILIZING ENVIRO-MASTER ONLINE STORE	54
	CHAPTER 18	PAPER PRODUCTS	70
	CHAPTER 19	HAND SOAPS AND SANITIZERS	74
	CHAPTER 20	PAPER AND SOAP DISPENSERS	77
	CHAPTER 21	ENVIRO-MASTER CHEMICALS	80
	CHAPTER 22	CAN LINERS	98
	CHAPTER 23	SPECIALIZED PRODUCT AND EQUIPMENT	102

SECTION IV

SECTION IV		
SERVICES		
CHAPTER 24	ADA COMPLIANCE STANDARDS	112
CHAPTER 25	REQUIRED TOOLS FOR SERVICE INSTALL	117
CHAPTER 26	THE SANI-SERVICE INSTALL	126
CHAPTER 27	PRODUCTS USED IN SANI-SERVICE	133
CHAPTER 28	PROVIDING SANI-SERVICE	147
CHAPTER 29	VIRUS VAPORIZER SERVICE (SANI-GUARD)	168
CHAPTER 30	THE SANI-SCRUB AND SCRUB	180
CHAPTER 31	DRAIN LINE PROGRAMS	193
CHAPTER 32	MICROFIBER FLOOR SERVICE	197
CHAPTER 33	RPM WINDOW PROGRAM	206
CHAPTER 34	SANI-POD ENVIRONMENTAL HEALTH SERVICE	226
CHAPTER 35	REFRESH, DUMPSTER, PATIO, FOH, BOH	233
SECTION V		
OPERATIONS – GRO	W, RETAIN	
CHAPTER 36	THE FIELD OPERATIONS MANAGER	281
CHAPTER 37	THE HEALTH AND SAFETY TECHNICIAN	304
CHAPTER 38	INVENTORY CONTROL (Weekly Usage)	322
CHAPTER 39	MONTHLY INVENTORY CONTROL	326
CHAPTER 40	WEEKLY OPERATIONAL PROCEDURES	328
CHAPTER 41	ACCOUNTS RECEIVABLE/COLLECTIONS	343
CHAPTER 42	FRANCHISEE PAYMENT POLICIES	346
CHAPTER 43	FINANCIAL GOALS – HOW TO ACHIEVE THEM	350
CHAPTER 44	OFFICE SPECIFICATIONS	361
CHAPTER 45	COMPANY VANS AND TRUCKS	367
CHAPTER 46	SAFETY AND HEALTH PROGRAM, OSHA	389
CHAPTER 47	INFORMATION TECHNOLOGY	424

APPENDICES

Ι	SALES LEADERSHIP AND CUSTOMER RETENTION	438
II	OFFICE MANAGEMENT GUIDE	475
III	ZOHO GUIDE (CRM)	557
IV	BRANDING GUIDELINES	596

EXHIBIT G CONFIDENTIALITY AGREEMENT

ENVIRO-MASTER: Enviro-Master International Franchise, LLC FRANCHISEE: DATE:

Enviro-Master International Franchise, LLC, ("Enviro-Master") has developed and possesses valuable confidential information concerning a program of a providing restroom hygiene, drain line management, paper, and chemical products and services (the "Program").

Franchisee desires to become a franchisee of Enviro-Master, and if Franchisee does so, Franchisee will learn confidential information concerning the Program. The disclosure or unauthorized use by Franchisee of Enviro-Master's confidential information would result in significant damage to Enviro-Master. Therefore, the parties desire to enter into an agreement protecting such confidential information.

AGREEMENT:

In consideration of the above and the mutual covenants and promises of the parties hereto, Enviro-Master and Franchisee agree as follows:

1. **Confidential Information Defined**. "Confidential Information" means any and all of the following used or developed by Enviro-Master: technical information; methods of operation; equipment; specifications; pricing; customers and potential customers; manuals; software, brochures; financial information; business plans; franchise information, and all information contained in such documents; identification of franchisees and employees; trade secrets; and any other information that Enviro-Master would reasonably be expected to deem confidential. Information that is otherwise "Confidential Information" shall not lose its confidentiality merely because it may be available from other sources, particularly if such information is not common knowledge or if the compilation of such information required an expenditure of time or money. "Confidential Information" does not include information known to Franchisee prior to the date hereof, or which (i) is separately obtained from a public source; (ii) is disclosed to Franchisee by a third party who is not an employee or agent of Enviro-Master; or (iii) is released with the prior written consent Enviro-Master.

2. <u>Confidentiality</u>. The parties agree that Franchisee and its employees, agents and other persons to whom Franchisee discloses the Confidential Information will not, without Enviro-Master's written permission, disclose or otherwise make available any of the Confidential Information to any Person (defined below) who is not an employee under Franchisee's control. Franchisee will have any person who will have access to any Confidential Information execute agreements (in a form enforceable by Enviro-Master) to adhere to the same obligations binding on Franchisee under this Agreement. For purposes of this Agreement, "Person" shall include any natural person, and any corporation, partnership, trust, or other organization or entity.

3. <u>Prohibition on Use</u>. Franchisee agrees that it will never, without the written consent of Enviro-Master, use any of the Program or any of Enviro-Master's Confidential Information. Specifically, and without limiting the foregoing, Franchisee agrees that it will not implement a

program similar to or based on the Program or identifiable components of the Program without Enviro-Master's written consent.

4. <u>Burden of Proof</u>. In any dispute arising under this Agreement, Enviro-Master shall have the burden of proving Franchisee's access to Confidential Information and the possibility of Confidential Information having been used or disclosed in violation of this Agreement. Once Enviro-Master meets it burden, the burden of proof shall be upon Franchisee to show that no such disclosure or use has been made in violation of this Agreement.

5. **Documents.** Except with the express permission of Enviro-Master, Franchisee shall not copy or permit any person to copy any documents (including papers, electronic or magnetic media, audio or visual recordings, or any other means on which Confidential Information is stored or recorded) provided to it by Enviro-Master. Within three (3) business days of written or oral request by Enviro-Master, Franchisee shall return all Documents and copies of Documents.

6. <u>**Remedies.**</u> If the terms of this Agreement are broken, a remedy at law would be inadequate and the aggrieved party would be entitled to injunctive relief. The provisions of this paragraph shall be in addition to and not exclusive of any remedy otherwise available to Enviro-Master in law or equity.

7. <u>Intellectual Property Rights</u>. Franchisee acknowledges Enviro-Master's rights in the Program and the Confidential Information, and expressly denies that Franchisee has any rights therein.

8. <u>General Provisions</u>. No failure of a party to act upon any breach shall impair the right of such party to act upon such breach or default (or any subsequent breach or default). If any portion of this Agreement is held to be unenforceable or invalid, the validity and enforceability of the remaining portions will not be affected thereby. This Agreement is governed by the laws of the State of North Carolina. Any proceeding relating to this Agreement shall be maintained in a court of competent jurisdiction in Mecklenburg County, North Carolina, and parties consent to jurisdiction and venue therein. This Agreement shall bind the parties and their successors and permitted assigns, and may not be assigned without the written consent of both parties. The terms of Sections 1 through 7 shall survive the termination of this Agreement. This Agreement replaces and merges all prior understandings and agreements between the parties relating to its subject matter. This Agreement may not be modified except by a written instrument executed by both parties. **IN WITNESS WHEREOF,** the parties have hereunto executed this Agreement the day and year first above written.

Enviro-Master:

Enviro-Master International Franchise, LLC

Ву:_____

Title: _____

Franchisee:

By:		
Title:		

EXHIBIT H CURRENT ENVIRO-MASTER FRANCHISEES AS OF 12/31/2023

[See attached]

ALABAMA

Enviro-Master of Birmingham

2700 7th Avenue South Birmingham, AL 35233 Gaston Capital Holdings LLC Luke Ellington 479-883-8841 Iellington@emgch.com

Enviro-Master of Mobile

26088 Capital Drive, Unit 8 Daphne, AL 36526 Ron Finch 251-219-6205 rfinch@emofmobile.com

ARIZONA

Enviro-Master of Phoenix East

638 W Broadway Rd, Suite 311 Mesa, AZ 85210 Scott Sheldon 480-264-2225 <u>Ssheldonc10@gmail.com</u>

Enviro-Master of Phoenix West

15847 W. Glenrosa Avenue Goodyear, AZ 85359 Thomas Bushong 602-820-0635 tbushong@emofphoenixwest.com

CALIFORNIA

Enviro-Master of Central Valley

9423 Plum Avenue Stockton, CA 95212 Danny Swanson 209-639-6570 dannyswanson@comcast.net

Enviro-Master of Los Angeles North

7054 Gerald Avenue, Suite A Van Nuys, CA John Raber 310-351-8628 jraber@enviro-masterla.com

Enviro-Master of Los Angeles South

24416 Main Street, Unit 303 Carson, CA 90745 David & John Devereux 310-620-7787 david@enviromasterla.com

Enviro-Master of Oakland/San Jose

3851 Charter Park Drive, Suite K San Jose, CA 95316 David Chang 678-756-5577 <u>david@emsfbayarea.com</u>

Enviro-Master of San Francisco/San Mateo

3851 Charter Park Drive, Suite K San Jose, CA 95316 David Chang 678-756-5577 david@emsfbayarea.com

Enviro-Master of San Bernardino

3931 Carter Street, Suite 405 Riverside, CA 92501 John Roddy 909-793-3680 jroddy@gmail.com

Enviro-Master of Sacramento

2655 Del Monte Street Sacramento, CA 95691 Harminder Sehmi 650-889-8098 hsehmi@emsacramento.com

Enviro-Master of San Diego

7888 Silverton Avenue, Suite F San Diego, CA 92126 Richard Zamberlan 619-586-5675 rjzamberlan@emsandiego.com

Enviro-Master of Ventura

2889 Bunsen Avenue, Unit C Ventura, CA 93003 Ronnie Vargas 805-635-8085 enviromasterronnie@gmail.com

COLORADO

Enviro-Master of Denver North

9421 Gunbarrel Ridge Boulder, CO 80301 Brad and Lyndee Davis 949-307-3626 bdavis@emofdenvernorth.com Idavis@emofdenvernorth.com

Enviro-Master of Denver South

7344 S. Alton Way, Unit 8F Centennial, CO 80112 Jim Malcom 303-390-0800 JimMalcolm@emofdenversouth.com

DC

Enviro-Master of DC Metro

14204 Cherry Lane Court Laurel, MD 20707 Neil Middleton 443-472-2748 neil@emofdcmetro.com

DELAWARE

Enviro-Master of Delaware

25 Maggies Way, Suite 3 Dover, DE 19901 Michael and John Kelly 302-735-5555 <u>mkelly@emofdelaware.com</u> <u>ikelly@emofdelaware.com</u>

FLORIDA

Enviro-Master of Gainesville

6950 Philips Highway, Suite 22 Jacksonville, FL 32216 Rich Sweat 904-434-9223 rich@enviromasterservices.com

Enviro-Master of Jacksonville

6950 Philips Highway, Suite 22 Jacksonville, FL 32216 Mike Wooden 847-445-6448 <u>mwooden@enviromasterservices.com</u>

Enviro-Master of Ft. Lauderdale

4350 SW 5th Ave, Ste I-1 Davie, FL 33055 Scott Murphy 954-336-9958 smurphy@emftlauderdale.com

Enviro-Master of Ft.

Myers 12750 Trade Center Drive Unit 1213 Bonita Springs, FL 34135 D.J. and Lindsay DeLuca 518-669-5335 Djdeluca71@gmail.com

Enviro-Master of Miami

4960 NW 165th Street, Suite B-6 Miami, FL 33137 Kryzia & Jerry Jimenez 787-244-2522 service@emofmiami.com

Enviro-Master of Orlando

100 Candace Drive, Unit 116 Maitland, FL 32751 Andrew Martin 321-972-3210 andrewmartin.em@gmail.com

Enviro-Master of Pinellas

6261 39th Street N, Suite D Pinellas County FL 33781 Jonathan Kelly 617-458-8041 <u>ikelly@emspinellas.com</u>

Enviro-Master of Sarasota

2331 63rd Avenue East, Suite S Bradenton, FL 34203 Jeff Engelskirch 904-672-6185 jeff@emsarasota.com

Enviro-Master of Space Coast

4260 Dow Road, Suite 401 Melbourne, FL 32934 Andrew Martin 321-972-3210 Andrewmartin.em@gmail.com

Enviro-Master of South Broward

4980 NW 165th St, Unit A17 Miami Gardens, FL 33014 Jerry Jimenez 787-244-2515 service@emofmiami.com

Enviro-Master of Treasure Coast

727 North Drive, Suite H Melbourne, FL 32934 David Edwards & Lou Nolen 832-794-7932 <u>davidedwards@emtreasurecoast.com</u> <u>lounolen@emtreasurecoast.com</u>

Enviro-Master of Tampa

8413 Laurel Fair Circle, Suite #103, Tampa, FL 33610 Jeff DeVengencie 813-246-5144 jeffd@emoftampa.com

Enviro-Master of West Palm Beach

2200 N Florida Mango Road, Suite 6 West Palm Beach, FL 33409 Ben & Stephanie Green 559-259-7310 teamgreen@emofpalmbeach.com

GEORGIA

Enviro-Master of Atlanta East

3070 Business Park Drive, Suite J Norcross, GA 30071 Andy Blakely 706-974-9687 blakely@enviro-masteratleast.com

Enviro-Master of Atlanta

Southeast 2650 Pleasantdale Road, Unit #11 Doraville, GA 30340 Bill Crabtree 404-985-7142 bcrabtree@ematlse.com

Enviro-Master of Atlanta West

1341 Capital Circle, Suite L Marietta, GA 30067 Steve Tucker 404-423-5628 <u>stucker@ematlanta.com</u>

Enviro-Master of Savannah/Charleston

200 Edsel Drive, Suite E Richmond Hill, GA 31324 Matt Merry 912-596-7624 <u>mmerry@emsavannah.com</u>

ILLINOIS

Enviro-Master of Chicago Land

205 Airport Drive Joliet, IL 60431 Ben Laster 708-439-4263 <u>benjaminlaster@me.com</u>

Enviro-Master of Chicago North

205 Airport Drive Joliet, IL 60431 Ben Laster 815-630-2367 <u>benjaminlaster@me.com</u>

INDIANA

Enviro-Master of Indianapolis

6330 East 75th Street, Suite 310 Indianapolis, IN 46250 Joel Wesley 317-567-0661 jwesley@emindianapolis.com

KANSAS

Enviro-Master of Kansas City

11761 W 86th Terrace Lenexa, KS 66214 Bill Lynch & Lanny Schlemeier 913-955-3330 <u>bill@emofkansascity.com</u> <u>lanny@emofkansascity.com</u>

Enviro-Master of Wichita

5817 N Broadway Ave Park City, KS 67219 Andy Titus 316-258-5669 andy@emwichita.com

KENTUCKY

Enviro-Master of Louisville

1831 Cargo Court Louisville, KY 40299 Gaston Capital Holdings Luke Ellington 704-906-0134 <u>lellington@emgch.com</u>

MARYLAND

Enviro-Master of Baltimore

611 Seminole Avenue Catonsville, MD 21228 Rodrigo Ferrua 479-899-8595 rodrigo.ferrua@embaltimore.com

MASSACHUSETTS

Enviro-Master of Boston North

14 Jefferson Rd Winchester, MA 01890 Mike Hamilton 781-812-8772 <u>mikehamilton@emofbostonsouth.co</u> <u>m</u>

Enviro-Master of Boston South

33 Forge Hill Road, Unit #2 Franklin, MA 02038 Mike Hamilton 781-812-8772 <u>mikehamilton@emofbostonsouth.co</u> <u>m</u>

MICHIGAN

Enviro-Master of Detroit 2323 E Nine Mile, Suite B Warren, MI 48091 Ketura Brandes 734-620-0641 kb1envriomaster@yahoo.com

Enviro-Master of Metro Detroit South

12060 Farmington Road Livonia, MI 48150 Matt Carstens 313-410-9519 carstens@emmds.com

Enviro-Master of Grand Rapids

848 West River Center Drive, Suite E Tom Stock Park, MI 49321 Nick and Krista Wolfe 616-550-8907 <u>kwolfe@emofwestmichigan.com</u> <u>nickw@emofwestmichigan.com</u>

Enviro-Master of Northern Michigan

12009 North Saginaw Street Mt. Morris, MI 48458 Nydia Rodriguez 248-881-2011 nydia@emnorthernmichigan.com

MINNESOTA

Enviro-Master of Minneapolis

4976 Highway 169 N New Hope, MN 55428 Steve Loosbrock 612-859-4030 steve@solutionsatwork.com

Enviro-Master of St. Paul

4976 Highway 169 N New Hope, MN 55428 Steve Loosbrock 612-859-4030 steve@solutionsatwork.com

MISSOURI

Enviro-Master of St. Louis 777 Merus Court Fenton, MO 63026 Warren Heuman 618-977-7841 wheuman@em-stl.com

NEBRASKA

Enviro-Master of Omaha 9995 J Street Omaha, NE 68127 Mike Whitaker

402-250-5693 MikeW@EMofOmaha.com

NEVADA

Enviro-Master of Las Vegas

4345 Corporate Center Drive, Suite 201 North Las Vegas, NV 89030 Brandon Fears, et al. 704-840-7488 Brandon.fears@gracchuspartners.com

Enviro-Master of Reno/Lake

Tahoe 2803 Green Head Court Reno, NV 89509 Robert Loveall 816-377-0157 rloveall@emofrenolaketahoe.com

NEW JERSEY

Enviro-Master of New Jersey -North 218 South Jefferson Street, Unit 7

Orange, NJ 07050 Paul Greene 973-885-3924 paulgreene7355@gmail.com

Enviro-Master of New Jersey – Central

23 Tara Way Pennington, NJ 08534 Pete Justo 917-540-0758 pbj@emsofncj.com

NEW MEXICO

Enviro-Master of Albuquerque

3750 Hawkins NE Albuquerque, NM 87109 Willie Jenkins 915-929-0129 eptenviromastermain@gmail.com

NEW YORK

Enviro-Master of Albany

118 Empie Lane Cobleskill, NY 12043 Justin Long (518)728-6158 justin@emofalbany.com

Enviro-Master of Rockland/Westchester

20 C Mountainview Road Orangeburg, NY 10962 Matt Kern 845-547-2196 mkern@emofrocklandwestchester.com

NORTH CAROLINA

Enviro-Master of Charlotte

415 Minuet Lane Charlotte, NC 28217 Bobby Skinner 704-258-6658 rskinner@enviro-master.com

Enviro-Master of Concord

415 Minuet Lane Charlotte, NC 28217 Aaron Jordan 704-807-3261 ajordan@emconcord.com

Enviro-Master of Greensboro

1323 South Park Drive Kernersville, NC 27284 Mike Hill 704-361-6064 mhill@enviro-masterofthecarolinas.com

Enviro-Master of Raleigh

3200 Lake Woodard Drive, Unit 105 Raleigh, NC 27604 Clay Whitehurst 910-619-0937 <u>cwhitehurst@enviromasterofraleigh.com</u>

Enviro-Master of Wilmington

3200 Lake Woodard Drive, Unit 105 Raleigh, NC 27604 Clay Whitehurst 910-619-0937 cwhitehurst@enviromasterofraleigh.c om

NORTH DAKOTA

Enviro-Master of The Dakotas

4501 15th Ave. South, Suite 107B Fargo, ND 58103 Archit Shah 917-945-0978 archit@architshah.com

OHIO

Enviro-Master of Cincinnati 9864 Crescent Park Drive West Chester, OH 45069 Bryan Michel 513-795-7531 bryan@emcincinnati.com

Enviro-Master of Cleveland

290 Alpha Drive Pittsburgh, PA 15238 Robert Weimer 412-252-5222 <u>bweimer@pestco.com</u>

Enviro-Master of Columbus

6475 East Main Street Columbus, OH 43068 Eric Poderys 614-284-5139 <u>ericpoderys@emcolumbusoh.com</u>

OKLAHOMA

Enviro-Master of Tulsa

3168 S. 108th E. Avenue, Suite 380 Tulsa, OK, 74146 Scott Estes 918-671-2790 sestes@emoftulsa.com

Enviro-Master of Oklahoma City

3150 E Reno Ave Oklahoma City, OK 73117 Taylor Hildenbrand 405-938-5690 taylor@emcleanokc.com

OREGON

Enviro-Master of Portland 4409 SE 24th Ave Portland, OR 97202 Matt Nixt 503-208-9751

Matt@emofportland.com

PENNSYLVANIA

Enviro-Master of Central PA

6325 Allentown Blvd. Unit 1005 Harrisburg, PA 17112 Heather Schlegel 717-315-9292 heathers@emofcentralpa.com

Enviro-Master of Philadelphia

10067 Philmont Ave, Suite 320 Philadelphia, PA 19006 Matt Kern 855-504-9443 <u>mkern@emofphilly.com</u>

Enviro-Master of Pittsburgh

290 Alpha Drive Pittsburgh, PA 15238 Robert Weimer 412-252-5200 <u>bwiemer@pestco.com</u>

RHODE ISLAND

Enviro-Master of Providence 205 Hallena Road, Unit 318-C Warwick, RI 02886 Todd Rywolt 401-744-6180 todd@enviromasterpvd.com

SOUTH CAROLINA

Enviro-Master of Columbia

1404 St. Andrews Road, Suite 370 Columbia, SC 29210 Heritage Hygiene Company Nancy Rhodes 803-673-5314 nrhodes@em-columbia.com

Enviro-Master of

Greenville 1200 Woodruff Road, Suite B6 Greenville, SC 29607 Gaston Capital Holdings LLC Luke Ellington 704-906-0134 <u>lellington@emgch.com</u>

Enviro-Master of Myrtle Beach

10910 Lethco Way Charlotte, NC 28277 Bobby Skinner 704-258-6658 <u>rskinner@enviro-master.com</u>

TENNESSEE

Enviro-Master of Chattanooga/Knoxville

4295 Cromwell Road, Suite 527 Chattanooga, TN 37421 Eric Muschel 423-580-9301 muschel@emofchattanooga.com

Enviro-Master of Memphis

3885 S. Perkins Road, Suite 10 Memphis, TN 38118 Thom and Tracy Whitley 901-277-4597 twhitley@emofmemphis.com

Enviro-Master of Nashville

2605-A Winford Ave Nashville TN 37211 Gaston Capital Holdings LLC Luke Ellington 704-906-0134 lellington@emgch.com

TEXAS

Enviro-Master of Austin

10201 McKalla Place, Suite C Austin, TX 78758 Larry Barde 512-633-5428 Ibarde@enviro-master.com

Enviro-Master of Dallas West

1529 Central Park Drive Hurst, TX 76053 Erik Rhein 214-505-3448 <u>erik@emnorthtexas.com</u>

Enviro-Master of Denton

1529 Central Park Drive Hurst, TX 76053 Erik Rhein 214-505-3448 erik@emnorthtexas.com

Enviro-Master of El Paso

6400 Airport Road, Bldg. C, Suite T El Paso, TX 79925 Willie Jenkins 915-929-0129 jenkinswk@hotmail.com

Enviro-Master of Fort Worth

1529 Central Park Drive Hurst, TX 76053 Joe & Bobbi de Sola 972-342-7441 bdesola@emfortworth.com

Enviro-Master of North Texas

605 S. Sherman St. Suite 605P Richardson, TX 75081 Erik Rhein 214-505-3448 erik@emnorthtexas.com

Enviro-Master of San Antonio

5504 Bandera Road, Suite 510 San Antonio, TX 78238 Seldon Hurley 210-521-2991 <u>seldon.hurley@alamocityhygiene.co</u> <u>m</u>

UTAH

Enviro-Master of Salt Lake City

380 W. Ironwood Drive South Salt Lake, UT Troy Porter 801-856-6196 tporter@em-utah.com

VIRGINIA

Enviro-Master of Northern Virginia

4280-B Henninger Court Chantilly, VA 20151 Wilfredo Corps 703-862-2506 Wcorps@enviromasterservicesofnova.com

Enviro-Master of Fredericksburg

4280-B Henninger Court Chantilly, VA 20151 Wilfredo Corps 703-862-2506 Wcorps@enviromasterservicesofnova.com

Enviro-Master of Richmond

1323 South Park Drive Kernersville, NC 27284 Mike Hill & Jay Beardslee 704-361-6064 <u>mhill@ems-nc-va.com</u>

WASHINGTON

Enviro-Master of Seattle

22315 NE 9th Drive Sammamish, WA 98074 Will Sadler 425-283-9796 will@emseattle.com

WISCONSIN

Enviro-Master of Madison

N 33 W. 29218 Millridge Drive Pewaukee, WI 53072 Greg Kredell 314-809-4902 gregkredell@emofmadison.com

Enviro-Master of Milwaukee

2626 S. 162nd Street New Berlin, WI 53151 David Poll 262-278-0454 dpoll@emofmilwaukee.com

Franchisees who Purchased in 2022 or 2023 but

Were Not Operational by December 31, 2023

Franchisee's Name	Address	City	State	Zip Code	Phone	Market
Aaron Spurlock, et al.	13811 Woodthorpe Lane	Houston	тх	77079	713-301- 8197	Houston Metro North
Andy Whittington	1052 Richelieu Lane	Houston	ТΧ	77018	337-344- 6942	Houston Metro West

EXHIBIT I

FORMER ENVIRO-MASTER FRANCHISEES

STORES TRANSFERRED DURING FISCAL YEAR 2023

Former Franchisee Name	Address	City	State	Zip Code	Phone
Michael Caputo New Jersey Central	23 Tara Way	Pennington	NJ	8534	917-325-3083
Sam Kenna Boston North	14 Jefferson Rd	Winchester	MA	1880	617-797-0385
Melissa Lippert, et al. San Diego	44750 Villa Del Sur	Temecula	CA	92592	951-757-6690
Michael McMahan Las Vegas	6450 South Cameron Street Suite 104	Las Vegas	NV	89118	315-558-2096
Jason Welz, et al. Pinellas County	1182 Brownell St	Clearwater	FL	33756	571-528-2765

FORMER ENVIRO-MASTER FRANCHISEES

WHO CEASED OPERATIONS DURING FISCAL YEAR 2023

Former Franchisee Name	Address	City	State	Zip Code	Phone
None	N/A	N/A	N/A	N/A	N/A

EXHIBIT J

FINANCIAL STATEMENTS

[See attached]

Enviro-Master International Franchise, LLC FDD May 6, 2024 #198798v1

GUARANTEE OF PERFORMANCE

For value received, Enviro-Master Holdings, LLC, a Delaware limited liability company (the "Guarantor"), located at 3060 Peachtree Road, NW, Suite 360, Atlanta, GA 30305, absolutely and unconditionally guarantees to assume the duties and obligations of Enviro-Master International Franchise, LLC, a North Carolina limited liability company, located at 5200 77 Center Drive, Suite 500, Charlotte, NC 28217 (the "Franchisor"), under its franchise registration in each state where the franchise is registered, and under its Franchise Agreement identified in its 2024 Franchise Disclosure Document, as it may be amended, and as that Franchise Agreement may be entered into with franchisees and amended, modified or extended from time to time. This guarantee continues until all such obligations of the Franchisor under its franchise Agreement has been completely discharged, whichever first occurs. The Guarantor is not discharged from liability if a claim by a franchisee against the Franchisor remains outstanding. Notice of acceptance is waived. The Guarantor does not waive receipt of notice of default on the part of the Franchisor. This guarantee is binding on the Guarantor and its successors and assigns.

The Guarantor signs this guarantee at Charlotte , NC on the 6^{th} day of May 2024.

Guarantor:

Enviro-Master Holdings, LLC By: Name: Tod Bierling Title: C.E. D.

Enviro-Master Holdings, LLC and Subsidiaries

Independent Auditor's Report and Consolidated Financial Statements

December 31, 2023 and January 1, 2023



Contents

Independent Auditor's Report	1
Consolidated Financial Statements	
Balance Sheets	3
Statements of Operations	4
Statements of Changes in Members' Equity	5
Statements of Cash Flows	6
Notes to Financial Statements	8

FORV/S

800 Shades Creek Parkway, Suite 500 / Birmingham, AL 35209 P 205.212.5300 / F 205.212.5380 forvis.com

Independent Auditor's Report

Board of Managers and Members Enviro-Master Holdings, LLC and Subsidiaries Charlotte, North Carolina

Qualified Opinion

We have audited the accompanying consolidated financial statements of Enviro-Master Holdings, LLC and Subsidiaries (collectively, the "Company"), which comprise the consolidated balance sheets as of December 31, 2023 and January 1, 2023 and the related consolidated statements of operations, changes in members' equity, and cash flows for the fiscal year ended December 31, 2023, the Successor Period from March 29, 2022 to January 1, 2023, and the Predecessor Period from January 3, 2022 to March 28, 2022, and the related notes to the consolidated financial statements.

In our opinion, except for the effects of not reporting the value of identifiable intangible assets separately from goodwill and the resulting effect on amortization expense, as discussed in the Basis for Qualified Opinion paragraph, the consolidated financial statements referred to above present fairly, in all material respects, the financial position of Enviro-Master Holdings, LLC and Subsidiaries as of January 1, 2023 and the results of their operations and their cash flows for the fiscal year ended December 31, 2023, the Successor Period from March 29, 2022 to January 1, 2023, and the Predecessor Period from January 3, 2022 to March 28, 2022, in conformity with accounting principles generally accepted in the United States of America.

Basis for Qualified Opinion

As more fully described in Note 1 to the consolidated financial statements, the Company has not identified and analyzed the value of intangible assets separately from goodwill. Accounting principles generally accepted in the United States of America require that assets acquired in a business combination be recorded at fair value, and that separately identifiable intangible assets be recognized separate from goodwill, which would result in an increase to intangible assets and decrease to goodwill and goodwill amortization expense. The effects on the consolidated financial statements have not been determined.

Responsibilities of Management for the Consolidated Financial Statements

Management is responsible for the preparation and fair presentation of the consolidated financial statements in accordance with accounting principles generally accepted in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of consolidated financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the consolidated financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern within one year after the date that the consolidated financial statements are available to be issued.



Auditor's Responsibilities for the Audit of the Consolidated Financial Statements

Our objectives are to obtain reasonable assurance about whether the consolidated financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance, but is not absolute assurance, and therefore is not a guarantee that an audit conducted in accordance with auditing standards generally accepted in the United States of America (GAAS) will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the consolidated financial statements.

In performing an audit in accordance with GAAS, we:

- Exercise professional judgment and maintain professional skepticism throughout the audit.
- Identify and assess the risks of material misstatement of the consolidated financial statements, whether due to fraud or error, and design and perform audit procedures responsive to those risks. Such procedures include examining, on a test basis, evidence regarding the amounts and disclosures in the consolidated financial statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control. Accordingly, no such opinion is expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluate the overall presentation of the consolidated financial statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern for a reasonable period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control-related matters that we identified during the audit.

FORVIS, LLP

Birmingham, Alabama April 22, 2024

Enviro-Master Holdings, LLC and Subsidiaries Consolidated Balance Sheets December 31, 2023 and January 1, 2023

	December 31, 2023			January 1, 2023
ACCETC				
ASSETS				
Current Assets	¢	1 000 705	¢	2 501 270
Cash and cash equivalents	\$	1,823,705	\$	2,501,370
Restricted cash		188,409		107,519
Accounts receivable, net		858,673		333,400
Notes receivable, current portion		-		7,789
Inventories		30,014		43,400
Prepaid expenses		269,991		460,775
Total current assets		3,170,792		3,454,253
Property and equipment, net		800,798		386,557
Goodwill, net		78,949,960		82,654,774
Capitalized software, net		925,223		-
Trademark, net		7,225		-
Right-of-use asset - operating lease		1,710,680		88,597
Notes receivable, non-current portion		105,000		-
Other assets		120,180		20,180
Total assets	\$	85,789,858	\$	86,604,361
LIABILITIES AND MEMBERS' EQUITY Current Liabilities	<u></u>	4 0 4 0 0 4 4	•	4 000 000
Accounts payable - trade	\$	1,343,941	\$	1,099,369
Accounts payable - franchisees		190,724		916,892
Deferred revenue		177,828		250,770
Accrued expenses and other current liabilities		1,141,813		827,452
Current portion of lease liability - operating lease		136,003		99,584
Current portion of lease liability - finance lease		7,170		7,493
Current portion of long-term debt		700,000		400,000
Current portion of related party note		199,837		
Total current liabilities		3,897,316		3,601,560
Operating lease liability, net of current portion		2,027,200		-
Finance lease liability, net of current portion		15,258		22,428
Notes Payable		38,602		,
Deferred tax liabilities, net		29,194		-
Long-term debt, net of current portion and debt issuance costs		39,652,308		38,553,078
Related party note, net of current portion		1,206,545		
Total liabilities		46,866,423		42,177,066
Members' Equity		38,923,435		44,427,295
Total liabilities and members' equity	\$	85,789,858	\$	86,604,361

Enviro-Master Holdings, LLC and Subsidiaries Consolidated Statements of Operations Fiscal Year Ended December 31, 2023 and Periods Ended January 1, 2023 and March 28, 2022

	Succe	Predecessor		
	Year Ended December 31, 2023	March 29, 2022 Through January 1, 2023	January 3, 2022 Through March 28, 2022	
Revenue				
Franchise sales Royalties and franchise fees Product sales, net Service revenue	\$ 721,109 12,784,934 10,381,553 3,539,641	\$ 611,895 9,940,378 8,275,399 -	\$	
Total revenue	27,427,237	18,827,672	5,888,133	
Cost of Revenue				
Products	9,233,819	7,658,104	2,168,468	
Labor	3,651,240	1,824,356	699,196	
Broker commissions on franchise sales	80,000	157,000_	27,000	
Total cost of revenue	12,965,059	9,639,460	2,894,664	
Gross Profit	14,462,178	9,188,212	2,993,469	
Operating Expenses Selling, general and administrative Salaries and wages Depreciation and amortization Transaction expenses	5,599,691 4,912,431 9,742,578 429,294	4,074,381 2,914,914 6,941,747 2,286,440	1,281,098 833,004 55,152 2,590,840	
Total operating expenses	20,683,994	16,217,482	4,760,094	
Loss from Operations	(6,221,816)	(7,029,270)	(1,766,625)	
Other (Expense) Income Interest expense Other income (expense)	(5,004,276) 106,283	(2,852,819) (1,023)	(3,188) 4,827	
Total other (expense) income	(4,897,993)	(2,853,842)	1,639	
Loss before income taxes	(11,119,809)	(9,883,112)	(1,764,986)	
Income tax expense	(95,907)			
Net Loss	<u>\$ (11,215,716)</u>	<u>\$ (9,883,112)</u>	<u>\$ (1,764,986)</u>	

Enviro-Master Holdings, LLC and Subsidiaries Consolidated Statements of Changes in Members' Equity Fiscal Year Ended December 31, 2023 and Periods Ended January 1, 2023 and March 28, 2022

	 Members' Equity		Accumulated Deficit		Total Members' Equity	
PREDECESSOR PERIOD Balance, January 3, 2022	\$ 2,162,868	\$	-	\$	2,162,868	
Distributions to members Net loss	 (1,620,577) -		- (1,764,986)		(1,620,577) (1,764,986)	
Balance, March 28, 2022	\$ 542,291	\$	(1,764,986)	\$	(1,222,695)	

	 Members' Accumulated Equity Deficit			Total Members' Equity	
SUCCESSOR PERIOD Balance, March 29, 2022	\$ -	\$	-	\$	-
Capital contributions Net loss	 54,310,407 -		- (9,883,112)		54,310,407 (9,883,112)
Balance, January 1, 2023	54,310,407		(9,883,112)		44,427,295
Capital contributions Additional paid-in capital: stock options Net loss	 5,560,000 151,856 -		- - (11,215,716)		5,560,000 151,856 (11,215,716)
Balance, December 31, 2023	\$ 60,022,263	\$	(21,098,828)	\$	38,923,435

Enviro-Master Holdings, LLC and Subsidiaries Consolidated Statements of Cash Flows Fiscal Year Ended December 31, 2023 and Periods Ended January 1, 2023 and March 28, 2022

		Succ	Di	edecessor		
		Year EndedMarch 29, 2022December 31,Through2023January 1, 2023		January 3, 2022 Through March 28, 2022		
Operating Activities						
Net loss	\$	(11,215,716)	\$	(9,883,112)	\$	(1,764,986)
Adjustments to reconcile net loss to net cash (used in) provided by operating activities						
Provision for bad debts		40,638		-		58,385
Depreciation and amortization		9,742,578		6,941,747		55,152
Amortization of debt issuance costs Loss on disposal of property		238,030		175,927		-
and equipment		-		3,287		-
Deferred income taxes		29,194		-		-
Non-cash lease expense		214,194		158,438		-
Stock compensation expense		151,856		-		-
Changes in operating assets and liabilities				<i></i>		
Accounts receivable		(53,049)		(111,791)		46,021
Inventories		35,740		12,250		(4,355)
Prepaid expenses		90,784		(411,203)		30,058
Accounts payable		(485,545)		1,299,095		(1,055,783)
Accrued expenses and		004 400		(0.4.4.0000)		770.000
other current liabilities		201,428		(344,020)		773,080
Accrued transaction fees		-		-		2,590,840
Lease liability - operating lease		227,342		99,584		-
Deferred revenue		(72,942)		233,640		(246,019)
Net cash (used in) provided						
by operating activities		(855,468)		(1,826,158)		482,393
Investing Activities						
Purchase of property and equipment		(493,094)		(89,935)		(9,107)
Issuance of notes receivable		(105,000)		(00,000)		(93,181)
Collections on notes receivable		7,789		116,670		(00,101)
Acquisition, net of cash acquired		(4,173,438)	(71,434,132)			-
Purchase of trademark	(4, 173, 430) (7, 225)		(11,404,102)			-
Purchase of capitalized software		(925,223)		-		-
Net cash used in investing						
activities		(5,696,191)		(71,407,397)		(102,288)

Enviro-Master Holdings, LLC and Subsidiaries Consolidated Statements of Cash Flows Fiscal Year Ended December 31, 2023 and Periods Ended January 1, 2023 and March 28, 2022

	Successor				Predecessor			
		ear Ended cember 31, 2023		rch 29, 2022 Through nuary 1, 2023	Jar	nuary 3, 2022 Through Irch 28, 2022		
Financing Activities								
Proceeds from long-term debt		-		40,000,000		-		
Debt issuance costs		(148,800)		(1,172,849)		-		
Payments on long-term debt		(400,000)		(300,000)		-		
Payments on related party note		(93,618)		-		-		
Net proceeds on revolving line of credit		1,710,000		250,000		-		
Payments on finance lease obligations		(7,493)		(245,114)		(132,026)		
Principle payments on notes payable		(15,205)		-		-		
Distributions paid		-		-		(1,620,577)		
Proceeds from capital contributions		4,910,000		37,310,407		-		
Net cash provided by (used in) financing activities		5,954,884		75,842,444		(1,752,603)		
Net Increase (Decrease) in Cash, Cash Equivalents, and Restricted Cash		(596,775)		2,608,889		(1,372,498)		
Orali Orali Englistado en d Destrictod		0.000.000				0 077 047		
Cash, Cash Equivalents, and Restricted Cash, Beginning of Period		2,608,889		<u>-</u>		3,877,317		
Cash, Cash Equivalents, and Restricted Cash, End of Period	\$	2,012,114	\$	2,608,889	\$	2,504,819		
Supplemental Disclosure of Noncash Investing and Financing Activities Membership interests issued as acquisition consideration	\$	650,000	\$	17,000,000	\$	-		

Note 1. Nature of Business and Summary of Significant Accounting Policies

Organization and Nature of Business

Enviro-Master Holdings, LLC and Subsidiaries (the "Company", the "Successor") is headquartered in Charlotte, North Carolina and provides hygiene, sanitation, and disinfection services to commercial businesses via its franchise agreement which grants franchisees across the United States and Canada the rights to use the Company's marketing, business techniques and systems under the "Enviro-Master" trademark. The Company was formed on March 16, 2022, for the purpose of acquiring the membership interests of Enviro-Master International Franchise, LLC (the "Operating Company", the "Predecessor") under the terms of the purchase agreement dated March 29, 2022 (the "Transaction"). Enviro-Master Holdings, LLC and Subsidiaries has two wholly owned franchisees: i) EMMB, LLC, a South Carolina limited liability company, and ii) Enviro-Master Service, LLC, a North Carolina limited liability company, and ii) Enviro-Master Service, LLC, a North Carolina limited liability company 3, 2022, to March 28, 2022 (the "Predecessor Period"), the period from March 29, 2022, to January 1, 2023 (the "Successor Period"), and for the fiscal year ended December 31, 2023.

Reporting Period

The Company's fiscal year is comprised of 13 four-week reporting periods. Fiscal year 2023 began January 2, 2023, and ended December 31, 2023. The Predecessor Period began January 3, 2022, and ended March 28, 2022, and the Successor Period began March 29, 2022, and ended January 1, 2023.

Principles of Consolidation

The accompanying consolidated financial statements include the accounts of the Company and its wholly owned subsidiaries. All significant intercompany balances and transactions have been eliminated.

Use of Estimates

The preparation of consolidated financial statements in conformity with accounting principles generally accepted in the United States of America ("GAAP") requires management to make estimates and assumptions that affect the reported amounts of assets, liabilities, and contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates and assumptions.

Reclassifications

Certain reclassifications have been made to the Predecessor Period and Successor Period consolidated financial statements included herein to conform to the 2023 presentation. Reclassifications had no impact on previously reported members' equity or net loss.

Cash and Cash Equivalents, Restricted Cash

The Company considers all cash and short-term maturities of three months or less to be cash equivalents. The Company maintains cash balances at financial institutions that are insured by the Federal Deposit Insurance Corporation ("FDIC"). At times, cash balances may be in excess of the FDIC insurance limit. The Company has not experienced any losses in such accounts.

The Company receives cash on behalf of franchisees that are typically paid out a week in arrears. Cash relating to franchisees totaled approximately \$191,000 and \$917,000 as of December 31, 2023 and January 1, 2023, respectively and is included within Accounts payable - franchisees within the consolidated balance sheets.

Restricted cash is held by the Company relating to advertising and can only be used for purposes as defined by the franchise agreement. The following is a reconciliation of cash and cash equivalents and restricted cash as presented in the accompanying consolidated statement of cash flows as of December 31, 2023 and January 1, 2023:

	De	cember 31, 2023	January 1, 2023		
Cash and cash equivalents Restricted cash	\$	1,823,705 188,409	\$	2,501,370 107,519	
Total cash, cash equivalents, and restricted cash shown in the consolidated statement of cash flows	<u>\$</u>	2,012,114	<u>\$</u>	2,608,889	

Accounts Receivable

Accounts receivable are carried at face value less an allowance for credit losses. Management monitors accounts receivable for delinquency and provides for estimated losses for specific receivables that are not likely to be collected. When determining the collectability of specific customer accounts, management considers customer credit-worthiness, past transaction history with the customer, current economic and industry trends, changes in customer payment trends, and reasonably supportable economic forecasts. The allowance for credit losses was immaterial to the consolidated financial statements as of December 31, 2023 and January 1, 2023.

Notes Receivable

Notes receivable are carried at face value less an allowance for credit losses. Management monitors notes receivable for delinquency and provides for estimated losses for specific receivables that are not likely to be collected. When determining the collectability of specific customer accounts, management considers customer credit-worthiness, past transaction history with the customer, current economic and industry trends, changes in customer payment trends, and reasonably supportable economic forecasts. No allowance for credit losses was deemed necessary as of December 31, 2023 and January 1, 2023.

Inventories

Inventories are stated at cost, determined on a first-in, first out ("FIFO") basis, and consist primarily of soap dispensers and sprayers.

Capitalized software

The Company internally develops its software application and capitalizes related costs incurred during the application's development stage, which include costs to design the software configuration and interfaces, coding, installation, and testing. Such capitalized costs include external direct costs utilized in developing and obtaining the applications and payroll and payroll-related expenses for employees who are directly associated with the development of the applications. Costs incurred during the preliminary and post-implementation stages of internal use computer software are expensed as incurred. Costs incurred to maintain existing software are expensed as incurred. Capitalized software costs are amortized over three years after the Company has entered the post-implementation stage. As of December 31, 2023, the Company's software application had not yet entered the post-implementation stage and therefore did not amortize any of its capitalized costs during the fiscal year.

Property and Equipment

Property and equipment is remeasured to the estimated fair value for assets acquired through business combinations and subsequently depreciated. Property and equipment acquired in the ordinary course of business is recorded at cost and subsequently depreciated. Major expenditures and those that substantially increase useful lives are capitalized. Depreciation is determined using the straight-line method over the estimated useful lives of the assets. Leasehold improvements are depreciated over the lesser of the lease term or the estimated useful life

of the related asset. When property and equipment is retired or sold, the cost and related accumulated depreciation is removed from the consolidated balance sheet, and the resulting gain or loss is included in the consolidated statements of operations. Expenditures for maintenance and repairs are expensed when incurred.

Goodwill

Goodwill, which represents the excess of the consideration paid over the fair value of recognized net assets acquired, includes, but is not limited to, the value of the workforce in place, the ability to generate profits and cash flows, an established going concern, and other unidentifiable intangible assets. The Company amortizes goodwill on a straight-line basis over 10 years and tests goodwill for impairment at the entity level. Goodwill is tested for impairment only when a triggering event occurs or circumstances change that indicate the fair value of the entity may be less than its carrying value. In accounting for business combinations, the Company does not recognize separately from goodwill customer-related intangible assets not capable of being sold or licensed independently from other assets of the business and noncompetition agreements. No impairment indicators exist as of December 31, 2023 and January 1, 2023.

In connection with the Transaction, the Company assigned all amounts paid in excess of identifiable assets to goodwill. GAAP requires that separately identifiable intangible assets acquired in a business combination be recorded at fair value. Management has not identified or established a value for the separately identifiable intangible assets acquired, which represents a departure from GAAP. If the Company had recorded separately identifiable intangible assets at fair value, as required, it would result in an increase to intangible assets and a decrease to goodwill and the related goodwill amortization expense. The impact on the consolidated financial statements has not been determined.

Impairment of Long-Lived Assets

The Company reviews the carrying value of long-lived assets for impairment whenever triggering events or changes in circumstances indicate the carrying value of an asset may not be recoverable from the estimated future cash flow expected to result from its use and eventual disposition. In cases where the estimated undiscounted future cash flow is less than the carrying value of an asset, an impairment loss is recognized equal to the amount by which the carrying value exceeds the fair value of the asset. Based on management's assessments, no impairment occurred during the fiscal year ended December 31, 2023, the Successor Period, or the Predecessor Period.

Debt Issuance Costs

Debt issuance costs are incurred by the Company in connection with the issuance of debt and are deferred and amortized to interest expense on the consolidated statements of operations using a method that approximates the effective interest method.

Revenue Recognition

The Company derives its revenues from franchise sales, product sales, royalties and other franchise fees, and from its wholly owned franchisees. Revenues are recognized when control of the promised goods or services is transferred to customers in an amount that reflects the consideration the Company expects to be entitled to receive in exchange for those goods or services. The Company applies the five-step model under FASB ASC Topic 606, *Revenue from Contracts with Customers*, to determine when revenue is earned and recognized.

The Company has elected the practical expedient available to private companies under ASU 2021-02, *Franchisors* - *Revenue from Contracts with Customers,* to account for pre-opening activities as one distinct performance obligation for new franchise sales. The initial franchise fees are payable based on contract terms prior to the franchise opening. The transaction price is recognized as revenue on a straight-line basis over a 2-month period, based on timing of when services are satisfied for pre-opening activities. Franchise renewals and existing franchise sales are recognized as revenue upon execution of the franchise agreement.

The Company receives a significant portion of its revenue from sales-based royalties, admin fees, advertising fees and other franchise fees from franchisees. These fees are determined as a percentage of sales and are recognized in the period earned by the Company.

The Company receives a significant portion of its revenue from product sales to franchisees. Revenues and the related cost of revenue are recognized upon delivery of the products to franchisees.

The Company receives a significant portion of its revenue from its wholly owned franchisees, which primarily consists of services performed for the franchisees' customers. These amounts are included in service revenue in the consolidated statements of operations. Revenues and the related cost of revenue are recognized upon delivery of products to customers or completion of services.

Selling, General, and Administrative Expenses

Selling, general, and administrative expenses consist of costs associated with administrative and support functions related to the Company's existing business as well as growth and development activities. These costs primarily consist of consultants, credit card and bank fees, advertising, and professional expenses.

Franchise Agreements

Franchise agreements are available for a territory over a specified period of time. The franchise agreement defines the territory the new franchise is able to serve. New franchisees are required to pay the Company an initial franchise fee plus a weekly royalty fee of 6%, an administrative fee of 5%, and an advertising fee of 2% of the franchisee's weekly gross receipts for the duration of the franchise agreement.

Advertising Costs

Advertising costs are expensed as incurred. The Company incurred advertising of costs of approximately \$730,000 during the fiscal year ended December 31, 2023, and approximately \$488,000 and \$189,000 during the Successor Period and Predecessor Period, respectively.

Income Taxes

Income taxes are provided for the tax effects of transactions reported in the consolidated financial statements. As a limited liability company ("LLC"), the Company is not subject to income taxes under current federal and some state tax laws. Enviro-Master Service, LLC and EMMB, LLC, however, are taxable entities. As such, the Company accounts for income taxes under the asset and liability method for these entities.

The Company has determined that it does not have any material unrecognized tax benefits or obligations as of December 31, 2023 and January 1, 2023. Interest and penalties, if any, are reflected in income taxes in the accompanying consolidated statements of operations. Tax years remain generally subject to examination by federal and state tax authorities for 3 years. See Note 9 for additional information regarding income taxes.

Stock-Based Compensation

The Company accounts for stock-based awards in accordance with ASC 718, Stock Compensation, which requires compensation cost related to share-based payments, classified as equity awards, to be measured based on the grant date fair value of the award. The Company recognizes compensation cost over the requisite service period.

Concentrations

The Company purchases products on behalf of its franchisees and sells them to the franchisees as needed. The largest vendor in the Company's purchasing portfolio represented approximately 63% of purchases during the fiscal year ended December 31, 2023. No such concentration was present during the Successor and Predecessor Periods.

Recently Adopted Accounting Standards

In June 2016, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update ("ASU") 2016-13, *Financial Instruments - Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments.* This update requires companies to measure credit losses utilizing a methodology that reflects expected credit losses and requires a consideration of a broader range of reasonable and supportable information to estimate credit losses. The Company adopted ASU 2016-13 on January 2, 2023. The adoption of this standard did not have a material impact on the consolidated financial statements.

Note 2. Business Combinations

Acquisition

During the fiscal year ended December 31, 2023, the Company completed an acquisition of two of its franchisees. The aggregate fair value of consideration paid was \$6,550,000.

Pursuant to the Interest and Stock Purchase Agreement, the Company acquired all interests of Enviro-Master Service, LLC and EMMB, LLC as of January 17, 2023. The purchase price allocation process was complete as of December 31, 2023.

The consideration paid was allocated to the assets acquired and liabilities assumed according to their estimated fair values at the time of acquisition as follows:

Cash Accounts receivable Inventories Property and equipment	\$ 226,562 512,862 22,354 120,764
Gross assets acquired, excluding goodwill Accounts payable and accrued expenses and other current liabilities Goodwill	 882,542 116,882 <u>5,784,340</u>
Total purchase price	\$ 6,550,000

As a result of the acquisition, approximately \$115,000 of transaction costs were expensed within the consolidated statements of operations during the fiscal year ended December 31, 2023.

Transaction

As discussed in Note 1, the Company purchased all the membership interests in the Operating Company in conjunction with the Transaction on March 29, 2022. The total purchase price was \$90,938,951, which was made up of the following:

Cash Rollover equity	\$ 73,938,951 17,000,000
Total consideration transferred	\$ 90,938,951

The purchase price was financed through a combination of debt borrowings of \$40,000,000 and equity contributions of \$54,310,407.

The Transaction was recorded in accordance with ASC 805: Business Combinations. Management has elected to apply pushdown accounting to the Company's consolidated financial statements. Accordingly, the Company has recorded all assets acquired and liabilities assumed at the acquisition date fair values with any excess recognized as goodwill.

The consideration paid was allocated to the assets acquired and liabilities assumed according to their estimated fair values at the time of acquisition as follows:

Assets Acquired		
Cash	\$	2,504,819
Accounts receivable		221,609
Prepaid expenses and other current assets		125,402
Note receivable		124,459
Property, plant, and equipment		540,439
Right of use asset		247,035
Gross assets acquired, excluding goodwill		3,763,763
Liabilities Assumed		
Accounts payable and accrued expenses		1,317,821
Other current liabilities		552,622
Lease liability - finance lease		35,324
Lease liability - operating lease		275,035
Total liabilities assumed		2,180,802
Net assets acquired, excluding goodwill		1,582,961
Goodwill		89,355,990
Total purchase price	<u>\$</u>	90,938,951

Goodwill, which represents the excess of the consideration paid over the fair value of recognized net assets acquired, includes, but is not limited to, the value of the workforce in place, the ability to generate profits and cash flows, an established going concern, customer-related intangibles, and other unidentifiable intangible assets.

Note 3. Goodwill

The change in the carrying amount of goodwill during the fiscal year ended December 31, 2023 is as follows:

Balance, January 2, 2023	\$	89,355,990
Goodwill related to the acquisition during the fiscal year ended December 31, 2023		5,784,340
Balance, December 31, 2023	<u>\$</u>	95,140,330

Goodwill consists of the following as of December 31, 2023 and January 1, 2023:

		Decembe	er 31, 2023	
	Useful Life	Carrying Amount	Accumulated Amortization	Goodwill, Net
Goodwill	10 years	\$ 95,140,330	\$ (16,190,370)	\$ 78,949,960
		January	y 1, 2023	
	Useful Life	Carrying Amount	Accumulated Amortization	Goodwill, Net
Goodwill	10 years	\$ 89,355,990	\$ (6,701,216)	\$ 82,654,774

Amortization expense of goodwill was approximately \$9,489,000 and \$6,701,000 during the fiscal year ended December 31, 2023 and Successor Period, respectively. Future amortization of goodwill is expected to be approximately \$9,514,000 per year for the next 5 years. The weighted average useful life of goodwill as of December 31, 2023 was approximately 8 years.

Note 4. Notes Receivable

Note receivables represents company-offered financing of initial franchise fees and working capital loans to certain Company franchisees. These are secured with the franchisee's business as pledged collateral. The notes receivable balance as of December 31, 2023 of approximately \$105,000 consisted of a 5-year note which earns interest of 8% per annum and is classified as a non-current asset on the Company's consolidated balance sheet. The notes receivable balance as of January 1, 2023 of approximately \$8,000 consisted of a note with a 3 year original term that was non-interest bearing and is classified as a current asset on the Company's consolidated balance sheet.

Note 5. Property And Equipment

Property and equipment consisted of the following as of December 31, 2023 and January 1, 2023:

	Estimated Useful Life	Dee	cember 31, 2023	Já	anuary 1, 2023
Software and website design Computer equipment Office furniture and fixtures Leasehold improvements Warehouse equipment Vehicles Construction in progress Accumulated depreciation and amortization	3 years 5 years 4 years 11 years 5 years 5 years	\$	368,910 172,356 56,373 306,111 131,878 74,074 178,595 (487,499)	\$	330,096 164,608 56,373 19,555 - - 50,000 (234,075)
Property and equipment, net		<u>\$</u>	800,798	<u>\$</u>	386,557

Depreciation expense was approximately \$253,000 during the fiscal year ended December 31, 2023 and approximately \$241,000 and \$55,000 during the Successor Period and Predecessor Period, respectively.

Note 6. Accrued Liabilities

Accrued expenses and other current liabilities consist of the following as of December 31, 2023 and January 1, 2023:

	Dec	ember 31, 2023	Ja	anuary 1, 2023
Accrued commissions Accrued credit card liability Accrued payroll and payroll taxes Other accrued expenses	\$	218,672 15,365 309,504 592,605	\$	131,731 21,282 222,085 452,354
	<u>\$</u>	1,136,146	\$	827,452

Note 7. Long-Term Debt and Revolving Line of Credit

Long-term debt and revolving line of credit consist of the following as of December 31, 2023 and January 1, 2023:

	De	ecember 31, 2023	 January 1, 2023
Revolving line of credit Term loan	\$	1,960,000 39,300,000	\$ 250,000 39,700,000
Total current and long-term debt Current portion Debt issuance costs, net		41,260,000 (700,000) (907,692)	 39,950,000 (400,000) (996,922)
Total long-term debt, net of current portion and debt issuance costs	\$	39,652,308	\$ 38,553,078

As part of the Transaction, the Company entered into a credit facility (the "Credit Agreement") which provides for (i) a \$40,000,000 term loan credit facility (the "Term Loan") and (ii) availability of a \$3,000,000 revolving credit facility (the "Revolver"). Borrowings under the Credit Agreement are secured by substantially all of the Company's assets. The Term Loan requires quarterly payments of principal and interest beginning June 30, 2022, with a balloon payment upon maturity on March 29, 2027. The Revolver and any accrued interest is also due upon maturity on March 29, 2027.

The Term Loan and Revolver accrue interest at an adjusted term Secured Overnight Financing Rate ("SOFR") plus the applicable margin of 6.5% per annum, as defined in the Credit Agreement. The interest rate associated with the Term Loan was approximately 11.97% and 10.76%, respectively, as of December 31, 2023 and January 1, 2023. The interest rate associated with the Revolver was approximately 11.97% and 10.92%, respectively, as of December 31, 2023 and January 1, 2023.

The Term Loan and Revolver, described above, are subject to certain financial covenants and restrictions specified in the Credit Agreement. In addition, the Company is subject to excess cash flow requirements, which require accelerated debt principal payments under certain conditions. As of December 31, 2023, the Company was in compliance with its financial covenants, and no acceleration of debt principal payments was required.

Principal maturities of long-term debt and the revolving line of credit consist of the following as of December 31, 2023:

2024 2025 2026 2027	\$	700,000 800,000 800,000 38,960,000
	<u>\$</u>	41,260,000

Note 8. Related Party Note

The related party note consists of the following as of December 31, 2023:

Related party note Current portion	\$ 1,406,382 (199,837)
Total seller note (related party), net of current portion	\$ 1,206,545

In connection with the acquisition during the fiscal year ended December 31, 2023, the Company entered into a promissory note agreement with a related party of the Company in the amount of \$1,500,000. The agreement requires monthly principal payments beginning March 1, 2023 and matures on January 17, 2033. Interest payments are required monthly on the unpaid principal balance at 6%.

Principal maturities of long-term debt consist of the following as of December 31, 2023:

2024 2025	\$ 199,83 199,83	7
2026 2027	199,83 199,83	
2028	199,83	57
Thereafter	407,19	1
	<u>\$ 1,406,38</u>	2

Note 9. Income Taxes

As a limited liability company ("LLC"), the Company is not subject to income taxes under current federal and some state tax laws. The Company's taxable income or loss is allocated to the members in accordance with the operating agreement and is reflected in their income tax returns.

Enviro-Master Service, LLC and EMMB, LLC are taxable entities which are consolidated for financial reporting. As such, the Company accounts for income taxes under the asset and liability method for Enviro-Master Service, LLC and EMMB, LLC. Deferred tax assets and liabilities are recognized for the future tax consequences attributable to differences between the financial statement carrying amounts of existing assets and liabilities and their respective tax basis, measured using enacted tax rates expected to apply to taxable income in the years in which those temporary differences are expected to be recovered or settled. The effect on deferred tax assets or liabilities of a change in tax rates is recognized in income in the period that includes the enactment date. Deferred income tax

expense represents the change during the period in the deferred tax assets or liabilities. Deferred tax assets are reduced by a valuation allowance when, in the opinion of management, it is more likely than not that some portion or all of the deferred tax assets will not be realized.

The income tax provision consists of the following during the fiscal year ended December 31, 2023:

Current Provision Federal State	\$ 43,297 <u>23,416</u>
Deferred Tax Expense	<u> </u>
Federal State	27,193 2,001
	29,194
Income tax Expense	<u>\$ 95,907</u>

The tax effect of the temporary differences giving rise to the Company's net deferred tax liabilities approximates the following as of December 31, 2023:

Deferred Tax Assets Bad debt reserves State tax loss carryforwards	\$ 6,775 <u>316</u>
Total deferred tax assets	7,091
Deferred Tax Liabilities Property and equipment Other	(18,162) (18,123)
Total deferred tax liabilities	(36,285)
Net deferred tax liabilities	<u>\$ (29,194)</u>

The difference between the Company's effective tax rate and the statutory federal income tax rate during the fiscal year ended December 31, 2023 is primarily due to goodwill amortization, state income taxes, and non-deductible expenses.

Note 10. Leases

Significant Lease Accounting Policies

The Company evaluates leases at contract inception to determine whether the Company has the right to control use of the identified asset for a period of time in exchange for consideration. If it is determined the Company has the right to obtain substantially all of the economic benefit from use of the identified asset and the right to direct the use of the identified asset, the Company recognizes a right-of-use ("ROU") asset and lease liability. Also, at contract inception, the Company evaluates leases to estimate their expected term which includes renewal options that the Company is reasonably assured will be exercised and evaluates the classification of the lease as either an operating lease or a finance lease. Operating lease liabilities represent the present value of lease payments not yet paid.

Operating lease ROU assets represent the Company's right to use an underlying asset and are based upon the operating lease liabilities adjusted for prepayments or accrued lease payments, initial direct costs, and lease incentives. Additionally, the Company uses the risk-free interest rate at the time of commencement or modification date in determining the present value of lease payments. The Company assesses the impairment of the ROU asset at the asset group level whenever events or changes in circumstances indicate that the carrying value of the asset may not be recoverable.

The Company's leases typically contain rent escalations over the lease term. The Company recognizes expense for these leases on a straight-line basis over the lease term. Variable lease costs, which may include common area maintenance, insurance, and taxes are not included in the lease liability and are expensed in the period incurred. The Company's lease agreements do not contain any material residual value guarantees or material restrictive covenants.

Operating Lease Obligations

The Company leases office space accounted for as an operating lease. The lease has a remaining lease term of 7 years. Leases with an initial term of twelve months or less are not recorded on the consolidated balance sheets. The Company recognizes lease expense for short-term leases on a straight-line basis over the lease term. The Company accounts for fixed lease and non-lease components together as a single, combined lease component. The depreciable life of related assets and leasehold improvements are limited by the expected lease term.

The lease cost and other required information are as follows for the fiscal year ended December 31, 2023 and the Successor Period and Predecessor Period:

	2023			Successor Period		Predecessor Period	
Operating lease cost	\$	256,112	\$	159,728	\$	69,272	

The Company's variable lease cost and costs related to short-term leases, those with a duration between one and twelve months, were immaterial to the consolidated financial statements during the fiscal year ended December 31, 2023 and the Successor Period and the Predecessor Period.

Other information	C	Fiscal ear Ended December 31, 2023	 uccessor Period	 decessor Period
Cash paid for amounts included in the measurement of lease liabilities Operating cash flows from operating leases	\$	99,725	\$ 176,331	\$ 57,636
Right-of-use assets obtained in exchange for new operating lease liabilities	\$	1,836,277	\$ -	\$ -

The following summarizes the weighted average remaining lease term and discount rates as of December 31, 2023 and January 1, 2023.

	December 31, 2023	January 1, 2023
Weighted-average remaining lease term Weighted-average discount rate	7 years 3.65%	
As of December 31, 2023, maturities of lease liabilities were as	s follows:	
2024 2025 2026 2027 2028 Thereafter	\$ 225,767 393,785 405,608 417,765 430,278 	
Total lease payments Tenant improvement allowance Imputed interest	2,965,523 (402,845 <u>(399,475</u>	
Present value of lease liabilities	<u>\$2,163,203</u>	

Finance Lease Obligations

The Company leases computer equipment accounted for as a finance lease. The lease has a remaining term of 3 years.

The Company's estimated future minimum lease payments required under the terms of the lease are as follows:

2024 2025 2026	\$ 7,962 8,686 7,238
Total minimum lease payments Amounts representing interest	 23,886 <u>(1,458</u>)
Present value of net minimum lease payments Current portion of net minimum lease payments	 22,428 <u>(7,170</u>)
Finance lease obligations, net of current portion	\$ 15,258

Note 11. Related Party Activities

The Company incurred a management fee from its financial sponsor of approximately \$308,000 and \$203,000 during the fiscal year ended December 31, 2023 and the Successor Period, respectively, which is recorded in selling, general, and administrative expenses on the accompanying consolidated statements of operations. The Company receives royalty revenue from franchisees owned by a related party. The total amount of revenue earned was approximately \$1,039,000 during the fiscal year ended December 31, 2023, and \$841,000 and \$244,000 during the Successor Period and Predecessor Period, respectively. During the fiscal year ended December 31, 2023, the Company entered into a promissory note agreement with a related party of the Company in the amount of \$1,500,000 (see Note 8). The Company paid approximately \$167,000 in principal and interest in relation to the promissory note during the fiscal year ended December 31, 2023. As of December 31, 2023, the principal balance of the promissory note was approximately \$1,406,000.

Note 12. Retirement Plan

The Company participates in a SIMPLE IRA plan, where it matches 100 percent of up to 3 percent of an eligible employee's contributions. All full-time employees are eligible to participate in the plan after ninety days of employment. In conjunction with the SIMPLE IRA plan, the Company contributed a total of approximately \$106,000 during the fiscal year ended December 31, 2023, and approximately \$71,000 and \$20,000 during the Successor Period and Predecessor Period, respectively.

Note 13. Stock-Based Compensation

The Company's parent, Enviro-Master Holdings, LLC, issued certain time and performance vesting options pursuant to the Enviro-Master Holdings, LLC 2023 Option Incentive Plan (the "Equity Plan"). The Company has accounted for the options within its consolidated financial statements as the option holders are providing services directly related to the Company and its subsidiaries. Unvested options issued to employees are forfeited in accordance with the Equity Plan upon termination of employment from the Company. Options granted under the plan have been allocated into two groups based on their vesting criteria: time vesting and performance vesting options.

Time vesting options are equity classified awards, vest over a 5-year period and will vest 20% on each anniversary date of the date of grant. The vesting commencement date for Units granted in 2023 commenced on March 29, 2022. The Company accounts for forfeitures in the period they occur resulting in a reversal of all previously recognized compensation expense for awards forfeited. There were no forfeitures of time vesting options during the fiscal year ended December 31, 2023. As such, the Company recognizes compensation expense over a straight-line period of 5 years and has recorded compensation expense of approximately \$151,856 net of forfeitures of \$0 for the fiscal year ended December 31, 2023. This expense is included in salaries and wages on the accompanying consolidated statement of operations.

Performance vesting options are equity classified awards, and vest subject to a liquidity event in which an Internal Rate of Return of 20% is achieved by the equity holders of the Company, provided that the option holder's continuous service has not terminated prior to such vesting date, which was not considered probable at the date of issuance or as of December 31, 2023. As a result, no compensation expense or equity has been recorded based on this criterion.

No portion of the option shall vest after the date the option holder's continuous service terminates for any reason, however options vested prior to termination without cause may be exercised within a prescribed time-period. The Company accounts for forfeitures in the period they occur resulting in a reversal of all previously recognized compensation expense for awards forfeited. There were no forfeitures of performance vesting options during the fiscal year ended December 31, 2023.

Information regarding activity under the Company's Equity Plan is summarized as follows:

		Time Ves	Performance Vesting Units			
	Number of Units	Number of Vested Units	Weighted Average Exercise Price per Unit	Weighted Average Fair Value per Unit at Grant Date	Number of Units	Weighted Average Exercise Price per Unit
Outstanding January 2, 2023 Granted Exercised Cancelled or forfeited	- 1,840.00 - -	290.50	\$ - 1,000.00 - -	\$ - 1,000.00 -	- 1,840.00 - -	\$ - 1,000 - -
Outstanding, December 31, 2023	1,840.00	-	<u>\$ 1,000.00</u>	<u>\$ 1,000.00</u>	1,840.00	<u>\$ 1,000</u>

The fair value of each stock option granted was estimated on the date of grant using the Black-Scholes option pricing model with the following average assumptions:

Weighted Averages	<u>2023 Grants</u>
Distribution yield	0.00%
Expected volatility	22.00%
Risk-free interest rate	3.51%
Expected term	5.00 years

As of December 31, 2023, there was \$340,116 of total unrecognized compensation cost related to non-vested time based awards granted under the Company's Equity Plan which is expected to be recognized over a weighted average period of 3.45 years.

Note 14. Subsequent Events

The Company evaluated all events or transactions that occurred after December 31, 2023 through April 22, 2024, the date that the consolidated financial statements were available to be issued. During this period, there have been no material events that would require an adjustment or disclosure in the consolidated financial statements other than as discussed below.

On March 28, 2024, the Company acquired Enviro-Master of Atlanta, LLC for approximately \$1,150,000. The initial accounting for the acquisition has not been finalized as of the date that the consolidated financial statements were available to be issued.

Enviro-Master Holdings, LLC and Subsidiaries

Independent Auditor's Report and Consolidated Financial Statements

As of January 1, 2023, for the Successor Period from March 29, 2022 through January 1, 2023, and for the Predecessor Period from January 3, 2022 through March 28, 2022

Table of Contents

Independent Auditor's Report	1
Consolidated Financial Statements:	
Consolidated Balance Sheet	3
Consolidated Statements of Operations	4
Consolidated Statements of Changes in Members' Equity (Deficit)	5
Consolidated Statements of Cash Flows	6
Notes to Consolidated Financial Statements	8

FORV/S

Independent Auditor's Report

Board of Managers and Members Enviro-Master Holdings, LLC and Subsidiaries Charlotte, North Carolina

We have audited the accompanying consolidated financial statements of Enviro-Master Holding, LLC and Subsidiaries (collectively, the "Company"), which comprise the consolidated balance sheet as of January 1, 2023, and the related consolidated statements of operations, changes in members' equity, and cash flows for the Successor Period from March 29, 2022 to January 1, 2023, and the Predecessor Period from January 3, 2022 to March 28, 2022, and the related notes to the consolidated financial statements.

In our opinion, except for the effects of not reporting the value of identifiable intangible assets separately from goodwill, and the resulting effect on amortization expense, as discussed in the Basis for Qualified Opinion paragraph, the consolidated financial statements referred to above present fairly, in all material respects, the financial position of Enviro-Master Holdings, LLC and Subsidiaries as of January 1, 2023, and the results of their operations and their cash flows for the Successor Period from March 29, 2022 to January 1, 2023, and the Predecessor Period from January 3, 2022 to March 28, 2022, in conformity with accounting principles generally accepted in the United States of America.

Basis for Qualified Opinion

As more fully described in Note 1 to the consolidated financial statements, the Company has not identified and analyzed the value of intangible assets separately from goodwill. Accounting principles generally accepted in the United States of America require that assets acquired in a business combination be recorded at fair value, and that separately identifiable intangible assets be recognized separate from goodwill, which would result in an increase to intangible assets and decrease to goodwill and goodwill amortization expense. The effects on the consolidated financial statements have not been determined.

Responsibilities of Management for the Consolidated Financial Statements

Management is responsible for the preparation and fair presentation of the consolidated financial statements in accordance with accounting principles generally accepted in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of consolidated financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the consolidated financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern within one year after the date that the consolidated financial statements are available to be issued.



Auditor's Responsibilities for the Audit of the Consolidated Financial Statements

Our objectives are to obtain reasonable assurance about whether the consolidated financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance, but is not absolute assurance, and therefore is not a guarantee that an audit conducted in accordance with auditing standards generally accepted in the United States of America (GAAS) will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the consolidated financial statements.

In performing an audit in accordance with GAAS, we:

- Exercise professional judgment and maintain professional skepticism throughout the audit.
- Identify and assess the risks of material misstatement of the consolidated financial statements, whether due to fraud or error, and design and perform audit procedures responsive to those risks. Such procedures include examining, on a test basis, evidence regarding the amounts and disclosures in the consolidated financial statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control. Accordingly, no such opinion is expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluate the overall presentation of the consolidated financial statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern for a reasonable period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control-related matters that we identified during the audit.

FORVIS, LLP

Birmingham, AL March 27, 2023

ASSETS

Current assets:		
	¢	2 501 270
Cash and cash equivalents	\$	2,501,370
Restricted cash		107,519
Accounts receivable, net		333,400
Notes receivable		7,789
Inventories		43,400
Prepaid expenses		460,775
Total current assets		3,454,253
Property and equipment, net		386,557
Goodwill, net		82,654,774
Right-of-use asset - operating lease, net		88,597
Other assets		20,180
Total assets	\$	86,604,361
LIABILITIES AND MEMBERS' EQUITY Current liabilities:		
Accounts payable - trade	\$	1,099,369
Accounts payable - franchisees		916,892
Deferred revenue		250,770
Accrued expenses and other current liabilities		827,452
Current portion of lease liability - operating lease		99,584
Current portion of lease liability - finance lease		7,493
Current portion of long-term debt		400,000
Total current liabilities		3,601,560
Finance lease liability, net of current portion		22,428
Long-term debt, net of current portion and debt issuance costs		38,553,078
Total liabilities		42,177,066
Members' equity		44,427,295
Total liabilities and members' equity	\$	86,604,361

Enviro-Master Holdings, LLC and Subsidiaries Consolidated Statements of Operations Periods Ending January 1, 2023 and March 28, 2022

	Successor	Predecessor
	March 29, 2022	January 3, 2022
	Through	Through
	January 1, 2023	March 28, 2022
Revenue:		
Franchise sales	\$ 611,895	\$ 79,500
Royalties and franchise fees	9,940,378	3,089,480
Product sales, net	8,275,399	2,719,153
Total revenue	18,827,672	5,888,133
Cost of revenue:		
Products	7,658,104	2,168,468
Labor	1,824,356	699,196
Broker commissions on franchise sales	157,000	27,000
	/	
Total cost of revenue	9,639,460	2,894,664
Gross profit	9,188,212	2,993,469
Operating expenses:		
Selling, general and administrative	4,074,381	1,281,098
Salaries and wages	2,914,914	833,004
Depreciation and amortization	6,941,747	55,152
Total operating expenses	13,931,042	2,169,254
(Loss) income from operations	(4,742,830)	824,215
Other (expense) income:		
Interest expense	(2,852,819)	(3,188)
Other (expense) income	(2,052,019) (1,023)	4,827
Transaction expenses	(2,286,440)	(2,590,840)
Total other expense	(5,140,282)	(2,589,201)
Net loss	¢ (0.002.110)	\$ (1,764,986)
	\$ (9,883,112)	\$ (1,764,986)

Enviro-Master Holdings, LLC and Subsidiaries Consolidated Statements of Changes in Members' Equity (Deficit) Periods Ending January 1, 2023 and March 28, 2022

PREDECESSOR PERIOD	Total Members' uity (Deficit)
Balance, January 3, 2022	\$ 2,162,868
Distributions to members	(1,620,577)
Net loss	 (1,764,986)
Balance, March 28, 2022	\$ (1,222,695)

SUCCESSOR PERIOD	Total Members Equity	
Balance, March 29, 2022	\$	-
Capital contributions	54,310,4	07
Net loss	(9,883,1	12)
Balance, January 1, 2023	\$ 44,427,2	95

Enviro-Master Holdings, LLC and Subsidiaries Consolidated Statements of Cash Flows Periods Ending January 1, 2023 and March 28, 2022

	Successor March 29, 2022 Through January 1, 2023	Predecessor January 3, 2022 Through March 28, 2022
	<u> </u>	
Cash flows from operating activities:		
Net loss	\$ (9,883,112)	\$ (1,764,986)
Adjustments to reconcile net loss to net cash provided by		
(used in) operating activities:		
Provision for bad debts	-	58,385
Depreciation and amortization	6,941,747	55,152
Amortization of debt issuance costs	175,927	-
Loss on disposal of property and equipment	3,287	-
Non-cash lease expense	158,438	-
Changes in operating assets and liabilities:	,	
Accounts receivable	(111,791)	46,021
Inventories	12,250	(4,355)
Prepaid expenses	(411,203)	30,058
Accounts payable	1,299,095	(1,055,783)
Accrued expenses and other current liabilities	(344,020)	773,080
Accrued transaction fees	-	2,590,840
Lease liability - operating lease	99,584	_,,
Deferred revenue	233,640	(246,019)
		(=:0,0:0)
Net cash (used in) provided by operating activities	(1,826,158)	482,393
Cash flows from investing activities:		
Purchase of property and equipment	(89,935)	(9,107)
Issuance of notes receivable	(00,000)	(93,181)
Collections on notes receivable	116,670	(00,101)
Acquisition, net of cash acquired	(71,434,132)	_
	(11,101,102)	
Net cash used in investing activities	(71,407,397)	(102,288)
Cash flows from financing activities:		
Proceeds from long-term debt	40,000,000	-
Debt issuance costs	(1,172,849)	-
Payments on long-term debt	(300,000)	-
Proceeds on revolving line of credit	250,000	-
Payments on finance lease obligations	(245,114)	(132,026)
Distributions paid	-	(1,620,577)
Proceeds from issuance of equity	37,310,407	-
Net cash provided by (used in) financing activities	75,842,444	(1,752,603)

Enviro-Master Holdings, LLC and Subsidiaries Consolidated Statements of Cash Flows Periods Ending January 1, 2023 and March 28, 2022

	Marc T	iccessor h 29, 2022 hrough ary 1, 2023	Jar	redecessor nuary 3, 2022 Through rch 28, 2022
Net increase (decrease) in cash, cash equivalents, and restricted cash Cash, cash equivalents, and restricted cash - beginning of period		2,608,889		(1,372,498) 3,877,317
Cash, cash equivalents, and restricted cash - end of period	\$	2,608,889	\$	2,504,819
Supplemental disclosure of non-cash investing and financing activities: Membership interests issued as acquisition consideration	\$	17,000,000	\$	

Notes to Consolidated Financial Statements

1. Nature of Business and Summary of Significant Accounting Policies

Organization and nature of business

Enviro-Master Holdings, LLC and Subsidiaries (the "Company", the "Successor") is headquartered in Charlotte, North Carolina and provides hygiene, sanitation, and disinfection services to commercial business via its franchise agreement which grants franchisees across the United States and Canada the rights to use the Company's marketing, business techniques and systems under the "Enviro-Master" trademark. The Company was formed on March 16, 2022, for the purpose of acquiring the membership interests of Enviro-Master International Franchise, LLC (the "Operating Company", the "Predecessor") under the terms of the purchase agreement dated March 29, 2022 (the "Transaction"). The accompanying consolidated financial statements are presented for the period from January 3, 2022, to March 28, 2022 (the "Predecessor Period") and includes only the results of the Operating Company from March 29, 2022, to January 1, 2023 (the "Successor Period") and includes the results of the Company which is inclusive of the Operating Company.

Reporting period

The Company has elected to report on a fiscal calendar year effective in 2022 for 13 four-week reporting periods. Fiscal year 2022 began January 3, 2022 and ended January 1, 2023.

Principles of consolidation

The accompanying consolidated financial statements include the accounts of the Company. All significant intercompany balances and transactions have been eliminated.

Use of estimates

The preparation of consolidated financial statements in conformity with accounting principles generally accepted in the United States of America ("GAAP") requires management to make estimates and assumptions that affect the reported amounts of assets, liabilities, and contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates and assumptions.

Cash and cash equivalents

The Company considers all cash and short-term maturities of three months or less to be cash equivalents. The Company maintains cash balances at financial institutions that are insured by the Federal Deposit Insurance Corporation ("FDIC"). At times, cash balances may be in excess of the FDIC insurance limit. The Company has not experienced any losses in such accounts.

The Company receives cash on behalf of franchisees that are typically paid out a week in arrears. Cash relating to franchisees totaled \$916,892 as of January 1, 2023. See Accounts payable – franchisees below.

Restricted cash

Restricted cash is held by the Company relating to advertising and can only be used for purposes as defined by the franchise agreement. Restricted cash relating to advertising totaled \$107,519 as of January 1, 2023.

Following is a reconciliation of cash and cash equivalents and restricted cash as presented in the accompanying consolidated statements of cash flows as of January 1, 2023:

Cash and cash equivalents Restricted cash, current	\$	2,501,370 107,519
Total cash, cash equivalents, and restricted cash shown in the consolidated statement of cash flows	<u>\$</u>	2,608,889

Accounts receivable

Accounts receivable are carried at face value less an allowance for doubtful accounts. On a periodic basis, management evaluates accounts receivable balances and establishes allowances for doubtful accounts, based on history of past write-offs and collections, as well as customer specific circumstances and credit conditions. No allowance was deemed necessary as of January 1, 2023.

Notes receivable

Notes receivable are carried at face value less an allowance for doubtful accounts. On a periodic basis, management evaluates notes receivable balances and establishes allowances for doubtful accounts, based on history of past write-offs and collections, as well as customer specific circumstances and credit conditions. No allowance was deemed necessary as of January 1, 2023.

Inventories

Inventory is stated at cost, determined on a first-in, first out ("FIFO") basis and consists primarily of soap dispensers and sprayers.

Property and equipment

Property and equipment are stated at fair value for assets acquired at the Transaction and stated at cost, less accumulated depreciation, for assets acquired in the ordinary course of business. Major expenditures and those that substantially increase useful lives are capitalized. Depreciation is determined using the straight-line method over the estimated useful lives of the assets. Leasehold improvements are amortized over the lease term or the estimated useful life of the related asset, whichever is shorter. When property and equipment is retired or sold, the cost and related accumulated depreciation or amortization are removed from the consolidated balance sheet, and the resulting gain or loss is included in the consolidated statements of operations. Ordinary maintenance and repair costs are expensed as incurred.

Goodwill

Goodwill represents the excess of the purchase price over the fair value of net assets (including separately recognized intangible assets) acquired. In 2022, the Company adopted ASU No. 2014-18, *Business Combinations* (Topic 805): Accounting for Identifiable Intangible Assets in a Business Combination. This accounting standard allows that under business combinations: (1) customer-related intangible assets that are not capable of being sold or licensed separately from the rest of the business and (2) noncompetition agreements be subsumed into goodwill. Adoption of this alternative also required the Company to adopt ASU No. 2014-02, *Intangibles — Goodwill and Other* (Topic 350): Accounting for Goodwill, which requires that goodwill be amortized on a straight-line basis over 10 years (or less, if another useful life is more appropriate). In 2022, the Company adopted ASU No. 2021-03, *Intangibles – Goodwill and Other* (Topic 350): Accounting Alternative for Evaluating Triggering Events, which permits the Company to amortize goodwill over 10 years and test for impairment at the reporting-unit level at the end of the reporting period when a triggering event occurs indicating the fair value of the reporting unit may be below its carrying amount. No impairment indicators exist as of January 1, 2023.

As discussed above, on March 29, 2022, Enviro-Master International Franchise, LLC was acquired through an equity purchase agreement. The Company assigned all amounts paid in excess of identifiable assets to goodwill. GAAP requires that separately identifiable intangible assets acquired in a business combination be recorded at fair value. Management has not identified or established a value for the separately identifiable intangible assets acquired, which represents a departure from GAAP. If the Company had recorded separately identifiable intangible assets at fair value, as required, it would result in an increase to intangible assets and a decrease to goodwill and the related goodwill amortization expense. The impact on the consolidated financial statements has not been determined.

Impairment of long-lived assets

The Company reviews the carrying value of long-lived assets for impairment whenever triggering events or changes in circumstances indicate the carrying value of an asset may not be recoverable from the estimated future cash flow expected to result from its use and eventual disposition. In cases where the estimated undiscounted future cash flow is less than the carrying value of an asset, an impairment loss is recognized equal to the amount by which the carrying value exceeds the fair value of the asset. Based on managements' assessments, no impairment occurred during the Successor Period or Predecessor Period.

Debt issuance costs

Costs incurred related to the Company's debt are capitalized and amortized over the loan periods using the straightline method (which approximates the effective interest method). The debt issuance costs are shown as a direct deduction from long-term debt on the consolidated balance sheet. As of January 1, 2023, debt issuance costs, net of accumulated amortization was \$996,922.

Accounts payable – franchisees

The Company collects accounts receivable on behalf of its franchisees and remits the balance to the franchisees weekly after deducting franchise service fees, royalties, advertising, administrative, telemarketing fees and other amounts which may be due to the Company. Accounts payable – franchisees include these collections and all other amounts due to the franchisees.

Revenue recognition

The Company derives is revenues from franchise sales, product sales, and royalties and other franchise fees. Revenues are recognized when control of the promised goods or services is transferred to customers in an amount that reflects the consideration the Company expects to be entitled to receive in exchange for those goods or services. The Company applies the five-step model under FASB ASC Topic 606, *Revenue from Contracts with Customers*, to determine when revenue is earned and recognized.

The Company has elected the practical expedient available to private companies under ASU 2021-02, *Franchisors*—*Revenue from Contracts with Customers,* to account for pre-opening activities as one distinct performance obligation for new franchise sales. The initial franchise fees are payable based on contract terms prior to the franchise opening. The transaction price is recognized as revenue on a straight-line basis over a 2-month period, based on timing of when services are satisfied for pre-opening activities. Franchise renewals and existing franchise sales are recognized as revenue upon execution of the franchise agreement.

The Company receives a significant portion of its revenue from sales-based royalties, admin fees, advertising fees and other franchise fees from franchisees. These fees are determined as a percentage of sales and are recognized in the period earned by the Company.

The Company receives a significant portion of its revenue from product sales to franchisees. Revenues and the related cost of revenue are recognized upon delivery of the products to franchisees.

Selling, general and administrative expenses

Selling, general and administrative expenses consist of costs associated with administrative and support functions related to the Company's existing business as well as growth and development activities. These costs primarily consist of consultants, credit card and bank fees, advertising, and professional expenses.

Franchise agreements

Franchise agreements are available for a territory over a specified period of time. The franchise agreement defines the territory the new franchise is able to serve. New franchisees are required to pay the Company an initial franchise fee plus a weekly royalty fee of 6%, administrative fee of 5%, and advertising fee of 2% of the franchisee's weekly gross receipts for the duration of the franchise agreement.

Advertising costs

Advertising costs are expensed as incurred. Advertising expense was \$488,032 and \$188,722 for the Successor Period and Predecessor Period, respectively.

Income taxes

The Company is a limited liability company treated as a partnership for federal and state income tax purposes with all income tax liabilities and/or benefits of the Company being passed through to the members. As such, no recognition of federal or state income taxes for the Company have been provided for in the accompanying consolidated financial statements.

Leases

In February 2016, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update ("ASU") 2016-02, *Leases* (Topic 842). This guidance applies to all entities that enter into leases and is intended to increase transparency and comparability among organizations by recognizing lease assets and lease liabilities on the consolidated balance sheet and disclosing key information about leasing arrangements. The new guidance requires lessees to recognize the assets and liabilities on the consolidated balance sheet for the rights and obligations created by leases with lease terms of more than twelve months, amends various other aspects of accounting for leases by lessees and lessors, and requires enhanced disclosures. Leases will be classified as finance or operating, with the classification affecting the pattern and classification of expense recognition within the consolidated statements of operations. See below for lease accounting policies and Note 8.

The Company transitioned to FASB Accounting Standards Codification ("ASC") Topic 842, *Leases* ("ASC 842"), from ASC Topic 840, *Leases* (the "Previous Standard") on January 3, 2022 using the effective date as the date of initial application. The consolidated financial statements reflect the application of ASC 842 guidance beginning in 2022. The Company's transition to ASC 842 represents a change in accounting principle.

The Company has elected to exclude short-term leases, or leases with a term of twelve months, from the balance sheet.

In addition, the Company has elected the transition package of practical expedients permitted within the new standard. This practical expedient permits the Company to carryforward the historical lease classification and not to reassess initial direct costs for any existing leases.

The Company has elected the practical expedient that allows lessees to avoid separating lease and non-lease components within a contract if certain criteria are met. The lessee's practical expedient election is limited to circumstances in which (i) the timing and pattern of revenue recognition are the same for the non-lease component and the related lease component and (ii) the combined single lease component would be classified as an operating lease. This practical expedient allows the Company the ability to combine the lease and non-lease components if the underlying asset meets the two criteria above.

Lastly, the Company has elected the practical expedient to use the risk-free interest rate as the discount rate at the time of commencement or modification date in determining the present value of lease payments.

2. Business Combination

As discussed in Note 1, the Company purchased all the membership interests in the Operating Company in conjunction with the Transaction on March 29, 2022. The total purchase price was \$90,938,951, which was made up of the following:

Cash Rollover equity	\$	73,938,951 17,000,000
Total consideration transferred	<u>\$</u>	90,938,951

The purchase price was financed through a combination of debt borrowings of \$40,000,000 and equity contributions of \$54,310,407.

The Transaction was recorded in accordance with ASC 805: *Business Combinations*. Management has elected to apply pushdown accounting to the Company's consolidated financial statements. Accordingly, the Company has recorded all assets acquired and liabilities assumed at the acquisition date fair values with any excess recognized as goodwill. The purchase price allocation process was complete as of January 1, 2023.

The consideration paid was allocated to the assets acquired and liabilities assumed according to their estimated fair values at the time of acquisition as follows:

Assets acquired:	
Cash	\$ 2,504,819
Accounts receivable	221,609
Prepaid expenses and other current assets	125,402
Note receivable	124,459
Property, plant, and equipment	540,439
Right of use asset	 247,035
Gross assets acquired, excluding goodwill	 3,763,763
Liabilities assumed:	
Accounts payable and accrued expenses	1,317,821
Other current liabilities	552,622
Lease liability – finance lease	35,324
Lease liability – operating lease	 275,035
Total liabilities assumed	 2,180,802

Net assets acquired, excluding goodwill	1,582,961
Goodwill	<u>89,355,990</u>
Total purchase price	<u>\$ 90,938,951</u>

Goodwill, which represents the excess of the consideration paid over the fair value of recognized net assets acquired, includes, but is not limited to, the value of the workforce in place, the ability to generate profits and cash flows, an established going concern, customer-related intangibles, and other unidentifiable intangible assets.

3. Goodwill

Goodwill, as a result of the acquisition disclosed in Note 2, consists of the following as of January 1, 2023:

	Amortization Period		January 1, 2023
Goodwill Accumulated amortization	10 Years	\$	89,355,990 <u>(6,701,216)</u>
Goodwill, net		<u>\$</u>	82,654,774

Amortization expense for goodwill was \$6,701,216 for the Successor Period. Estimated amortization expense of goodwill for each of the five years subsequent to January 1, 2023 and thereafter are as follows:

2023 2024 2025 2026 2027 Thereafter	\$	8,935,700 8,935,700 8,935,700 8,935,700 8,935,700 37,976,274
	<u>\$</u>	82,654,774

4. Notes receivable

Note receivables represents company-offered financing of initial franchise fees and working capital loans. The term of the note is for three years, with non-interest bearing. The note is secured with the franchisee's business as pledged collateral. The notes receivable balance as of January 1, 2023, in the amount of \$7,789 is classified as a current asset on the accompanying consolidated balance sheet.

5. Property and equipment, net

Property and equipment consist of the following as of January 1, 2023:

	Estimated Useful Life (in years)	Ja	nuary 1, 2023
Software and website design Computer equipment Office furniture and fixtures Leasehold improvements Construction in progress	3 years 5 years 4 years 11 years	\$	330,096 164,608 56,373 19,555 50,000
Total property and equipment Accumulated depreciation			620,632 <u>(234,075</u>)
Property and equipment, net		<u>\$</u>	386,557

Depreciation expense was \$240,531 and \$55,152 for the Successor Period and Predecessor Period, respectively.

6. Accrued liabilities

Accrued expenses and other current liabilities consist of the following as of January 1, 2023:

Accrued commissions	\$ 131,731
Accrued credit card liability	21,282
Accrued payroll and payroll taxes	222,085
Other accrued expenses	452,354
	<u>\$ 827,452</u>

7. Long-term debt and revolving line of credit

Long-term debt consists of the following as of January 1, 2023:

Revolving line of credit	\$ 250,000
Term loan	<u>39,700,000</u>
Total current and long-term debt	39,950,000
Current portion	(400,000)
Debt issuance costs, net	<u>(996,922)</u>
Total long-term debt, net of current portion and debt issuance costs	<u>\$ 38,553,078</u>

Term loan and revolving line of credit

As part of the "Acquisition," the Company entered into a credit facility (the "Credit Agreement") which provides for (i) a \$40,000,000 term loan credit facility (the "Term Loan") and (ii) availability of up to \$3,000,000 of a revolving credit facility (the "Revolver"). Borrowings under the Credit Agreement are secured by substantially all of the Company's assets as well as Members' equity. The Term Loan requires calendar quarterly payments of principal and interest beginning June 30, 2022, with a balloon payment upon maturity on March 29, 2027. The Revolver and any accrued interest is due upon maturity on March 29, 2027.

The Term Loan and Revolver accrue interest at an adjusted term SOFR for the interest therefor plus the applicable margin of 6.5% per annum, as defined in the Credit Agreement. As of January 1, 2023, the interest rates associated with the Term Loan and Revolver were approximately 10.76% and 10.92%, respectively.

The Term Loan and Revolver, described above, are subject to certain financial covenants and restrictions specified in the Credit Agreement. In addition, the Company is subject to excess cash flow requirements, which require accelerated debt principal payments under certain conditions. As of January 1, 2023, the Company was in compliance with its financial covenants.

Principal maturities of long-term debt consist of the following as of January 1, 2023:

2023	\$ 400,000
2024	700,000
2025	800,000
2026	800,000
2027	800,000
Thereafter	 36,450,000

8. Commitments and contingencies

Operating lease obligations

The Company leases office space accounted for as an operating lease. Leases with an initial term of 12 months or less are not recorded on the consolidated balance sheets; the Company recognizes lease expense for these leases on a straight-line basis over the lease term. For leases commencing after January 3, 2022, the Company accounts for fixed lease and non-lease components together as a single, combined lease component. Variable lease costs, which may include common area maintenance, insurance, and taxes are not included in the lease liability and are expensed in the period incurred.

\$

39,950,000

The office lease expires on April 29, 2023 and does not include a renewal option. The depreciable life of assets and leasehold improvements are limited by the expected lease term.

The Company evaluates leases at contract inception to determine whether the Company has the right to control use of the identified asset for a period of time in exchange for consideration. If it is determined the Company has the right to obtain substantially all of the economic benefit from use of the identified asset and the right to direct the use of the identified asset, the Company recognizes a right-of-use ("ROU") asset and lease liability. Also, at contract inception, the Company evaluates leases to estimate their expected term which includes renewal options that we are reasonably assured that we will exercise, and the classification of the lease as either an operating lease or a finance lease. Operating lease liabilities represent the present value of lease payments not yet paid. Operating lease liabilities adjusted for prepayments or accrued lease payments, initial direct costs and lease incentives. The Company assesses the impairment of the ROU asset at the asset group level whenever events or changes in circumstances indicate that the carrying value of the asset may not be recoverable.

Changes in these assumptions and management judgments may produce materially different amounts in the recognition of the right-of-use assets and lease liabilities. Additionally, any loss resulting from an impairment of the right-of-use assets is recognized by a charge to income, which could be material.

The Company's lease contains rent escalations over the lease term. The Company recognizes expense for these leases on a straight-line basis over the lease term.

The Company's lease agreement does not contain any material residual value guarantees or material restrictive covenants.

Consolidated balance sheet information related to right-of-use assets and lease liabilities are as follows:

Classifications	Ja 	nuary 1, 2023
Right-of-use asset, net	<u>\$</u>	88,597
Current portion of lease liabilities	\$	99,584
Lease liabilities, net of current portion		
	\$	<u>99,584</u>
	Right-of-use asset, net Current portion of lease liabilities	Classifications Right-of-use asset, net \$ Current portion of lease liabilities \$

The following summarizes the weighted average remaining lease term and discount rate as of January 1, 2023.

Remaining lease term - operating leases 4 months

Discount rate - operating leases 0.59%

For the Successor period ended January 1, 2023, the components of lease cost, which are included within selling, general, and administrative expenses on the consolidated statements of operations, were as follows:

Operating lease cost Variable lease cost	\$ 159,728 -
Total lease cost	\$ 159,728

For the Predecessor period ended March 28, 2022, the components of lease cost, which are included within selling, general, and administrative expenses on the consolidated statements of operations, were as follows:

Operating lease cost Variable lease cost	\$	69,272 -
Total lease cost	<u>\$</u>	69,272
The following summarizes supplemental cash flow information for the Successor Period:		
Cash paid for amounts included in the measurement of lease liabilities: Operating cash flows from operating leases	<u>\$</u>	176,331
The following summarizes supplemental cash flow information for the Predecessor Period:		
Cash paid for amounts included in the measurement of lease liabilities: Operating cash flows from operating leases	<u>\$</u>	57,636

The Company's costs related to short-term leases, those with a duration between one and twelve months, were immaterial.

As of January 1, 2023, maturities of lease liabilities were as follows:

2023 Imputed interest	\$ 99,732 <u>(148</u>)
Present value of lease liabilities	\$ 99,584

Finance lease obligations

The Company leases equipment accounted for as a finance lease under a non-cancellable finance leases, included in property and equipment, net, as of January 1, 2023 are as follows:

Computer equipment Accumulated Depreciation	\$ 35,141 <u>(5,220</u>)
	\$ <u>29,921</u>

The Company's estimated future minimum lease payments required under the terms of the leases is as follows:

2023 2024 2025 2026	\$	7,962 8,686 8,686 7,239
Total minimum lease payments Amounts representing interest		32,573 (2,652)
Present value of net minimum lease payments Current portion of net minimum lease payments		29,921 (7,493)
Finance lease obligations, net of current portion	<u>\$</u>	22,428

9. Related party activities

The Company entered into a lease agreement with a related party through common ownership prior to the transaction on March 29, 2022. The lease agreement is for the use of a boat and the Company is required to cover all repairs and maintenance expenses. During the Predecessor Period, total lease payments were \$16,030 and repairs and maintenance expenses were \$39,042. The lease was terminated in March 2022.

The Company incurred a management fee from a related party. The total amount incurred was approximately \$203,000 and \$0 for the Successor Period and Predecessor Period, respectively, which is recorded in Selling, general, and administrative expenses on the accompanying consolidated statement of operations.

10. Retirement plan

January 1, 2016, the Company established a SIMPLE IRA plan. All full-time employees are eligible to participate in the plan after ninety days of employment, as well as an annual election period from November 2 through December 31. During the Successor Period and Predecessor Period, the Company matched 100 percent of up to three percent of an eligible employee's contributions, totaling \$71,353 and \$20,140, respectively.

11. Subsequent Events

The Company evaluated all events or transactions that occurred after January 1, 2023 through March 27, 2023, the date that the consolidated financial statements were available to be issued. During this period, there have been no material events that would require an adjustment or disclosure in the consolidated financial statements other than as discussed below.

On January 18, 2023, the Company acquired Enviro-Master Charlotte, LLC for approximately \$6,700,000. Enviro-Master Charlotte, LLC has franchise rights for the Charlotte, NC and Myrtle Beach, SC territories. The initial accounting for the acquisitions has not been finalized as of the date that the consolidated financial statements were available to be issued. In connection with the acquisitions, the Company signed a note agreement with the seller for \$1,500,000 to fund the acquisition maturing in January 2028, with six percent interest per annum. Additionally, rollover equity for approximately \$650,000 was included in the purchase price.



ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC

AND SUBSIDIARIES

CONSOLIDATED FINANCIAL STATEMENTS

DECEMBER 31, 2021 AND 2020

ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC AND SUBSIDIARIES INDEX TO CONSOLIDATED FINANCIAL STATEMENTS DECEMBER 31, 2021 AND 2020

Page

INDEX TO CONSOLIDATED FINANCIAL STATEMENTS	1
INDEPENDENT AUDITORS' REPORT	2 - 3
CONSOLIDATED FINANCIAL STATEMENTS:	
Consolidated Balance Sheets	4 - 5
Consolidated Statements of Income	6
Consolidated Statements of Equity	7
Consolidated Statements of Cash Flows	8
Notes to Consolidated Financial Statements	9 -19
Supplemental Schedule I – Consolidating Balance Sheets	20 -23
Supplemental Schedule II – Consolidating Statements of Income	24 -25
Supplemental Schedule III – Consolidating Statements of Equity	26
Supplemental Schedule IV – Consolidating Statements of Cash Flows	27-28
Supplemental Schedule V – Consolidating Statements of General and	
Administrative Expenses	29 - 30



4600 Park Road, Suite 112 Charlotte, NC 28209 704 372-1167 704 377-3259 fax scharfpera.com

Board of Managers and Members Enviro-Master International Franchise, LLC and Subsidiaries Charlotte, North Carolina

INDEPENDENT AUDITORS' REPORT

We have audited the accompanying consolidated balance sheets of Enviro-Master International Franchise, LLC and its subsidiaries as of December 31, 2021 and 2020, which comprise the consolidated statements of income, consolidated statements of equity, consolidated statements of cash flows and the related notes to the consolidated financial statements for the years ended December 31, 2021 and 2020.

Management's Responsibility for the Financial Statements

Management is responsible for the preparation and fair presentation of these financial statements in accordance with accounting principles generally accepted in the United States of America; this includes the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of financial statements that are free from material misstatement, whether due to fraud or error.

Auditor's Responsibility

Our responsibility is to express an opinion on these financial statements based on our audits. We conducted our audits in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free from material misstatement.

An audit involves performing procedures to obtain audit evidence about the amounts and disclosures in the financial statements. The procedures selected depend on the auditor's judgment, including the assessment of the risks of material misstatement of the financial statements, whether due to fraud or error. In making those risk assessments, the auditor considers internal control relevant to the entity's preparation and fair presentation of the financial statements in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the entity's internal control. Accordingly, we express no such opinion. An audit also includes evaluating the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluating the overall presentation of the financial statements.

We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Opinion

In our opinion, the financial statements referred to above present fairly, in all material respects, the consolidated financial position of Enviro-Master International Franchise, LLC and its subsidiaries as of December 31, 2021 and 2020, and the results of its operations and cash flows for the years then ended, in accordance with accounting principles generally accepted in the United States of America.

Supplemental Information

Our audit was conducted for the purpose of forming an opinion on the financial statements as a whole. The consolidated schedules of consolidated balance sheets, consolidated statements of income, cash flows, and general and administrative expenses are presented for the purposes of additional analysis and are not a required part of the financial statements. Such information is the responsibility of management and was derived from and relates directly to the underlying accounting and other records used to prepare the financial statements. The information has been subjected to the auditing procedures applied in the audit of the financial statements and certain additional procedures, including comparing and reconciling such information directly to the underlying accounting and other records used to prepare the financial statements or to the financial statements themselves, and other additional procedures in accordance with auditing standards generally accepted in the United States of America. In our opinion, the information is fairly stated in all material respects in relation to the financial statements as a whole.

Schaf Pur. C. Pucc

February 24, 2022 Charlotte, North Carolina

ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC AND SUBSIDIARIES <u>CONSOLIDATED BALANCE SHEETS</u> <u>DECEMBER 31, 2021 AND 2020</u>

	2021	2020	
ASSETS			
CURRENT ASSETS			
Cash	\$ 3,877,317	\$ 4,145,114	
Accounts receivable - franchisees, net	326,015	402,942	
Inventory	51,295	119,499	
Current portion of notes receivable	23,490	33,752	
Prepaid expenses and other current assets	78,530	99,207	
Total current assets	4,356,647	4,800,514	
PROPERTY AND EQUIPMENT - net of accumulated depreciation	508,553	427,887	
OTHER ASSETS Notes receivable - net of current portion Intangible assets - net of amortization	7,789 4,308	80,839 75,999	
Other assets	21,280	21,230	
	33,377	178,068	
	\$ 4,898,577	\$ 5,406,469	

	2021		2020	
LIABILITIES AND MEMBERS' EQUITY				
CURRENT LIABILITIES				
Accounts payable - trade Accounts payable - franchisees Deferred revenue Accrued expenses	\$	821,847 951,102 263,149 472,688	\$	933,173 1,929,100 147,101 509,564
Current portion of capital lease obligations		78,218		40,880
Total current liabilities		2,587,004		3,559,818
NON-CURRENT DEBT				
Capital lease obligations - net of current portion		116,722		29,488
Other liabilities		31,983		46,165
		148,705		75,653
MEMBERS' EQUITY				
Enviro-Master International Franchise, LLC				
members' equity		2,162,868		1,637,372
Non-controlling interest		-		133,626
		2,162,868		1,770,998
	\$	4,898,577	\$	5,406,469

ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF INCOME YEARS ENDED DECEMBER 31, 2021 AND 2020

	2021	2020
REVENUE		
Franchise sales, net	\$ 598,567	\$ 1,211,125
Royalties and franchise fees	13,348,482	12,606,533
Product sales, net	9,719,291	10,358,485
	23,666,340	24,176,143
COST OF REVENUE		
Products	8,364,323	9,019,704
Labor	3,210,409	2,872,048
Broker commissions on franchise sales	138,044	245,178
	11,712,776	12,136,930
GROSS PROFIT	11,953,564	12,039,213
GENERAL AND ADMINISTRATIVE EXPENSES	7,423,783	6,667,018
	4,529,781	5,372,195
OTHER INCOME (EXPENSE)		
Interest income	4,319	10,835
Other income	9,834	-
Loss on sale of property and equipment	(5,739)	(2,146)
Settlement (expense) income	(56,200)	82,062
Loss on foreign currency exchange	(8,368)	(381)
Interest expense	(8,503)	(25,261)
	(64,657)	65,109
CONSOLIDATED NET INCOME	4,465,124	5,437,304
Less: Net loss attributable to non-controlling interest	(14,470)	(19,191)
NET INCOME ATTRIBUTABLE TO ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC	\$ 4,479,594	\$ 5,456,495
-		

ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF EQUITY YEARS ENDED DECEMBER 31, 2021 AND 2020

	TOTAL	Enviro-Master International <u>Franchise, LLC</u>	Non- Controlling <u>Interest</u>
As of the year ended December 31, 2019	\$ 465,981	\$ 313,164	\$ 152,817
Net income (loss)	5,437,304	5,456,495	(19,191)
Distributions to members	 (4,132,287)	(4,132,287)	
As of the year ended December 31, 2020	1,770,998	1,637,372	133,626
Net income (loss)	4,465,124	4,479,594	(14,470)
Distributions to members	 (4,073,254)	(3,954,098)	(119,156)
As of the year ended December 31, 2021	\$ 2,162,868	\$ 2,162,868	\$

See Notes to Consolidated Financial Statements

ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC AND SUBSIDIARIES <u>CONSOLIDATED STATEMENTS OF CASH FLOWS</u> <u>YEARS ENDED DECEMBER 31, 2021 AND 2020</u>

	2021	2020
Net income	\$ 4,465,124	\$ 5,437,304
Adjustments to reconcile net income		
to net cash provided by operating activities:		
Depreciation and amortization	275,308	262,644
Loss on disposal of property and equipment	5,739	2,146
Bad debt recovery	(12,088)	-
Changes in operating assets and liabilities:		
Decrease (increase) in accounts		
receivable - franchisees	89,015	(160,763)
Decrease in prepaid expenses and		
other current assets	20,677	89,230
Decrease (increase) in inventory	68,204	(119,499)
(Increase) decrease in other assets	(50)	4,140
(Decrease) increase in accounts payable - trade	(111,326)	180,285
(Decrease) increase in accounts		
payable - franchisees	(977,998)	949,399
Increase in deferred revenue	101,866	21,276
(Decrease) increase in accrued expenses	(36,876)	165,130
Total adjustments	(577,529)	1,393,988
Cash provided by operating activities	3,887,595	6,831,292
Cash flows from investing activities:		
Purchases of property and equipment	(100,766)	(157,419)
Net proceeds from notes receivable	83,312	293,126
Cash (used in) provided by investing activities	(17,454)	135,707
Cash flows from financing activities:		
Principal payments on notes payable	-	(12,284)
Principal payments on notes payable - members	-	(160,565)
Principal payments on capital lease obligations	(64,684)	(44,416)
Distributions to members	(4,073,254)	(4,132,287)
Cash used in financing activities	(4,137,938)	(4,349,552)
NET (DECREASE) INCREASE IN CASH	(267,797)	2,617,447
CASH AT BEGINNING OF YEAR	4,145,114	1,527,667
CASH AT END OF YEAR	\$ 3,877,317	\$ 4,145,114

ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS YEARS ENDED DECEMBER 31, 2021 AND 2020

Note 1 – Summary of Significant Accounting Policies:

The Company:

Enviro-Master International Franchise, LLC ("EMIF") was organized in North Carolina on December 30, 2013. Originally founded in 2009 as Enviro-Master Services, the Company provides hygiene service franchises to franchisees throughout the United States and Canada. The Company awards exclusive franchises via its franchise agreement which grants franchisees the right to use the Company's marketing, business techniques and systems under the "Enviro-Master" trademark. The Company provides franchisees with initial training, ongoing training and support, business products at discounted prices, as well as revenue collection services. The Company remits the collections to franchisees after deducting the franchise service fees, royalties, advertising, administrative, telemarketing fees and other amounts which may be due.

EM Capital Services, LLC ("EMCS") was formed in North Carolina on September 7, 2016 to hold and sell the rights to use customer lists that were acquired as part of an asset purchase from a competitor in the restroom cleaning business. EMIF owns 80 percent of membership interests in EMCS. As of December 31, 2021, EMCS ceased operations as customer rights were fully amortized, the Company had completed collection activities for all notes receivable related to the sales of customer lists and there were no further ongoing obligations.

Principles of Combination:

Enviro-Master International Franchise, LLC and EM Capital Services, LLC (collectively the "Company") have common ownership and management. The financial statements of these companies have been consolidated in accordance with Financial Accounting Standards Board ("FASB") Accounting Standards Codification ("ASC") Topic 810 "Consolidation". All intercompany receivables and payables have been eliminated in these consolidated financial statements.

Cash and Cash Equivalents:

The Company considers all highly liquid investments having an original maturity of three months or less to be cash equivalents. Amounts invested may exceed federally insured limits at any given time. As of December 31, 2021 and 2020, the Company did not have any investments with maturities greater than three months.

Accounts Receivable and Revenue Recognition:

The Company adopted FASB ASC 606 and Accounting Standards Update 2021-02 "Franchisors-Revenue from Contracts with Customers (Subtopic 952-606)" effective for the year ended December 31, 2020. FASB ASC 606 prescribes a five-step model that includes: (1) identify the contract; (2) identify the performance obligations; (3) determine the transaction price; (4) allocate the transaction price to the performance obligations; and (5) recognize revenue as performance obligations are satisfied. The practical expedients under ASU 2021-02 aim to simplify the application about identifying performance obligations by permitting franchisors that are not public business entities to account for pre-opening services provided to a franchisee as distinct from the franchise license.

EMIF is a franchisor and derives revenue from three sources. The Company adopted the practical expedient to enable recording of revenue on initial franchise fees as certain pre-opening performance obligations as set forth in the franchise agreement are completed. Revenue from product sales is recognized at the time the franchise takes possession of the products. Royalties and franchise fee revenue, which include royalties, marketing and administrative charges that are primarily based on a percentage of franchisee revenue, are accrued as the underlying franchisee revenue is earned. The revenue of EMIF represents 100 percent of the consolidated total revenue from continuing operations for 2021 and 2020.

Accounts receivable are carried at face value less an allowance for doubtful accounts. On a periodic basis, management evaluates accounts receivable balances and establishes allowances for doubtful accounts, based on history of past write-offs and collections, as well as customer specific circumstances and credit conditions. At December 31, 2021 and 2020, the allowance for doubtful accounts from franchisees was \$0.

EMCS is a financing company whose clients are the franchisees of EMIF. Revenue generated from the sale of rights to use certain assets of the Company are recognized upon receipt of the applicable fee or an executed note receivable. No revenue was generated by EMCS for the years ended December 31, 2021 and 2020.

Notes Receivable:

Notes receivable are carried at face value less an allowance for doubtful accounts. Interest accrues on notes receivable as stated in each note agreement. On a periodic basis, management evaluates notes receivable balances and establishes allowances for doubtful accounts, based on history of past write-offs and collections as well as customer specific circumstances and credit conditions.

Property and Equipment:

Property and equipment are recorded at cost. Depreciation is recorded using the straight-line method over the estimated useful lives of three years for computer software and website; five years for computers; five to seven years for furniture and fixtures; seven years for office equipment; and fifteen years for leasehold improvements. Repairs and maintenance costs are expensed as incurred. Any gain or loss on disposition of property and equipment is reported in the year of disposition.

Intangible assets:

Intangible assets primarily include customer lists, all of which are amortized on a straightline basis over their estimated useful lives of five years. The Company evaluates the recoverability of identifiable intangible assets whenever events or changes in circumstances indicate that an intangible asset's carrying amount may not be recoverable. No impairment was recorded for the years ended December 31, 2021 and 2020.

The Company has adopted the guidance provided in FASB ASC Topic 350 "Accounting for Goodwill." In accordance with this accounting alternative, the Company amortizes goodwill on a straight-line basis over ten years and evaluates goodwill for impairment at the entity level when a triggering event occurs. No impairment has been recorded through December 31, 2021.

Inventory:

Inventory is stated at cost, determined on a first-in, first out ("FIFO") basis and consists primarily of soap dispensers and sprayers.

Accounts Payable - Franchisees:

The Company collects accounts receivable on behalf of its franchisees and remits the balance to the franchisees weekly after deducting franchise service fees, royalties, advertising, administrative, telemarketing fees and other amounts which may be due to the Company. Accounts Payable – Franchisees includes these collections and all other amounts due to the franchisees.

Deferred Revenue:

The Company recognizes revenue at the time the franchisee takes possession of products purchased. When the Company receives payment from franchisees for products ordered before the franchisee has taken possession of the products, the amount received is recorded as deferred revenue until the sale is complete. At December 31, 2021 and 2020, deferred revenue from products was \$231,393 and \$123,101, respectively. The cost of goods sold associated with these transactions are approximately \$196,545 and \$98,570, as of December 31, 2021 and 2020, respectively. These costs will be recognized when the franchisee has taken possession of the products, indicating the sale transaction is completed, and the revenue is recognized as earned. Additionally, \$31,756 and \$24,000 of deferred revenue related to new franchise training and sales management programs has been recorded as of December 31, 2021 and 2020, respectively.

Freight Charges:

The Company passes freight charges through to its franchisees.

Advertising:

The Company expenses advertising costs as incurred. Advertising costs related to continuing operations for the years ended December 31, 2021 and 2020, totaled \$363,098 and \$431,213, respectively.

Income Taxes:

The Company is a limited liability company that has elected to be taxed under the partnership provisions of the Internal Revenue Code. Under these provisions, the members are taxed on their share of the Company's taxable income. The Company bears no liability or expense for income taxes, and none is reflected in these financial statements. Similar provisions apply for state income taxes.

The Company accounts for income taxes in accordance with FASB ASC 740-10 "Income Taxes". This standard clarifies the accounting for income taxes, by prescribing a minimum recognition threshold a tax position is required to meet before being recognized in the balance sheet. It also provides guidance on derecognition, measurement and classification of amounts related to uncertain tax positions, accounting for and disclosure of interest and penalties, accounting in interim period disclosures and transition relating to the adoption of new accounting standards. Under FASB ASC 740-10, the recognition for uncertain tax positions should be based on a more-likely-than-not threshold that the tax position will be sustained upon audit. The tax position is measured as the largest amount of benefit that has a greater than fifty percent probability of being realized upon settlement. The Company believes it is no longer subject to U.S. federal or state income tax examinations by tax authorities for the years before 2018.

Fair Value of Financial Instruments:

The Company's financial instruments include cash, accounts receivable, inventory, accounts payable, accrued liabilities and notes payable. The carrying amounts of these instruments approximate fair value due to their short maturities and variable rates of interest. The carrying amounts of long-term debt approximate their fair values based on current rates available for similar types of instruments.

As of December 31, 2021 and 2020, the Company did not have any outstanding financial derivative instruments.

Use of Accounting Estimates:

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of certain assets and liabilities and disclosures. Accordingly, the actual amounts could differ from those estimates. Any adjustments applied to estimated amounts are recognized in the year in which such adjustments are determined.

Note 2 – Notes Receivable:

Notes receivable represents company-offered financing of initial franchise fees and working capital loans. The terms of the notes range from six months to five years, with interest up to twelve percent. The notes are generally secured with the franchisee's business as pledged collateral.

The following represents notes receivable as of December 31, 2021 and 2020:

	2021		2020	
Notes receivable	\$	31,279	\$	114,591
Less: allowance for doubtful accounts		_		_
		31,279		114,591
Less: current portion of notes receivable		(23,490)		(33,752)
	\$	7,789	\$	80,839

Note 3 – Property and Equipment:

Property and equipment at December 31, 2021 and 2020, is summarized as follows:

	2021		2020	
Computer equipment and software	\$	627,104	\$	456,709
Website development in process		-		1,500
Leasehold improvements		26,822		26,822
Office furniture and fixtures		251,849		238,024
Website design		197,521		197,521
		1,103,296		920,576
Less: accumulated depreciation		(594,743)		(492,689)
	\$	508,553	\$	427,887

Depreciation for the years ended December 31, 2021 and 2020, was \$203,616 and \$167,613, respectively.

Note 4 – Intangible Assets:

Customer Lists:

The customer lists are amortized on a straight-line basis over their estimated useful lives of five years. Customer lists are evaluated for impairment when a triggering event occurs. At December 31, 2021, the customer lists were fully amortized. For the years ended December 31, 2021 and 2020, amortization expense was \$71,691 and \$95,032, respectively. No impairment losses were recorded during the years ended December 31, 2021 or 2020.

Trademarks:

The Company had \$4,308 of indefinite lived intangible assets related to trademarks as of December 31, 2021 and 2020. No amortization or impairment losses have been recorded as of December 31, 2021.

Note 5 – Accrued Expenses:

Accrued expenses at December 31, 2021 and 2020, consisted of the following:

	2021		2020	
Accrued commissions	\$	141,585	\$	189,461
Accrued credit card liability		23,477		11,612
Accrued payroll and payroll taxes		212,327		219,752
Other accrued expenses		95,299		88,739
	\$	472,688	\$	509,564

<u>Note 6 – Line of Credit:</u>

During December 2017, the Company signed a line of credit note with a company related to one of EMIF's members. The credit line is available to the Company in the event that the Company would be unable to meet its obligations to its franchisees as the result of a cyber-attack that compromises the Company's bank account and for which the Company would make a claim with its insurance carrier but would not have received proceeds from insurance prior to requiring funds. The amount available to the Company is \$750,000. For any advances received on the line of credit, interest is payable monthly at the Prime Interest Rate, as reported by The Wall Street Journal, on the date of any advance received and is adjusted on the first day of the subsequent month following any changes to the Prime Interest Rate. All unpaid principal and interest are due one hundred eighty days after receipt of the advance. The credit line was renewed and is available to the Company until December 29, 2022, and is renewable annually thereafter. As of December 31, 2021, the Company has not taken any advances on the line of credit.

Note 7 – Long-Term Debt:

During December 2018, EMIF assumed the debt of the abandoned Seattle franchise from EMCS. Loan payments vary based on account receivable collections. The loan was payable in 260 weekly installments of \$141, including principal and interest at 12.00 percent per year. During 2020, the remaining balance of the note was paid off in full.

Note 8 - Notes Payable - Members:

The Company borrowed money from its members in September 2016. During the year ended December 31, 2020, these notes were paid in full.

Note 9 – Capital Lease Obligations:

At December 31, 2021 and 2020, capital lease obligations are as follows:

	2021		2020	
Capital lease dated December 2017, due in 60 monthly installments of \$1,558, including interest at 9.05 percent, secured by office furniture.	\$	17,814	\$	34,093
Capital lease dated December 2017, due in 60 monthly installments of \$537, including interest at 9.05 percent, secured by office furniture.		6,629		12,199
Capital lease dated July 2017, due in 63 monthly installments of \$120, including interest at 10.07 percent, secured by office equipment.		_		2,298
Capital lease dated June 2021, due in 60 monthly installments of \$724, including interest at 4.50 percent, secured by office equipment.		37,085		-
Capital lease dated July 2021, due in 36 monthly installments of \$4,582, including interest at 7.00 percent, secured by software.		133,412		_
Capital lease dated September 2017, due in 60 monthly installments of \$350, including interest at 10.07 percent, secured by office equipment.		_		7,583
Capital lease dated July 2018, due in 36 monthly installments of \$1,834, including interest at 12.00				
percent, secured by software.		-		14,195
		194,940		70,368
Less: current portion of capital lease obligations		(78,218)		(40,880)
	\$	116,722	\$	29,488

Note 9 – Capital Lease Obligations (continued):

At December 31, 2021, capital lease obligations due in aggregate annual installments are as follows:

Year ending December 31,	
2022	\$ 78,218
2023	58,523
2024	43,523
2025	8,166
2026	6,510
Thereafter	
	\$ 194,940

As of December 31, 2021 and 2020, fixed assets capitalized related to the above leases were \$289,786 and \$180,672, respectively. For the years ended December 31, 2021 and 2020, the Company recorded depreciation related to these assets of \$55,669 and \$38,932, respectively.

<u>Note 10 – Franchise Sales:</u>

Upon the execution of a franchise agreement, the franchisee will pay the Company an initial franchise fee. This initial franchise fee ranges up to \$90,000 depending on the population in the franchisee's territory. Revenue from franchise sales is recognized when the franchise agreement is signed and either a non-refundable initial fee has been received or a note receivable from the franchisee has been executed and the Company has completed certain obligations as set out in the franchise agreement.

The Company has experienced substantial growth, with increases in the number of franchises sold each year. However, as a territory is assigned to each franchise sold, the Company may reach the point where existing markets become saturated and initial franchise sales revenue declines. Franchise sales after market saturation will come primarily from franchise resales and renewal fees for existing franchises.

The number of initial franchises sold during the year and in operation as of the years ended December 31, 2021 and 2020, are as follows:

	2021	2020
Sold	5	9
In operation at December 31,	88	80

<u>Note 11 – Royalty Income:</u>

During the initial term of the franchise, the franchisee will pay minimum annual royalty fees to EMIF as follows; year 1 is \$0, year 2 is \$12,000, year 3 is \$24,000, year 4 is \$36,000 and year 5 is \$48,000.

After the fifth year of operation, the franchisee will pay minimum annual royalty fees based on the size of the franchisee's territory. The franchisee's territory will be categorized according to the levels of population, based on the most recent information from the National Association of Counties, or a comparable source on the population of persons in the territory. For purposes of minimum annual royalty fees territories may be combined. After the tenth year the minimum annual royalty fee will be fixed in perpetuity.

		3	<u>lear 6</u>		<u>Year 7</u>		<u>Year 8</u>		<u>Year 9</u>	<u>Y</u>	<u>ear 10</u>
		M	inimum	1	<u>Minimum</u>]	<u>Minimum</u>	l	<u>linimum</u>	M	<u>inimum</u>
Level	Population	Roya	alty Fees	Roy	<u>alty Fees</u>	<u>Roy</u>	<u>alty Fees</u>	Roy	<u>alty Fees</u>	<u>Roya</u>	<u>lty Fees</u>
1	0 to 500,000	\$	48,000	\$	48,000	\$	48,000	\$	48,000	\$	48,000
2	500,001 to 1,000,000		60,000		72,000		72,000		72,000		72,000
3	1,000,001 to 1,500,000		60,000		72,000		84,000		84,000		84,000
4	1,500,001 to 2,000,000		60,000		72,000		84,000		96,000		96,000
5	2,000,001 to 2,500,000		60,000		72,000		84,000		96,000		108,000
б	2,500,001 to 3,000,000		60,000		72,000		84,000		96,000		108,000
7	3,000,001 and over		60,000		72,000		84,000		96,000		108,000

Note 12 - Commitments and Contingencies:

Operating Leases:

EMIF entered into a lease agreement dated November 2, 2017, for its headquarters. EMIF leases office equipment, office space and a boat under various operating leases that expire through 2024. Rent paid during the years ended December 31, 2021 and 2020, for these leases was \$330,424 and \$281,889, respectively. Future minimum lease payments for operating leases that had initial or remaining non-cancelable lease terms in excess of one year as of December 31, 2021, are as follows:

Year ending December 31, 2022 \$ 330,938 2023 195,905 2024 88,165 Thereafter \$ 615,008

Note 12 – Commitments and Contingencies (continued):

Related Party Operating Lease:

EMIF has entered into lease agreements with a related party through common ownership in which it rents a boat that the Company uses to entertain current and prospective franchisees and current and prospective customers. At December 31, 2021, the lease calls for the Company to make monthly payments of \$8,015 through November 2024. Total payments made on these leases were \$96,180 and \$38,260 for the years ended December 31, 2021 and 2020, respectively.

Note 13 – Retirement Plans:

On January 1, 2016, EMIF established a SIMPLE IRA plan. All full-time employees are eligible to participate in the plan after ninety days of employment, as well as an annual election period from November 2 through December 31. For the years ended December 31, 2021 and 2020, the Company matched 100 percent of up to three percent of an eligible employee's contributions, totaling \$88,898 and \$79,835, respectively.

Note 14 – Other Matters:

Concentrations:

The Company places its cash and cash equivalents on deposit with North Carolina financial institutions. The balances are insured by the Federal Deposit Insurance Corporation up to \$250,000. At December 31, 2021 and 2020, the Company's cash balances exceeded this limit by \$3,265,314 and \$3,523,176, respectively. From time to time during the course of the year, the Company's cash balances may exceed this insured limit. The Company periodically reviews the financial condition of the institutions and believes the risk of loss to be minimal.

Product purchases for the years ended December 31, 2021 and 2020, include amounts from three major suppliers that represent approximately 84 percent and 74 percent, respectively, of the Company's purchases. The Company believes no risk is present under this arrangement due to other suppliers being readily available.

Legal Matters:

The Company is subject to various claims and legal proceedings in the ordinary course of its business activities. Management believes that any liability that ultimately results from the resolution of these matters will not have a material effect on the financial condition or results of operations of the Company.

COVID-19:

The Company experienced no significant adverse effects resulting from the COVID-19 pandemic during the years ended December 31, 2021 and 2020.

Note 15 – Subsequent Events:

The Company evaluated all events and transactions through February 24, 2022, the date the financial statements were issued. During this period, there were no material recognizable or non-recognizable subsequent events which occurred prior to issuance of the financial statements.

ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC AND SUBSIDIARIES SUPPLEMENTAL SCHEDULE I – CONSOLIDATING BALANCE SHEETS DECEMBER 31, 2021

	Env	iro-Master						
	Int	ernational	EM Cap	pital				
	<u>Fran</u>	<u>chise, LLC</u>	<u>Service</u>	s, LLC	<u>Elimi</u>	<u>nations</u>	Co	nsolidated
ASSETS								
CURRENT ASSETS								
Cash		\$ 3,877,317	\$	-	\$	-	\$	3,877,317
Accounts receivable - franchisees		326,015	·	-	·	_		326,015
Inventory		51,295		-		-		51,295
Current portion of notes receivable		23,490		-		-		23,490
Prepaid expenses and other								
current assets		78,530						78,530
Total current assets		4,356,647	<u>.</u>	-				4,356,647
DRADEDWY AND FOULDWENT and of								
PROPERTY AND EQUIPMENT - net of								
accumulated depreciation		508,553	. <u> </u>					508,553
OTHER ASSETS								
Notes receivable - net of								
current portion		7,789		-		-		7,789
- Customer lists - net of amortization		4,308		-		_		4,308
Other assets		21,280		-		_		21,280
		33,377		-				33,377
	ė		Ċ		Ċ		~	
	Ş	4,898,577	Ş		Ş		Ş	4,898,577

ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC AND SUBSIDIARIES SUPPLEMENTAL SCHEDULE I – CONSOLIDATING BALANCE SHEETS (CONTINUED) DECEMBER 31, 2021

LIABILITIES AND MEMBERS' EQUITY	Inte	ro-Master rnational chise, LLC	EM Cap <u>Service</u>		<u>Elimi</u>	nations	<u>Cor</u>	nsolidated
CURRENT LIABILITIES								
Accounts payable - trade	\$	821,847	\$	-	\$	-	\$	821,847
Accounts payable - franchisees		951,102		-		-		951,102
Deferred revenue		263,149		-		-		263,149
Accrued expenses		472,688		-		-		472,688
Current portion of capital lease								
obligations		78,218		-		-		78,218
Total current liabilities		2,587,004						2,587,004
NON-CURRENT DEBT								
Capital lease obligations -								
net of current portion		116,722		-		-		116,722
Other liabilities		31,983		-		-		31,983
		148,705						148,705
MEMBERS' EQUITY								
Enviro-Master International Franchise, L	LC							
members' equity		2,162,868		_		_		2,162,868
Non-controlling interest				_		_		
		2,162,868						2,162,868
		2,102,000						2,102,000
	\$	4,898,577	\$	_	\$	_	\$	4,898,577

ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC AND SUBSIDIARIES SUPPLEMENTAL SCHEDULE I - CONSOLIDATING BALANCE SHEETS DECEMBER 31, 2020

	Enviro-Master International Franchise, LLC	EM Capital <u>Services, LLC</u>	Eliminations	<u>Consolidated</u>
ASSETS				
CURRENT ASSETS				
Cash	\$ 4,144,203	\$ 911	\$ -	\$ 4,145,114
Accounts receivable - franchisees	402,942	-	-	402,942
Inventory	119,499	-	-	119,499
Current portion of notes receivable	33,752	-	-	33,752
Prepaid expenses and other current assets	99,207			99,207
Total current assets	4,799,603	911		4,800,514
PROPERTY AND EQUIPMENT - net of				
accumulated depreciation	427,887			427,887
OTHER ASSETS				
Notes receivable - net of current portion	80,839	-	-	80,839
Customer lists - net of amortization	8,478	67,521	-	75,999
Other assets	21,230	-	-	21,230
Investment in subsidiaries	(66,904)		66,904	
	43,643	67,521	66,904	178,068
	\$ 5,271,133	\$ 68,432	\$ 66,904	\$ 5,406,469

ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC AND SUBSIDIARIES SUPPLEMENTAL SCHEDULE I – CONSOLIDATING BALANCE SHEETS (CONTINUED) DECEMBER 31, 2020

		iro-Master ernational	EM	Capital				
	Fra	nchise, LLC		vices, LLC	Elir	<u>minations</u>	Co	nsolidated
LIABILITIES AND MEMBERS' EQUITY								
CURRENT LIABILITIES								
Accounts payable - trade	\$	933,173	\$	_	\$	-	\$	933,173
Accounts payable - franchisees		1,929,100		-		-		1,929,100
Accounts payable - related parties		4,290		(4,290)		-		-
Deferred revenue		147,101		-		-		147,101
Accrued expenses		503,564		6,000		-		509,564
Current portion of capital								
lease obligations	_	40,880		-		-		40,880
Total current liabilities		3,558,108		1,710		-		3,559,818
NON-CURRENT DEBT Capital lease obligations - net of current portion Other liabilities		29,488		_		_		29,488
Other Hadilities		46,165						46,165
		75,653						75,653
MEMBERS' EQUITY								
Enviro-Master International Franchise, LLC								
members' equity Non-controlling interest		1,637,372		(66,904) 133,626 66,722		66,904 - 66,904		1,637,372 133,626 1,770,998
	\$	5,271,133	\$	68,432	\$	66,904	\$	5,406,469

ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC AND SUBSIDIARIES SUPPLEMENTAL SCHEDULE II – CONSOLIDATING STATEMENTS OF INCOME YEAR ENDED DECEMBER 31, 2021

	Enviro-Master International	EM Capital		
	Franchise, LLC	<u>Services, LLC</u>	<u>Eliminations</u>	<u>Consolidated</u>
REVENUE				
Franchise sales, net	\$ 598,567	\$ –	\$ -	\$ 598,567
Royalties and franchise fees	13,348,482	-	-	13,348,482
Product sales, net	9,719,291			9,719,291
	23,666,340			23,666,340
COST OF REVENUE				
Products	8,364,323	-	-	8,364,323
Labor	3,210,409	-	-	3,210,409
Broker commissions on franchise sales	138,044	-	_	138,044
	11,712,776			11,712,776
GROSS PROFIT	11,953,564	-	-	11,953,564
GENERAL AND ADMINISTRATIVE EXPENSES	7,351,555	72,228		7,423,783
	4,602,009	(72,228)		4,529,781
OTHER INCOME (EXPENSE) FROM CONTINUING OPERATIONS	1			
Interest income	4,319	-	-	4,319
Settlement expense	(56,200)	-	-	(56,200)
Other income	9,834	-	-	9,834
Loss on sale of property and equipment	(5,739)	-	-	(5,739)
Loss on foreign currency exchange	(8,368)	-	-	(8,368)
Interest expense	(8,380)	(123)	-	(8,503)
Net equity losses of subsidiaries	(57,881)		57,881	
	(122,415)	(123)	57,881	(64,657)
CONSOLIDATED NET INCOME (LOSS)	4,479,594	(72,351)	57,881	4,465,124
Less: Net (loss) attributable to				
non-controlling interest		(14,470)		(14,470)
NET INCOME (LOSS) ATTRIBUTABLE TO ENVIRO-MASTER				
INTERNATIONAL FRANCHISE, LLC	\$ 4,479,594	\$ (57,881)	\$ 57,881	\$ 4,479,594

ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC AND SUBSIDIARIES SUPPLEMENTAL SCHEDULE II – CONSOLIDATING STATEMENTS OF INCOME YEAR ENDED DECEMBER 31, 2020

	International	EM Capital		
	Franchise, LLC	<u>Services, LLC</u>	<u>Eliminations</u>	<u>Consolidated</u>
REVENUE				
Franchise sales, net	\$ 1,211,125	\$ -	\$ -	\$ 1,211,125
Royalties and franchise fees	12,606,533	-	-	12,606,533
Product sales, net	10,358,485			10,358,485
	24,176,143			24,176,143
COST OF REVENUE				
Products	9,019,704	_	_	9,019,704
Labor	2,872,048	-	-	2,872,048
Broker commissions on franchise sales	245,178	-	_	245,178
	12,136,930		_	12,136,930
GROSS PROFIT	12,039,213	-	-	12,039,213
GENERAL AND ADMINISTRATIVE EXPENSES	6,572,288	94,730		6,667,018
	5,466,925	(94,730)		5,372,195
OTHER INCOME (EXPENSE)				
Interest income	10,835	-	-	10,835
Settlement income	82,062	-	-	82,062
Loss on sale of property and equipment	(2,146)	-	-	(2,146)
Loss on foreign currency exchange	(381)	-	-	(381)
Interest expense	(24,033)	(1,228)	-	(25,261)
Net equity earnings of subsidiaries	(76,767)		76,767	
	(10,430)	(1,228)	76,767	65,109
CONSOLIDATED NET INCOME (LOSS)	5,456,495	(95,958)	76,767	5,437,304
Less: Net (loss) income attributable to				
non-controlling interest		(19,191)		(19,191)
NET INCOME (LOSS) ATTRIBUTABLE TO ENVIRO-MASTE	R			
INTERNATIONAL FRANCHISE, LLC	\$ 5,456,495	\$ (76,767)	\$ 76,767	\$ 5,456,495

ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC AND SUBSIDIARIES SUPPLEMENTAL SCHEDULE III – CONSOLIDATING STATEMENTS OF EQUITY YEARS ENDED DECEMBER 31, 2021 AND 2020

	Enviro-Master			
	International	EM Capital		
	Franchise, LLC	Services, LLC	<u>Eliminations</u>	<u>Consolidated</u>
As of the year ended December 31, 2019	\$ 313,164	\$ 143,306	\$9,511	\$ 465,981
Net income (loss)	5,456,495	(95,958)	76,767	5,437,304
Capital contributions	-	19,374	(19,374)	-
Distributions to members	(4,132,287)			(4,132,287)
As of the year ended December 31, 2020	1,637,372	66,722	66,904	1,770,998
Net income (loss)	4,479,594	(72,351)	57,881	4,465,124
Capital contributions	-	5,629	(5,629)	-
Distributions to members	(3,954,098)		(119,156)	(4,073,254)
As of the year ended December 31, 2021	\$ 2,162,868	<u>\$ </u>	\$	\$ 2,162,868

ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC AND SUBSIDIARIES SUPPLEMENTAL SCHEDULE IV – CONSOLIDATING STATEMENTS OF CASH FLOWS YEAR ENDED DECEMBER 31, 2021

	Enviro-Master International <u>Franchise, LLC</u>	EM Capital Services, LLC	<u>Eliminations</u>	<u>Consolidated</u>
Net income (loss) from continuing operations	\$ 4,479,594	\$ (72,351)	\$ 57,881	\$ 4,465,124
Adjustments to reconcile net income (loss) to net				
cash provided by (used in) operating activities:				
Depreciation and amortization	207,787	67,521	-	275,308
Equity loss in subsidiaries	57,881	-	(57,881)	-
Loss on disposal of property and equipment	5,739	-	-	5,739
Bad debt recovery	(12,088)	-	-	(12,088)
Changes in operating assets and liabilities:				
Decrease in accounts receivable - franchisees	89,015	-	-	89,015
Decrease in inventory	68,204	-	-	68,204
Decrease in prepaid expenses and other				
current assets	20,677	-	-	20,677
Increase in other assets	(50)	-	-	(50)
Decrease in accounts payable - trade	(111,326)	-	-	(111,326)
Decrease in accounts payable - franchisees	(977,998)	-	-	(977,998)
Increase in deferred revenue	101,866	-	-	101,866
Decrease in accrued expenses	(35,166)	(1,710)		(36,876)
Total adjustments	(585,459)	65,811	(57,881)	(577,529)
Cash provided by (used in) operating activities	3,894,135	(6,540)		3,887,595
Cash flows from investing activities:				
Purchases of property and equipment	(100,766)	-	-	(100,766)
Contributions to LLC member interests	(5,629)	-	5,629	-
Net proceeds from notes receivable	83,312			83,312
Cash (used in) provided by investing activities	(23,083)		5,629	(17,454)
Cash flows from financing activities:				
Principal payments on capital lease obligations	(64,684)	-	-	(64,684)
Distributions/contributions to/from members	(4,073,254)	5,629	(5,629)	(4,073,254)
Cash (used in) provided by financing activities	(4,137,938)	5,629	(5,629)	(4,137,938)
NET DECREASE IN CASH	(266,886)	(911)	_	(267,797)
CASH AT BEGINNING OF YEAR	4,144,203	911		4,145,114
CASH AT END OF YEAR	\$ 3,877,317	\$	\$	\$ 3,877,317

ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC AND SUBSIDIARIES SUPPLEMENTAL SCHEDULE IV – CONSOLIDATING STATEMENTS OF CASH FLOWS YEAR ENDED DECEMBER 31, 2020

	Enviro-Master			
	International	EM Capital		
	Franchise, LLC	<u>Services, LLC</u>	<u>Eliminations</u>	<u>Consolidated</u>
Net income (loss)	\$ 5,456,495	\$ (95,958)	\$ 76,767	\$ 5,437,304
Adjustments to reconcile net income (loss)				
to net cash provided by				
(used in) continuing operating activities:				
Depreciation and amortization	172,617	90,027	-	262,644
Equity loss from subsidiaries	76,767	-	(76,767)	-
Loss on disposal of property and equipment Changes in operating assets and liabilities:	2,146	-	-	2,146
Increase in accounts receivable - franchisees	(160,763)	-	-	(160,763)
Decrease in prepaid expenses and other current assets	89,230	_	_	89,230
Increase in inventory	(119,499)	_	_	(119,499)
(Increase) decrease in other assets	(2,652)	6,792	_	4,140
Increase in accounts payable - trade	180,285	-	-	180,285
Increase in accounts payable - franchisees	949,399	-	-	949,399
Increase in deferred revenue	21,276	-	-	21,276
Increase (decrease) in accrued expenses	186,313	(21,183)		165,130
Total adjustments	1,395,119	75,636	(76,767)	1,393,988
Cash provided by (used in) operating activities	6,851,614	(20,322)		6,831,292
Cash flows from investing activities:				
Purchases of property and equipment	(157,419)	-	-	(157,419)
Net proceeds (payments) from notes receivable	293,126	-	-	293,126
Contributions to LLC member interests	(19,374)		19,374	
Cash (used in) provided by investing activities	116,333		19,374	135,707
Cash flows from financing activities:				
Principal payments on notes payable	(12,284)	-	-	(12,284)
Principal payments on notes payable - members	(160,565)	-	-	(160,565)
Principal payments on capital lease obligations	(44,416)	-	-	(44,416)
Distributions/contributions to/from members	(4,132,287)	19,374	(19,374)	(4,132,287)
Cash (used in) provided by financing activities	(4,349,552)	19,374	(19,374)	(4,349,552)
NET (DECREASE) INCREASE IN CASH	2,618,395	(948)	-	2,617,447
CASH AT BEGINNING OF YEAR	1,525,808	1,859		1,527,667
CASH AT END OF YEAR	\$ 4,144,203	\$ 911	\$ –	\$ 4,145,114

ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC AND SUBSIDIARIES

SUPPLEMENTAL SCHEDULE V – CONSOLIDATING STATEMENTS OF GENERAL AND ADMINISTRATIVE EXPENSES YEAR ENDED DECEMBER 31, 2021

	In	viro-Master ternational anchise, LLC	EM Capital rvices, LLC	<u>Eli</u>	<u>minations</u>	<u>Co</u>	nsolidated
Advertising and promotion	\$	363,098	\$ _	\$	_	\$	363,098
Awards		87,507	-		-		87,507
Bad debt recovery		(12,088)	-		-		(12,088)
Bank charges and credit card fees		884,159	-		-		884,159
Computer and technology expense		298,313	-		-		298,313
Contributions		7,097	-		-		7,097
Depreciation and amortization		207,787	67,521		-		275,308
Dues and subscriptions		76,241	-		-		76,241
Insurance		85,001	-		-		85,001
Other expenses		227,327	-		-		227,327
Office supplies		35,852	-		-		35,852
Payroll and payroll taxes		3,386,705	-		-		3,386,705
Postage expense		43,881	-		-		43,881
Professional services		711,210	4,707		-		715,917
Recruiting		75,104	-		-		75,104
Rent		211,154	-		-		211,154
Repairs and maintenance		7,844	-		-		7,844
Retirement plan match		88,898	-		-		88,898
Taxes and licenses - other		37,950	-		-		37,950
Telephone		29,306	-		-		29,306
Travel and entertainment		466,856	-		-		466,856
Training and continuing education		32,353	 				32,353
	\$	7,351,555	\$ 72,228	\$		\$	7,423,783

<u>ENVIRO-MASTER INTERNATIONAL FRANCHISE, LLC AND SUBSIDIARIES</u> <u>SUPPLEMENTAL SCHEDULE V – CONSOLIDATING STATEMENTS OF GENERAL AND ADMINISTRATIVE EXPENSES</u> <u>FOR THE YEAR ENDED DECEMBER 31, 2020</u>

	Enviro-Master			
	International	EM Capital		
	Franchise, LLC	<u>Services, LLC</u>	<u>Eliminations</u>	<u>Consolidated</u>
Advertising and promotion	\$ 431,213	\$ -	\$ -	\$ 431,213
Awards	36,778	-	-	36,778
Bank charges and credit card fees	847,651	(5)	-	847,646
Computer and technology expense	261,333	-	-	261,333
Contributions	7,767	-	-	7,767
Depreciation and amortization	172,617	90,027	-	262,644
Dues and subscriptions	57,398	-	-	57,398
Insurance	82,682	-	-	82,682
Other expense	131,061	-	-	131,061
Office supplies	40,223	-	-	40,223
Payroll and payroll taxes	2,947,171	-	-	2,947,171
Postage expense	54,172	-	-	54,172
Professional services	563,349	4,708	-	568,057
Recruiting	114,729	-	-	114,729
Rent	228,216	-	-	228,216
Repairs and maintenance	6,952	-	-	6,952
Retirement plan match	79,835	-	-	79,835
Taxes and licenses - other	13,388	-	-	13,388
Telephone	32,586	-	-	32,586
Travel and entertainment	371,537	-	-	371,537
Training and continuing education	91,630			91,630
	\$ 6,572,288	\$ 94,730	\$ –	\$ 6,667,018

EXHIBIT K

STATE SPECIFIC ADDENDA

We are required to provide you with additional information as a condition of registering our franchise offering in certain states. The additional disclosures are set out below. These additional disclosures apply only if the jurisdictional requirements of the applicable state franchise law are met.

REQUIRED BY THE STATE OF CALIFORNIA

SECTION 31125 OF THE CALIFORNIA CORPORATIONS CODE REQUIRES US TO GIVE YOU A DISCLOSURE DOCUMENT, IN A FORM CONTAINING THE INFORMATION THAT THE COMMISSIONER MAY BY RULE OR ORDER REQUIRE, BEFORE A SOLICITATION OF A PROPOSED MATERIAL MODIFICATION OF AN EXISTING FRANCHISE.

THE CALIFORNIA FRANCHISE INVESTMENT LAW REQUIRES THAT A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE BE DELIVERED TOGETHER WITH THE FDD.

See the cover page of the FDD for our URL address. OUR WEBSITE HAS NOT BEEN REVIEWED OR APPROVED BY THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION AND INNOVATION. ANY COMPLAINTS CONCERNING THE CONTENTS OF THIS WEBSITE MAY BE DIRECTED TO THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION AND INNOVATION AT WWW.DFPI.CA.GOV.

Item 3, Additional Disclosure. The following statement is added to Item 3:

Neither we nor any person listed in Item 2 is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities Exchange Act of 1934, 15 U.S.C.A. 78a *et seq.*, suspending or expelling such parties from membership in such association or exchange.

Item 5, Additional Disclosures: The following statements are added to Item 5:

The Department has determined that we, the franchisor, have not demonstrated that we are adequately capitalized and/or that we must rely on franchise fees to fund our operations. The Commissioner has imposed a free deferral condition, which requires that we defer the collection of all initial fees from California franchisees until we have completed all of our pre-opening obligations and you are open for business.

Item 17, Additional Disclosures. The following statements are added to Item 17:

California Business and Professions Code Sections 20000 through 20043 provide rights to the franchisee concerning termination, transfer or non-renewal of the franchise. If the Franchise Agreement contains a provision that is inconsistent with the law, the law will control.

The Franchise Agreement provide for termination upon bankruptcy. These provisions may not be enforceable under federal bankruptcy law (11 U.S.C.A. § 101, *et seq.*).

The Franchise Agreement provides for application of the laws of North Carolina. These provisions may not be enforceable under California law.

The Franchise Agreement contains a covenant not to solicit which extends beyond the termination of the franchise. These provisions may not be enforceable under California law.

You must sign a general release if you receive a refund of initial fees, and if you transfer your franchise or execute a successor Franchise Agreement. These provisions may not be enforceable under California law. California Corporations Code Section 31512 voids a waiver of your rights under the Franchise Investment Law (California Corporations Code Section 21000 voids a waiver of your rights under the Franchise Relations Act (Business and Professions Code Sections 20000 through 20043).

The Franchise Agreement requires that any action you bring be commenced in North Carolina. This provision may not be enforceable under California law. Prospective franchisees are encourage to consult private legal counsel to determine the applicability of California and federal laws (such as Business and Professions Code Section 20040.5, Code of Civil Procedure Section 1281, and the Federal Arbitration Act) to any provisions of a franchise agreement restricting venue to a forum outside the State of California.

Any arbitration will be conducted in accordance with AAA Rules of Commercial Arbitration, before an arbitrator appointed by AAA.

No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

FRANCHISE AGREEMENT ADDENDUM

REQUIRED BY THE STATE OF CALIFORNIA

TO BE EXECUTED SIMULTANEOUSLY WITH CALIFORNIA FRANCHISE AGREEMENTS

1. No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

2. The Department has determined that we, the franchisor, have not demonstrated that we are adequately capitalized and/or that we must rely on franchise fees to fund our operations. The Commissioner has imposed a free deferral condition, which requires that we defer the collection of all initial fees from California franchisees until we have completed all of our pre-opening obligations and you are open for business.

This _____ day of ______, 20____

Franchisee:

Enviro-Master:

Enviro-Master International Franchise, LLC

Date:

By:_____

Title:_____

Date:_____

REQUIRED BY THE STATE OF HAWAII

THESE FRANCHISES HAVE BEEN FILED UNDER THE FRANCHISE INVESTMENT LAW OF THE STATE OF HAWAII. FILING DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION OR ENDORSEMENT BY THE DIRECTOR OF COMMERCE AND CONSUMER AFFAIRS OR A FINDING BY THE DIRECTOR OF COMMERCE AND CONSUMER AFFAIRS THAT THE INFORMATION PROVIDED HEREIN IS TRUE, COMPLETE AND NOT MISLEADING.

THE FRANCHISE INVESTMENT LAW MAKES IT UNLAWFUL TO OFFER OR SELL ANY FRANCHISE IN THIS STATE WITHOUT FIRST PROVIDING TO THE PROSPECTIVE FRANCHISEE, OR SUBFRANCHISOR, AT LEAST SEVEN DAYS PRIOR TO THE EXECUTION BY THE PROSPECTIVE FRANCHISEE, OF ANY BINDING FRANCHISE OR OTHER AGREEMENT, OR AT LEAST SEVEN DAYS PRIOR TO THE PAYMENT OF ANY CONSIDERATION BY THE FRANCHISEE, OR SUBFRANCHISOR, WHICHEVER OCCURS FIRST, A COPY OF THE FDD, TOGETHER WITH A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE.

THIS FDD CONTAINS A SUMMARY ONLY OF CERTAIN MATERIAL PROVISIONS OF THE FRANCHISE AGREEMENT. THE CONTRACT OR AGREEMENT SHOULD BE REFERRED TO FOR A STATEMENT OF ALL RIGHTS, CONDITIONS, RESTRICTIONS AND OBLIGATIONS OF BOTH THE FRANCHISOR AND THE FRANCHISEE.

NO STATEMENT, QUESTIONNAIRE, OR ACKNOWLEDGEMENT SIGNED OR AGREED TO BY A FRANCHISEE IN CONNECTION WITH THE COMMENCEMENT OF THE FRANCHISE RELATIONSHIP SHALL HAVE THE EFFECT OF (i) WAIVING ANY CLAIMS UNDER APPLICABLE STATE FRANCHISE LAW, INCLUDING FRAUD IN THE INDUCEMENT, OR (ii) DISCLAIMING RELIANCE ON ANY STATEMENT MADE BY ANY FRANCHISOR, FRANCHISE SELLER, OR OTHER PERSON ACTING ON BEHALF OF THE FRANCHISOR. THIS PROVISION SUPERSEDES ANY OTHER TERM OF ANY DOCUMENT EXECUTED IN CONNECTION WITH THE FRANCHISE.

Registered agent in the state authorized to receive service of process: Commissioner of Securities, State of Hawaii, 335 Merchant Street, Honolulu, Hawaii 96813.

REQUIRED BY THE STATE OF ILLINOIS

1. <u>Item 17, Additional Disclosures</u>. The following statements are added to Item 17:

The Illinois Franchise Disclosure Act provides that any provision in the Franchise Agreement that designates jurisdiction or venue in a forum outside of Illinois is void with respect to any action which is otherwise enforceable in Illinois.

The Illinois Franchise Disclosure Act requires that Illinois law apply to any claim arising under the Illinois Franchise Disclosure Act.

The conditions under which your Franchise Agreement can be terminated and your rights upon nonrenewal may be affected by Sections 19 and 20 of the Illinois Franchise Disclosure Act.

No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

Each provision of these Additional Disclosures shall be effective only to the extent, with respect to such provision, that the jurisdictional requirements of the Illinois Franchise Disclosure Act are met independently without reference to these Additional Disclosures. The Additional Disclosures shall have no force or effect if such jurisdictional requirements are not met.

FRANCHISE AGREEMENT ADDENDUM REQUIRED BY THE STATE OF ILLINOIS

TO BE EXECUTED SIMULTANEOUSLY WITH ILLINOIS FRANCHISE AGREEMENTS

1. Sections 21.2 and 22.9 of the Franchise Agreement, along with any other provisions of the Franchise Agreement, are hereby amended to reflect, as required under Section 41 of the Illinois Franchise Disclosure Act, (the "Act") any condition, stipulation, or provision purporting to waive compliance with any provision of the Act or Illinois law is void.

2. Any provision in the Franchise Agreement including Sections 22.1, 22.6 and 22.7, designating jurisdiction or venue in a forum outside of Illinois is void with respect to any action which is otherwise enforceable in Illinois.

3. Illinois law shall apply to any claim arising under the Act.

4. Sections 19 and 20 of the Act supersede any contrary provision of the Franchise Agreement with respect to termination or nonrenewal of the Franchise Agreement.

5. No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

This _	day of	. 20
	••••) ••	,,,

Franchisee:

Enviro-Master:

Enviro-Master International Franchise, LLC

Date:_____

By:			

Title:_____

Date:_____

Enviro-Master International Franchise, LLC FDD May 6, 2024 #198798v1 K-7

REQUIRED BY THE STATE OF INDIANA

Item 17, Additional Disclosures. The following statements are added to Item 17:

- The Indiana Deceptive Franchise Practices Law (Indiana Code 23-2-2.7 et seq.) in general governs the relationship between the franchisor and the franchisee by forbidding certain provisions in the franchise agreement and related documents and by preventing the franchisor from engaging in certain acts and practices which could be considered coercive or oppressive to the franchisee. If any of the provisions of the Franchise Agreement conflicts with this law, this law will control.
- 2. Any provisions requiring you to sign a general release of claims against us, including upon execution of a successor Franchise Agreement, refund of initial fees, or transfer, does not release any claim you may have under the Indiana Deceptive Franchise Practices Law.
- 3. The Franchise Agreement provides that suit must be brought in North Carolina. These provisions may not be enforceable under Indiana law.
- 4. Indiana franchise laws will govern the Franchise Agreement and any and all other related documents.
- 5. No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

Each provision of these Additional Disclosures shall be effective only to the extent, with respect to such provision, that the jurisdictional requirements of the Indiana Deceptive Franchise Practices Law are met independently without reference to these Additional Disclosures. The Additional Disclosures shall have no force or effect if such jurisdictional requirements are not met.

REQUIRED BY THE STATE OF MARYLAND

1. <u>**Risk Factors, Additional Disclosures**</u>. The following RISK FACTOR is added to Special Risks to Consider About this Franchise:

<u>Sales Performance Required</u>. You must maintain minimum sales performance levels. Your inability to maintain these levels may result in loss of any territorial rights you are granted, termination of your franchise, and loss of your investment.

- 2. <u>Item 5, Additional Disclosures.</u> Item 5 is hereby amended as follows: Based upon the franchisor's financials condition, the Maryland Securities Commissioner has required a financial assurance. Therefore, all initial fees and payments owed by franchisees shall be deferred until the franchisor completes its pre-opening obligations under the franchise agreement.
- 3. <u>Item 17, Additional Disclosures</u>. The following statements are added to Item 17:

The Franchise Agreement provides for termination upon bankruptcy. These provisions may not be enforceable under federal bankruptcy law.

Any provisions requiring you to sign a general release of claims against us, including upon execution of the Franchise Agreement or a successor Franchise Agreement, refund of initial fees, or transfer, does not release any claim you may have under the Maryland Franchise Registration and Disclosure Law.

Any claims arising under the Maryland Franchise Registration and Disclosure Law must be brought within 3 years after the grant of the franchise.

A franchisee may bring a lawsuit in Maryland for claims arising under the Maryland Franchise Registration and Disclosure Law.

No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

Each provision of these Additional Disclosures shall be effective only to the extent, with respect to such provision, that the jurisdictional requirements of the Maryland Franchise Registration and Disclosure Law are met independently without reference to these Additional Disclosures.

MARYLAND FRANCHISE AGREEMENT ADDENDUM REQUIRED BY THE STATE OF MARYLAND

1. Based upon the franchisor's financials condition, the Maryland Securities Commissioner has required a financial assurance. Therefore, all initial fees and payments owed by franchisees shall be deferred until the franchisor completes its pre-opening obligations under the franchise agreement.

2. All representations requiring prospective franchisees to assent to a release, estoppel or waiver of liability are not intended to nor shall they act as a release, estoppel or waiver of any liability incurred under the Maryland Franchise Registration and Disclosure Law.

3. Section 2.3(e) of the Franchise Agreement is amended to add the following statement: "Pursuant to COMAR 02.02.08.16L, the general release required pursuant to this subparagraph or otherwise as a condition for renewal, sale, and/or assignment or transfer shall not apply to any liability incurred under the Maryland Franchise Registration and Disclosure Law."

4. Section 11.3(e) of the Franchise Agreement is amended to add the following statements after the last parenthetical: "...and Franchisor must execute a similar General Release of claims against Franchisee. Pursuant to COMAR 02.02.08.16L, the general release required pursuant to this subparagraph or otherwise as a condition for renewal, sale, and/or assignment or transfer shall not apply to any liability incurred under the Maryland Franchise Registration and Disclosure Law."

5. Section 22.7(b)(ii) of the Franchise Agreement is amended to add the following statement: "Provided, however, that nothing herein shall prevent Franchisee from bringing a lawsuit in Maryland for claims arising under the Maryland Franchise Registration and Disclosure Law."

6. Section 22.5 of the Franchise Agreement is amended to add the following statement: "Notwithstanding the foregoing, any claims arising under the Maryland Franchise Registration and Disclosure Law must be brought within three years after the grant of the franchise."

7. Article 22 is amended to add the following, new Section 22.10:

22.10 NO WAIVER OF STATE LAW OBLIGATIONS. All representations requiring Franchisee to assent to a release, estoppel or waiver of liability are not intended to nor shall they act as a release, estoppel or waiver of any liability incurred under the Maryland Franchise Registration and Disclosure Law.

8. No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller,

or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

This _____ day of ______, 20____

Franchisee:

Enviro-Master:

Enviro-Master International Franchise, LLC

Date:_____

By:_____

Title:_____

Date:_____

REQUIRED BY THE STATE OF MINNESOTA

1. **<u>Trademarks</u>**. The following statement is added to Item 13:

Notwithstanding the foregoing, we will indemnify you against liability to a third party resulting from claims that your use of a Marks infringes trademarks rights of a third party; provided, that we will not indemnify against the consequences of your use of the Marks unless the use is in accordance with the requirements of the Franchise Agreement and the System. Minnesota considers it unfair to not protect your right to use the trademarks. Refer to Minnesota Statues, Section 80C.12, Subd. 1(g).

2. **Choice of Forum and Law.** The following statement is added to Item 17:

Minnesota Statute § 80C.21 and Minnesota Rule 2860.4400J prohibit us from requiring litigation to be conducted outside Minnesota requiring waiver of a jury trial, or requiring you to consent to liquidated damages, termination penalties or judgment notes. In addition, nothing in the FDD or Franchise Agreement can abrogate or reduce (1) any of your rights as provided for in Minnesota Statutes, Chapter 80C, or your rights to any procedure, forum, or (2) remedies provided for by the laws of the jurisdiction.

3. <u>General Release</u>. The following statement is added to Item 17:

Minnesota Rule 2860.4400D prohibits us from requiring you to assent to a release, assignment, novation, or waiver that would relieve any person from liability imposed by Minnesota Statute §§ 80C.01 - 80C.22.

4. Waiver of Right to Jury Trial and Consent to Liquidated Damages or Termination Penalties: The following statement is added to Item 17:

You cannot consent to us obtaining injunctive relief. We may <u>seek</u> injunctive relief. See Minnesota Rule 2860.4400J, which among other things, prohibits us from requiring you to waive your rights to a trial or to consent to liquidated damages, termination penalties, or judgment notes.

5. **Notice of Termination**. The following statement is added to Item 17:

With respect to franchises governed by Minnesota law, we will comply with Minnesota Statute § 80C.14, subdivisions 3, 4, and 5 which require, except in certain specified cases, (1) that a franchisee be given 90 days notice of termination (with 60 days to cure) and 180 days notice for non-renewal of the Franchise Agreement and (2) that consent to the transfer of the franchise will not be unreasonably withheld.

6. The following statement is added at the end of Item 17:

No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

The provision of this Additional Disclosure shall be effective only to the extent, with respect to such provision, that the jurisdictional requirements of the Minnesota franchise statutes are met independently without reference to these Additional Disclosures.

REQUIRED BY THE STATE OF NEW YORK

1. <u>The following information is added to the cover page of the Franchise Disclosure</u> <u>Document</u>:

INFORMATION COMPARING FRANCHISORS IS AVAILABLE. CALL THE STATE ADMINISTRATORS LISTED IN EXHIBIT A OR YOUR PUBLIC LIBRARY FOR SERVICES OR INFORMATION. REGISTRATION OF THIS FRANCHISE BY NEW YORK STATE DOES NOT MEAN THAT NEW YORK STATE RECOMMENDS IT OR HAS VERIFIED THE INFORMATION IN THIS FRANCHISE DISCLOSURE DOCUMENT. IF YOU LEARN THAT ANYTHING IN THIS FRANCHISE DISCLOSURE DOCUMENT. IF YOU LEARN THAT ANYTHING IN THIS FRANCHISE DISCLOSURE DOCUMENT IS UNTRUE, CONTACT THE FEDERAL TRADE COMMISSION AND THE APPROPRIATE STATE OR PROVINCIAL AUTHORITY. THE FRANCHISOR MAY, IF IT CHOOSES, NEGOTIATE WITH YOU ABOUT ITEMS COVERED IN THE FRANCHISE DISCLOSURE DOCUMENT. HOWEVER, THE FRANCHISOR CANNOT USE THE NEGOTIATING PROCESS TO PREVAIL UPON A PROSPECTIVE FRANCHISEE TO ACCEPT TERMS WHICH ARE LESS FAVORABLE THAN THOSE SET FORTH IN THIS FRANCHISE DISCLOSURE DOCUMENT.

2. <u>The following is added at the end of Item 3:</u>

Except as provided above, with regard to the franchisor, its predecessor, a person identified in Item 2, or an affiliate offering franchises under the franchisor's principal trademark:

A. No such party has an administrative, criminal or civil action pending against that person alleging: a felony, a violation of a franchise, antitrust, or securities law, fraud, embezzlement, fraudulent conversion, misappropriation of property, unfair or deceptive practices, or comparable civil or misdemeanor allegations.

B. No such party has pending actions, other than routine litigation incidental to the business, which are significant in the context of the number of franchisees and the size, nature or financial condition of the franchise system or its business operations.

C. No such party has been convicted of a felony or pleaded nolo contendere to a felony charge or, within the 10 year period immediately preceding the application for registration, has been convicted of or pleaded nolo contendere to a misdemeanor charge or has been the subject of a civil action alleging: violation of a franchise, antifraud, or securities law; fraud; embezzlement; fraudulent conversion or misappropriation of property; or unfair or deceptive practices or comparable allegations.

D. No such party is subject to a currently effective injunctive or restrictive order or decree relating to the franchise, or under a Federal, State, or Canadian franchise, securities, antitrust, trade regulation or trade practice law, resulting from a concluded or pending action or proceeding brought by a public agency; or is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities and Exchange Act of 1934, suspending or expelling such person from membership in such association or exchange; or is subject to a currently effective injunctive or restrictive order relating to any other business activity as a result of an action brought by a public agency or department, including, without limitation, actions affecting a license as a real estate broker or sales agent.

3. The following is added to the end of the "Summary" sections of Item 17(c), titled "**Requirements for franchisee to renew or extend**," and Item 17(m), entitled "**Conditions for franchisor approval of transfer**":

However, to the extent required by applicable law, all rights you enjoy and any causes of action arising in your favor from the provisions of Article 33 of the General Business Law of the State of New York and the regulations issued thereunder shall remain in force; it being the intent of this proviso that the non-waiver provisions of General Business Law Sections 687(4) and 687(5) be satisfied.

4. The following language replaces the "Summary" section of Item 17(d), titled **"Termination by franchisee":**

You may terminate the agreement on any grounds available by law

5. The following is added to the end of the "Summary" sections of Item 17(v), titled "**Choice of forum**", and Item 17(w), titled "**Choice of law**":

The foregoing choice of law should not be considered a waiver of any right conferred upon the franchisor or upon the franchisee by Article 33 of the General Business Law of the State of New York.

6. The following is added at the end of Item 17:

No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

REQUIRED BY THE STATE OF NORTH DAKOTA

- 1. Item 17(c) of the Disclosure Document and Section 2.3(e) of the Franchise Agreement are amended so that no general release shall be required upon renewal of that Agreement.
- 2. Item 17(i) of the Disclosure Document and Sections 18 and 19 of the Franchise Agreement provide for certain rights upon termination or expiration of the Franchise Agreement. You are not providing that you consent to termination upon expiration of the Franchise Agreement, but your rights as a franchisee will expire nonetheless. We also cannot require that you pay liquidated damages upon termination.
- 3. Item 17(r) of the Disclosure Document and Section 20.3 of the Franchise Agreement contain provisions relating to covenants not to compete; covenants not to compete such as those mentioned above are generally unenforceable in North Dakota. However, any other business covenants contained within Section 20.3, such as covenants not to solicit or covenants respecting confidential information, may still be enforced.
- 4. Item 17(u) of the Disclosure Document and Sections 22.6 and Section 22.7 of the Franchise Agreement are amended to provide that the site of any dispute resolution shall be agreeable to all parties and shall not be remote from Franchisee's place of business.
- 5. Item 17(v) of the Disclosure Document and Section 22.7 of the Franchise Agreement are amended to eliminate the requirement that Franchisee consent to the jurisdiction of the courts of North Carolina or any court outside of North Dakota.
- 6. Item 17(w) of the Disclosure Document and Section 22.1 of the Franchise Agreement are amended to reflect that North Dakota law shall apply to this Agreement.
- 7. Item 17 of the Disclosure Document and Franchise Agreement are amended to reflect the following language: No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 8. Section 22.8 of the Franchise Agreement is amended to reflect that Franchisee does not waive its right to trial by jury.

- 9. Section 22.9 of the Franchise Agreement is amended to not waive exemplary or punitive damages.
- 10. Section 22.5 of the Franchise Agreement is amended to delete reference to a limitation of claims being brought after one year; the statute of limitations under North Dakota law shall apply.

REQUIRED BY THE STATE OF RHODE ISLAND

Item 17, Additional Disclosure. The following statements are added to Item 17:

Section 19-28.1-14 of the Rhode Island Franchise Investment Act provides that: "A provision in a franchise agreement restricting jurisdiction or venue to a forum outside this state or requiring the application of the laws of another state is void with respect to a claim otherwise enforceable under this Act."

No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

The provision of this Additional Disclosure shall be effective only to the extent, with respect to such provision, that the jurisdictional requirements of the Rhode Island Franchise Investment Law are met independently without reference to these Additional Disclosures.

REQUIRED BY THE STATE OF SOUTH DAKOTA

No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

REQUIRED BY THE STATE OF VIRGINIA

Item 17, Additional Disclosure. The following statement is added to Item 17:

Pursuant to Section 13.1-564 of the Virginia Retail Franchising Act, it is unlawful for a franchisor to cancel a franchise without reasonable cause. If any ground for default or termination stated in the franchise agreement does not constitute "reasonable cause," as that term may be defined in the Virginia Retail Franchising Act or the laws of Virginia, that provision may not be enforceable.

No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

REQUIRED BY THE STATE OF WASHINGTON

In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, Chapter 19.100 RCW will prevail.

RCW 19.100.180 may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise. There may also be court decisions which may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise.

In any arbitration or mediation involving a franchise purchased in Washington, the arbitration or mediation site will be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration or mediation, or as determined by the arbitrator or mediator at the time of arbitration or mediation. In addition, if litigation is not precluded by the franchise agreement, a franchisee may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.

A release or waiver of rights executed by a franchisee may not include rights under the Washington Franchise Investment Protection Act or any rule or order thereunder except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable.

Transfer fees are collectable to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.

Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provisions contained in the franchise agreement or elsewhere that conflict with these limitations are void and unenforceable in Washington.

RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) soliciting or hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such provisions contained in the franchise agreement or elsewhere are void and unenforceable in Washington.

The following language replaces the "Summary" section of Item 17(d), titled "**Termination by Franchisee**":

You may terminate the agreement on any grounds available by law.

K-21

The following language is included at the end of Item 17:

No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

The undersigned does hereby acknowledge receipt of this addendum.

Dated this	day of	20
Prospective Franchisee	r:	Enviro-Master International Franchise, LLC
By <u>:</u>		Ву:
Title:		Title:
Date:		Date:

K-22

WASHINGTON ADDENDUM TO THE FRANCHISE AGREEMENT

In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, Chapter 19.100 RCW will prevail.

RCW 19.100.180 may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise. There may also be court decisions which may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise.

In any arbitration or mediation involving a franchise purchased in Washington, the arbitration or mediation site will be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration or mediation, or as determined by the arbitrator or mediator at the time of arbitration or mediation. In addition, if litigation is not precluded by the franchise agreement, a franchisee may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.

A release or waiver of rights executed by a franchisee may not include rights under the Washington Franchise Investment Protection Act or any rule or order thereunder except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable.

Transfer fees are collectable to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.

Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provisions contained in the franchise agreement or elsewhere that conflict with these limitations are void and unenforceable in Washington.

RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) soliciting or hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such provisions contained in the franchise agreement or elsewhere are void and unenforceable in Washington.

Section 23 of the Franchise Agreement is amended to add the following language:

Nothing in this section should be construed as a waiver of any of Franchisee's rights under the Washington Franchise Investment Protection Act, RCW 19.100, and the rules adopted thereunder.

USE OF FRANCHISE BROKERS. The franchisor may use the services of franchise brokers to assist it in selling franchises. A franchise broker represents the franchisor and is paid a fee for referring prospects to the franchisor and/or selling the franchise. Do not rely only on the information provided by a franchise broker about a franchise. Do your own investigation by contacting the franchisor's current and former franchisees to ask them about their experience with the franchisor.

No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

The undersigned does hereby acknowledge receipt of this addendum.

Dated this	day of	_ 20	
	· · · · · · · · · · · · · · · · · · ·	_	

Prospective Franchisee:

Enviro-Master International Franchise, LLC

Ву:	Ву:
Title:	Title:
Date:	Date:

EXHIBIT L

FRANCHISEE DISCLOSURE QUESTIONNAIRE

THIS DOCUMENT WILL NOT BE SIGNED BY YOU, AND WILL NOT APPLY, IF THE OFFER OR SALE OF THE FRANCHISE IS SUBJECT TO THE STATE FRANCHISE REGISTRATION/DISCLOSURE LAWS IN THE STATES OF CALIFORNIA, HAWAII, ILLINOIS, INDIANA, MARYLAND, MICHIGAN, MINNESOTA, NEW YORK, NORTH DAKOTA, RHODE ISLAND, SOUTH DAKOTA, VIRGINIA, WASHINGTON, OR WISCONSIN.

As you know, Enviro-Master International Franchise, LLC ("we", "us" or "our") and you are preparing to enter into a Franchise Agreement for the operation of an Enviro-Master franchise. The purpose of this Questionnaire is to determine whether any statements or promises were made to you that we have not authorized or that may be untrue, inaccurate or misleading, to be certain that you have been properly represented in this transaction, and to be certain that you understand the limitations on claims you may make by reason of the purchase and operation of your franchise. You cannot sign or date this Questionnaire the same day as the Receipt for the Franchise Disclosure Document, but you must sign and date it the same day you sign the Franchise Agreement and pay your franchise fee. Please review each of the following questions carefully and provide honest responses to each question. If you answer "No" to any of the questions below, please explain your answer on the back of this sheet.

- Yes___ No___ I. Have you received and personally reviewed the Franchise Agreement and each exhibit and schedule attached to them?
- Yes___ No___ 2. Have you received and personally reviewed the Franchise Disclosure Document ("Disclosure Document") we provided?
- Yes___ No___ 3. Did you sign a receipt for the Disclosure Document indicating the date you received it?
- Yes___ No___ 4. Do you understand all the information contained in the Disclosure Document and the Franchise Agreement?
- Yes___ No___ 5. Have you reviewed the Disclosure Document and the Franchise Agreement with a lawyer, accountant or other professional advisor?
- Yes___ No___ 6. Have you discussed the benefits and risks of developing and operating an Enviro-Master franchise with an existing Enviro-Master franchisee?
- Yes___ No___ 7. Do you understand the risks of developing and operating an Enviro-Master franchise?
- Yes___ No___ 8. Do you understand the success or failure of your franchise will depend in large part upon your skills, abilities and efforts and those of the persons you employ, as well as many factors beyond your control such as weather, competition, interest rates, the economy, inflation, labor and supply costs, lease terms and the marketplace?

- Yes___ No___ 9. Do you understand that the only restriction concerning where another franchised or company unit may open is the limited protected territory specified in your Franchise Agreement?
- Yes___ No___ 10. Do you understand that most disputes or claims you may have arising out of or relating to the Franchise Agreement must be litigated in the courts closest to our principal executive office or arbitrated at the office of the American Arbitration Association closest to our principal executive office?
- Yes___ No___ 11. Do you understand that you (and your manager if you will employ one full-time) must satisfactorily complete our initial training course before we will allow your Enviro-Master unit to open?
- Yes___ No___ 12. Has any employee or other person speaking on our behalf made any statement or promise regarding the costs involved in operating a Enviro-Master franchise that is not contained in the Disclosure Document or that is contrary to, or different from, the information contained in the Disclosure Document?
- Yes___ No___ 13. Has any employee or other person speaking on our behalf made any statement or promise or agreement, other than those matters addressed in the Franchise Agreement, concerning advertising, marketing, media support, marketing penetration, training, support service or assistance that is contrary to, or different from, the information contained in the Disclosure Document?
- Yes___ No___ 14. Has any employee or other person speaking on our behalf made any statement or promise regarding the actual, average or projected profits or earnings, the likelihood of success, the amount of money you may earn, or the total amount of revenue a Enviro-Master franchise will generate, that is not contained in the Disclosure Document or that is contrary to, or different from, the information contained in the Disclosure Document?
- Yes___ No___ 15. Do you understand that the Franchise Agreement contains the entire agreement between us and you concerning the franchise for the Enviro-Master unit, meaning any prior oral or written statements not set out in the Franchise Agreement will not be binding?

YOU UNDERSTAND THAT YOUR ANSWERS ARE IMPORTANT TO US AND THAT WE WILL RELY ON THEM. BY SIGNING THIS QUESTIONNAIRE, YOU ARE REPRESENTING THAT YOU HAVE CONSIDERED EACH QUESTION CAREFULLY AND RESPONDED TRUTHFULLY TO THE ABOVE QUESTIONS.

Signature of Franchise Applicant	Signature of Franchise Applicant
Name (please print)	Name (please print)
Dated	Dated
Signature of Franchise Applicant	Signature of Franchise Applicant
Name (please print)	Name (please print)
Dated	Dated

Special note for residents of the State of Maryland and franchised businesses located in <u>Maryland</u>: Nothing in this Franchisee Disclosure Questionnaire shall act as a release, estoppel, or waiver of any liability incurred under the Maryland Franchise Registration and Disclosure Law.

EXPLANATION OF ANY NEGATIVE RESPONSES [REFER TO QUESTION NUMBER AND USE ADDITIONAL PAPER IF NECESSARY]:

State Effective Dates

The following states have franchise laws that require that the Franchise Disclosure Document be registered or filed with the state, or be exempt from registration:

California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, and Wisconsin.

This document is effective and may be used in the following states, where the document is filed, registered or exempt from registration, as of the Effective Date stated below:

State	Effective Date
California	Pending
Hawaii	Pending
Illinois	Pending
Indiana	Pending
Maryland	Pending
Michigan	Pending
Minnesota	Pending
New York	Pending
North Dakota	Pending
Rhode Island	Pending
South Dakota	Pending
Virginia	Pending
Wisconsin	Pending

Other states may require registration, filing, or exemption of a franchise under other laws, such as those that regulate the offer and sale of business opportunities or seller-assisted marketing plans.

RECEIPT

Enviro-Master Copy

This disclosure document summarizes certain provisions of the franchise agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If Enviro-Master International Franchise, LLC offers you a franchise, it must provide this disclosure document to you 14 calendar-days before you sign a binding agreement with, or make a payment to, the franchisor or an affiliate in connection with the proposed franchise sale. However, some state franchise laws require Enviro-Master to provide this disclosure document to you at the first personal meeting held to discuss the franchise sale or at least 10 business days before you sign a binding agreement with, or make a payment to, the franchisor or an affiliate in connection with the proposed franchise sales.

If Enviro-Master International Franchise, LLC does not deliver this disclosure document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580 and the state agency located in **Exhibit A**.

The name, principal business address, and telephone number of each franchise seller is as follows:

d Bierling	Name:	Larry Rushing
00 77 Center Drive, Ste. 500	Address:	5200 77 Center Drive, Ste. 500
arlotte, NC 28217		Charlotte, NC 28217
lephone: (704) 302-1016	Telephone:	(704) 302-1016
	d Bierling 00 77 Center Drive, Ste. 500 arlotte, NC 28217 lephone: (704) 302-1016	00 77 Center Drive, Ste. 500 Address: arlotte, NC 28217

Issuance Date: May 6, 2024

I have received a disclosure document dated May 6, 2024 that included the following Exhibits:

Agent for Service of Process	Exhibit A	Confidentiality Agreement	Exhibit G
Franchise Agreement	Exhibit B	Current Franchisees	Exhibit H
Form of Multi-Unit Development		Former Franchisees	Exhibit I
Addendum to Franchise Agreement	Exhibit C	Financial Statements	Exhibit J
Market Reservation and Deposit		State Specific Addenda	Exhibit K
Agreement	Exhibit D	Franchisee Disclosure	
Promissory Note	Exhibit E	Questionnaire	Exhibit L
Operations Manual			
Table of Contents	Exhibit F		

Date

Franchisee

Date

Franchisee

TO BE RETURNED TO:

Larry Rushing Enviro-Master 5200 77 Center Drive, Suite 500 Charlotte, NC 28217

RECEIPT Applicant Copy

This disclosure document summarizes certain provisions of the franchise agreement and other information in plain language. Read this Disclosure Document and all agreements carefully.

If Enviro-Master International Franchise, LLC offers you a franchise, it must provide this disclosure document to you 14 calendar-days before you sign a binding agreement with, or make a payment to, the franchisor or an affiliate in connection with the proposed franchise sale. However, some state franchise laws require Enviro-Master to provide this disclosure document to you at the first personal meeting held to discuss the franchise sale or at least 10 business days before you sign a binding agreement with, or make a payment to, the franchisor or an affiliate in connection with the proposed franchise sales.

If Enviro-Master International Franchise, LLC does not deliver this disclosure document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580 and the state agency located in **Exhibit A**.

The name, principal business address, and telephone number of each franchise seller is as follows:

Name: Tod Bierling Address: 5200 77 Center Drive, Ste. 500 Charlotte, NC 28217 Telephone: (704) 302-1016

Name: Larry Rushing Address: 5200 77 Center Drive, Ste. 500 Charlotte, NC 28217 Telephone: (704) 302-1016

Issuance Date: May 6, 2024

I have received a disclosure document dated May 6, 2024 that included the following Exhibits and Other Attachments:

Agent for Service of Process	Exhibit A	Confidentiality Agreement	Exhibit G
Franchise Agreement	Exhibit B	Current Franchisees	Exhibit H
Form of Multi-Unit Development		Former Franchisees	Exhibit I
Addendum to Franchise Agreement	Exhibit C	Financial Statements	Exhibit J
Market Reservation and Deposit		State Specific Addenda	Exhibit K
Agreement	Exhibit D	Franchisee Disclosure	
Promissory Note	Exhibit E	Questionnaire	Exhibit L
Operations Manual			
Table of Contents	Exhibit F		

Date

Franchisee

Date

Franchisee

RETAIN THIS COPY FOR YOUR RECORDS